Nutrition Assistance Program Report Series The Office of Analysis, Nutrition and Evaluation

Special Nutrition Programs

Report No. CN-04-CFABT

Measuring Competitive Foods in Schools: An Inventory Approach



United States Food and Department of Nutrition Agriculture Service

May 2004

"The U.S. Department of Agriculture (USDA) prohibits discrimination in all its programs and activities on the basis of race, color, national origin, gender, religion, age, disability, political beliefs, sexual orientation, or marital or family status. (Not all prohibited bases apply to all programs) Persons with disabilities who require alternative means for communication of program information (Braille, large print, audiotape, etc.) should contact USDA's TARGET Center at (202) 720-2600 (voice and TDD)"

To file a complaint of discrimination, write USDA, Director, Office of Civil Rights, Room 326-W, Whitten Building, 14th and Independence Avenue, SW, Washington, DC 20250-9410 or call (202) 720-5964 (voice and TDD). USDA is an equal opportunity provider and employer.



United States Food and Department of Nutrition Agriculture Service May 2004 Special Nutrition Programs Report No. CN-04-CFABT

Measuring Competitive Foods in Schools: An Inventory Approach

Authors:

Mary Kay Crepinsek Nancy Cole Joanna Golding

Submitted by: Abt Associates, Inc. 55 Wheeler Street Cambridge, MA 02138 Submitted to:

Office of Analysis, Nutrition and Evaluation USDA, Food and Nutrition Service 3101 Park Center Drive 10th Floor Alexandria, VA 22302-1500

Project Director: Mary Kay Crepinsek

Project Officer: Patricia McKinney

This study was conducted under Contract number 53-3198-2-026/43-3198-3-3705 with the Food and Nutrition Service.

This report is available on the Food and Nutrition Service web site: http://www.fns.usda.gov/oane.

Suggested Citation:

U.S. Department of Agriculture, Food and Nutrition Service, Office of Analysis, Nutrition and Evaluation, Measuring Competitive Foods in Schools: An Inventory Approach. Mary Kay Crepinsek, Nancy Cole, Joanna Golding. Project Officer, Patricia McKinney. Alexandria, VA: 2004.

Acknowledgments

The Feasibility Study to Develop a Methodology to Monitor the Impact of Changes in Competitive Food Policies was a fast-paced, six-month research project that required assistance from three schools. The authors are indebted to all of the school staff who provided information for the study, including cafeteria managers and other school food service staff, teachers, principals, and School Food Authority directors. We also greatly appreciate the efforts of the SFA directors, principals, and state Child Nutrition directors who helped facilitate the collection of information in their sites. Without the cooperation and hard work of all of these individuals, the study could not have been completed.

The authors also wish to recognize other staff members at Abt Associates Inc. who played important roles on the project. Neil Berquist entered survey data, and helped research the nutrient composition of new foods. Patty Connor led the coding and nutrient analysis effort, and contributed to the final report. Ellie Lee created the analytic database, and Lisa Litin conducted the food and nutrient analyses. Susan Byers Paxson was responsible for production of all study documents. Gail Langeloh, a nutritionist and independent consultant, participated in the design, implementation, and analysis phases of the study. She also contributed to the final report.

Pat McKinney served as the federal project officer for the Office of Analysis, Nutrition, and Evaluation, Food and Nutrition Service. She and other reviewers at FNS provided valuable input throughout the study.

Contents

Executiv	e Summary	1
Ba	ickground	1
Pu	rpose of the Study	1
Da	ata Collection and Analysis Plan	2
Ke	ey Findings for Three Schools	
	Implementation of the Data Collection Plan	
	Competitive Food Policies and Practices	4
	Competitive Foods Offered and Sold	5
Pr	oposed Methodology for Examining the Impact of Changes in Competitive Foo	od Policies 7
Co	onclusions	
Chapter	1: Introduction	9
Ov	verview of the Study	
	Study Goals and Objectives	
	Schools Recruited for the Study	
Or	ganization of the Report	13
Chapter	2: Design and Implementation of the Data Collection Plan	
Da	ata Collection Instruments	
	Principal Survey and SFA Director Survey	
	School Foods Data Collection Instruments	
Da	ata Collection Procedures	
	Advance Meetings with School Staff	
	Training and Technical Assistance	
	Survey Completions	
	State Director Interviews	
	Debriefing Meetings	
	Data Preparation and Data Entry in Nutrient Database	
Da	ata Collection Issues	
	Overall Burden	
	Data Collection Period	
	Identifying Respondents	
	Tracking Number of Items Sold in Vending Machines	
	Designing Cafeteria Instruments to Collect Data on All Foods	
Su	mmary	
	Findings from Data Collection	
	Limitations	
Chapter	3: Competitive Foods in the School Environment: Policies and Practices i	in Three
Ea	od Venues	
гu	Food Venues Available in School Cafeterias	
	Food Venues Available Outside Cafaterias	50 ۱۸
٨٠	vailability of Food and Reverages by Time of Day	40 / 41
Po Do	licies About Types of Food Available	
10	110103 110001 1 ypts 01 1 000 rivaliaolo	

Operation and Administration of Food Venues	43
Operation of Food Venues Available During the School Day	43
Operation of Food Venues Available Outside the School Day	45
Overall Administration of Food Venues	46
Exclusive Beverage Contract	46
SFA Pricing Strategies for Competitive Foods	47
Other Sources of Food	49
Off-Campus Eating Establishments	50
Nutrition Initiatives	51
Summary	53
Chapter 4: Competitive Foods in the School Environment: Nutritional Characteristics	55
Foods Offered and Sold	56
Number and Types of Food Offered	57
Foods Most Commonly Sold	60
Nutrient Composition of Competitive Foods	63
Food Energy and Macronutrients	64
Micronutrients	67
Food Guide Pyramid Servings of Competitive Foods	70
Foods Meeting FDA Labeling Criteria	72
Foods of Minimal Nutritional Value	74
Summary	77
Unweighted Tables	79
Weighted Tables	97
Chapter 5: Methodology for Assessing the Impact of Competitive Food Policies	113
Overall Design Considerations	113
Cross-Sectional Design	114
Longitudinal Design	115
Measuring Competitive Food Policies	116
Identifying the Respondent	116
Defining Food Sources	117
Lack of Clear Policies	118
Measuring Nutritional Outcomes	119
Analyzing Nutritional Outcomes	120
What Is the Impact of Changes in Competitive Food Policies on the Nutritional	
Quality of Competitive Foods Available at School?	121
Do Competitive Foods "Crowd Out" the Selection of USDA Reimbursable Meals?.	121
How Much Do Competitive Foods Add to (or Subtract from) Total Food Energy,	
Key Nutrients, Added Sugars, and Fat Served to Students During the School	
Day?	122
Conclusions	125
References	127
Appendix A: Instrument Design and Data Quality Issues	. A-1
Principal Survey	. A-1
Data Quality	. A-1

Item Nonresponse	
Instrument Design Issues	
Recommendations	
SFA Director Survey	
Data Quality	
Item Nonresponse	
Instrument Design Issues	
Recommendations	
Cafeteria Survey	A-4
Data Quality	
Item Nonresponse	A-5
Instrument Design Issues	A-6
Recommendations	A-7
Inventory of Snack Bar Prepackaged Food and Beverage Items	A-7
Data Quality/Item Nonresponse	A-7
Instrument Design Issues	A-7
Inventory of School Store Food and Beverage Items	
Data Quality	
Item Nonresponse	
Inventory of Vending Machine Items	
Data Quality	
Item Nonresponse	A-9
1	
Appendix B: Validation Study Findings	B-1
Collection of Validation Data	
A la Carte-Only and Snack Bar Items	B-1
School Store Items	B-2
Validation Results	B-2
Appendix C: Nutrient Database Issues	
Entry Procedures	
Entry Issues	
Brand Name and Flavor Specificity	
Missing Foods	C-3
Ice Cream Amounts	C-5
Comparison of FIAS Nutrients with Replacement Values.	C-5
Summary of FIAS Entry Issues	C-6
Comparison of FIAS Data to Label Nutrition Information	C-8
Appendix D: Reimbursable Meal Menus	D-1
Appendix E: Competitive Foods Available During the School Day	
rr · · · · · · · · · · · · · · · · · ·	_ _
Appendix F: Competitive Foods Available Outside the School Day	
Appendix G: Instruments	

Executive Summary

Background

The present epidemic of overweight and obesity among children has focused increased attention on the foods available to children at school. The U.S. Department of Agriculture (USDA) sets standards for the nutritional quality of meals served daily to 28 million children as part of the National School Lunch Program (NSLP) and to 8 million children as part of the School Breakfast Program (SBP). In a large number of schools, however, foods and beverages are available to students as alternatives to the federally reimbursed meals, as a la carte items in school cafeterias; as items from vending machines, snack bars, and school stores; or as part of special profit-making events (Wechsler *et al.*, 2001). These "competitive foods" tend to be high in calories, fat, and added sugars, and low in essential nutrients. Thus, the availability of competitive foods may undermine the nutritional goals of the USDA school meal programs.

Under current regulations, USDA has only limited ability to influence foods that compete with reimbursable meals. Regulation 7 CFR 210.11 prohibits the sale of foods of minimal nutritional value, including beverages, in school food service areas during meal periods. USDA is unable, however, to regulate the availability and nutritional quality of competitive foods outside of the school food service area and outside of meal periods. As a result of this lack of regulatory power, recent efforts to influence competitive foods have come in the form of initiatives to encourage changes in the school nutrition environment, and a proposal to Congress to offer financial incentives to schools that offer healthful food options in competitive food venues. Some states have passed legislation to limit the types of food that can be sold at school.

The availability of competitive foods in schools has changed over time and will continue to change in response to calls for more healthful food options for schoolchildren. Currently, USDA lacks a mechanism for monitoring these changes. With an interest in monitoring these changes, the USDA Food and Nutrition Service (FNS) contracted with Abt Associates Inc. to design and test a data collection plan for obtaining information on competitive food policies and the nutritional characteristics of all foods offered and served in schools. That field experience is described in this report, along with analyses of collected data and recommendations for a methodology for use in ongoing monitoring of the impact of changes in competitive food policies.

Purpose of the Study

The Feasibility Study to Develop a Methodology to Monitor the Impact of Changes in Competitive Food Policies had two main goals:

- 1. Design and test a plan for collecting and analyzing data about competitive food policies and all foods offered and served to students at school; and
- 2. Provide a methodology for assessing the impact of changes in competitive food policies on the nutritional quality of all foods served at school.

To achieve the first goal, data collection instruments and procedures were developed and data were collected from three schools during May of 2003. The study collected data about the availability of

competitive foods (locations and time periods), school nutrition policies, and financial arrangements related to the availability of competitive foods, and descriptions of all food and beverage items offered and served to students during the school day. Data about competitive foods were entered in a nutrient analysis system, and food group and nutrient analyses of these items were conducted.

Lessons learned from the data collection and analyses were then used to address the second goal of the study. Recommendations include a description of overall design considerations, methodological and operational issues involved in collecting information about competitive food policies and competitive foods offered and served, alternative outcome measures, and methods of analysis.

Data Collection and Analysis Plan

This feasibility study collected data from three secondary schools purposively selected from two states (Massachusetts and Virginia). The schools included two high schools and one middle school, all having multiple food venues. In addition to cafeteria service, all three schools have vending machines, one high school has a snack bar, and both high schools have school stores.

Information about the school nutrition environment and school nutrition policies was collected through mail surveys of principals and SFA directors. The Principal Survey collected data about the availability of competitive foods (number and types of food venues, locations, times of availability); operation and administration of food venues; school policies about competitive foods; off-campus eating; use of food for fundraising; and nutrition initiatives. The SFA Director Survey collected information about a la carte service and pricing strategies; food service-operated vending machines; district policies about competitive foods; and exclusive beverage contracts. Both surveys also asked about the amount of profit earned and use of profits from competitive food sales.¹

Data on reimbursable school meals and all competitive foods offered and served to students was collected through food lists and inventories completed by cafeteria staff and other school personnel who operate food venues. For each food and beverage item, data were collected on manufacturer, brand, and name of the item; package or serving size; recipe, if prepared from scratch; and the number served or sold during the data collection week.

Most of the burden of collecting data for both reimbursable meals and competitive foods fell on the cafeteria staff. Several strategies were implemented to reduce this burden, including the introduction of an "inventory approach" for tracking food selections and sales. The inventory approach was used for prepackaged foods and beverages. Cafeteria staff simply counted the number of cases and individual items on hand at the beginning and end of the data collection week, and they recorded deliveries. In addition, data collection forms allowed for counts of total food servings, rather than requiring separate tracking of servings in reimbursable meals *versus* a la carte.

The inventory approach was well suited to the tracking of sales of individual items in vending machines stocked by school staff. It was unclear, however, how best to obtain these data for machines operated and stocked by outside distributors. Data collection, therefore, was attempted by

¹ The Principal and SFA Director Survey data were revised and supplemented, based on information obtained by the researchers during on-site training and debriefing visits to the schools.

multiple methods: SFA directors and principals were asked for available invoices, study forms were sent to distributors, and researchers contacted distributors directly.

All respondents were compensated for the time spent on data collection activities.

Analysis of data on foods offered and served was limited to competitive foods and beverages available "a la carte only", and in snack bars, vending machines, and school stores available to students during the school day.² Mean food energy, nutrients, and servings from each of the five main Food Guide Pyramid food groups were calculated using the USDA Survey Nutrient Database and Pyramid Servings Database, respectively. Nutritional characteristics were reported for broadly defined groups of food—baked goods/desserts, beverages, bread/grain products, candy, frozen desserts, salads, and snacks—as well as for subgroups (e.g., cookies, fruit juice, pretzels).

Key Findings for Three Schools

This feasibility study was designed to test a data collection and analysis plan. Findings include lessons learned from implementation of the data collection plan, as well as a demonstration of the types of qualitative and quantitative data that can be obtained about school nutrition policies and the nutritional characteristics of competitive foods available during the school day. All results are suggestive and cannot be generalized to a larger population due to the small sample and purposive selection of schools in the study.

Implementation of the Data Collection Plan

Principal and SFA Director Surveys

- SFA directors found the survey easy and were motivated to provide complete surveys.
- School principals did not have knowledge of all food operations within their schools, and did not have time to investigate the answers to survey questions.
- Respondents to both surveys sometimes misinterpreted the questions about competitive food policies, due to a lack of formal policies at the SFA and school levels.

Food Lists and Inventories

- Schools used different methods of food delivery to sell similar foods. For example, the same items sold in cafeteria vending machines at one school were offered on the a la carte line at another school. As a result, all competitive foods offered in the cafeteria were grouped together for nutrition analyses.³
- It was feasible to use an inventory approach to collect data on prepackaged foods and beverages. The approach was well accepted by school staff because it was straightforward and moved data collection activities outside of meal periods. Validation data

² Vending machines not available during the school day were excluded from analysis, and foods served a la carte and also in reimbursable meals were excluded.

³ The alternative was to group foods by sources of competitive foods, such as a la carte, snack bars, and vending machines. We concluded that cafeteria foods should be grouped together because different methods of delivery might be primarily related to space and staffing constraints. Foods available outside the cafeteria, such as in school stores, were grouped separately.

collected for one day during the data collection period indicated that school staff accounted for 95 percent of all competitive food items offered.

- Collection of data on **all** foods offered and served in both reimbursable meals and competitive food venues imposed significant burden on the cafeteria managers/staff. We attempted to obtain estimates of this burden, but the data were not of high quality.
- Reporting of servings data was a very burdensome task, regardless of whether the task was implemented using the traditional daily servings approach or a weekly inventory approach. This burden introduced potential bias due to (a) reporting of "portions prepared" rather than "portions served" for reimbursable meal components, (b) changes in stocking procedures, and (c) missing data from food venues operated by non-food service organizations.
- Data on foods offered and sold in vending machines operated by non-school personnel (i.e., machines filled by distributors) were difficult to obtain. Invoices did not contain sales information for individual items, and only one of four distributors contacted directly responded to requests for data.⁴

Competitive Food Policies and Practices

Availability of Competitive Foods

- More competitive food items were offered in school cafeterias (a la carte, snack bar, vending machines) than in venues located outside the cafeteria. (The only food venues outside the cafeterias and available during the school day were school stores.)
- Competitive foods were available during lunch periods in all three schools.
- School stores and vending machines in the cafeteria offered competitive foods before classes in the two high schools. None of the schools provided a la carte cafeteria service during the breakfast period.
- Carbonated soft drinks were not available during the school day. School stores sold candy and other "foods of minimal nutritional value" in the high schools, but were open for limited time periods.

Operation of Competitive Food Venues

- The school food service department was responsible for all of the competitive food venues in the cafeterias (a la carte, snack bar, vending machines).
- In the high schools, academic clubs and athletic departments ran the food venues located outside the cafeteria and available during school (school stores), as well as those available only after school (vending machines).
- Purchasing and stocking for one or more vending machines in all three schools was handled by outside distributors (Coca-Cola, Pepsi, and smaller local companies).

⁴ It is important to note that three of the distributors operated vending machines only available after school. The missing vending machine data from the fourth distributor affected nutrition analyses for one machine in only one school (the middle school).

Policies and Practices Affecting Students' Food Choices

- Other than federal and state regulations prohibiting the sale of foods of minimal nutritional value, nutrition standards or criteria were not applied to competitive foods, and there were few restrictions or requirements on the types of foods sold to students.
- Pricing and other strategies were used to discourage purchase of a la carte items in lieu of a full reimbursable meal. For example, one high school priced the entrée-plus-milk the same as the full-price reimbursable meal. The middle school only allowed students to purchase a la carte items if they had purchased a full meal or brought a lunch from home.
- Two schools reported conducting promotional activities to encourage healthful food choices.

Competitive Foods Offered and Sold⁵

Number and Types of Competitive Foods Offered

- The number of available competitive food items varied widely between schools: 23 items were offered at the middle school; 43 and 118 items were offered at the high schools.
- Beverages, including juice-based drinks, 100 percent fruit juice, flavored milk, and spring water, accounted for the largest share of competitive foods offered in all three school cafeterias (30 to 48 percent).
- Snacks, primarily chips and popcorn, comprised another one quarter of the competitive items offered in the cafeterias (24 to 26 percent).
- Candy and snack items were the most commonly offered foods outside the cafeteria (in school stores).

Number and Types of Competitive Foods Sold

- In the two high schools, the average daily number of competitive foods sold per student was 0.87 and 0.66. (In other words, if students purchase only one competitive food item each day, then 87 percent and 66 percent of students purchased a competitive food item.)
- In two of the three schools, chocolate chip cookies, fruit punch, spring water, and tortilla chips were among the top five selling competitive food items in the cafeterias during the data collection week.
- Fruit punch, spring water, and "penny candy" were the most popular items sold outside the cafeteria.

⁵ Excludes items in reimbursable meals that are also available a la carte.

Food Energy and Nutrients in Competitive Foods Offered and Sold⁶

- The food energy content of competitive food items for the three schools ranged from zero (spring water) to over 400 calories (snack cakes and 14- to 16-ounce whole or 2 percent flavored milk).
- Baked goods and snacks, as offered, provided the highest mean amounts of calories and total fat. The average baked good contained 12 to 18 percent of the 1989 Recommended Energy Allowance (REA) for schoolchildren, and 25 percent of the Daily Reference Value for total fat.
- Beverages and bread/grain products (e.g., cereal bars, pretzels, rice cakes) were more likely than other competitive foods to meet Dietary Guidelines-based benchmarks of 30 percent or less of calories from total fat and less than 10 percent of calories from saturated fat.
- The average calorie and fat content of beverages and snacks "as selected" was somewhat lower than the average "as offered." Students were not as likely to buy milk as juice drinks and bottled water, and tended to purchase chips in smaller *versus* "big grab" packages.
- As a group, beverages offered the most significant amounts of vitamin A, vitamin C, and calcium relative to other competitive foods (6, 100 and 20 percent of Reference Daily Intake, respectively). Mean amounts of these nutrients were lower, however, after taking into account students' relative purchases of milk, juice-based beverages, and water.
- The mean vitamin and mineral content of competitive foods was influenced by the availability of highly fortified items, including milk, juice-based drinks, cereal bars and mixes, toaster pastries, and some candy.
- In the two high schools, the average daily food energy provided by the sale of competitive foods was found to be 141 calories per student, when averaged over all students in attendance.⁷ The average daily caloric contribution of competitive foods was equal to 5.6 percent of the 1989 REA. In combination with reimbursable meals, competitive foods raised average daily food energy per student by 17 percent (relative to the NSLP standard) or 18 percent (relative to findings from the School Nutrition Dietary Assessment Study-II).

Food Guide Pyramid Servings in Competitive Foods Offered and Sold

- Few foods offered in the school stores made a substantial contribution to dailyrecommended servings of foods from the main Pyramid food groups.
- Baked goods, bread/grain products, and snacks offered an average of 1 to 1³/₄ servings from the grain group.

⁶ Data presented for foods "as offered" were obtained by computing a simple mean (each competitive food item weighted equally). Data for competitive items "as selected" or "as sold" were weighted by servings or sales to students, within school, with schools weighted equally.

⁷ A similar estimate for the third school in the study could not be obtained due to missing data on the number of items sold from vending machines.

- Among the beverages offered, milk and 100 percent fruit juice offered close to two servings from the dairy and fruit groups, respectively. As selected, however, the mean servings of fruit and dairy were considerably lower (less than ½ serving each).
- Although baked goods and beverages contributed positively to the grain, fruit, and dairy groups, they also contained the highest mean amounts of added sugars (7 teaspoons).

Proposed Methodology for Examining the Impact of Changes in Competitive Food Policies

There are two reasons to monitor competitive foods available in schools: to assess the nutritional quality of available foods, and to evaluate the impact of changes in competitive food policies. One goal of this feasibility study was to recommend a methodology for evaluating the impact of changes in competitive food policies, informed by the lessons learned from implementation of the data collection plan.

The impact of changes in competitive food policies may be evaluated by the outcome measures listed below.

- 1. Changes in the nutritional quality of competitive foods available at school.
- 2. Changes in the percent of students receiving reimbursable meals. (In other words, do competitive foods "crowd out" the selection of USDA reimbursable meals?)
- 3. Changes in the contribution of competitive foods to total food energy, key nutrients, added sugars, and fat served to students during the school day.

Measuring the impact of changes in competitive food policies, however, requires longitudinal data on both policies and competitive foods offered and sold at individual schools. The current feasibility study was not designed to measure the impact of changes in competitive food policies; data were collected from only three schools at a single point in time. Instead, the study demonstrated the measurement of "baseline" outcomes (presented above under "Findings").

One of the key recommendations of this study is that the impact of changes in competitive food policies should be assessed by a focused collection of data on competitive foods, rather than collecting data on **all** foods offered and served during the school day, as was done for this study. This recommendation is based on both implementation and methodological concerns.

- *Implementation issues:* Collection of data on **all** foods offered and served during the school day (competitive foods and reimbursable meals) was found to pose a large burden on respondents. In contrast, limiting data collection to competitive foods reduces both the numbers and types of foods included in the data collection. Competitive foods consist largely of prepackaged foods and beverages, and this study was successful in implementing a newly developed inventory approach to collect data on prepackaged items.
- *Methodological issues:* This study demonstrated that the contribution of competitive foods to total energy served to students during the school day can be assessed using readily available benchmarks (and the same is true for other key nutrients). Two

alternative benchmarks are (1) the National School Lunch Program (NSLP) standards for reimbursable meals, and (2) the findings of the School Nutrition Dietary Assessment Study (SNDA-II).

The main limitations of the proposed approach are:

- It is not possible, using the inventory approach, to identify separately competitive food sales during the school day and sales after the school day, for venues available in both time periods.
- It is difficult to obtain data from vending machines operated by non-school personnel.
- By focusing data collection on competitive foods, rather than collecting data on all foods available at school, it is not possible to investigate the relationship between the availability of competitive foods and the nutritional quality of reimbursable meals, if this interaction exists.

Conclusions

This feasibility study developed and implemented a plan for collecting data to describe school nutrition policies and the nutritional characteristics of all foods offered and served to students during the school day. Findings indicate that, for the three schools surveyed, there is little in the way of formal policies about competitive foods at either the school or district levels. If this is more generally observed, future monitoring efforts should assess observable practices that could be verified by an outside observer. Alternatively, future surveys should allow for open-ended narrative responses so that the distinction between policy and practice can be determined. It may also be preferable to rely primarily on SFA directors and cafeteria managers for information on competitive food practices, rather than principals. They were the most knowledgeable respondents in this study, and were motivated to provide complete surveys.

Despite the introduction of some innovations in the data collection methodology and use of incentives, the burden of collecting information simultaneously on both reimbursable meals and competitive foods was problematic for school food service staff. This burden raises questions about the quality of the data that can be obtained on all school foods (for purposes of nutritional analysis) in a self-administered survey. On the other hand, this study demonstrated that the inventory approach was well accepted and produced high quality data on prepackaged competitive foods. Thus, one of the key outcomes of this study is the recommendation that the impact of changes in competitive food policies be assessed by focusing data collection on only those competitive foods offered outside of the reimbursable meal programs.

Three main limitations of the recommended data collection methodology were identified: (1) missing data from food venues not operated by school personnel, (2) the inability to distinguish between competitive food sales during and after the school day, and (3) the inability to identify interactions between competitive food availability and the nutritional quality of reimbursable meals. To address these limitations, future studies might request that respondents collect data on vending machine items and sales from vendors during stocking, and investigate the possibility of collecting data over a period without any after-school events. To reduce the burden of collecting servings data for all foods, on-site assistance for school food service staff at the points of sale may be necessary. These and other potential solutions could be investigated further within a larger study design.

Chapter 1 Introduction

The U.S. Department of Agriculture (USDA) sets standards for the nutritional quality of reimbursable meals served to schoolchildren.¹ Aside from reimbursable meals, however, USDA has limited ability to influence food and beverages available to K-12 students during the school day. In a large number of schools, foods are available to students as alternatives to reimbursable meals, as a la carte items in school cafeterias; items from vending machines, snack bars, and school stores; or as part of special profit-making events. Some students may leave the school during meal periods and obtain food from off-campus eating establishments. Additionally, one half of all school districts contract with companies that buy rights to sell soft drinks at schools in the district (Wechsler *et al.*, 2001).

The current epidemic of overweight and obesity among children, and the goals of Healthy People 2010,² has focused increased attention on the foods available to children during the school day. One concern is that the nutritional goals of the USDA school meal programs may be undermined by the availability of "competitive foods".³ Although the impact of competitive foods on the overall nutritional quality of schools foods is unknown, the following statements pertain: (1) competitive foods are frequently of lower nutrient density (i.e., low in nutrients relative to food energy content) than components of a reimbursable meal; (2) their availability may send a conflicting message to students about the value of healthful foods; and (3) they divert student participation in, and thus income from, the reimbursable meal programs, jeopardizing their financial status and ability to make improvements (Cline and White, 2000; USDA, 2001).

Only one tool is currently available to USDA to reduce the influence of foods that compete with reimbursable meals. The requirement of Section 10(a) of the Child Nutrition Act (42 U.S.C. 1799(1)) directs USDA to regulate the service of foods in competition with reimbursable meals. This regulation (7 CFR 210.11) prohibits the sale of foods of minimal nutritional value (FMNV), including beverages, in the food service area during school meal periods. Federal regulations allow states and local School Food Authorities (SFAs) to impose additional restrictions.⁴ USDA is not, however, currently able to regulate the availability and nutritional quality of competitive foods available to students outside of the school food service area and outside of meal periods. As a result of this lack

¹ In 1995, the Department launched the School Meals Initiative (SMI), which includes nutrition standards for the National School Lunch Program (NSLP) and School Breakfast Program (SBP). The standards call for school meals to provide on average, one third of students' daily nutrition needs for food energy and key nutrients at lunch and one fourth at breakfast. They also include goals for fat and saturated fat that are consistent with the *Dietary Guidelines for Americans* recommendations (USDHHS/USDA, 1995).

² Healthy People 2010 is a set of health goals and objectives developed under the direction of the Secretary of Health and Human Services. Healthy People 2010 Objective 19.15 focuses specifically on the school environment, calling for USDA to work to "increase the proportion of children and adolescents, ages 6 to 19 years, whose intake of meals and snacks at school contributes proportionally to good overall dietary quality" (USDHHS, 2000).

³ USDA defines competitive foods as any "foods offered at school, other than meals served through USDA's school meal programs—school lunch, school breakfast, and after-school snack programs" (USDA, 2001).

⁴ Information compiled by FNS in September 2002 indicates that competitive food policies in 32 of 50 states (64 percent) are the same as the USDA regulation, without additional restrictions. Eighteen states specify additional provisions that further restrict the sale of and/or income from all foods sold in competition with reimbursable meals (USDA, 2002).

of regulatory power, recent efforts to influence competitive foods have come in the form of initiatives to encourage changes in the school environment.

To foster healthy school nutrition environments, USDA launched two initiatives that support the development of healthful eating habits among schoolchildren. *Changing the Scene: Improving the School Nutrition Environment* and *Eat Smart. Play Hard* provide resources for schools to promote healthy eating and physical activity (USDA, 2000). USDA (Team Nutrition) is also collaborating with the Centers for Disease Control (CDC) to develop materials for schools to use in improving the school nutrition environment. At the same time, states are working to implement the Action Goals of the national Action for Healthy Kids initiative (AFHK, 2003).

To further promote a healthy school food environment, USDA recently proposed to offer financial incentives to schools to "offer healthful food options in vending machines, school canteens, and their a la carte menu service." In recent testimony to Congress, USDA Undersecretary Bost described this USDA proposal to "establish a Healthy School Environment that supports the President's HealthierUS and No Child Left Behind initiatives" (USDA, 2003). The proposed program recognizes the importance of influencing competitive foods in schools. Undersecretary Bost described an incentive system based on the four keystones of HealthierUS:

- Nutrition—Eat a nutritious diet;
- Physical fitness—Be physically active each day;
- Prevention—Get preventive screening; and
- Avoid risk behaviors—Make healthy choices.

The current feasibility study is related to recent initiatives to improve the school food environment. Information about the availability and nutritional quality of competitive foods in schools is needed by USDA, states, and policymakers to evaluate *Changing the Scene* and other initiatives and policy changes regarding competitive foods. Furthermore, in working to put an incentive system into operation, FNS will need data to inform the development of criteria for identifying healthful food options, and a methodology to monitor changes in the school environment.

Overview of the Study

The feasibility study⁵ addresses USDA's interest in developing an approach that could be used on an ongoing basis to monitor the impact of changes in competitive food policies. This study designed and tested a data collection plan for obtaining information on competitive food policies and on all foods offered and served in schools. That field experience is described in this report, along with analyses of collected data. Based on the findings of the field experience, recommendations are presented for a methodology for use in monitoring the impact of changes in competitive food policies.

⁵ The study was referred to as the "Study of Foods in the School Environment" on all correspondence and instruments completed by respondents.

Study Goals and Objectives

This feasibility study had two main goals:

- 1. Design and test a plan for collecting and analyzing data about competitive food policies and all foods offered and served to students at school; and
- 2. Provide a methodology for assessing the impact of changes in competitive food policies on the nutritional quality of all foods served at school.

To achieve the first goal, the study developed data collection instruments and procedures, and collected data from three schools during May 2003. Data collection included mail surveys of school principals and SFA directors, and self-administered school food lists and inventories completed by cafeteria and other school staff for a one-week period. Abt Associates staff provided on-site training and technical assistance, collected one day of data for validation purposes, and conducted in-person debriefing meetings with respondents. Issues were identified and analyses conducted on competitive foods available to students **during the school day**. Although foods may be sold on school property after school hours or during weekend sporting or other events, they were not the focus of this study.⁶

The data collection and analysis plan met the following specific objectives:

- Provide information on the locations and times competitive foods are available, school food policies, and financial arrangements related to the availability of competitive foods.
- Conduct food group and nutrient analyses for each source of competitive foods (a la carte only,⁷ snack bars, school stores, vending machines).
- Identify the potential problems and issues surrounding the collection and analysis of data using the data collection instruments designed for this study.

Lessons learned from the implementation of the study design were then used to address the second goal of the study: provide recommendations for a methodology for analyzing the impact of changes in competitive food policies. These recommendations include a description of overall design considerations, methodological and operational issues involved in collecting information about competitive food policies and competitive foods offered and served, alternative outcome measures, and methods of analysis.

Schools Recruited for the Study

Three schools were recruited to participate in the study; they were not randomly selected, and are not representative of any larger population of schools. Table 1.1 shows the characteristics of the three schools that participated in the study. The three schools are located in Massachusetts and Virginia.⁸

⁶ There is one exception—data were collected about foods and beverages sold in vending machines only available after school in one of the participating schools. Information on these items is presented in an appendix to this report.

⁷ The analyses of a la carte foods in this study were limited to those foods **not** also included in reimbursable meals; these items are referred to as "a la carte-only" foods.

⁸ Schools in Massachusetts were targeted for the study so that data collection would occur in close proximity to Abt headquarters in Cambridge. One school in Virginia was included to provide a site in close proximity to FNS headquarters so that the project officer could participate in debriefings.

Massachusetts and Virginia provided different environments, in terms of regional food preferences and also in terms of the state policy for competitive foods in schools. All states operate within the USDA policy limiting the availability of FMNV in the school cafeteria during school meal periods. Massachusetts has not supplemented the USDA policy. Virginia expanded the USDA policy to restrict the sale of FMNV anywhere in the school during school meal periods.

Table 1.1

School	1	2	3
Location	Massachusetts	Massachusetts	Virginia
Grade span	6-8	9-12	9-12
Total enrollment	570	3,626	1,518
Percent free/reduced	89%	41%	7%
Competitive foods:			
A la carte foods (including meal components)		Yes	Yes
A la carte foods (non- reimbursable items)	Yes		Yes
Snack bar		Yes	
School store		Yes	Yes
Vending machines (outside contract)	Yes	Yes	Yes
Vending machines (food service)		Yes	Yes
Cafeteria checkouts	POS (LunchBox)	POS (LunchBox)	POS (Café Terminal)
Menu planning system	Assisted Nutrient	Traditional Food-	Enhanced Food-Based
	Standard Menu Planning	Based	
On-site food preparation	Yes	Yes	Yes
Number of cafeterias	1	2	1
Food service management	Yes		
company			
School breakfast (SBP)	Yes (universal-free)	Yes	No

Schools Recruited for the Feasibility Study

The three schools include one middle school and two high schools. The focus on secondary schools was intended to provide more complex settings for testing the feasibility of data collection relative to elementary schools. The schools vary in size, the percent of students with free and reduced price lunch eligibility, the type of menu planning system, and the number of competitive food sources. All schools have on-site kitchens, serve a la carte foods, and have vending machines; one school has a snack bar, and two have a school store.

Operation of the vending machines varies. Two of the schools have some vending machines operated by the school food service and some by contract with an outside distributor. The school food service

in the third school does not operate any vending machines. All three schools use POS systems; two serve breakfast, with one school serving universal free breakfast.

Organization of the Report

The remaining chapters in this report present the following information:

- Chapter 2 describes the design and implementation of data collection instruments and procedures, and summarizes the main data collection problems and successes.
- Chapter 3 presents the results of the surveys of competitive food policies and practices.
- Chapter 4 presents analyses of the nutritional characteristics of competitive foods offered and sold in the three schools.
- Chapter 5 provides a discussion of the operational and methodological issues involved in collecting data and analyzing the impact of changes in competitive food policies. Recommendations, informed by the findings of this feasibility study, are provided for large-scale monitoring.

Appendices provide additional analyses, detailed tables, and the data collection instruments.

Chapter 2 Design and Implementation of the Data Collection Plan

The main charge for this study was to develop data collection instruments and methods to test the feasibility of capturing two types of information:

- 1. Information about student access to competitive foods in the school environment, and
- 2. Information about the types and nutritional characteristics of foods offered and sold to students—both within and outside school cafeterias.

Instruments and procedures used for previous studies of the school meal programs provided a starting point for the study design, but prior studies did not collect quantitative information about foods available at school in addition to reimbursable meals.¹

This chapter describes the instruments and procedures used for the study, the data collection issues that were anticipated and addressed during the design and implementation of the data collection plan, and findings from the fielding of the instruments. Instrument design and data quality issues, as well as some recommendations for future testing, are discussed in Appendix A.

Data Collection Instruments

The instruments developed for this study are described in Table 2.1. The Principal Survey and SFA Director Survey collected data about the school food environment and policies related to the availability of competitive foods during the school day. The Cafeteria Survey contained seven data collection instruments to collect data on all foods served in the school food service area, including reimbursable meals and a la carte foods. The "other competitive foods" instruments collected data about foods and beverages available from alternative food sources that may be located anywhere on school property.

¹ Instruments to collect detailed food descriptions, portion sizes, and numbers of items sold in reimbursable meals and a la carte were developed and pre-tested by Abt Associates staff for the second School Nutrition Dietary Assessment Study (SNDA-II). The time required to collect these data was prohibitive, especially in secondary schools, when combined with the same task for reimbursable meals. As a result, detailed data collection for a la carte foods was dropped from the study.

Table 2.1

Data Collection Instruments

Category / Instrument		Topics Description		
Pr	incipal Survey	Competitive food availability (locations, hours, types of foods); school policies for competitive food; off-campus eating; use of food for fundraising; nutrition initiatives.		
SFA Director Survey		A la carte food service; food service-operated vending machines; district policies for competitive foods; exclusive beverage contracts; menu planning; pricing strategies for competitive foods.		
C a 1.	a feteria Survey Daily Meal Counts Form	Reimbursable meal counts for each day, by meal price		
2.	Every Day Cafeteria Foods Form	Foods served every day : full description of food item and either the count of servings per day (reimbursable and a la carte) or check that food is on the Inventory of Prepackaged Foods or the Self-Serve Bar Form. ^a		
3.	Cafeteria Foods Form	Foods not served every day: full description of food item and count of servings per day (reimbursable and a la carte), or check that food is entered on the Self-Serve Bar Form.		
4.	Inventory of Every Day Prepackaged Foods	Prepackaged foods served daily and requiring no preparation, including prepackaged condiments. Full description of food item, starting inventory, deliveries, and ending inventory.		
5.	Self-Serve Bar Form	Salad bar and other self-serve food bar items (e.g., condiment bar, sandwich bar): full description of item and portion size (if pre-portioned) or starting amount, amount added, and amount leftover, for one day.		
6.	Recipe Forms	For foods prepared from scratch or by combining two or more foods or ingredients, and listed on the Cafeteria, Every Day Foods, or Self-Serve Bar Forms. Full description of ingredients, amount used, servings prepared, and preparation methods.		
7.	Food Outside Meal Periods Checklist	List of all foods and beverages sold in the cafeteria outside of the breakfast and lunch meal periods during the target week. Includes the times of day the items are available to students.		
Other competitive foods forms				
1.	Inventory of Prepackaged Snack Bar Items	Description and inventory of all prepackaged foods and beverages requiring no preparation and sold at snack bar. ^b		
2.	School Store Food and Beverage Inventory Form	Description and inventory of all foods and beverages sold at school store.		
3.	Vending Machine Inventory Form	Description and inventory of all foods and beverages sold in vending machines.		
a	If either the inventory or self-serve forms a foods forms.	re used, the respondent need only list the name of the food on the cafeteria		

b For snack bars offering prepared items, a snack bar foods form is used to record counts of servings.

Principal Survey and SFA Director Survey

The **Principal Survey** and **SFA Director Survey** were designed as mail surveys with telephone follow-up. The surveys have a similar structure, with a series of questions about each source of competitive foods (a la carte, vending machines, snack bar/canteen/food cart, school store, other sources). The instruments contain skip patterns so that, for the most part, the SFA Director responded to questions about food service-operated venues while principals responded to questions about nonfood service-operated venues. Both respondents, however, were asked about a la carte foods, because a la carte policies may vary within SFA.

The format of questions about competitive food sources differed slightly on the two instruments. Principals were asked about competitive food policies and practices "at your school"; SFA directors were asked about district-level competitive food policies and practices, and the types of schools for which they apply (e.g., elementary, middle, or high schools). In addition, principals were asked about school policies and practices (mealtime policies, school initiatives, etc.), and SFA directors were asked about SFA operations (a la carte pricing strategies, availability of nutrition information, etc.). The main data elements collected in the Principal and SFA Director Surveys are listed in Table 2.2.

School Foods Data Collection Instruments

The school foods forms were used to collect data about all foods offered and served, during the school day, for a one-week period (the "target week"). These forms have two basic structures:

- 1. Servings approach (designed for prepared foods). This type of form is used to describe and count the number of items selected by students on a daily basis.
- 2. Inventory approach (designed for prepackaged foods). This type of form is used to describe and count the number of items selected by students using an inventory approach for the target week (i.e., respondents record starting and ending inventory and all deliveries).

The two approaches were intended to be used in combination for cafeteria foods (reimbursable and a la carte) and for snack bars offering both prepared and prepackaged items; only the inventory approach is used for school stores and vending machines.

Cafeteria Survey

The Cafeteria Survey is comprised of seven forms (see Table 2.1). This survey collected data on all foods and beverages offered and served in reimbursable meals or a la carte in the cafeteria during a one-week period. The main focus was to obtain sufficient detail to describe the types of foods available, the types of foods selected, and the nutrient content of those foods. Detailed data are collected for both the breakfast period (where offered) and lunch periods. Qualitative data about foods offered between meals were collected via data collection checklists.² The primary respondent for the Cafeteria Survey was the cafeteria manager or other food service staff person.

² The checklist approach was chosen because (1) it was not known whether cafeteria foods are offered to students outside regular meal periods, and (2) space on the data collection forms was extremely limited.

Table 2.2

Data Collected by Principal and SFA Director Surveys

Торіс	Questions on Principal Survey	Questions on SFA Director Survey		
Questions about competitive food sources				
Food venues	Number, type, location, and times competitive foods are available for sale; restrictions on vending machine access.	Sources of competitive foods under the responsibility of the school food service, by school type.		
Standards for foods	Rules governing the types of compe- titive foods available to students at school (e.g., nutrition standards, price criteria, prohibited or required foods).	District-level rules governing the types of competitive foods available to students (e.g., nutrition standards, price criteria, prohibited or required foods).		
Operation of food venues	Purchasing and stocking arrangements for vending machines in the school	Purchasing and stocking arrangements for vending machines under the responsibility of the school food service. ^a		
Exclusive contracts	Whether school covered by "pouring rights" contract ^b	Exclusive beverage contracts covering any schools in district		
Availability of sales data	Documentation of product sales (number sold) from vending machines and school stores, if available.	Documentation of product sales (number sold) from vending machines under the responsibility of the school food service.		
Profits	Amount of profit earned from non-food service-operated competitive food sales, and use of profits.	Amount of profit earned from food service-operated competitive food sales, and use of profits.		
Other topics (specific to each survey) Parties responsible for determining competitive food availability (hours, location, types of food).				
	Mealtime policies.			
	Open campus policies and types of nearby off-campus eating establishments.			
	Types and frequency of fundraisers that sell food or beverages.			
	Initiatives and activities to promote healthful food choices.			
	Daily student attendance for target week.			
		A la carte pricing strategies.		
		Availability of nutrition information for competitive foods.		
		Use of computerized systems for cafeteria operations.		
		Use of branded foods (e.g., McDonald's, Pizza Hut, Subway) for a la carte sales.		
		Food service management company involvement in competitive food sales.		
		Menu planning system used for reimbursable meals.		

a Includes vending machines for which the food service staff purchase and stock food and beverage items, as well as those under contract with an outside distributor.

b A long-term contract with a beverage company that establishes the company as a sole source vender for some types of beverages.

The Cafeteria Survey was packaged in a single booklet containing the multiple data collection forms. A separate instruction manual provided detailed guidelines and sample completed forms. The booklet and instruction manual were packaged with recipe forms and reference guides in a large accordion folder, with labeled pockets to assist respondents in locating and organizing materials.³ Color-coded forms, color printing, tabs, and other special formatting features were used to create an attractive, user-friendly package.

For this study, the **Every Day Cafeteria Foods Form** used for prior studies was modified to reduce burden. Respondents list the foods offered every day on this form. They enter the number of servings of prepared foods on a daily basis, but servings of prepackaged items need not be tracked on a daily basis; instead, the form contains a checkbox to indicate that the item is tracked on an inventory form. This approach reduces the number of foods that must be tracked daily to determine servings.

The **Cafeteria Foods Form** collects information on prepared and prepackaged items not served every day. The inventory approach was not used for these foods. This form was not substantially modified from versions used in prior studies.

Both cafeteria foods forms request complete food descriptions, including, for pre-prepared items, brand name, manufacturer, and product code. This information was intended to be used, in conjunction with bid lists or other documents obtained from the SFA director, to determine whether additional information on nutrient content is required when entering the food in a nutrient analysis system. Most pre-prepared food items used for school food service are not included in nutrient databases.

The **Inventory of Every Day Prepackaged Foods** was developed and used to track servings of prepackaged items for the week.⁴ This form called for the starting inventory of items on Monday (or the previous Friday, end of day), deliveries during the week, and the ending inventory on Friday.⁵ Determination of starting and ending inventories, and transcription of delivery invoices, may be performed outside of normal cafeteria hours, potentially easing the burden that this type of data collection places on cafeteria staff during meal preparation and serving hours. It was not expected that this approach could be used in schools that share a stockroom with other food sources in the school (e.g., snack bar, vending machines), or in schools where inventory may be transferred out to other schools in the district during the target week.

The remaining forms for the Cafeteria Survey (forms 5 through 7 in Table 2.1) have been used or tested for other studies (SNDA-II or the Integrated Study of School Meal Costs and Outcomes) and were not substantially modified for this study.

³ The Recipe Forms are bound separately to allow respondents to attach copies of their recipes in lieu of re-copying them onto study forms. The Product Code Guide provides guidelines for identifying and recording manufacturer product codes, and the foods for which a product code is needed. The Daily Reminder List provides tips for getting organized, and a summary of day-by-day activities for the target week.

⁴ Foods requiring some preparation, including simple heating, are not considered candidates for the inventory method, because some portions removed from the package may be wasted.

⁵ The daily average servings can then be computed from the weekly total during analysis.

Other Competitive Foods Forms

New data collection instruments were developed for snack bars, school stores, and vending machines. It was expected that school food service staff would complete these forms if the school food service operates these food venues; otherwise, the school principal (or, in two cases, the SFA director) was asked to designate a respondent prior to the data collection period (for example, the principal would provide contact information so that the faculty supervisor for the school store could be recruited).

The Inventory of Vending Machine Items and Inventory of School Store Food and Beverage

Items collected information on prepackaged snack and beverage items for a one-week period. Instructions emphasized the need for respondents to enter the complete food name, manufacturer, brand, and package size for each item offered at the food venue. For vending machines, it was assumed that principals, SFA directors, or other school personnel could obtain information on the number of items sold for machines contracted out to distributors. The availability of this information is discussed later in this chapter.

The procedures for snack bars differed from vending machines and school stores. The inventory approach was used for vending machines and school stores because the majority of items typically sold through these venues are commercially available, individually packaged snacks and beverages. If a school prepared items from scratch for sale in a vending machine or school store, they would be asked to attach a recipe for the items. For snack bars, however, food preparation was expected to be more common; therefore, both a foods form (based on servings approach, with request for product code and recipes if needed) and an inventory form were provided for collection of snack bar information. These forms are similar to the every day foods and inventory forms used for foods served in the school cafeteria.

Separate forms are completed for each vending machine, school store, and snack bar in the school, so that competitive foods offered and served can be analyzed by location. The locations of all competitive food sources were requested on the data collection forms. Forms for each type of competitive food source were bound together with a page of instructions and a sample completed form. It was requested that the inventories of competitive foods be completed for the same target week as the Cafeteria Survey, whenever possible.

Data Collection Procedures

The data collection effort for the feasibility study spanned an eight-week period from mid-April to mid-June 2003.⁶ This time frame included the scheduling of data collection activities with respondents; advance meetings with school food service staff prior to finalizing the instruments; on-site training, technical assistance, and collection of validation data by Abt Associates staff; completion of surveys and school foods forms by SFA and school staff; and debriefing meetings with all respondents. State Child Nutrition (CN) directors were also interviewed by telephone during this period. Table 2.3 lists all data collection activities for the study.

⁶ The target week for data collection in all three schools was May 19-23, 2003.

Table 2.3

Data Collection Activities

	School #1	School #2	School #3
Advance meeting with school staff	Yes	Yes	No
On-site training:			
Time spent training	70 min	90 min	90 min
Number trained	2	5	4
On-site technical assistance:			
Time spent reviewing forms	15 min	85 min	25 min
Time spent talking with respondents	15 min	45 min	30 min
Validation data collected by Abt Associates staff	Yes	Yes ^a	Yes
Instruments completed and respo	ondent(s):		
SFA Director Survey	SFA Director	SFA Director	SFA Director
Principal Survey	Principal and teacher	Headmaster	Assistant Principal
Cafeteria Survey	Cafeteria manager	Food service staff person (not manager), assisted by manager and other staffperson	Cafeteria manager
Vending Machine Inventory (food service)	N/A	Cafeteria manager and other food service staffperson	Cafeteria manager
Vending Machine Inventory (outside contract)	Pending (beverage distributor)	Pending (beverage, snack, and ice cream distributor)	Athletic director (items offered—no sales information)
Snack Bar Forms	N/A	Food service staff persons (not manager)	N/A
School Store Inventory	N/A	Faculty supervisor for student-run store (items offered and "sold")	Faculty supervisor for student-run store
State CN director interview conducted ^b	Yes	Yes	Yes
Debriefing meetings completed	Yes, all respondents	Yes, except with headmaster, store supervisor	Yes, all respondents

N/A = not applicable

a These data could not used; respondent provided food and beverage information for a week prior to the target week.

b Brief telephone interview—no formal instrument was developed.

Advance Meetings with School Staff

Advance meetings were held with school staff in two of the three schools.⁷ The advance meetings provided an opportunity to meet the respondents, learn more about school food service operations and locations of competitive food venues, obtain feedback on the draft data collection instruments, and discuss a strategy for obtaining information about vending machines stocked by an outside distributor. During the meeting with the SFA director and cafeteria manager, Abt Associates staff reviewed typical menus, production records, and point-of-service (POS) system reports generated by the food service staff to identify information that might be useful in completing the study forms. For example, reimbursable meal counts were readily available on daily POS system reports.

Training and Technical Assistance

On the Thursday prior to the target week, one member of the Abt Associates study team visited each of the schools to train respondents on completing the school foods forms. Actual training time ranged from a little more than an hour at the middle school (School #1) to approximately 1½ hours at the high schools. The SFA directors and cafeteria managers participated in all three trainings. At School #2, snack bar staff, the person responsible for reimbursable meal and a la carte meal counts/ production records, and their field supervisor were also trained.⁸ At School #3, the training included the athletic director responsible for vending machines outside the cafeteria and a marketing teacher who runs the school store. For all three schools, the majority of training time was devoted to reviewing the instructions and procedures for the Cafeteria Survey (about 1 hour).

During the target week, Abt Associates staff visited each school on Tuesday or Wednesday to review the study forms completed thus far, answer questions, provide additional training, if needed, and collect validation data. The middle school cafeteria manager (School #1) had been out on Monday of the target week, so there was little recorded in her survey to review. This was unfortunate because there were omissions and problems with the data from this school that might have been rectified if caught early on. As with the training visits, most of the time spent during the technical assistance visits was devoted to the Cafeteria Survey (30 minutes to 2 hours).⁹ There were fewer questions and problems with the competitive foods inventory forms at this point in the data collection.

Validation Data

On the day that technical assistance visits took place in the three schools, Abt Associates staff collected information to validate, qualitatively, the information respondents provided on study forms. Using the same forms, study staff listed all items sold a la carte only, in vending machines, at the snack bar, and in school stores. They also documented (to the extent they could be identified) package size and price. The main purpose of validation was to corroborate the identity of competitive foods, provide a way to check that school staff provided a comprehensive list of items available, and document the ability of an outside observer to collect these data without disrupting school operations.

⁷ The third school was identified late in the recruitment phase of the study, so an advance meeting could not be scheduled at this school.

⁸ The faculty supervisor for the student-run school store (head of the business education department) at the Massachusetts high school was not trained, because he would not agree to complete the study form. At the time of the training visit, he provided an invoice for a typical weekly food order, which he said represented the items sold the previous week. These data were treated as valid for purposes of analysis.

⁹ At all schools, there was a need to review the options for recording condiments.

The most common problem with the collection of these data was identifying package sizes for some vending machine items because items are not always fully visible. In addition, some a la carte, school store, and snack bar foods were accessible only with the assistance of school staff, and prices were not always posted. Results of the validation analysis are reported in Appendix B.

During the same school visit, also for validation purposes, a random one-third sample of all competitive food items was purchased from each of the food venues. The labels from purchased items were used for comparison with the nutrient information obtained from the nutrient database.¹⁰ This analysis is reported as part of Appendix C.

Survey Completions

Principal and SFA Director Surveys were mailed (or hand-delivered, during training visit) to principals and SFA directors the week prior to the target data collection week. A cover letter asked respondents to return the surveys within two weeks of receipt in the prepaid overnight delivery envelope provided. Two of the three principals, and one SFA director were contacted by telephone when their surveys were not received on time (or prior to scheduled debriefing meetings).

All school foods data collection booklets were distributed to respondents during the pre-target week training visit.¹¹ Respondents were asked to return the completed booklets within one week of the last day of the target week. SFA directors were asked to collect and return the Cafeteria Surveys and other forms completed by school food service personnel by prepaid overnight delivery; other school personnel completing competitive foods forms were provided with separate return mailers. Two SFA directors returned the materials on or close to the due date (Cafeteria Surveys and one Vending Machine Inventory). Abt Associates staff obtained the remaining materials on the day of the debriefing meeting at the schools.

The main problem in obtaining completed survey forms was encountered for vending machines operated by outside distributors. In order to obtain complete information on vending machine items, the SFA directors in Schools #1 and #2 contacted the outside distributors who stock machines located in the school cafeterias. The SFA director for School #1 worked through the business manager for the city, because the city held a contract with a major beverage distributor. The SFA director for School #2 contacted his local distributor directly. In both cases the SFA director provided the distributor with a copy of the Inventory of Vending Machine Items form and asked that they either complete the form for a one-week period or provide the information in whatever format it was available. The athletic director in School #3 was also asked to contact his distributors (one local and one national distributor), but it was not clear that he actually did so. After several follow-up requests by study staff, the requested data were received from one of the SFA directors (from one of the four distributors).¹²

¹⁰ Beverages were not purchased, primarily because the Nutrition Facts information was thought to be readily available locally or on-line.

¹¹ Ideally, the Cafeteria Survey would be mailed to respondents two weeks in advance of the target week. This allows additional time for respondents to review the instruction booklet and sample completed forms prior to the training visit. The same lead time may not be needed for the other foods forms, because only one type of form is involved and instructions are relatively brief.

¹² Abt Associates staff also attempted to collect information from vending machine distributors, but with little success. One local distributor reported that they only tracked certain categories of snack items in their machines (e.g., chips, cookies), and that this information is considered proprietary. Coca-Cola Enterprises is able to report sales for

State Director Interviews

During the data collection period, brief telephone interviews were conducted with the Massachusetts and Virginia state CN directors to collect information about state policies and initiatives related to competitive foods. They were also asked about the availability of nutrition information for commodity diversion products used in their states.

Debriefing Meetings

The Abt Associates project director and one other member of the study team debriefed respondents during in-person meetings at all three schools. FNS staff attended the debriefing for School #3 (in Virginia). All respondents participated in the debriefing meetings, except the headmaster and school store supervisor at School #2.¹³

During the meetings, each respondent was asked to confirm or estimate the time required to complete the questionnaire or school foods forms. They were also asked about the clarity of questions and instructions, the reasonableness of the questions or task, problems encountered, and suggestions for improving the questionnaires/data collection forms. Additional, instrument-specific topics were also explored with each respondent. Key findings from the debriefing meetings are discussed in the section on data collection issues and in Appendix A.

Data Preparation and Data Entry in Nutrient Database

Upon receipt, Abt Associates staff reviewed the surveys and all school foods instruments for legibility and completeness. The Principal and SFA Director Surveys were essentially complete, except for some item non-response. To the extent possible, missing data and clarifications were obtained from respondents during the debriefing meetings. If these meetings had not taken place, follow-up calls would have been made to most of the respondents. After editing and assigning school identification numbers, responses to the Principal and SFA Director Surveys were entered into an Excel database.

Project nutritionists reviewed the Cafeteria Surveys and noted problems, but editing was limited to the forms containing a la carte-only foods.¹⁴ For the three schools in the study, this included the Inventory of Every Day Prepackaged Foods Forms (two schools) and the Every Day Cafeteria Foods Form (one school).¹⁵ Because of the short period of time for data preparation and analysis, study staff relied primarily on the validation data and on-line sources (rather than follow-up telephone calls) to supplement respondents' descriptions of the competitive food and beverage items. Again, debriefing

individual items by machine (product name, size, units sold—not price), but could not do so on an ongoing basis. Their smaller distributors may not have the same capability. Coca-Cola agreed to fax the information for the target week for School #1, but it was never received. Calls made to the local distributor in Virginia for Pepsi were not returned.

¹⁵ None of the a la carte-only foods were part of a self-serve bar or prepared from a recipe, and none of the schools sold a la carte items outside of meal periods.

¹³ The headmaster had not completed the Principal Survey when the debriefing meeting was initially scheduled. Subsequent efforts to meet or interview him by telephone after receiving his completed survey proved unsuccessful. The faculty supervisor for the school store was not asked to participate in the debriefing because he did not complete the study form.

¹⁴ Data on reimbursable meals collected in the Cafeteria Survey were not entered, in accordance with guidance provided by FNS in the Statement of Work for the feasibility study. Because time and resources were limited, the study was designed to focus on identifying the issues involved with the collection and analysis of a la carte-only foods.

meetings also provided the opportunity to clarify some data with cafeteria staff, such as questions about how servings and inventory data were obtained. Follow-up calls were made to both cafeteria staff and SFA directors to obtain nutrient information,¹⁶ and missing bulk packaging and package size information.

The competitive foods data were coded using the Food Intake and Analysis System (FIAS), maintained by the University of Texas' Health Science Center. Data entry procedures and issues for the nutrient data are discussed in Appendix C.

Data Collection Issues

The main challenges anticipated during the design of the data collection plan were:

- Overall burden;
- Determining the data collection period;
- Identifying appropriate respondents;
- Tracking the numbers of foods sold in vending machines; and
- Designing cafeteria instruments to collect data on all foods.

Some of these challenges are obviously related (for example, the performance of new instruments depends on identification of appropriate respondents). The following sections discuss each of these challenges and how they were addressed during the study design and implementation. Appendix A includes a detailed discussion of the quality of the data collected and design issues that impacted data quality. It also provides some recommendations about instrument design that should be considered prior to future testing.

Overall Burden

It was anticipated that the burden of collecting data on all school foods and competitive food policies and practices would be great. Specific ways in which burden was addressed are discussed in the section below on designing cafeteria instruments to collect data on all foods. In addition, all respondents were offered compensation for the time spent completing the instruments and participating in the debriefing meetings, as follows:

- Principal and SFA Director Surveys: each \$50
- Cafeteria Survey: \$200
- Inventories of Competitive Foods: each \$50

The estimated response burden for each instrument used in the feasibility study is provided in Table 2.4. Overall, the principals and SFA directors did not find the surveys to be too long, whereas the cafeteria staff were required to devote a large amount of time and effort to complete the survey, especially for staff also completing one of the other competitive foods forms.

¹⁶ Nutrient information was needed for only one pre-prepared item: ready-to-bake chocolate chip cookies. The manufacturer and product code were provided, but the item could not be located on line.

Table 2.4

Respondent Estimates of Data Collection Burden, in Hours^a

	School #1	School #2	School #3
Principal Survey			
Total hours	2.33	2.33	1.25
Gathering info	0.75	0.0	0.0
Consulting with other staff	1.25	1.0	0.25
Completing questionnaire	0.33	1.33	1.0
Total number of respondents	2.0	1.0	1.0
SFA Director Survey			
Total hours	1.0	1.83	0.33
Gathering info	0.5	0.33	0.0
Consulting with other staff	0.25	0.0	0.0
Completing questionnaire	0.33	1.5	0.33
Total number of respondents	1.0	1.0	1.0
Cafeteria Survey			
Total hours	6.0	47.0	11.75
Prior to target week ^b	1.0	12.0	4.0
During target week	5.0	35.0	7.75
Total number of respondents	1.0	3.0	1.0
Snack Bar Inventory			
Total hours	N/A	7.83	N/A
Prior to target week ^b		1.0	
During target week		6.83	
Total number of respondents	N/A	2.0	N/A
Vending Machine Inventory			
Total hours	N/A	7.0 ^c	5.83
Prior to target week ^b		0.0	1.0
During target week		7.0	4.83
Total number of respondents	N/A	2.0	1.0
School Store Inventory			
Total hours	N/A	N/A	1.83
Prior to target week ^b			0.0
During target week			1.83
Total number of respondents	N/A	1.0	1.0
All Instruments			
Total hours	9.33	66.0	21.0
Total number of (unique) respondents	4.0	8.0	4.0

a Excludes time spent collecting data for foods available after school only.

b Excludes training time.

c Excludes time spent by outside vending distributor.

N/A = not applicable/available

Principal and SFA Director Surveys

Although none of the principals felt the survey was too long $(1\frac{1}{4}$ to 2-1/3 hours), the high school principals found it more difficult than the middle school respondents. One high school (assistant) principal noted that she did not know the answers to many of the questions about her school. The headmaster of the second high school reported that he did not have time to "research" the answers to questions he could not answer, such as vending machine profits and policies about a la carte foods.

SFA directors found the survey "easy" and relatively brief (20 minutes to just under 2 hours). One director commented that the survey seemed overly focused on sales from vending machines relative to a la carte items sold in the cafeteria. Another was surprised the survey did not include any questions about SFA director opinions, or about why certain foods are selected to be sold.

Cafeteria Survey and Inventories of Competitive Foods

Most of the data collection burden for this study falls on the cafeteria staff. Cafeteria managers are responsible for preparing and serving reimbursable and a la carte meals for large numbers of students within short periods of time. In addition, the cafeteria manager may be responsible for purchasing and stocking items for vending machines, a snack bar or canteen, or food carts/kiosks.

The Cafeteria Survey required a total amount of time that ranged from 6 hours for the middle school to 47 hours for the larger high school (School #2) (see Table 2.4). The latter estimate may not be reliable. The main respondent did not keep a daily log of time spent during the target week, and reported spending 12 hours prior to the target week reading instructions. Based on probing by study staff during the debriefing, Abt's estimate of the maximum burden for this school is closer to 30 hours. This is still considerably more time than was reported by the manager in the second high school (about 12 hours). Both schools offered a large number of entrée choices, but the school that spent the most time completed more recipe forms, a self-serve bar form, and forms for breakfast as well as lunch. Still, the Cafeteria Survey for School #3 included all of the a la carte-only foods, whereas for School #2 these were recorded by another respondent on the snack bar forms.

Two cafeteria managers had to take the Cafeteria Survey home because they did not have time to complete it during the school day. The third Cafeteria Survey respondent (School #2) came to work early and stayed late each day of the target week, and enlisted the help of two other staff for an hour each day. In addition, the manager at School #1 had to be paid overtime, in accordance with union rules.

Cafeteria managers in the two high schools found the Inventory of Every Day Prepackaged Foods form time-consuming to complete, citing multiple deliveries coming in during the course of the week. Milk was in fact delivered daily. According to the data recorded on the forms, however, other items were delivered on only one or two days during the target week.

The Inventory of Prepackaged Snack Bar Items, completed only for School #2, was considered particularly burdensome due to the large number of items offered at the snack bar. Respondents felt that taking the inventory and recording deliveries was "a two-person job". The total time required to complete the form, excluding SFA director time, was approximately 8 hours.¹⁷ This burden was found to impact the choices available to students: the respondent reported during the debriefing that

¹⁷ The SFA director set up an initial spreadsheet so that cafeteria staff would not have to write out the product names. This form was used instead of the study forms, and led to problems that are discussed in Appendix A.

she tried to avoid opening new cases of items on Friday because she didn't want to have to count individual packages for the ending inventory.

The Inventory of Vending Machine Items seemed to work quite well for the two high schools with food service-operated vending machines (Schools #2 and 3). Respondents reported that the forms were easy to complete; the data collection burden was similar in both schools—6 to 7 hours for two machines. One school restocked items on a daily basis, and the other restocked on only three days of the target week.

The test of the Inventory of School Store Food and Beverage Items was based on only one respondent. The faculty supervisor for the school store in School #3 found the form easy to complete, and did so in just under 2 hours. Unfortunately, because this was a testing week at the school, she did not place any orders or receive the usual weekly deliveries, so the burden estimate may be low.

Note that burden for some of the other competitive food sources may be underestimated because of the time of year the study was conducted. The cafeteria managers were not replenishing stock to the extent they would have been earlier in the school year. In addition, respondents in both high schools reported that student food sales were lower than usual due to end-of-school-year activities, including testing periods, during which breakfast and snacks were served by parent-teacher organizations.

Data Collection Period

The standard method used to analyze the nutrient content of reimbursable school meals is based on the average nutrient content of the foods offered and sold over a five-day (one-week) data collection period. This study used the same one-week time period for competitive foods in order to capture the variability in student selections across days of the week, and the possible relationship between competitive food selections and reimbursable meal offerings.

One exception to the five-day period was for vending machines stocked by outside distributors. Sales data for these machines were requested for "whatever time period it was available". One distributor provided two weeks of sales data for vending machines in School #2. These data revealed that sales were quite variable from week to week, possibly because the machines ran out of stock of some items during the period.

Identifying Respondents

While designing the data collection plan, it was not always clear who the most appropriate respondent would be for each instrument or data item.¹⁸ This was a particular issue in the larger schools with multiple sources of competitive foods for which several different district or school personnel were responsible. For school food service-operated venues (a la carte, snack bar, and vending machines in the cafeteria), the SFA directors and cafeteria managers were the primary contacts with Abt Associates and they designated appropriate respondents.

¹⁸ For example, prior to the initial contact with the school, it is not possible to know the persons responsible for vending machines located outside the cafeteria and school stores.

For non-food service operated food venues (school stores and certain vending machines), it was assumed that principals could be consulted to determine potential respondents. Principals were not always willing, however, to get involved with the recruitment of these staff.

For questions about the school environment and school food policies, the study relied largely on school principals. It was assumed that principals would be best able to answer (or obtain the answers to) these questions, but they had difficulty with many questions on the Principal Survey. Findings for the small sample of schools in this study indicate that principals were not always aware of the types, locations, and policies concerning competitive foods available to students (especially in the high schools). The degree to which principals were motivated to obtain the information from more knowledgeable district or school personnel varied.

Both the Principal Survey and SFA Director Survey included questions to determine whether any other personnel needed to be consulted when completing the questionnaire. Two principals reported consulting with the cafeteria managers; one SFA director consulted with the cafeteria manager and one SFA director consulted with the city business manager (about the exclusive beverage contract).

Tracking Number of Items Sold in Vending Machines

Of particular interest for this study was the ability to track the sales of individual items in vending machines available during the school day. The inventory approach could be adopted for this task in schools where the school food service or other school staff stock the machines. A major question, however, was how feasible or difficult it would be to obtain sales data for machines that are not stocked by school staff. Thus, the study was designed to explore multiple potential sources of information on the types and sales of foods and beverages stocked in these machines. These included attempts to have distributors complete the inventory forms, requests for copies of relevant documents from distributors (through the principal and SFA director surveys), and direct contacts by Abt Associates staff with distributors.

This feasibility study found that information on the types of products and number sold from vending machines operated by outside distributors was not readily available. It could only be obtained from one of four distributors contacted.

The SFA directors for two schools each spent an hour making telephone contacts to the distributors to obtain the necessary information. The SFA director for School #2 was successful in obtaining the data for four vending machines. The distributor recorded on the Inventory of Vending Machine Items form the slot number, item description, price, package size, and the number of each item added to the machines during a two-week period. Because the drivers replace what has been sold during the previous week, it seemed reasonable to assume that these data were representative of sales and could be used in the analysis.¹⁹ Although it was not working during the target week, the distributor reported that the drivers typically use a scanning device that performs an electronic inventory of the items in the vending machines prior to restocking. Apparently this device is currently being used by only 1 to 2 percent of vending distributors nationwide.

¹⁹ There was some evidence that sales varied over the two weeks from one of the machines; therefore, an average of the numbers sold during this period was used in the analysis.
The athletic director at School #3 is responsible for six vending machines (these were available to students after school only). He was able to partially complete the vending inventory form, recording the slot numbers, names, and package sizes of the items in the machines,²⁰ but was unable to obtain information from his two distributors about the number of individual items sold. The statements the athletic director receives on a monthly basis include total sales (in dollars). For one machine, the statement also includes the total number of units sold (e.g., 180 snacks), and for the beverage machines, the total number of cases sold. The athletic director indicated that the distributor decides which items go into the machines, and restocks only once every two weeks.

As noted earlier, efforts by Abt Associates staff to obtain information from the vending distributors were not successful.

Designing Cafeteria Instruments to Collect Data on All Foods

In designing data collection instruments for this study, four main features were implemented to reduce the burden on cafeteria managers:²¹

- 1. Collection of servings data combined an "inventory approach" with the traditional "servings approach" used for reimbursable meals;
- 2. Data collection did **not** require separate tracking of servings in reimbursable meals *versus* a la carte;
- 3. The requirement that cafeteria staff collect labels and nutrition information for preprepared food items was eliminated; and
- 4. Plans were made to discuss alternative methods of data collection with school staff during advance meetings.

The main findings are:

- Collecting detailed data simultaneously about reimbursable meals and a la carte (and other competitive foods) was extremely burdensome for school cafeteria staff. It could not be accomplished without considerable overtime or working on their "own time", especially at the high school level.
- Cafeteria Survey data quality was a major problem for two of the three schools. The most significant issues for one school (#1) were omissions of reimbursable items from all meal component categories, plus condiments; recording the number of portions **prepared** versus number **served**; and incomplete/inaccurate information recorded for the inventory of prepackaged a la carte only items. The second school (#2) also recorded the amount **prepared** versus **served** for many foods, and sometimes expressed the amount in terms

²⁰ This information was recorded while the driver was filling the machines; that made it possible to "see" the package sizes for most of the items.

²¹ The use of daily menu forms, preprinted with meal component categories and descriptions of commonly offered foods, similar to those designed for the Integrated Study, was considered to reduce burden further. Their use was rejected for two reasons: (1) the preprinted forms were still undergoing testing, and (2) use of the blank forms would allow for comparisons of the data collection burden due to a la carte foods with that observed for SNDA-II.

of cases, gallons, pounds, etc., instead of individual portions. Incomplete food descriptions were a problem, to varying degrees, in all three schools.

• Cafeteria managers in two of the three schools had significant difficulty and were unsuccessful at providing manufacturer product codes for pre-prepared foods (as an alternative to collecting labels), which are needed to obtain nutrition information.

Inventory Approach

The "inventory method" for obtaining information on the number of competitive items sold was well accepted by food service staff. Three inventory forms were used in the cafeteria by one or more schools: Inventory of Every Day Prepackaged Foods, Inventory of Vending Machine Items, and Inventory of Prepackaged Snack Bar Items.^{22,23} Although the inventory approach saved time during the busy meal periods, data collection in the cafeteria was still time-consuming.

The inventory method was found to have one limitation—this data collection method may mask dayto-day variations in food offerings that are due to shortfalls in stock. Most competitive foods are "everyday items"—that is, the intention is to offer them every day. Stocks of foods run out during the week, however, influencing the choices faced by students on any particular day. Respondents noted that some of the stock "run out" observed in this study was due to the timing of data collection at the end of the school year.

Separate Tracking of Reimbursable Meal and a la Carte Servings

The second feature of the data collection plan was that, for foods served in reimbursable meals and also offered a la carte, cafeteria staff were not asked to separately track reimbursable and a la carte servings. For these foods, the cafeteria manager could report the **total** number of servings when completing the cafeteria foods forms. Based on experience in previous studies, this feature was expected to be integral to the feasibility of collecting data on all foods offered in the cafeteria.²⁴ Separate counts of servings were not needed for the two main goals of this study, as discussed in Chapter 5. Furthermore, separate tracking of servings is not possible in many schools. Putting special procedures in place to do so would impose an enormous burden that might interfere with operations and influence the foods available to students during the data collection period.²⁵

Eliminating Labels and Nutrition Information

The third way in which the data collection burden was intended to be reduced for cafeteria staff was by eliminating the task of providing product labels and nutrition information for pre-prepared items

²² Two of the three schools used the inventory method for milk, which was available both as part of the reimbursable meal and a la carte. The middle school manager felt more comfortable including milk on the daily Cafeteria Foods Forms because "it goes with the meal," thus, she used the servings method for milk. Because milk is typically ordered and delivered daily in many schools, it is not clear whether the inventory method is less burdensome than the servings approach for this item.

²³ The only a la carte-only item that was not tracked on the Inventory of Every Day Prepackaged Foods form was a readyto-bake chocolate chip cookie, because theoretically any wasted servings could not be accounted for using the inventory method. (In reality, this item is so popular that it usually sells out completely, so there is no waste.)

²⁴ Some schools may be able to separately count the numbers of items served in reimbursable meals and items sold a la carte. None of the schools in the feasibility study would have been able to do this without putting special (manual) procedures in place, and not without disrupting usual operations.

²⁵ There was concern that cafeteria staff might offer fewer choices or choose not to restock some food items if they were unable to keep track as they replenished stocks.

listed on the cafeteria foods forms. (Nutrient information is needed to establish a definitive match with items in the nutrient database, or to enter new items in the database.) Cafeteria staff were instead asked to provide the manufacturer's product codes, but only for specially formulated school food service items (e.g., Rib-BQs, commodity chicken patties) and foods with a nutritional claim (e.g., vitamin-fortified juice, low-fat cheese, Super Donuts). Rather than burdening cafeteria staff with the collection of product labels, it was assumed that this information could be obtained in a centralized fashion after the data collection week, either on-line or from SFA directors.²⁶

Alternative Methods of Data Collection

Alternative methods for obtaining detailed information about foods served in the school cafeteria were also explored. During the instrument development phase of the study, Abt Associates staff reviewed production records kept by cafeteria managers and found they contained useful information, which would have to be duplicated on the study forms. One school's production records were quite detailed, with food descriptions, recipe codes, manufacturer names and product codes, portion sizes, and complete production information (numbers prepared, leftover, and used) for each food item served. On the other hand, food descriptions on records kept by another school were incomplete for purposes of nutrient analysis, and production information for similar foods was combined (e.g., hamburger and cheeseburger, kielbasa and hotdog). At two of the three schools, the production information for milk was not separated out by flavor or fat content (e.g., servings of chocolate skim and whole milk were counted together). Thus, it was preferable to proceed with all schools completing the study forms, transcribing information from production records, when appropriate.

The use of POS system reports was also explored, as a possible source of information on the number of a la carte items sold and/or the numbers of students purchasing a la carte items throughout the week. Neither of the two systems used in the three schools were set up to provide a count of the students who made a la carte purchases. Both systems did provide counts for any food or group of foods for which there was a separate register key. Most of the register keys in all three schools, however, were assigned to groups of a la carte items by type or price (e.g., cookies @ \$0.75, chips @ \$0.50). Therefore, the POS system reports used in the three schools could not be substituted for counts of servings or inventories collected on study forms.

Summary

This chapter describes the data collection instruments and procedures used for the study, and discusses problems and successes encountered with the collection of data. This section summarizes the main findings from the fielding of the data collection instruments, and the limitations of what can be learned from the sample of three schools.

Findings from Data Collection

The findings of the data collection portion of the study are summarized below:

²⁶ This is similar to the process Abt Associates has used in past studies when required labels or nutrition information were not provided or not available through the cafeteria manager. This centralized process also recognizes that the most efficient way to collect nutrition information in a future national study is to centralize the process (because many schools serve the same items).

Data Collection Instruments

- Respondents to the Principal and SFA Director survey had difficulty with the format of the questions about competitive food policies. These questions were sometimes misinterpreted due to a lack of formal policies at the SFA and school levels.²⁷
- Respondents to the Principal and SFA Director Surveys were generally unable to respond to questions about profit earned from competitive food operations.
- Instruments should be revised to collect detailed information about each vending machine. The current survey collected information on each topic in the aggregate (number of machines by location, hours, operation, food items), and it was not possible to provide a comprehensive description of each individual machine.
- A data collection period of one-week may not be sufficient for obtaining data on competitive food selections. Compared to the components of the reimbursable meal, competitive foods are more likely to run out of stock during a week. As a result, the data on selections can be highly variable from week-to-week depending on whether stocks are replenished on a timely basis.
- The Cafeteria Survey presented problems regarding the accounting for condiments, the accuracy of servings data, and obtaining manufacturers' product codes. These issues would have required significant follow-up if the data on reimbursable meals had been included in the analysis.

Burden

- Data collection imposed significant burden on the cafeteria managers/staff. Cafeteria staff at the two high schools spent 18 and 45 hours completing data forms for the target week. Respondents reported that data collection had to be done outside of work hours, with assistance of other staff, or as paid overtime.
- In addition to the time burden, two of the three schools had difficulty with the data collection process. Future studies may need to plan for more and better timed technical assistance.

Response Issues

- The Principal Survey had significant item non-response, especially regarding competitive food policies and financial arrangements.
- The Principal Survey yielded data that did not agree with on-site observations by Abt staff observations (e.g., the existence of vending machines and what is in them).
- The Principal Survey would have required extensive telephone follow-up to obtain completed surveys had debriefing visits not been scheduled to follow-up with respondents.
- SFA directors were knowledgeable about food policies and motivated to complete the survey; there was no item non-response from SFA directors.

²⁷ Due to the lack of information about the prevalence and variety of competitive food policies at the district and schoollevel, future surveys should included more opportunity for open-ended narrative responses on these topics.

- The study forms for the snack bar and school store inventories were not used by some of the respondents because of perceived burden. This led to incomplete data from these food venues.
- Inventory data for vending machines operated by non-school personnel (i.e., machines filled by distributors) were difficult to obtain. For this study, only one of four distributors responded to requests for data.
- Validation of competitive food data showed that school staff accounted for nearly all competitive food items on the cafeteria and inventory forms. Across all food venues, only 5 percent of offered items were not entered on forms, suggesting that missing data was not a large problem.

Data Entry Issue

• Obtaining accurate nutrient data requires significant effort beyond data entry of product names in a nutrient database. More than 10 percent of the competitive food and beverage items seen in this study were new products or had brand-specific fortifications.

Limitations

The testing of instruments and procedures was limited in several ways by the small sample included in this feasibility study:

- All three schools had just one serving location for lunch and a la carte sales. As a result, the data collection instruments and procedures were not tested in a school environment with multiple serving locations and storage areas.
- All three schools used POS systems, so the burden estimates do not represent the experiences of respondents in schools that rely on manual systems for tracking meal counts and a la carte sales. This study did not find, however, that POS systems, as used in the sample schools, were particularly helpful in reducing the burden of reporting servings of individual food items, either for reimbursable meals or competitive food service.
- The study was unable to test the Foods Outside Meal Periods Checklist. None of the three schools offered a la carte foods or beverages for sale to students outside of the breakfast or lunch periods.
- The study was unable to test the collection of data from canteens, concession stands, and snack bars that offer prepared foods (in addition to prepackaged snack and beverage items). Prepared foods impose greater burden on data collection because these foods may need to be described by completing recipe forms, although the process is similar to that required for reimbursable meals.
- The study had limited opportunity to test the feasibility of obtaining nutrition information for specially formulated school foodservice items. These items were not sold a la carte-only. Information was collected for these items when offered as components of the reimbursable meal, but by design, the study did not enter these data into a nutrient

database. The issue to be investigated in further studies is the ability to accurately identify products from the food descriptions provided by cafeteria managers.²⁸

²⁸ Requesting manufacturer product codes in lieu of collecting labels for components of the reimbursable meal was not particularly successful. Recommendations for future data collection include (a) request purchasing specifications from SFA directors, and emphasize that cafeteria managers provide complete food descriptions; (b) request nutrient information from SFA directors when it is not available on-line (SFA directors reported that they keep this information on file or can obtain it easily from distributors); (c) obtain nutrition information for commodity diversion products from state CN agencies.

Chapter 3 Competitive Foods in the School Environment: Policies and Practices in Three Schools

This feasibility study tested an approach for collecting data about the school environment, school policies, and financial arrangements related to the availability of competitive foods. Two survey instruments were fielded to collect these data: the Principal Survey and SFA Director Survey. The instruments were discussed in Chapter 2, and an assessment of the success of the instruments was provided in that chapter.

This chapter presents a picture of school food policies and practices in three schools. The Principal Survey and SFA Director Survey are the primary data sources. The survey data were revised and supplemented, however, based on information obtained during on-site training and debriefing visits to the schools. In addition, the narrative provided in this chapter includes information obtained from the Cafeteria Survey and the inventories of competitive foods, thus allowing a rich picture of the school environment found at the three sample schools. It is important to recognize that the schools were not randomly selected, and are not representative of schools nationwide.

Supplementing and revising the survey data with information from the site visits allows for the most complete and accurate presentation of school environments for the purpose of this report. For future monitoring efforts, however, it may not be feasible to conduct on-site interviews in all schools. Chapter 5 provides recommendations for improved surveys that may be tested for future data collection efforts.

The Principal and SFA Director Surveys collected information about four sources of competitive foods:

- A la carte
- Snack bars
- Vending machines
- School stores

In addition, the SFA Director Survey included questions about menu planning, computer systems, pouring rights contracts, availability of nutrition information for competitive foods sold by the school food service, and whether pricing strategies were used by the school food service to encourage healthful food choices. The Principal Survey included additional questions about off-campus eating establishments, use of foods for fundraising activities, and school initiatives to increase healthful food choices.

The two surveys had minimal overlap: the Principal Survey asked for school information, whereas the SFA Director Survey primarily asked for district-level information (with some questions specific to the sample school). With this structure, information was obtained about competitive food policies at the district and school levels. Information about state policies was obtained via telephone conversations with the Massachusetts and Virginia state CN directors.

The primary goal of this chapter is to characterize the availability of competitive foods, the operation of food venues where students may obtain competitive foods, and the formal policies that influence the school food environment—albeit based on a small sample of schools.

Food Venues

Table 3.1 provides a summary of the food venues available to students at the three sample schools. At all three schools, food is available in the cafeteria from reimbursable meal service, a la carte service (including snack bars), and vending machines. There are no food venues located outside the cafeteria in School #1 (the middle school). At Schools #2 and #3, students may also obtain food outside the cafeteria from the school store, and from vending machines available only after school.

Food Venues Available in School Cafeterias

In all three schools, reimbursable meals are served in the cafeteria. Cafeterias contain multiple serving locations and food venues, described as follows:

- School #1 (middle school). Students form a single line to pick up a reimbursable meal; after selecting the reimbursable lunch, students can purchase a la carte snack items at a separate register. A single vending machine is located in the cafeteria.
- School #2 (high school). Reimbursable lunch offerings are set up as food stations in the cafeteria. In addition, the cafeteria has a separate snack bar and seven vending machines.
- School #3 (high school). Cafeteria has three lines where students can purchase the same items, including reimbursable lunches and/or a la carte items. Two vending machines are located in the cafeteria.

A la carte foods are offered during lunch (but not breakfast) at all three schools. A la carte foods may be components of the reimbursable meal or "a la carte-only foods" that are never sold as part of a reimbursable meal. As shown in Table 3.1, two of the three schools allow a la carte purchase of reimbursable meal entrees and other meal components; School #2 does not allow a la carte purchase of reimbursable entrees alone. Additional rules governing the purchase of a la carte items are:

- School #1. Students may purchase a la carte items only after the purchase of a reimbursable meal, or if they bring a bag lunch.
- School #2. Components of the reimbursable meal can only be purchased a la carte as a complete meal, with the exception of milk and juice, which can be purchased individually.
- School #3. French fries may only be purchased a la carte after a reimbursable meal has been taken **and** when an additional entrée is purchased.

A la carte-only foods are available in all three schools: at a separate register in School #1, from a snack bar in School #2, and on the line in School #3. In School #1, a la carte-only foods include prepackaged snacks, such as chips, doughnuts, cupcakes; and beverages, such as spring water, sports drinks, and juice-based drinks. In School #2, a la carte-only foods include salads, ice cream, and snacks such as chips, cookies, snack cakes, and granola bars; beverages include milk, juice drinks, and spring water. School #3 sells a la carte-only foods that include ice cream and snacks, such as

pretzels, popcorn, cookies, and snack cakes; and beverages, including spring water and juice-based drinks.

Table 3.1

Description of Food Venues Available to Students

	School #1	School #2	School #3
School cafeterias		0011001.112	
Cafeteria service for reimbursable	1 cafeteria,	2 cafeterias,	1 cafeteria,
meals and a la carte	single line	food stations	3 lines
Breakfast hours	7:00 am -8:00 am	6:50 am -7:30 am	N/A
Lunch hours	10:30 am -1:00 pm	10:18 am -12:48 pm	11:39 am -1:45 pm
	(4 periods)	(6 periods)	(4 periods)
Components of reimbursable lunch offered a la carte?			
Entrees	yes	no	yes
Other components	yes	yes	yes
A la carte only availability			
During breakfast	no	no	N/A
During lunch	yes	yes	yes
Outside meal periods	no	no	no
Location in cafeteria	On the line (separate register)	Separate snack window	On the line
Restrictions	Purchased only with reimbursable meal or bag	none	none
Vanding machines	Iuncn	7	0
vending machines	Ι	7	2
Outside cafeterias			
Vending machines			
Number available during school	-	-	-
Number available only after school	-	3	6
School store			
Number of stores	-	1	1
Availability	N/A	Before classes and during lunch	Before classes

N/A = not applicable

Sources: Principal and SFA Director Surveys, and on-site observations.

Vending machines are located in the cafeteria at all three schools. The types of foods and beverages available from cafeteria vending machines vary by school (see Table 3.2). School #1 (middle school) has only one machine, which sells fruit juice (100% juice) and juice drinks. School #3 has two machines: one machine selling 100% juice and juice-based drinks, and one machine selling snacks. In contrast, the seven machines located in the cafeteria at School #2 offer considerable variety: spring water, juice-based drinks, milk, snacks, and ice cream.

Table 3.2

Types of Vending Machine	s Available to Students	, Inside and Outside the Cafeteria
--------------------------	-------------------------	------------------------------------

	School #1	School #2	School #3
	Nu	umber of machin	es
In cafeteria, available during school day			
Water only	_	1	_
Juice and juice-based drinks ^a	1	2	1
Milk and other non-carbonated beverages	_	2	_
Snacks	_	1	1
Ice cream and other frozen desserts	-	1	-
Outside cafeteria/available only after school ^b			
Snacks	_	_	1
Carbonated beverages	_	3	_
Sports and juice-based drinks	_	_	1
Water and juice-based drinks	_	_	1
Carbonated and non-carbonated beverages	-	-	3
Total vending machines	1	10	8

a Only available during lunch periods. Access at other times is visually monitored.

b Timers are used to restrict access.

Sources: Principal Survey and on-site observations.

Food Venues Available Outside Cafeterias

Food and beverages are available outside the cafeteria in the two high schools surveyed. School #2 has a school store available to students before classes and during lunch. School #3 has a school store available before classes only.

The two high schools also have vending machines that sell beverages after school. Three machines sell carbonated soft drinks in School #2, whereas six machines are available after school hours in School #3 and sell snacks, sports drinks, juice-based drinks, and spring water, in addition to carbonated soft drinks. Access to these machines is restricted by timers that lock the machines during the day. Because these machines do not provide food and beverages during school hours, they are not discussed in detail in the main body of this report. Appendix F provides a list of the items available in each of these vending machines in School #3.

Availability of Foods and Beverages by Time of Day

Table 3.3 shows the periods of the day when food and beverages are available from each food venue. School #1 has the least number of food venues available, and students have the least access during the day to those foods. Reimbursable meals are offered for both breakfast and lunch, and a la carte and vending machine items are available during lunch. The principal in School #1 reported that students are not allowed access to vending machine items during breakfast, or at any time outside of lunch periods. The vending machine is visually monitored by cafeteria and other school staff.

Students in the two high schools have access to foods during meal times and throughout the day. At School #2, students can obtain food during breakfast time from the cafeteria reimbursable meal service, vending machines, and the school store. These same sources are available during lunch, along with a la carte service at the snack window. In addition, vending machines located in the cafeteria may be accessed throughout the school day.

School #3 does not offer reimbursable breakfasts, but food is available before classes from vending machines located in the cafeteria and from the school store. Throughout the remainder of the day, School #3 looks much like School #2, with the exception that the school store does not operate during the lunch period.

Table 3.3

Types of Food Available to Students, by Time of Day

		Before Classes	Before Lunch	During Lunch	After Lunch	After Classes
1#1	Reimbursable meals					
00	A la carte					
Sch	Vending in cafeteria					

~	Reimbursable meals			
#	A la carte			
00	Vending in cafeteria			
Sch	Vending outside cafeteria			
•,	School store			

~	Reimbursable meals			
¥	A la carte			
00	Vending in cafeteria			
Sch	Vending outside cafeteria			
	School store			
Sourc	e: Principal Survey.			

Policies About Types of Food Available

Access to food and beverages during the school day is determined by both the times that food venues are available and also by the types of products offered for sale. As discussed in Chapter 1, USDA regulations prohibit the sale of foods of minimum nutritional value (FMNV) in school cafeterias during meal periods. State policy in Virginia (School #3) supplements federal regulations in three ways: Virginia regulations prohibit FMNVs from being sold anywhere in schools during meal periods, and in addition, prohibit the sale of iced or hot coffee or tea at all times and allow the sale of non-carbonated water anytime. There are no supplementary state policies in Massachusetts regarding competitive foods in schools.

This feasibility study sought information about district and school-level policies related to competitive foods. The SFA Director Survey included questions about the use of nutrition standards for approving a la carte items; and about district policies that prohibit or require foods or beverages for sale a la carte, beyond policies specified by federal or state regulations. The Principal Survey included the same questions regarding school-level policies.

Table 3.4 shows that nutrition standards are not used in any of the three schools to approve food and beverage items offered in the a la carte service, vending machines, or school stores. With respect to prohibitions and restrictions on certain food items, however, the three schools surveyed had different approaches. School #1 prohibits certain products from the a la carte service and vending machines; School #2 requires that a la carte service include certain items and permits only candy to be sold in the school store; and School #3 has no restrictions or requirements, other than the Virginia state policy.

School #1 reported that coffee, tea, caffeinated beverages, soft drinks, candy, and gum are prohibited from a la carte service. During the debriefing, however, it became clear that no formal policies prohibit these foods, as was discussed in Chapter 2.

In School #2 there are no foods prohibited from a la carte service or vending machines. In this school the principal reported a policy to sell only candy in the school store; it is likely that this policy is related to the logistics of stocking and displaying food in the store.¹ Data from the school store, however, showed an inventory of snack items (chips), in addition to candy. In School #3 there are no foods prohibited from a la carte service or vending machines; the principal did not know whether any foods are prohibited from being sold in the school store.

¹ This policy may also be in place to avoid competition between the school store and the school food service for sale of other snack items.

	School #1	School #2	School #3
Nutrition standards applied to food items?			
A la carte	no	no	no
Vending machines	no	no	no
School stores	N/A	no	don't know
Foods prohibited from being sold A la carte	Coffee, tea, caffeinated beverages, carbonated soft drinks, candy, gum	none	none
Vending machines	Coffee, tea, caffeinated beverages, candy, gum	none	none
School stores	N/A	Only candy is allowed	don't know
Foods required to be sold			
A la carte	none	Water, 100% juice, milk	none
Vending machines	none	none	none
School stores	N/A	none	don't know

School Policies Governing the Types of Competitive Foods^a

a Respondents were asked about school policies in addition to USDA and state regulations or guidelines.

N/A = not applicable

Sources: Principal and SFA Director Surveys.

Operation and Administration of Food Venues

This study collected data about the operation and administration of competitive food venues, including information about the organization responsible for the venue, the purchasing and stocking arrangements, the use of profits, and the persons responsible for approving key aspects of operation (see Table 3.5). These issues are important for the feasibility study because they help us understand the persons within a school who have knowledge of the competitive food operations, and vested interests in those operations. As discussed in Chapter 2, the school principals had difficulty responding to the survey and future survey efforts may need to consider alternative respondents.

Operation of Food Venues Available During the School Day

In all three schools, the school food service is responsible for most of the "competitive" foods and beverages available during the school day. All schools have a la carte service operated by the school food authority (SFA). The profits from a la carte sales go into a school food service general fund, and the profits are used to subsidize reimbursable meals.

Operation and Administration of Competitive Food Venues

	School #1	School #2	School #3
Organization responsible for food			
A la carte	SFA	SFA	SFA
Vending machines during school	City	SFA	SFA
Vending machines after school	N/A	Athletic dept. and academic clubs	Athletic dept.
School store	N/A	Faculty supervisor (student run)	Faculty supervisor (student run)
Purchasing and stocking arrangements			
for vending machines Vending machines during school	Distributor	SFA/Distributor	SFA
Vending machines after school	N/A	Academic clubs/ Distributor	Distributor
Any pouring rights contract?	yes	no	no
Use of profits			
A la carte	SFA general fund ^ª	SFA general fund ^ª	SFA general fund ^a
Vending machines during school	School general fund	SFA general fund	SFA general fund
Vending machines after school	N/A	School general fund	Athletic dept.
School store	N/A	School dept.	School dept.
Persons responsible for determining Vending machine			
Location	School board superintendent, District business manager	Principal SFA director	Cafeteria manager
Hours of availability	School board superintendent, District business manager	Principal SFA director	Principal
Types of foods/beverages sold	School board superintendent, District business manager	Principal SFA director	SFA director

	School #1	School #2	School #3
Persons responsible for determining School store			
Location	N/A	Principal	Faculty supervisor (marketing teacher)
Hours of availability	N/A	Principal	Faculty supervisor (marketing teacher)
Types of foods/beverages sold	N/A	Principal	Faculty supervisor (marketing teacher)

Operation and Administration of Competitive Food Venues

a A la carte profits are used to subsidize USDA reimbursable meals in all three schools.

N/A = not applicable

Sources: Principal and SFA Director Surveys.

School food service also operates cafeteria vending machines at schools #2 and #3. The school food service purchases and stocks items for all cafeteria vending machines in School #3; the school food service stocks three vending machines and a local distributor stocks four machines in the cafeteria of School #2. Profits from these machines go to a general SFA fund.

In School #1, SFA food operations are supplemented by a single non-SFA food venue—the single vending machine in the cafeteria. This machine is operated by the city, under a citywide exclusive beverage contract. The distributor purchases and stocks this machine and the profits from sales are added to a general school fund. The details of this contract are discussed in the next section.

In schools #2 and #3, the only non-SFA food venues available during the school day are the school stores. In both schools, the stores are student-run operations managed by a faculty supervisor. The profit from school stores goes to the business education and marketing departments.

Operation of Food Venues Available Outside the School Day

Schools #2 and #3 have non-SFA-operated vending machines for use after school hours. These machines are operated by the athletic department and academic clubs in School #2, and by the athletic department in School #3. The purchasing and stocking of most after-school machines is contracted out with distributors in School #2, but academic clubs stock some of the machines; in School #3, purchasing and stocking is done by distributors. The distributors operating these after-school machines include large companies such as Coca-Cola and Pepsi, and a smaller local company. Profits from these machines go to a general school fund in School #2, and to the athletic department in School #3.

Overall Administration of Food Venues

The day-to-day operation of food venues, described above, is possible only after a responsible party authorizes the installation and operation of a particular venue. As shown in Table 3.5, decisions related to vending machine location, hours, and contents involve many people.

In School #1 the school board, superintendent, district business manager, and food service director were all involved in determining the location, hours of availability, and content of the vending machine. In School #2 those decisions were made jointly by the principal and the food service director; whereas in School #3 the cafeteria manager made location decisions, the principal made decisions about hours of availability, and the food service director makes decisions about the types of food and beverages available in the machines.

In contrast to vending machines, school stores appear to operate somewhat autonomously. Decisions related to school store location, hours, and contents were made by the principal in School #2, and by the faculty supervisor in School #3.

Exclusive Beverage Contract

One school in our sample, School #1, has a vending machine operated under an exclusive beverage contract (a.k.a., pouring rights contract) (Table 3.6). The contract is between the Coca-Cola Company and the city; all schools and city buildings are included in the contract. The three-year contract covers all carbonated beverages, non-carbonated beverages, fruit juices, and water sold through vending machines in the school district.

Revenue from vending machine sales under the exclusive beverage contract is equal to 48 percent of sales, and payment is made to the school district. Non-monetary compensation is made in the form of items awarded to the athletic department. Although this contract does not include incentives, other types of pouring rights contracts might include incentives (e.g., a lump sum payment for exceeding sales goal). The SFA director for School #1 reported that after entering into the contract the number of vending machines in the district increased, but the number of other venues selling beverages remained the same (i.e., no snack bars or concessions were added to sell the particular beverages covered under the contract). The principal at School #1 was not aware of the beverage contract.

Beverage company:	Coca Cola
Other participants in contract:	Elementary schools Middle schools High schools Other city buildings
Contracting agent:	City
Terms of contract:	
Time period	3 years
Beverages ^a	Carbonated beverages Non-carbonated beverages Fruit juices Water
Payments: Percent of sales paid to school district Non-monetary payments Fixed annual commission	48% Athletic equipment N/A
Incentives: Payment for exceeding sales target Payment for each machine placed	N/A N/A
Impact of beverage contract on school district: Did number of vending machines increase?	yes
Did number of other venues selling beverages increase?	no
a These beverages are covered by the city contract; carbon vending machines.	nated beverages, however, are not supplied to middle school
N/A = not applicable	

Characteristics of Exclusive Beverage Contract at School #1

Source: SFA Director Survey.

SFA Pricing Strategies for Competitive Foods

As noted above, most foods and beverages available to students during the school day are operated by the SFA. As a result, SFAs may influence the food and beverage choices of students through the choice of items they offer for sale, and also through pricing strategies. The SFA Director Survey included questions about overall strategies for pricing a la carte items, and strategies for pricing a la carte items relative to reimbursable meals. SFA directors were also asked about pricing strategies for vending machine items. Responses are shown in Table 3.7.

SFA Pricing Strategies for a la Carte Meals and Vending Machines

	School #1	School #2	School #3		
Single standard price for a la carte entrees?	no	N/A	no		
Price of a la carte entrees relative to price of full-price reimbursable meal	lower	N/A	lower ^a		
Factors considered in setting a la					
Food costs Production costs ^b	\checkmark	√ √	\checkmark		
Administrative or indirect costs Incentive for reimbursable meal			\checkmark		
Incentive for healthful choices Ease of collecting payments			✓		
Factors considered in setting					
Food costs Production costs ^b	4	√ √	✓		
Incentive for reimbursable meal			\checkmark		
program participation Incentive for healthful choices			\checkmark		
a Price of entree plus milk is equal to price of	full reimbursable meal.				
b May include wages, benefits, utilities, equipment, supplies, etc.					
N/A = not applicable					

Source: SFA Director Survey.

The hallmark of the traditional reimbursable meal is that it provides good nutrition by providing foods from all the major food groups.² Consequently, one of the dangers of a la carte service is that students may purchase a la carte entrées and forgo the nutrition from other components of the meal. Of the three schools in this study, School #2 does not offer entrées a la carte. Schools #1 and #3 offer a la carte entrées, priced variably, and priced lower than a full reimbursable meal. School #3, however, encourages purchase of the full reimbursable meal by pricing the entrée plus milk equal to the full meal; and in School #1 a la carte purchase of entrées is rare due to a high percentage of free/reduced price-eligible students.

² Although not all schools plan their menus around the food-based meal patterns since the introduction of nutrient-based menu planning, all reimbursable meals must meet minimum nutrition standards based on Recommended Dietary Allowances (RDAs) for children and the *Dietary Guidelines for Americans*.

Pricing strategies may be applied to other a la carte items, in addition to entrées, and to vending machine items. Among all the factors that could be used as the basis for pricing, however, only food costs are considered when setting a la carte and vending prices in School #1. School #2 considers both food and production costs in setting prices for a la carte and vended items. School #3 considers food costs, incentives for participation in USDA reimbursable meals, and incentives for healthy choices. Notably, responses about pricing factors differed across the three districts, but were exactly the same for a la carte and vended items within districts.

An example of a Virginia incentive for choosing healthful food options is their pricing of a la carte french fries. Students can only purchase extra french fries after they have purchased a complete reimbursable meal, and the extra fries must then be purchased with another entrée. The price of a la carte fries is set at the price of a reimbursable meal. This pricing policy is set in order to encourage healthier eating along with the french fries; this policy, however, may be unintentionally encouraging students to overeat, or may lead to more plate waste.

Other Sources of Food

In addition to foods sold a la carte, in vending machines, and at school stores, there may be other sources of food sold to or provided for students (see Table 3.8). For example, it seems likely that parents might provide treats for birthdays or other special occasions in some schools. In our survey of three principals, each principal reported that parents and teachers are not permitted to offer students food or beverages in the classroom. The debriefing discussions revealed, however, that at least two principals misinterpreted the question; teachers have discretion to serve food in the classroom (such as pizza parties), and parents and students may bring in food for special occasions. In addition, bringing food into the classroom may be more common in elementary or middle schools, and therefore may not be captured by the sample of middle and high schools in this study.

Another source of food may be fundraisers in the schools. All three schools report that foods are allowed for fundraising and that there are no restrictions on the types of foods used for fundraising. In all three schools, student clubs and service organizations use food for fundraising. Candy and baked goods are used in fundraisers in schools #1 and #2, and candy and other purchased baked goods, such as doughnuts and muffins, are used in fundraisers in School #3. The reported frequency of fundraisers varied quite a lot between the three sampled schools. In School #1, fundraisers are never conducted during breakfast periods and rarely during lunch periods (less than once per month). Compared with School #1, fundraisers are reported as more frequent in both high schools. In School #2 fundraisers are held one or two times per month during breakfast and during lunch. In School #3 fundraisers are held one or two times per week in the morning, before classes.

Other Foods During the School Day: Availability and School Policies

	School #1	School #2	School #3
Foods in the classroom			
Parents/Teachers permitted to offer students food/beverages in classroom?	no	no	no
Foods used for fundraisers Does school allow foods to be	ves	ves	ves
used for fundraising?	,	yee	yee
Any restrictions on types of foods used for fundraising?	no	no	no
Organizations using food for fundraising	Student clubs/ Service organizations	Student clubs/ Service organizations, music or art dept., parent-teacher organizations	Student clubs/ Service organizations
Types of foods	Baked goods, candy	Baked goods, candy	Candy, purchased donuts, bagels, muffins
Frequency of food fundraisers			
During breakfast/Before classes	never	1 or 2 times per month	1 or 2 times per week
During lunch	less than once per month	1 or 2 times per month	never
Source: Principal Survey.			

Off-Campus Eating Establishments

Off-campus eating establishments are a potential source of food and beverages for some school students. The opportunity for students to eat off-campus is affected by school policy and the presence of off-campus eating establishments. In all three schools principals reported that students cannot leave campus during lunch or free periods. Students may still access off-campus eating establishments before or after school, however, and some students are likely to leave campus and access these establishments despite school policy.

Table 3.9 shows the number of off-campus eating establishments within walking distance and within driving distance of each school. School #1 is within walking distance of one restaurant and two convenience stores. School #1 did not report the number of stores within driving distance because students are not of driving age. At School #2 many fast food restaurants, other restaurants, and convenience stores are located both within walking and driving distance. The principal in School #3 reported two fast food restaurants, three other restaurants, and two food markets are within walking

distance of the school. There are more than 12 off-campus eating establishments within driving distance of the school, with over three fast food restaurants, over three other restaurants, five food markets, and one food court.

Table 3.9

Off-Campus Eating Establishments: Availability and School Policies

	School #1	School #2	School #3
	Nu	mber of establishme	ents
Within walking distance			
Fast food restaurants	—	many	2
Other restaurants, cafeterias,	1	many	3
Food markets, grocery, or convenience stores	2	many	2
Lunch wagons or push carts	_	_	_
Food courts	_	—	_
Other	—	—	—
Within driving distance, beyond walking distance	N/A ^a		
Fast food restaurants		many	3 or more
Other restaurants, cafeterias, diners		many	3 or more
Food markets, grocery, or convenience stores		many	5
Lunch wagons or push carts		_	
Food courts		_	1
Other		_	_
Are students allowed to leave			
school campus			
During lunch period	no	no	no
During free periods	N/A ^b	no	no
a Students are not of driving age.			
b Students do not have free periods.			
N/A = not applicable			

— = zero

Nutrition Initiatives

Principals were asked to indicate whether their school was implementing any of the school food environment initiatives shown in Table 3.10. Principals in schools #1 and #2 reported that promotional activities are conducted to encourage healthful food choices among students. The principal in School #3 encourages teachers and parent groups to raise funds by selling non-food items, although, as noted above, fundraisers selling food are still allowed. All three schools appear to

make an effort to either avoid advertising of less nutritious foods and beverages or, at least, not market them more highly than reimbursable school meals.

Table 3.10

Nutrition Initiatives Implemented by the Schools

	School #1	School #2	School #3
School has a health or nutrition advisory council made up of school staff, students, and parents that provides input about types of foods available			
School food service staff participate in making decisions and policies that affect types of foods available outside of school meal programs		\checkmark	
School groups are encouraged to raise funds by selling non-food items.			\checkmark
Food is not used by teachers or other school staff as a reward or punishment (e.g., withholding snacks for misbehavior) for students	\checkmark		\checkmark
Parents are encouraged to provide healthful food choices for bag lunches brought from home			
School has nutrition criteria for foods and beverages offered at parties, celebrations, and social events held after school or on weekends.			
Foods and beverages sold outside the school meal program are not more highly marketed than reimbursable school meals.	\checkmark		\checkmark
School conducts promotional activities to encourage healthful food choices, and does not permit advertising of less nutritious foods and beverages.	\checkmark	\checkmark	
School bans soda, other soft drinks, or sweetened fruit beverages (less than 100% juice) that may be sold to students in school or on school grounds.			
Source: Principal Survey.			

Two of the three principals reported that food is not used by teachers or other school staff to reward or punish students. During debriefing discussions, however, it was learned that teachers do use candy, pizza, and other foods to reward students in both schools.

None of the schools has a health or nutrition advisory council, and only one principal reported that the school food service is involved in decision-making about the types of foods sold outside the school meal programs. (Presumably, the principals interpreted this as foods sold in venues not operated by the SFA.)

The state CN directors were also asked about Massachusetts' and Virginia's involvement in state and local initiatives to promote a healthy environment in schools. Both states have held meetings to communicate the goals of Action for Healthy Kids to SFA directors. Massachusetts has also been conducting regional training sessions using *Changing the Scene* materials. SFA directors, superintendents, and principals are invited and must agree to attend as a team. Virginia has recently developed monthly health and nutrition-related activities for schools that will be disseminated soon. Although principals were not asked directly about these initiatives, they did not indicate their involvement in the open-ended response option provided on the survey.

Summary

Information collected about the school food environments shows that, in the three schools surveyed, the school food service is responsible for most food offered to students outside of the reimbursable meal service. SFAs operate a la carte service, snack bars, and vending machines in school cafeterias. Aside from school food service operations, food and beverages are available on a limited basis during the school day—in school stores with restricted hours, and from fundraising activities that are held on an *ad hoc* basis.

Unfortunately, although SFAs are in charge of most competitive food offered to students, they have not used nutrition standards to evaluate the foods offered and thereby ensure the availability of healthful food options. In addition, only one of the three SFAs reported use of pricing strategies to encourage healthful choices. Information presented in Chapter 4 and the appendices confirms the availability of many high-calorie, high-fat snack options from school food service operations.

In debriefings with school food service staff, they discussed the need to sell items that students will buy—both to satisfy their customers and to generate revenue to subsidize the reimbursable meal program. The need to generate revenue through competitive food sales is less important in schools serving a high percentage of free and reduced-price meals.

The information provided in this chapter also shows that the middle school differs from high schools in many important ways. There are fewer food venues at the middle school, and food and beverages are not available outside of meal periods.

Chapter 4 Competitive Foods in the School Environment: Nutritional Characteristics

This chapter provides a description of the food and nutrient composition of competitive foods offered to and selected by students during the school day **in the three sample schools**. The data for these analyses were obtained from a survey of cafeteria foods (the Cafeteria Survey) and inventories of other sources of competitive food (vending machines, school stores, etc.). The instruments and an assessment of their success were discussed in Chapter 2. Cafeteria managers, other school food service staff, school store supervisors, and a vending distributor provided detailed information about all foods and beverages available to students. The information included:

- Manufacturer, brand, and name of each item;
- Package size or serving size; and
- Numbers of each item sold during the target week.

Competitive foods from all sources available during the school day were included. The analysis of a la carte service was limited to food and beverages served only on an a la carte basis and **not** in reimbursable meals.¹

The approach to analysis and reporting of nutritional characteristics was twofold. First, data were combined across the three schools and the various venues offering competitive food items. This provides an overall picture of the average food and nutrient composition of all competitive foods offered in the school environment during the school day (in the three sample schools). Second, separate analyses were conducted for the items available **within** the cafeteria and **outside** the cafeteria. This grouping was used because USDA has the ability to regulate foods available in the cafeteria during meal periods.

Nutrient analyses for this study are fundamentally different from prior studies of school foods. Prior studies analyzed the nutrient content of reimbursable meals, presenting information about mean nutrient composition relative to USDA standards for school meals. The standards for school lunch are the same as national recommendations for healthful eating applied to a full-day diet, divided by three. Unlike school menu analyses, no similar standards for individual foods exist. It was therefore important to examine several dimensions of the nutritional composition of competitive foods, and to select reasonable, available benchmarks to facilitate interpretation of the findings.

The nutritional characteristics of competitive foods assessed for this study include:

¹ There were several reasons for this approach. First the issues involved in analyzing reimbursable meal items have already been identified, and data on their nutritional characteristics are available from prior studies (e.g., SNDA-I and SNDA-II). Second, to reduce burden, cafeteria staff were not asked to report servings data separately for foods offered both in reimbursable meals and a la carte. Sales data for these items, therefore, were not available. Finally, data processing and nutrient analysis of all foods was not a requirement for the feasibility study, due to the brief period of the project.

- Mean amounts of food energy (calories) and macronutrients: total fat, saturated fat, carbohydrate, protein, sodium, and dietary fiber;
- Mean percentages of calories from total fat and saturated fat;
- Mean amounts of micronutrients: vitamin A, vitamin C, thiamin, riboflavin, niacin, vitamin B₆, calcium, and iron;
- Mean numbers of Food Guide Pyramid servings: grains, vegetables, fruit, dairy, and meat; and
- Mean amounts of discretionary fat and added sugars.

The average nutrient content of competitive foods was also analyzed relative to criteria used for product labeling, as specified in the U.S. Food and Drug Administration's *Food Labeling Guide*. The criteria used were applied to each item to determine whether it could be classified as "free" or "low" in calories, total fat, saturated fat, cholesterol, and sodium. In addition, each competitive food item was evaluated to determine if the item fits the definition of a food of minimum nutrition value (FMNV).

The average nutritional characteristics are shown for broadly defined groups of foods (e.g., beverages, baked goods/desserts, snacks), and for subgroups (e.g., 100% fruit juice, cookies, chips). Separate tables present unweighted data (each competitive food item weighted equally) and weighted data (items weighted by servings or sales, within school, with schools weighted equally). These tables have been placed at the end of the text to make the chapter easier to navigate.

To help interpret the nutrient data presented in the tables, findings are discussed relative to the contribution competitive foods make to Daily Values (DVs). DVs are the dietary reference values FDA has adopted for use on food labels, to assist consumers in planning a healthful diet.² The DVs for nutrients, as well as Food Guide Pyramid servings recommendations, are included at the beginning of the corresponding sections in this chapter.

The chapter begins with a discussion of the number and types of competitive foods offered to students in each of the three schools, and the top selling items in each school. The remaining sections of the chapter discuss various aspects of the nutritional composition of the competitive foods offered and sold to students, and their relative contribution to reference values. Appendix E provides detailed tables of the nutritional characteristics of all competitive food items available during the school day, by school and source.

Foods Offered and Sold

Table 4.1 provides a review of the sources, locations, and time periods competitive foods are available to students during the school day. All three schools offer a la carte-only and vended foods and beverages in the cafeteria during lunch. Schools #2 and #3 also offer food and beverages outside the lunch period (vending machines), and outside the cafeteria in school stores. Competitive foods are not available outside the cafeteria or outside the lunch period in School #1.

² DVs comprise two sets of reference values: Daily Reference Values (DRVs), for macronutrients, and Reference Daily Intakes (RDIs) for vitamins and minerals (USFDA, 1993).

Table 4.1

	School #1		Scho	ool #2	School #3		
	During	Outside	During	Outside	During	Outside	
	Lunch	Lunch	Lunch	Lunch	Lunch	Lunch	
In cafeteria							
A la carte only	\checkmark				\checkmark		
Snack bar			\checkmark				
Vending machine (food service)			\checkmark	\checkmark	\checkmark	\checkmark	
Vending machine (outside contract)	~		~	~			
Outside cafeteria							
School store			\checkmark	\checkmark		\checkmark	
Sources: Principal and SFA Direct	tor Surveys.						

Sources of Competitive Foods Available to Students During the School Day

Number and Types of Foods Offered

Data on the number and types of competitive food and beverage items offered show the amount of choice available to students when selecting a snack, meal alternative, or beverage at school. As shown in Table 4.2, the total number of food and beverage items available in cafeterias varies considerably by school. As expected, fewer competitive foods are available at the middle school (School #1) than the high schools. Of the 23 items offered at School #1, about half are fruit juice and juice-based drinks available in the vending machine. The a la carte service at this school also offers juice-based drinks, plus a sports drink and spring water. A la carte food items include several types of baked goods, chips, and a cereal bar.

School #2 offers the largest number of items (118) overall, with approximately equal proportions available in vending machines and at the a la carte snack window. Beverages, mainly juice, juice-based drinks and flavored milk, comprise one third of all items. A wide variety of baked goods, frozen desserts, and snacks are offered, with especially large selections of chips (18 items) and ice cream novelties (18 items). Two types of salad are also available, with a choice of dressing (including one low-fat dressing). Beverages are more likely to be found in vending machines in School #2, whereas the largest selection of snacks and baked goods are sold at the a la carte snack window.

Forty-three competitive food items are available in the cafeteria at School #3—only a third as many available at School #2. Note also that in School #3 some of the same items offered a la carte are stocked in the vending machines. The number of unique items available to students is therefore somewhat fewer than the totals shown in Table 4.2. The largest selection of items in School #3 was observed among the frozen desserts, which are only sold on the a la carte line. School #3 also offers baked goods, fruit juice and juice-based drinks, and snacks such as chips and popcorn.

Sale 4.2

Numbers of Competitive Foods Available in the Cafeteria During the School Day, Overall and by Source

		School #1			School #2			School #3	
			Vending		Snack	Vending			Vending
Category/Food	Total	A la Carte	Machine	Total	Bar	Machine	Total	A la Carte	Machine
All items	23	18	15	118	58	60	43	27	16
Baked goods/Dessert	5	5		13	10	3	7	4	3
Cookies	1	1		8	6	1	1	1	
Doughnuts	2	2		—	—	—	—	—	—
Pastries (danish, toaster pastries, pies)	1	1		2	—	2	2	1	1
Snack cakes	1	1		3	3	—	4	2	2
Beverages	11	6	5	39	7	32	13	7	6
Fruit juice, 100%	2	—	2	4	_	4	1	—	1
Iced tea drinks, fortified	—	—	—	2	—	2	—	_	
Juice-based drinks, 11.5-12 oz	3		3	_	_	10		_	_
Juice-based drinks, 16-20 oz	3	3	_	3	_	3		_	_
Juice-based drinks, fortified, 11.5-12 oz	—	—	—	14	4	—	10	5	5
Juice-based drinks, fortified, 16-20 oz	1	1	_	_	_	_	1	1	_
Milk, flavored, skim or 1% fat, fortified			_	2	_	2		_	_
Milk, flavored, whole or 2% fat, fortified			_	9	2	7		_	_
Milk, whole	_	_	_	1	_	1	_	_	_
Sports drinks	1	1	_	_	_	_		_	_
Spring water	1	1	_	2	1	1	1	1	_
Yogurt drinks	—	—	—	2	—	2	—	—	—
Bread or grain products	1	1		14	10	4	1	1	
Cereal bars/Cereal mixes	1	1		1	1	_	_	_	
Crackers/Cracker sandwiches	_	_		6	4	2	_	_	
Granola bars	—			5	4	1	_	_	
Pretzels	_	_		1	_	1	1	1	
Rice or corn cakes	—	—		1	1	—	—	—	

(cont.)

Table 4.2

Numbers of Competitive Foods Available in the Cafeteria During the School Day, Overall and by Source

		School #1			School #2			School #3	
			Vending		Snack	Vending			Vending
Category/Food	Total	A la Carte	Machine	Total	Bar	Machine	Total	A la Carte	Machine
Frozen desserts				22	9	13	11	11	
Ice cream				18	8	10	10	10	
Non-dairy frozen desserts				4	1	3	1	1	
Salads				2	2				
Side salads ^a				2	2				
Snacks	6	6		28	20	8	11	4	7
Chips	5	5		18	11	7	7	2	5
Fruit roll-ups/Fruit snacks	_	_		5	5	_	_	_	_
Chips, baked or reduced fat	1	1		4	3	1	_	_	_
Popcorn	—			1	1	—	4	2	2

a Six varieties of salad dressing are offered with the salads.

Sources: Cafeteria Survey and Inventories of Snack Bar and Vending Machine Items.

In School #3, a larger variety of items is available a la carte than in the vending machines. Based on the inventory and validation data, it was also observed that six of the slots in the vending machine were never filled during the target week. It was not clear if this was typical or related to decreased ordering that week because of the proximity of data collection to the end of the school year.

Students have the option of selecting food and beverages generally considered healthful at all three schools. Spring water and 100% fruit juice are available at all three schools, and Schools #1 and #2 offer baked or reduced-fat chips. School #2 offers skim or 1% fat milk as an alternative to higher fat options. Other potentially more healthful snack options available at Schools #2 and #3 include pretzels, rice cakes, and frozen 100% fruit juice bars.

Table 4.3 shows the number of competitive food items available outside the cafeteria. Food and beverages are only available outside the cafeteria at the two high schools, in their school stores. The available data suggest that both school stores carry a similar number of items (about 40), but the mix of selections differs.³ The vast majority of foods available in the store at School #2 are candy items (88 percent). Beef jerky and chips are also sold; beverages and gum are not available. In contrast, the store at School #3 offers toaster pastries, cereal bars, sandwich crackers, juice drinks, and water, in addition to candy, gum, and chips. According to the store manager at School #3, because the school does not have a breakfast program and the store is open before school, some students purchase their breakfast from this food venue.

Foods Most Commonly Sold

Servings data from the Cafeteria Survey and inventories of competitive foods were used to identify the types and numbers of items actually purchased by students during the target week. These data were not available for all foods, for example, neither the SFA director nor study staff could obtain information on the number of beverages sold from the vending machine in School #1 during the target week. Table 4.4 shows the number and percentage of items for which sales data were available overall, and for each school. It is important to recognize the extent of missing data, especially for School #1, when interpreting findings for the types of foods purchased by students and their nutrient composition.

Tables 4.5 and 4.6 present the five top selling foods and beverages at each school, both inside and outside the cafeteria. Interestingly, despite differences between schools in the number and variety of choices available in the cafeterias, there are some striking similarities with regard to the foods and beverages sold most frequently.

Fruit punch, the top seller at the middle school, is also among the five top-selling items at the two high schools. The other most commonly purchased beverage is spring water, the second highest seller at both high schools. Chips, especially Doritos, were among the most popular items sold in Schools #1 and #2, and chocolate chip cookies were purchased frequently at both School #1 and School #3.

³ Note that data from School #2 were incomplete. The validation data collection revealed that the school store at School #2 offered a total of 59 items. In addition to the types of foods shown in Table 4.3, granola bars and cookies were also available for sale.

Table 4.3

Numbers of Competitive Foods Available Outside the Cafeteria During the School Day

Category/Food	School #1	School #2	School #3
All items		41	39
Baked goods/Dessert			2
Pastries (danish, toaster pastries, pies)			2
Beverages			3
Juice-based drinks. 16-20 fl oz			2
Spring water			1
Bread or grain products			5
Cereal bars/Cereal mixes			4
Crackers/Cracker sandwiches			1
Candy		36	22
Candy with chocolate		6	3
Candy with chocolate, single piece or pop		2	1
Candy without chocolate		11	3
Candy without chocolate, single piece or pop		17	2
Chewing gum		—	13
Snacks		5	7
Chips		4	6
Meat snacks (jerky)		1	1
Source: Inventory of School Store Food and Beverage	Items.		

Table 4.4

Competitive Food Sales During the School Day

		During Target Week			
School	Number of Products	Percent of Items with Sales Data	Number of Items Sold		
School #1	23	57%	1,102		
School #2	159	91	12,232		
School #3	82	96	4,257		
Total	264	89%	17,591		

Table 4.5

				Package	Number
School	Source	Food Category	Product	Size	Sold ^b
School #1	A la carte	Beverages	Snapple fruit punch	20 fl oz	216
	A la carte	Snacks	Doritos Cool Ranch	1.75 oz	122
	A la carte	Snacks	Cheetos Curls	1.75 oz	122
	A la carte	Snacks	Doritos Nacho	1.75 oz	122
	A la carte	Baked goods/Dessert	Grandma's chocolate chip cookies	2.75 oz	120
School #2	Snack bar	Bread or grain products	Goldfish crackers, cheddar	1.5 oz	585
	Vending	Beverages	Spring water	16 fl oz	373
	Snack bar	Snacks	Doritos Cool Ranch	1.75 oz	368
	Snack bar	Snacks	Lays Classic potato chips	1.75 oz	305
	Vending	Beverages	Veryfine fruit punch	11.5 oz	254
School #3	A la carte	Baked goods/Dessert	Chocolate chip cookie (school- baked)	2.5 oz	1,526
	A la carte	Beverages	Spring water	16.9 fl oz	811
	Vending	Beverages	Veryfine fruit punch Chocolate chip ice cream	11.5 fl oz	108
	A la carte	Frozen desserts	sandwich	4.5 oz	79
	A la carte	Frozen desserts	Snickers ice cream bar	2.0 oz	72

Top Five Selling Competitive Food Items Available Inside the Cafeteria^a

a Sales data were not available for vended beverages in School #1.

b Total for target week.

Sources: Cafeteria Survey and Inventories of Snack Bar and Vending Machine Items.

Table 4.6

Top Five Selling Competitive Food Items Available Outside the Cafeteria

School	Source	Food Category	Product	Package Size	Number Sold ^a
School #2	School store	Candy	Neon laser straw	.11 oz	1,200
	School store	Snacks	Slim Jim	.33 oz	600
	School store	Candy	Jolly Rancher pop	1 pop	300
	School store	Candy	Charms sweet and sour pop	1 pop	240
	School store	Candy	Charms fluffy cotton candy pop	1 рор	240
School #3	School store	Beverages	Hawaiian Punch Juicy Red	20 fl oz	88
	School store	Candy	Air Heads watermelon	0.55 oz	74
	School store	Candy	Winterfresh gum	5 sticks	61
	School store	Beverages	Hawaiian Punch Green Berry	20 fl oz	55
	School store	Beverages	DejaBlue purified water	20 fl oz	49
T (1 C	· · · 1 . D ·	C C 1 1//0		1	

a Total for target week. Data for School #2 were taken from invoices for previous week's order.

Source: Inventory of School Store Food and Beverage Items.

There are also some differences in the items sold at the sample schools. The most commonly purchased item at School #2 was goldfish crackers, which are not offered at either of the other two schools. In addition, although both high schools offer ice cream novelties, they appear to be top sellers only in School #3. At School #3 the top-selling item overall (chocolate chip cookies) is almost twice as likely to be purchased as the top-selling beverage (spring water), and nine times as likely to be purchased as the next most popular food (ice cream cookie sandwich). Students clearly have a strong preference for the chocolate chip cookies at School #3, which are served warm.

Finally, although some food and beverage items generally considered healthful are available in the cafeterias at all three schools (e.g., 100% fruit juice, lower fat snack chips, skim or 1% fat milk), they are not among the items students are most likely to purchase.

Outside the cafeteria, in the school store, students at School #2 are buying candy—especially the single piece or lollipop types (see Table 4.6).⁴ The other top-selling competitive food item in School #2 is beef jerky. Most likely, students buy more than one of these items; but this could not be confirmed, because sales cannot be tracked for individual students.

The top-selling items at the school store in School #3 include fruit punch, candy (not chocolate), gum, and bottled water. Sales of fruit punch may have been higher than usual during the data collection week. The store supervisor reported that the school was participating in a contest that would award \$3,000 to the school store that sold the most Hawaiian Punch during the month. The price of the punch had been lowered to \$0.75 to encourage sales.

Nutrient Composition of Competitive Foods

The food energy, macronutrient, and micronutrient composition of competitive foods available during the school day are presented in this section. Mean nutrient values for all foods **as offered** are discussed first, followed by the nutrient composition of foods **as served**. All nutrient values are reported on a per-package basis.

To determine nutrients as served (or selected), the average nutrient content of items in each school was weighted, using available data on actual sales to students; each school's data were then weighted equally in combining results across schools. As shown in Table 4.4, sales data were missing for about 40 percent of the competitive items offered in School #1. It was not possible to impute values for these items (fruit juice and juice-based drinks); rather than drop the school from the analysis altogether, however, results are tabulated using the data available. The weighted data are therefore biased by the exclusion of vended beverages at School #1. The potential impact on findings would be an underestimate of the weighted mean amounts of nutrients such as vitamins A and C, and an overestimate of calcium, fat, and sodium.

Tables 4.7 through 4.13 present results of unweighted analyses, and Tables 4.14 through 4.16 show weighted results.

⁴ One reason some items sell is low cost. The range of prices is greater for items available outside *versus* inside the cafeteria.

Food Energy and Macronutrients

Competitive foods were analyzed for food energy (calorie) and seven macronutrients (Tables 4.7 and 4.14). The text discussion, however, is limited to calories, total fat, saturated fat, and sodium. These dietary components are most likely to contribute to problems with overweight or obesity, or have been shown to be problematic in children's diets.⁵ The DRVs for the macronutrients analyzed for this study are shown below.

- Total fat: 65 g
- Saturated fat: 20 g
- Carbohydrate: 300 mg
- Protein: 50 g
- Cholesterol: 300 mg
- Sodium: 2,400 mg
- Dietary fiber: 25 g

The 1989 REAs for children age 7 to 18 are used as a point of reference when discussing the food energy content of competitive foods.⁶

All Food Available During School Day

Food energy. The mean food energy, or caloric content, of competitive foods overall ranges from zero (spring water) to just over 400 calories. Foods in the 400-calorie range include snack cakes and whole or 2% fat flavored milk (14- to 16-ounce size). One serving of these items provides as much as 20 percent of the average food energy allowance (1989 REA) for children age 7 to 10, and from 13 to 18 percent of REA for boys and girls age 11 to 18.

As a group, baked goods and desserts are highest in calories (mean of 363 calories) among all competitive foods available during the school day. Their average caloric content exceeds that of other snacks and beverages by more than 130 calories (60 percent). Candy and gum, on average, are lowest in calories (mean of 142 calories or 5 to 7 percent of REA). This can be attributed in part to the relatively low fat content of candy items without chocolate, and the small size of many of the items sold. The relative ranking of the mean caloric contribution of baked goods/desserts and candy/gum remains the same when weighted by student selections. Changes in the average caloric content as a result of the weighted analysis were not nutritionally significant, amounting to only 2 percent or less of REA.

Total fat and saturated fat. Not surprisingly, baked goods/desserts also contain the most total fat (16 g, on average) and a substantial amount of saturated fat (mean of 4 g). This group of competitive food items provides about one fourth of the DRV for fat and one fifth the DRV for saturated fat. Snacks and salad with dressing also contribute substantial amounts of total fat (mean of 13 g for both

⁵ Prior studies of schoolchildren's dietary intake have found they consume in excess of the *Dietary Guidelines* recommendations for fat and saturated fat, and more than the National Research Council's recommendation for sodium (Gleason and Suitor, 2001). They also show that a substantial number of children consume large amounts of soda and juice-based drinks high in added sugar. At the same time, many children fail to consume enough fruit and vegetables, and low intakes of some vitamins and minerals are a particular concern for teenage girls.

⁶ Average energy requirements for children age 7 to 10 are 2,000 calories; for children age 11 to 18 the range is from 2,200 to 3,000 calories per day (NRC, 1989).

groups, or 20 percent of DRV), and frozen desserts offer the most saturated fat (5.5 g, or 28 percent of DRV, on average). Alternatively, beverages and candy/gum offer the least fat, with both groups providing no more than 5 percent of the DRV for either total fat or saturated fat.

There was little change in the mean fat or saturated fat content (or percent of DRV) of competitive foods when examined as selected, for any of the major food categories. The largest difference was a drop in the average saturated fat for beverages of 6.5 percent DRV. This is consistent with the finding that fruit punch was one of the top-selling items in all three schools, whereas milk was not among the items most commonly purchased.

Percent of calories from fat and saturated fat. When measured relative to caloric content, the fat and saturated fat composition of competitive foods look similar to mean amounts. Compared to other food groups, salads with dressing have the largest share of calories from total fat (77 percent), and frozen desserts contain the greatest percentage of calories from saturated fat (24 percent). Beverages, as a group, derive the smallest percentages of calories both from total and saturated fat (6 percent and 3.5 percent, respectively).⁷ This is representative of results of analyses of the percentage of calories from fat and saturated in competitive foods both as offered and as selected.

Tables 4.8 and 4.9 show, for each school, the number and percentage of competitive food items with 30 percent or less of calories from total fat and less than 10 percent of calories from saturated fat. These benchmarks are the same as the *Dietary Guidelines*-based standards for the maximum percent of calories from fat in saturated fat in school meals. Although not typically applied to foods, they offer one way of identifying items that might enhance or detract from achieving total daily intakes consistent with the *Dietary Guidelines*.

Among the competitive foods offered in the cafeteria, beverages were most likely to meet the total fat and saturated fat benchmarks at all three schools (72 to 100 percent of beverage items). The beverages with more than 30 percent of calories from fat and 10 percent or more of calories from saturated fat include flavored and whole milk, which were offered only in School #2. None of the snacks or baked goods available at Schools #1 and #3, and only a small share at School #2, contains 30 percent or fewer calories from total fat. More snacks and baked goods offered meet the less-than-10 percent cutoff for saturated fat, but this varies by school. Frozen desserts available at Schools #2 and #3 are least likely to contain either 30 percent or fewer calories from total fat or less than 10 percent of calories from saturated fat.

Outside the cafeteria, almost all of the baked goods, beverages, and bread/grain products meet the total fat and saturated fat benchmarks. These items are small in number and are offered only at School #3. In addition, a large share of the candy (including gum) selections available at both School #2 and School #3 offer 30 percent or fewer calories from total fat and less than 10 percent of calories from saturated fat.

Sodium. The mean amount of sodium in competitive foods ranges from a low of 2 mg for chewing gum to 729 mg for pretzels (approximately 30 percent of DRV for sodium) (see Tables 4.7 and 4.14). Salad with dressing has the highest average sodium content (364 mg), followed by snacks (312 mg) and bread/grain products (296 mg). One serving from these food groups provides from 12 to 15

⁷ Results for beverages are somewhat misleading, because the fat contents of milk and other, non-dairy beverages, are so different.

percent of the sodium DRV. Candy and gum combined contain only 28 mg sodium, on average. Frozen desserts and beverages also provide relatively small amounts of sodium (means of 58 and 100 mg, respectively, which are less than 5 percent of DRV).

There are only minor shifts in the relative ranking of competitive foods when the analysis of the amount of sodium is weighted by student selections. For example, bread/grain products are second to salads in mean sodium content (380 mg and 406 mg, respectively) instead of snacks (279 mg).

Food Inside the Cafeteria

Food energy. The food energy and macronutrient composition of competitive foods available in school cafeterias looks much the same as for all food venues combined. Most of the highest calorie items are available in the cafeteria, including snack cakes, doughnuts, and pastries (367 to 414 calories), as well as milk (300 to 407 calories). At the same time, there are numerous choices with considerably fewer calories. For example, rice cakes and non-dairy frozen desserts (e.g., frozen 100% juice bars) have less than 100 calories, on average (5 percent or less of REA), and many items provide fewer than 200 calories (or 7 to 10 percent of REA): cereal bars/mixes, popcorn, pretzels, fruit rollups, reduced-fat chips, and salads. Although some beverages, especially in 16- to 20-ounce sizes, tend to be more caloric, others offer less than 200 calories: water, 100% fruit juice, iced tea, some juice-drinks (11.5- to 12-ounce), and sports drinks. The calorie content of competitive beverage choices in the cafeteria is somewhat lower when examined as served *versus* as offered.

Total fat and saturated fat. As for all foods, the foods available in the cafeteria that contain the most total fat are snack cakes, doughnuts, cookies, and chips (15 to 23 g of fat, on average). Whole and 2% fat milk contain the highest mean amounts of total fat among the available beverages (14 to 16 g). One serving of these competitive items provides from 21 to 35 percent of the DRV for total fat. Generally, the items with the most fat also have the highest amounts of saturated fat.⁸ In the case of whole milk, it also has the most saturated fat relative to calories (30 percent of calories) of all items offered.

Students are purchasing the higher fat baked goods and beverages, as evidenced by the small increases in mean fat content for these items after weighting (chocolate chip cookies were among the top sellers). They are also more likely to select chips with less fat, that is, those in smaller packages—not necessarily the reduced-fat varieties. This may relate to issues of price or convenience, rather than a consideration of nutritional quality.

Percent of calories from fat and saturated fat. Aside from the salads mentioned above, many competitive foods provide more than 30 percent of calories from total fat. These are generally the same items highest in mean total fat, but also include popcorn, ice cream, and crackers. The higher fat items tend to provide more than 10 percent of calories from saturated fat, ranging from 11 percent for crackers to 30 percent for whole milk. Of note is the relatively high saturated fat content of granola bars: 20 percent of calories compared to 32 percent of calories from total fat. After weighting by student selections, somewhat more foods have average percentages of calories from fat and saturated fat that exceed these benchmarks.

⁸ One notable exception is the salads with dressing, where the largest share of total fat is unsaturated.

A number of the competitive items offered in the cafeteria contain little to no fat, including water, yogurt drinks (skim-milk based), juice and sports drinks, non-dairy frozen desserts, rice cakes, pretzels, and fruit rollups/snacks. These items also derive the lowest percentages of total calories from fat and saturated fat relative to other selections (when examined as offered and as served). Although baked or reduced-fat chips are about 50 percent lower in total fat than their "regular" counterparts, they still provide 8 g (12 percent of DRV for fat) and almost 40 percent of total calories as fat. The reduced-fat chips, however, do contain substantially less saturated fat as a percent of calories than the regular varieties.

Sodium. Students can select food items with as much as 30 percent of the DRV for sodium, such as pretzels (mean of 729 mg). Doughnuts, chips, and crackers are other higher sodium options, providing from 16 to 18 percent of the sodium DRV. Some of the lower sodium choices available include 100% fruit juice, water, and non-dairy frozen desserts. These items provide as little as 10 to 15 mg sodium. Even ice cream contributes less than 5 percent of the DRV for sodium (mean of 66 mg). With the exception of a decrease in mean sodium for chips (about 4 percent of DRV), these findings are consistent regardless of whether data are examined as offered or as served. The lower mean sodium value for chips as selected can be attributed to less frequent purchases of chips in 3- to 4-ounce packages versus the 1- to 2-ounce sizes.

Food Outside the Cafeteria

Food energy. The food energy and macronutrient content of competitive foods offered outside the cafeteria, in school stores, is shown in Tables 4.7 and 4.14. Items that can only be purchased outside the cafeteria include candy and gum, and meat snacks (beef jerky). Gum is relatively low in calories (mean value of 97 calories is for full packages, ranging from 5 to 17 sticks), and contributes virtually nothing else nutritionally. The caloric content of candy varies, however, mainly due to differences in package size. Candy with and without chocolate provides as much as 10 percent of the REA (211 to 245 calories). Beef jerky is relatively low in calories (mean of 76 calories), although more than half of the calories are from total fat and about one quarter are from saturated fat. There are no important differences between the mean calories as offered and as sold for items outside the cafeteria.

Total fat and saturated fat. Sources of fat and saturated fat differ somewhat for items available outside compared with inside the cafeteria. For example, chips and chocolate candy provide the most fat among the selections offered at school stores (12 to 13 g, or about 20 percent of DRV for total fat). These items are also the biggest contributors of saturated fat. Chocolate candy, with a mean saturated fat content of 5 g (25 percent of DRV), derives about 45 percent of its calories from total fat and 18 percent from saturated fat. This does not change when weighted by student sales.

Sodium. Competitive foods are available in school stores with mean sodium values just over 400 mg, or 17 to 18 percent of the DRV (toaster pastries and beef jerky). Lower sodium items are found mainly among the candy and gum selections. The increase in mean sodium for the cereal bar/mixes group when analyzed as sold (136 mg, or 6 percent of DRV), reflects student purchases of the highest sodium option in this food category (cereal snack mix).

Micronutrients

Competitive foods were analyzed for their micronutrient content, specifically, vitamins and minerals for which standards exist for school meals (vitamins A and C, calcium, and iron), and selected B
vitamins (thiamin, riboflavin, niacin, and vitamin B_6).⁹ The mean nutrient values for each food category are shown in Tables 4.10 and 4.15. In addition, findings are discussed relative to the contribution competitive foods make to RDIs.¹⁰ RDIs for the nutrients assessed are as follows:

- Vitamin A: 5,000 IU
- Vitamin C: 60 mg
- Thiamin: 1.5 mg
- Riboflavin: 1.7 mg
- Niacin: 20 mg
- Vitamin B6: 2 mg
- Calcium: 1,000 mg
- Iron: 18 mg

Results for B vitamins are included in tables but not discussed in detail.

All Food Available During School Day

Vitamin A. The mean vitamin A content of competitive foods accessible during the school day varies across food categories. Salads with dressing contribute the most significant amounts of vitamin A (mean of 250 μ g retinol equivalents (RE)), whereas candy/gum contain virtually no vitamin A. As a group, beverages also provide substantial amounts of vitamin A (97 μ g RE, on average), especially milk and juice drinks with added vitamin A. Salads and beverages offer an average of 17 percent and 6 percent of the RDI for vitamin A, respectively.¹¹ None of the other food groups offer vitamin A at levels greater than 5 percent of the RDI. After weighting by student selections, frozen desserts contribute more vitamin A, on average, than beverages. This reflects a small reduction (approximately 4 percent of RDI) in the mean vitamin A content of beverages. Students are less likely to select milk and juice-based drinks fortified with vitamin A than other beverages.

Vitamin C. Beverages and salads are also the main source of vitamin C in competitive foods. The mean amount of vitamin C in beverages overall (58 mg) approximates 100 percent of the RDI. Salads with dressing provide about one fourth as much vitamin C as beverages (mean of 16 mg, or 27 percent of RDI). The only other group of items that contributes an amount of vitamin C that exceeds 5 percent of the RDI is snacks (mean of 7 mg, or 12 percent of RDI for vitamin C). Although beverages as selected remain the most important source of vitamin C, the mean contribution drops by 18 percent of the vitamin C RDI.

Calcium. In contrast to findings for vitamins A and C, most of the categories of competitive foods provide calcium in amounts over 5 percent of the RDI. The exceptions are candy/gum and snacks.

⁹ The B vitamins included in the analysis represent those used to identify FMNV and/or used to fortify certain juicebased beverages available as competitive items.

¹⁰ RDIs are a population-weighted average of the RDAs for vitamins and minerals for healthy individuals over 4 years old. RDIs were chosen over the age- and gender-specific RDAs because SFA directors and others making decisions about foods to offer students may not have this information readily available. FMNVs are also determined by their relative contribution to the RDI.

¹¹ Note that the RDI for vitamin A is expressed as International Units (IU), whereas vitamin A values in the tables appear as retinol equivalents (RE). Reported percentages of RDA for vitamin A are based on the following conversion: 1 μ g RE = 3.33 IU vitamin A.

Beverages offer by far the most calcium of all items. The average calcium content of the available beverages is 198 mg, or almost 20 percent of the calcium RDI. Bread/grain products, baked goods, salads, and frozen desserts offer from 58 to 72 mg calcium (6 to 7 percent of RDI). As expected, the mean calcium content of beverages decreases after taking into account students' relative purchases of milk and non-dairy beverages. The change amounts to a mean of 134 mg calcium, or 13 of RDI.

Iron. The grain-based items are the only substantial sources of iron among all competitive foods offered. Baked goods and bread/grain products, either processed with enriched flour or whole grains, provide an average of 2 mg iron or 11 percent of the iron RDI. Salads with dressing are the only other foods offered with more than 5 percent of the RDI for iron (1 mg iron, or 5.5 percent of RDI). Results for iron are similar whether examined as offered or selected.

Food Inside the Cafeteria

Vitamin A. The micronutrient composition of competitive foods for sale in the cafeteria differs little from the micronutrient picture for all foods combined. Major sources of vitamin A include salads, milk, particularly fortified skim or 1% lowfat milk, and juice-based drinks with added vitamin A. The vitamin A contribution for these beverages averages 19 percent and 14 percent of RDI, respectively. As mentioned above, students are not purchasing these beverages as often as alternative beverage choices.

Vitamin C. Fruit juice (100% juice), whether examined as offered or as selected, is still the richest source of vitamin C available in the cafeteria (178 to 233 percent of RDI), despite fortification of some of the other beverage choices.

Calcium. There is a small reduction in the mean calcium and iron contributed by bread/grain products in the cafeteria. It appears that the cereal bars/mixes offered here are substantially lower in calcium and iron content than their calcium- and iron-fortified counterparts for sale in the school stores. Nonetheless, the most important sources of calcium are milk and 100% fruit juice, which are only available in the cafeteria. The milk choices provide from 583 to 817 mg calcium, on average, (58 to 82 percent of RDI) and 100% fruit juice contributes a mean of 185 mg calcium (18 percent of RDI). The relative contributions of calcium from fruit juice and milk remain the same when weighted by student selections.

Iron. In the cafeteria, baked goods/desserts offer the most iron (when examined as offered and as selected). Snack cakes contain the highest mean amount of iron at 2.3 mg (13 percent of RDI). One reason why the baked goods contain more iron than other grain-based items is their size. For example, most snack cakes and pastries for sale weigh between 3 and 4 ounces, whereas cereal and granola bars, crackers, and pretzels weigh from 1 to 2 oz.

Food Outside the Cafeteria

Vitamin A. The micronutrients provided by foods available outside the cafeteria are limited, and generally reflect the offerings of a few items that have been fortified. In contrast to foods and beverages offered in the cafeteria, the only substantial contributor of vitamin A in the school store (as offered and as sold) is toaster pastries (mean of 299 μ g RE, or 20 percent of RDI). These items are also fortified with B vitamins and iron, at approximately 20 percent of the RDI for these nutrients. Toaster pastries also contain about 400 calories per serving, or 13 to 20 percent of the REA.

Vitamin C. Juice drinks (Hawaiian punch) are the single significant source of vitamin C offered outside the cafeteria. A 20-ounce serving offers three times the RDI for vitamin C (and 290 calories). Some candy items also provide vitamin C, specifically, fruit-flavored candies that have been fortified with vitamin C (e.g., Skittles, Starburst fruit chews and sours, taffy).

Calcium and iron. Few competitive food items offered outside the cafeteria provide nutritionally important amounts of calcium or iron. In addition to the toaster pastries noted above, fortified cereal bars and cereal mixes are the exceptions. The mean calcium and iron values for these items are 133 mg and 4.8 mg, respectively. On average, the cereal bars/mixes available offer 13 percent of the RDI for calcium, and 27 percent of the RDI for iron. They also contain about 5 percent of the RDI for vitamin A, and from 85 to 213 calories. After weighting, the average calcium (and vitamin A) value for cereal bars/mixes drops slightly, and iron increases more substantially (about 8 percent of the iron RDI). This reflects the finding that students were more likely to select a cereal snack mix, with almost half the RDI for iron, than cereal bars.

Food Guide Pyramid Servings from Competitive Foods

Dietary guidance for health promotion includes daily recommendations for foods to eat from each of the five main Food Guide Pyramid food groups. The analyses presented in this section (Tables 4.11 and 4.16) show the mean number of servings contributed by competitive foods to the grain, vegetable, fruit, dairy, and meat/meat substitute groups. Competitive foods were also analyzed for mean amounts of discretionary fat and added sugars.¹² Results for the relative amounts of discretionary fat roughly parallel the findings for total fat presented above, so they are not discussed in detail here.

To provide a context for interpreting the Pyramid servings analysis, the daily-recommended numbers of servings for each of the main Food Guide Pyramid groups are reviewed below. The ranges of servings shown are based on calorie levels of 1,600 to 2,800 calories per day.

- Grain group: 6 to 11 servings
- Vegetable group: 3 to 5 servings
- Fruit group: 2 to 4 servings
- Dairy group: 2 to 3 servings
- Meat group: 5 to 7 ounces

Food Pyramid recommendations also include guidelines for maximum daily intakes of discretionary fat and added sugars:

- Discretionary fat: 53 to 93 g
- Added sugars: 6 to 18 teaspoons

These recommendations are intended to apply to children age 7 and above (USDA/ARS, 2000).

¹² Discretionary fat includes all "excess" fat from the five major food groups beyond amounts that would be consumed if only the lowest fat forms were eaten, as well as fats added to foods in preparation or at the table. Added sugars include sugar and sweeteners eaten separately or used as ingredients in processed or prepared foods. They do not include naturally occurring sugars, such as lactose in milk or the fructose in fruit (USDA/ARS, 2000).

All Food Available During School Day

A large share of the competitive foods available to students during the school day offers little in the way of main Pyramid food group servings. Some foods and beverages, however, do make a positive contribution. The types of foods that contribute at least one-half serving of a food from a Pyramid food group are discussed here.

Main food groups. Foods contributing to the grain group fall mainly in the baked goods, bread/grain products, and snack categories. These foods provide, on average, between 1 and $1\frac{3}{4}$ servings of grains. Salads with dressing provide an average of $1\frac{1}{2}$ vegetable servings, and beverages contribute to both the fruit and dairy groups. Overall, beverages provide a mean of $\frac{1}{2}$ serving of both fruit and dairy, although milk and 100% fruit juice offer considerably more (about 2 servings from the dairy and fruit groups, respectively). Because students were not as likely to select milk or fruit juice as other beverages, however, the weighted mean number of servings for beverages drops below $\frac{1}{2}$ serving of both dairy and fruit.

Added sugars. Baked goods and beverages contain the highest amounts of added sugars among all competitive foods offered, averaging about 7 teaspoons each (more than the daily recommended maximum for a 1,600 calorie diet). The added sugar content of breads/grains and snacks is considerably lower (means of $1\frac{1}{2}$ and $\frac{1}{2}$ teaspoons, respectively), and salads with dressing contain less than 1/2 teaspoon added sugar. The effect of considering student selection patterns in the analysis is to reduce mean discretionary fat, but increase the added sugar contribution from beverages.

Food Inside the Cafeteria

Main food groups. The contribution to daily recommended Pyramid servings from competitive foods available in the cafeteria is essentially the same as for all foods, as described above. Foods are available that provide from 1 to 2 servings from the grain group, $\frac{1}{2}$ to $\frac{1}{2}$ servings from the vegetable group, and 1 to 2 servings from each of the fruit and dairy groups. There are no significant sources of meat/meat substitutes offered outside the meal programs in the cafeteria. These findings represent results of analyses for competitive foods both as offered and as selected.

Foods in the cafeteria that offer the most servings from the grain group include pretzels (mean of 2.6 servings) and pastries (mean of 2.1 servings). On the other hand, the items that provide the most **whole grain** servings, on average, are rice/corn cakes ($1\frac{1}{2}$ servings), popcorn ($1\frac{1}{4}$ servings), and granola bars and regular chips (each 1 serving) (data not shown). Chips—primarily potato chips—are also the only competitive foods in the cafeteria, other than salads, that make a contribution to the vegetable group. Regular and baked or reduced fat chips provide between $\frac{1}{2}$ and $\frac{3}{4}$ servings of white potato from the vegetable group (data not shown).

Added sugars. The main sources of added sugars in the competitive beverages available in the cafeteria are juice-based drinks and sports drinks (9 to 13 teaspoons, on average). These beverages also contain from 0 to 1 serving of fruit, much less than an average of almost $2\frac{1}{4}$ servings of fruit from 100% juice. Conversely, although yogurt drinks contain almost $7\frac{1}{2}$ teaspoons added sugar, they contribute an important amount of fruit (mean of $1\frac{1}{2}$ servings) and almost $\frac{1}{2}$ serving of dairy. Baked goods and non-dairy frozen desserts also offer substantial amounts of added sugar, ranging from 5 to 9 teaspoons. Crackers, granola bars, salads, and chips offer lower sugar options among the items available in the cafeteria. Except for beverages, weighting by student selections does not change the relative amounts of added sugars for any of these competitive foods.

Food Outside the Cafeteria

Main food groups. Few foods available outside the cafeteria make a positive contribution to daily servings of Pyramid food groups, when examined both as offered and as sold. The only item among the candy and gum offerings that approximates a ¹/₂-serving contribution to any of the Pyramid food groups is chocolate candy (specifically, Twix caramel cookie bar with ¹/₂ grain serving). Chocolate candy also provides one of the highest amounts of discretionary fat among all items offered in the school stores. One item available in both high schools provides a small amount (¹/₂ ounce) of meat/ meat substitutes: beef jerky. As discussed earlier, this food also contains significant amounts of fat and saturated fat relative to calories, and a substantial amount of sodium.

Foods available in the school stores that offer at least 1 serving from one of Pyramid groups include toaster pastries, which contribute an average of 3 servings of grain, and cereal bars/mixes, which provide 1 grain serving. The chips offered in school stores contribute almost 1 serving of vegetable (white potato).

Added sugars. Among the items available outside the cafeteria, the major contributors of added sugar include juice-based drinks (mean of 16 teaspoons) and candy without chocolate (mean of 12 teaspoons). Note that the juice drinks available outside the cafeteria at 16- to 20 ounces contain more sugar, on average, than those offered in the cafeteria. Lower sugar options in the school stores include gum (some of which is sugar-free), chips, cereal bars, and crackers. This is does not vary when the analysis takes into account student sales.

Foods Meeting FDA Labeling Criteria

In addition to examining food groups and nutrient composition, competitive foods were analyzed relative to FDA criteria for nutrient content claims. FDA defines three categories for nutrient content labeling claims: "free", "low", and "reduced". The first two categories were applied to study data; these categories are defined by absolute standards—for example, the number of grams of fat per serving or per reference amount.¹³ The "reduced" category was not used for this study because it is defined relative to an appropriate reference product, and it was beyond the scope of the study to apply those standards. The FDA nutrient content claims criteria used are listed below.¹⁴

Calories

- Calorie free: less than 5 calories per reference amount and per labeled serving
- Low calorie: 40 calories or less per reference amount (and per 50 g if reference amount is small)

Total fat

- Fat free: less than 0.5 g per reference amount and per labeled serving
- Low fat: 3 g or less per 100 g per reference amount (and per 50 g if reference amount is small)

¹³ The reference amount is the amount customarily consumed per eating occasion, as defined by FDA (21 CFR 101.12). A small reference amount is a reference amount of 30 g or less or 2 tablespoons or less (USFDA, 1999).

¹⁴ FDA also specifies criteria for labeling claims about sugars. The criterion for "sugar free" requires that the product have less than 0.5 mg total sugars per labeled serving. Because the nutrient database used for the feasibility study does not provide values for total sugars, this criterion could not be utilized.

Saturated fat

- Saturated fat free: less than 0.5 g saturated fat and less than 0.5 g trans fatty acids per reference amount and per labeled serving
- Low saturated fat: 1 g or less per reference amount and not more than 15 percent of calories from saturated fat

Cholesterol

- Cholesterol free: less than 2 mg per reference amount and per labeled serving
- Low cholesterol: not more than 20 mg per reference amount (and per 50 g if reference amount is small)

Note: cholesterol claims are only allowed when food contains 2 g or less saturated fat per reference amount.

Sodium

- Sodium free: less than 5 mg per reference amount and per labeled serving
- Low sodium: not more than 140 mg per reference amount (and per 50 g if reference amount is small)

Table 4.12 shows the proportions of competitive foods that meet the FDA labeling criteria for calories, total fat, saturated fat, cholesterol, or sodium. The analysis used the same broad food categories shown in tabulations of nutrient composition and Pyramid food groups; some foods within these categories, however, were combined due to small numbers of items. Foods meeting multiple labeling criteria are identified in the detailed tables in Appendix E.

All Food Available During School Day

Calories. The only competitive food available during the school day that meets the **calorie-free** criterion is spring water. None of the items offered is considered **low-calorie**, according to FDA labeling criteria. A 40-calorie cutoff may be too limiting for purposes of evaluating competitive foods and beverages (e.g., rice/corn cakes are generally considered low calorie yet contain more than 40 calories per serving).

Total fat. Competitive foods identified as **fat-free** include most of the beverage selections (80 percent overall), about half of the candy/gum choices (48 percent), and 15 percent of all frozen desserts. Few of the remaining items in these food categories meet the **low-fat** criterion, although about one fourth of bread/grain products (24 percent) and a small share of the snacks (9 percent) offered are low fat. None of the baked goods/desserts meet the FDA definitions for fat free or low fat.

Saturated fat. With regard to saturated fat, beverages and candy/gum are among the foods most likely to be low in saturated fat. Over 80 percent of the items in these food categories meet the **low saturated fat** criterion. Additionally, more than half (52 percent) of the bread/grain products are low in saturated fat. In contrast, much smaller proportions (less than 20 percent) of baked goods, snacks, and frozen desserts are low in saturated fat.

Cholesterol. All candy and gum, and most bread/grain products and beverages (90 and 83 percent) are either **cholesterol-free** or **low-cholesterol**. For all other food categories, the percentage of

cholesterol-free or low-cholesterol items is lower, ranging from 42 to 58 percent. In all cases, findings for cholesterol claims vary considerably within food categories. For example, among the beverages available, all juice and juice-based drinks are cholesterol-free, but none of the dairy items are cholesterol-free, and only 21 percent are low-cholesterol. Results are similar for the non-dairy frozen dessert and ice cream items. In the snacks category, all of the fruit rollups/snacks are cholesterol-free, but not all of the chips or the beef jerky.¹⁵

Note that FDA limits cholesterol claims to foods that have no more than 2 g of saturated fat per reference amount. This explains why all baked and reduced fat chips meet the cholesterol-free criterion, but not the regular chips or popcorn.

Sodium. Few competitive foods overall meet the FDA criterion for **sodium-free**, although a substantial share of items can be classified as **low-sodium**. Low-sodium items may be found among the frozen desserts (91 percent), beverages (76 percent), and candy/gum (72 percent) categories. Salads with dressing and the vast majority of snacks are not considered low-sodium, according to FDA criteria.

Food Inside the Cafeteria

Other than eliminating the candy and gum, there are only minor shifts in the percentages of competitive foods that meet FDA labeling criteria when the criteria are applied to only those items available in the cafeteria. The most notable change is seen for snacks, where slightly more (6.5 percent) items in the cafeteria are low in saturated fat compared to all foods available at school. This difference is mainly attributable to meat snacks (beef jerky), which are only available outside the cafeteria.

Food Outside the Cafeteria

Given the different mix of competitive items available inside and outside the cafeteria, the disparities in the share of items meeting FDA labeling criteria between the two locations are not surprising. With regard to total fat, all of the beverages (n=3) offered in school stores are fat-free, compared with about 80 percent in the cafeterias. The difference is due to milk and yogurt-based drinks offered only in the cafeteria. Additionally, none of the snacks available outside the cafeteria is either fat-free or low fat—fruit rollups or similar items are not sold in the stores. The majority of fat-free or low-fat candy/gum items in the school stores are either chewing gum or candy without chocolate.

A fairly large proportion of items available in the school stores can be classified as low in saturated fat and cholesterol-free, including pastries. This finding can be explained, in part, by the relatively small number of selections among food categories other than candy, gum, and snacks. Foods that do not meet the saturated fat and cholesterol criteria include beef jerky, crackers, and most chocolate candy; chips and popcorn cannot be classified as low in saturated fat, and only 40 percent of the items offered are cholesterol-free.

Foods of Minimal Nutritional Value

USDA policy on competitive foods prohibits the sale of FMNVs, including beverages, in the food service area during school meal periods (7 CFR 210.11). These regulations do not prohibit the sale of

¹⁵ In addition, one of two salads with dressing (Caesar salad) did not meet the cholesterol criteria.

FMNVs outside the food service area during meal periods or at any other time during the school day. As noted in Chapter 3, however, Virginia policy prohibits the sale of FMNVs during meal periods anywhere in the school.

FMNVs are defined in two ways: (1) based on their nutrient composition relative to RDIs, and (2) in terms of specific categories of foods. The regulations are as follows:

- 1. In the case of artificially sweetened foods, a food that provides less than 5 percent of the RDIs for each of eight specified nutrients per serving; and
- 2. In the case of all other foods, a food that provides less than 5 percent of the RDI per 100 calories and per serving. The eight nutrients to be assessed for this purpose are protein, vitamin A, vitamin C, niacin, riboflavin, thiamin, calcium, and iron.

The categories of FMNV are soda water, water ices, chewing gum, certain candies, hard candy, jellies and gums, marshmallow candies, fondant, licorice, spun candy, and candy coated popcorn.

Although the USDA regulations effectively prohibit chewing gum, many types of candy, and carbonated soft drinks, they do not address the sale of all types of candy or sports drinks, and allow such items as fortified juice-based drinks high in sugar content. They also do not specifically mention whether non-carbonated water is a FMNV. Because Virginia policy allows water to be sold at any time during the school day, it was not classified as a FMNV for this analysis.

An analysis was conducted to identify the number of FMNV items available during the school day at all three schools. Table 4.13 shows the results for each source of competitive foods and for all sources combined. As expected, FMNVs are generally only available outside the cafeteria in the school stores. The exceptions are a handful of foods sold in the cafeteria at School #2:

- Fruit roll-ups
- Italian ices
- Popsicles

Note that the analysis for FMNV items was based on nutrient values derived from the FIAS nutrient database. Results may have been different if the classifications were made from nutrient values provided by the manufacturer or listed on the Nutrition Facts label for the item. For example, the package labels for fruit roll-ups show that both sizes have 25 percent of the RDI for vitamin C (15 mg), whereas FIAS values are only 1 mg vitamin C for these foods.¹⁶

With regard to the competitive foods available outside the cafeteria in School #2, approximately 37 percent of all items (foods only) offered in the store are FMNV items. That estimate rises to 52 percent when weighted by student sales data. In contrast, more than 40 percent of all school store items at School #3 are FMNV items; this drops to 28 percent of store items after factoring in what students are actually purchasing.

¹⁶ In addition, manufacturers may request and be granted a waiver for their particular brand of fruit roll-up.

Foods of Minimum Nutritional Value (FMNV) Available During the School Day^a

	Foods	and Beverages	Combined		Foods		Beverages			
School/Source	Number FMNV Items	Percent of All Items from Source	Weighted Percent of All Items ^b	Number FMNV Items	Percent of All Foods from Source	Weighted Percent of All Items ^b	Number FMNV Items	Percent of All Beverages from Source	Weighted Percent of All Items ^b	
School #1										
A la carte	0	0%	0%	0	0%	0%	0	0%	0%	
Vending machine	0	0	0	0	0	0	0	0	0	
All sources	0	0	0	0	0	0	0	0	0	
School #2										
Snack bar	2	3.5%	2.5%	2	3.9%	2.6%	0	0%	0%	
School store	15	36.6	51.7	15	36.6	51.7	0	0	0	
Vending machine	2	3.4	1.1	2	7.1	3.7	0	0	0	
All sources	19	12.1	24.8	19	15.8	30.3	0	0	0	
School #3										
A la carte	0	0%	0%	0	0%	0%	0	0%	0%	
School store	16	42.1	27.6	16	44.4	34.1	0	0	0	
Vending machine	0	0	0	0	0	0	0	0	0	
All sources	16	20.0	6.1	16	24.2	7.2	0	0	0	

a Excludes water.

b Weighted by total sales during target week.

Competitive foods sold in the school stores that did **not** meet the definition for FMNVs were taffy and other newer candies (e.g., Skittles, Air Heads), and Hawaiian fruit punch. Taffy is not listed among the types of candy categorically considered an FMNV, and it contains 6 percent of the RDI for vitamin C per piece; Skittles contain 68 percent of RDI for vitamin C per package; and 20 ounces of Hawaiian punch has 300 percent of RDI for vitamin C and 5 percent of the RDI for calcium.

Summary

This chapter examined the number, types, and nutritional characteristics of competitive foods offered and selected by students in three schools. Seven main food categories and 34 subcategories were used to summarize findings. The number of available competitive foods varied between the schools, with fewer offered at the middle school. Beverages, snacks, and frozen desserts accounted for most of competitive items offered in the cafeterias. Candy and snack items were the most commonly offered foods outside the cafeteria, in school stores. Based on available data, fruit punch, chocolate chip cookies, tortilla chips, and spring water were the most popular items sold to students.

Nutritional characteristics of the competitive food items were examined with the following measures:

- Mean food energy, macronutrient, and micronutrient content
- Percent of items with $\leq 30\%$ of calories from total fat/< 10% from saturated fat
- Mean servings of Food Guide Pyramid main food groups
- Percent of items meeting FDA regulations for labeling claims
- Percent of items meeting USDA criteria for food of minimal nutritional value (FMNV).

As expected, competitive food offerings included a large number of items with substantial amounts of food energy, total fat, and added sugar. Some lower calorie, lower fat items were also available, and FMNV were found mainly in school stores. The mean vitamin and mineral content of competitive foods was influenced by the availability of highly fortified food and beverage items. After weighting to take into account student selection patterns, the average calorie and fat content of beverages and snacks was somewhat lower, as were the number of servings of dairy and fruit. Students were not as likely to buy milk as juice drinks and bottled water.

Mean values of nutrients and Pyramid servings were difficult to interpret, except relative to some benchmarks. We chose to discuss the nutrient content of the various categories and subcategories of items in relation to Daily Values and FDA criteria for nutrition labeling claims. Both of these benchmarks can be easily observed on product labels and may influence choices made by students.

To be useful for monitoring purposes, criteria for classifying items as "healthful" or "less healthful" could be developed based on the mean percent of Daily Value for food energy and key nutrients in a single serving (package).¹⁷ Use of FDA labeling criteria for this purpose has some drawbacks: the criteria are somewhat complex to apply, they are probably too stringent for identifying low calorie items (less than 40 calories), and they do not always apply to the full package size.

¹⁷ Ideally, the criteria would consider food energy and nutrient content together, along with fat, added sugar, and possibly sodium content.

The competitive food items were also analyzed in relation to the Dietary Guidelines recommendations for daily intake of total fat and saturated fat. Although not typically applied to individual foods, these benchmarks can help identify food and beverage items that make a positive contribution toward meeting fat and saturated fat goals for the day. They are also the basis for standards for the fat and saturated fat content of USDA reimbursable schools meals.

One of the Ten Keys to Promote Healthy Eating in Schools included with the *Changing the Scene* materials recommends that food sold in addition to NSLP meals are "from the five major food groups of the Food Guide Pyramid." Many of the foods analyzed for this study contained only a fraction of a serving from one or more of these food groups. In addition, some foods that contributed one or more servings of a Pyramid group were also among the highest in calories and fat (e.g., baked goods). Although criteria for healthful competitive food items could be established based on relative contribution to daily-recommended servings of Pyramid groups, it seems likely that food energy and/or fat content would also need to be assessed.

Finally, USDA may want to consider revising the categories of FMNV to include sports drinks and taffy.

Unweighted Tables

Table 4.7Food Energy and Macronutrient Composition of Competitive Foods Available During the School Day, AllSchools and Sources

	Number	Mean			Mean	amount p	er item			Mean p calori	ercent of es from
	of items	calories per item	Total fat (g)	Satura- ted fat (g)	Carbo- hydrate (g)	Protein (g)	Choles- terol (mg)	Sodium (mg)	Dietary fiber (g)	Total fat	Saturated fat
Pakad gooda/doopart						-		-			
All items	27	362.7	15.8	3.8	53.0	4.0	16.7	265.5	1.6	38.5	9.5
Cookies	10	306.4	15.0	4.6	42.0	3.5	6.0	235.0	1.7	42.7	13.0
Doughnuts	2	380.6	22.9	5.0	41.6	4.2	40.4	414.6	1.5	53.9	11.6
Pastries (danish, toaster pastries,	_										
pies)	7	378.9	12.2	2.1	63.5	4.8	24.7	336.5	2.0	28.2	5.0
Shack cakes	8	414.3	18.2	4.0	60.7	3.9	17.0	204.2	1.2	38.6	8.6
Beverages											
All items	66	225.0	2.5	1.5	48.8	3.4	10.0	99.8	0.7	5.9	3.5
Fruit juice, 100% juice	7	179.4	0.4	0.0	43.3	1.6	0.0	11.9	0.6	2.3	0.3
Iced tea drinks, fortified	2	130.0	0.0	0.0	33.0	0.0	0.0	9.9	0.0	0.0	0.0
Juice-based drink, fortified, 16-20											
fl oz	2	237.2	0.0	0.0	58.2	0.0	0.0	345.0	0.0	0.0	0.0
Juice-based drinks, 11.5-12 fl oz	4	1/0.0	0.0	0.0	43.4	0.2	0.0	62.8	0.3	0.4	0.0
Juice-based drinks, 16-20 fl oz	8	272.0	0.0	0.0	69.5	0.2	0.0	87.9	0.5	0.3	0.0
Juice-based driftks, fortilied,	22	208.0	0.0	0.0	52.6	0.6	0.0	546	0.4	0.2	0.0
Milk flavored skim or 1% fat	23	200.0	0.0	0.0	52.0	0.0	0.0	54.0	0.4	0.2	0.0
fortified	2	313.9	6.2	3.8	54.7	16.6	21.2	271.2	2.7	16.9	10.3
Milk, flavored, whole or 2% fat,											
fortified	9	407.3	14.3	8.9	56.0	15.0	57.4	258.8	1.6	31.3	19.5
Milk, whole	1	299.8	16.3	10.2	22.7	16.1	66.4	239.1	0.0	48.9	30.5
Sports drinks	1	150.0	0.0	0.0	37.8	0.0	0.0	240.0	0.0	0.0	0.0
Spring water	5	0.0	0.0	0.0	0.0	0.0	0.0	15.2	0.0	0.0	0.0
Yogurt drinks	2	240.0	0.0	0.0	59.5	4.9	16.0	77.3	2.8	0.0	0.0
Bread or grain products											
All items	21	173.4	6.0	2.1	26.7	3.8	0.9	295.8	0.8	29.0	10.1
Cereal bars/cereal mixes	6	157.1	4.2	0.9	27.7	3.4	0.2	209.1	0.6	23.6	4.8
Crackers/Cracker Sandwiches	7	210.2	9.7	2.7	26.3	4.4	2.6	415.9	0.6	40.8	11.2
Granola bar	5	160.8	5.8	3.8	24.6	3.6	0.0	102.5	1.1	31.5	20.1
Pretzels	2	162.0	1.5	0.3	33.7	3.9	0.0	729.3	1.4	8.2	1.8
Rice or corn cakes	1	99.1	0.8	0.1	20.7	2.5	0.0	74.3	0.8	7.3	1.1
Candy/gum											
All items	58	142 1	3.0	10	30.7	07	07	28.0	02	14.9	54
Candy with chocolate	9	245.2	12.3	5.0	30.7	3.5	3.7	82.7	1.3	44.7	18.5
Candy with chocolate, single											
piece or pop	3	63.2	2.5	1.5	10.3	0.5	1.7	9.0	0.1	35.8	21.6
Candy without chocolate	14	211.1	2.1	0.5	49.5	0.2	0.0	34.1	0.0	8.1	1.8
Candy without chocolate, single											
piece or pop	19	86.2	1.2	0.3	19.8	0.1	0.0	18.2	0.0	11.9	2.7
Cnewing gum	13	96.6	0.1	0.0	30.8	0.0	0.0	2.1	0.0	1.1	0.2
Frozen desserts											
All items	33	188.3	9.5	5.5	24.6	2.9	21.7	57.5	0.5	41.5	23.6
Ice cream	28	204.3	11.2	6.4	24.6	3.2	25.6	65.8	0.6	48.8	27.8
Non-dairy frozen desserts	5	99.2	0.0	0.0	25.1	0.7	0.0	11.3	0.0	0.4	0.0
Salads with dressing	0	150.1	10.0	0.0	6 F	0.5	01.0	064.4	1.0	70.0	10.0
All liellis Side salads	2	150.1	12.0	2.2	0.0	3.5	21.0 21.0	364.4	1.0	76.0 76.9	12.9
Give salaus	2	150.1	12.0	2.2	0.0	5.5	21.0	004.4	1.0	10.0	12.3
Snacks											
All items	57	232.5	13.1	3.0	26.5	3.1	0.9	311.8	2.0	49.3	11.7

Food Available Inside and Outside the School Cafeteria

Table 4.7Food Energy and Macronutrient Composition of Competitive Foods Available During the School Day, AllSchools and Sources — Continued

	Number	Mean				Mean percent of calories from					
	of items	calories per item	Total fat (g)	Satura- ted fat (g)	Carbo- hydrate (g)	Protein (g)	Choles- terol (mg)	Sodium (mg)	Dietary fiber (g)	Total fat	Saturated fat
Snacks											
Chips	40	265.8	16.2	3.7	27.9	3.5	0.5	346.5	2.1	55.0	12.9
Chips - baked or reduced fat	5	182.7	8.1	0.9	25.1	2.2	0.0	254.6	2.0	38.9	4.4
Fruit roll-ups/fruit snacks	5	167.6	1.6	0.4	38.6	0.1	0.0	125.6	1.6	8.2	2.1
Meat snacks (jerky)	2	76.1	4.8	2.0	2.0	6.2	8.9	410.5	0.3	56.2	23.8
Popcorn	5	142.8	8.4	2.2	14.8	2.9	2.5	237.6	2.8	52.8	14.2

Food Available Inside and Outside the School Cafeteria

Table 4.7Food Energy and Macronutrient Composition of Competitive Foods Available During the School Day, AllSchools and Sources — Continued

	Number	Mean			Mean	amount p	er item			Mean p calori	ercent of es from
	of items	calories per item	Total fat (g)	Satura- ted fat (g)	Carbo- hydrate (g)	Protein (g)	Choles- terol (mg)	Sodium (mg)	Dietary fiber (g)	Total fat	Saturated fat
Baked goods/dessert											
All items	25	359.0	16.2	4.0	51.4	3.9	18.0	251.9	1.6	39.8	10.0
Cookies	10	306.4	15.0	4.6	42.0	3.5	6.0	235.0	1.7	42.7	13.0
Doughnuts	2	380.6	22.9	5.0	41.6	4.2	40.4	414.6	1.5	53.9	11.6
Pastries (danish, toaster pastries,											
pies)	5	367.0	12.8	2.4	59.3	4.7	34.5	296.7	1.9	30.1	5.6
Snack cakes	8	414.3	18.2	4.0	60.7	3.9	17.0	204.2	1.2	38.6	8.6
Beverages											
All items	63	226.6	2.6	1.6	48.8	3.5	10.4	99.9	0.7	6.2	3.7
Fruit juice, 100% juice	7	179.4	0.4	0.0	43.3	1.6	0.0	11.9	0.6	2.3	0.3
Iced tea drinks, fortified	2	130.0	0.0	0.0	33.0	0.0	0.0	9.9	0.0	0.0	0.0
Juice-based drink, fortified, 16-20											
fl oz	2	237.2	0.0	0.0	58.2	0.0	0.0	345.0	0.0	0.0	0.0
Juice-based drinks, 11.5-12 fl oz	4	170.0	0.0	0.0	43.4	0.2	0.0	62.8	0.3	0.4	0.0
Juice-based drinks, 16-20 fl oz	6	265.8	0.1	0.0	68.2	0.2	0.0	71.8	0.5	0.4	0.0
Juice-based drinks, fortified,											
11.5-12 fl oz	23	208.0	0.0	0.0	52.6	0.6	0.0	54.6	0.4	0.2	0.0
Milk, flavored, skim or 1% fat,											
fortified	2	313.9	6.2	3.8	54.7	16.6	21.2	271.2	2.7	16.9	10.3
Milk, flavored, whole or 2% fat,											
fortified	9	407.3	14.3	8.9	56.0	15.0	57.4	258.8	1.6	31.3	19.5
Milk, whole	1	299.8	16.3	10.2	22.7	16.1	66.4	239.1	0.0	48.9	30.5
Sports drinks	1	150.0	0.0	0.0	37.8	0.0	0.0	240.0	0.0	0.0	0.0
Spring water	4	0.0	0.0	0.0	0.0	0.0	0.0	14.6	0.0	0.0	0.0
Yogurt drinks	2	240.0	0.0	0.0	59.5	4.9	16.0	77.3	2.8	0.0	0.0
Bread or grain products											
All items	16	174.2	6.2	2.4	26.7	3.5	1.1	307.6	0.8	29.4	11.5
Cereal bars/cereal mixes	2	141.0	3.7	0.7	26.9	1.1	0.0	161.5	0.1	23.8	4.7
Crackers/Cracker Sandwiches	6	213.0	9.7	2.8	27.0	4.3	2.9	425.6	0.7	40.2	11.4
Granola bar	5	160.8	5.8	3.8	24.6	3.6	0.0	102.5	1.1	31.5	20.1
Pretzels	2	162.0	1.5	0.3	33.7	3.9	0.0	729.3	1.4	8.2	1.8
Rice or corn cakes	1	99.1	0.8	0.1	20.7	2.5	0.0	74.3	0.8	7.3	1.1
Frozen desserts											
All items	33	188.3	9.5	5.5	24.6	2.9	21.7	57.5	0.5	41.5	23.6
Ice cream	28	204.3	11.2	6.4	24.6	3.2	25.6	65.8	0.6	48.8	27.8
Non-dairy frozen desserts	5	99.2	0.0	0.0	25.1	0.7	0.0	11.3	0.0	0.4	0.0
Salads with dressing											
All items	2	150.1	12.8	2.2	6.5	3.5	21.0	364.4	1.6	76.8	12.9
Side salads	2	150.1	12.8	2.2	6.5	3.5	21.0	364.4	1.6	76.8	12.9
Snacks											
All items	45	245.2	13.5	3.0	29.0	3.1	0.7	321.2	2.2	47.3	10.8
Chips	30	285.7	17.2	3.9	30.4	3.8	0.6	378.8	2.2	54.3	12.8
Chips - baked or reduced fat	5	182.7	8.1	0.9	25.1	2.2	0.0	254.6	2.0	38.9	4.4
Fruit roll-ups/fruit snacks	5	167.6	1.6	0.4	38.6	0.1	0.0	125.6	1.6	8.2	2.1
Popcorn	5	142.8	8.4	2.2	14.8	2.9	2.5	237.6	2.8	52.8	14.2

Food Available In the School Cafeteria

Table 4.7Food Energy and Macronutrient Composition of Competitive Foods Available During the School Day, AllSchools and Sources — Continued

		Mean			Mean	amount p	er item			Mean p calorie	ercent of es from
	of items	calories per item	Total fat (g)	Satura- ted fat (g)	Carbo- hydrate (g)	Protein (g)	Choles- terol (mg)	Sodium (mg)	Dietary fiber (g)	Total fat	Saturated fat
Baked goods/dessert											
All items	2	408.9	10.6	16	74 0	49	0.0	436.0	22	23.4	35
Pastries (danish toaster pastries	-	100.0	10.0	1.0	7 1.0	1.0	0.0	100.0		20.1	0.0
pies)	2	408.9	10.6	1.6	74.0	4.9	0.0	436.0	2.2	23.4	3.5
Beverages											
All items	3	193.6	0.0	0.0	49.0	0.0	0.0	96.6	0.4	0.0	0.0
Juice-based drinks, 16-20 fl oz	2	290.5	0.0	0.0	73.5	0.0	0.0	136.0	0.6	0.0	0.0
Spring water	1	0.0	0.0	0.0	0.0	0.0	0.0	17.8	0.0	0.0	0.0
Bread or grain products											
All items	5	170.9	5.4	1.2	26.9	4.7	0.4	257.8	0.7	27.8	5.9
Cereal bars/cereal mixes	4	165.2	4.4	0.9	28.1	4.6	0.3	232.9	0.8	23.6	4.9
Crackers/Cracker Sandwiches	1	193.7	9.6	2.1	22.0	5.0	0.8	357.6	0.4	44.7	10.0
Candy/gum											
All items	58	142.1	3.0	1.0	30.7	0.7	0.7	28.0	0.2	14.9	5.4
Candy with chocolate Candy with chocolate, single	9	245.2	12.3	5.0	30.7	3.5	3.7	82.7	1.3	44.7	18.5
piece or pop	3	63.2	2.5	1.5	10.3	0.5	1.7	9.0	0.1	35.8	21.6
Candy without chocolate Candy without chocolate, single	14	211.1	2.1	0.5	49.5	0.2	0.0	34.1	0.0	8.1	1.8
piece or pop	19	86.2	1.2	0.3	19.8	0.1	0.0	18.2	0.0	11.9	2.7
Chewing gum	13	96.6	0.1	0.0	30.8	0.0	0.0	2.1	0.0	1.1	0.2
Snacks											
All items	12	184.6	11.9	3.0	17.4	3.1	1.6	276.4	1.4	56.8	15.1
Chips	10	206.3	13.4	3.2	20.4	2.4	0.1	249.6	1.6	57.0	13.4
Meat snacks (jerky)	2	76.1	4.8	2.0	2.0	6.2	8.9	410.5	0.3	56.2	23.8

Food Available Outside the School Cafeteria

Number and Percent of Competitive Food and Beverage Products with no more than 30% of Calories from Fat

		School #1			School #2		School #3			
	T	Total fa	t ≤ 30%	T	Total fa	t ≤ 30%		Total fa	t ≤ 30%	
	l otal items	Number	Percent	l otal items	Number	Percent	I otal items	Number	Percent	
In the cafeteria										
Baked goods/dessert	5	0	0.0	13	4	30.8	7	0	0.0	
Beverages	11	11	100.0	39	31	79.5	13	13	100.0	
Bread/grain products	1	1	100.0	14	5	35.7	1	1	100.0	
Frozen desserts	-	-	-	22	6	27.3	11	1	9.1	
Salads with dressing	-	-	-	2	0	0.0	-	-	-	
Snacks	6	0	0.0	28	5	17.9	11	0	0.0	
Outside the cafeteria										
Baked goods/dessert	-	-	-	-	-	-	2	2	100.0	
Beverages	-	-	-	-	-	-	3	3	100.0	
Bread/grain products	-	-	-	-	-	-	5	4	80.0	
Candy	-	-	-	36	28	77.8	22	19	86.4	
Snacks	-	-	-	5	0	0.0	7	0	0.0	

- No items in category.

Source: Cafeteria Survey and and Inventories of Competitive Foods.

Table 4.9 Number and Percent of Competitive Food and Beverage Products with less than 10% of Calories from Saturated Fat

		School #1			School #2		School #3			
	T	Saturated	fat < 10%	T	Saturated	fat < 10%	T	Saturated	fat < 10%	
	l otal items	Number	Percent	- I otal items	Number	Percent	I otal items	Number	Percent	
In the cafeteria										
Baked goods/dessert	5	3	60.0	13	8	61.5	7	6	85.7	
Beverages	11	11	100.0	39	28	71.8	13	13	100.0	
Bread/grain products	1	1	100.0	14	6	42.9	1	1	100.0	
Frozen desserts	-	-	-	22	4	18.2	11	1	9.1	
Salads with dressing	-	-	-	2	0	0.0	-	-	-	
Snacks	6	3	50.0	28	16	57.1	11	2	18.2	
Outside the cafeteria										
Baked goods/dessert	_	_	_	_	_	_	2	2	100.0	
Beverages	_	_	_	_	_	_	3	3	100.0	
Bread/grain products	-	-	-	-	-	-	5	5	100.0	
Candy	-	-	-	36	28	77.8	22	19	86.4	
Snacks	_	-	_	5	0	0.0	7	3	42.9	

- No items in category.

Micronutrient Composition of Competitive Foods Available During the School Day, All Schools and Sources

	Mean amount per item								
	Number of items	Vitamin A (RE)	Vitamin C (mg)	Thiamin (mg)	Riboflavin (mg)	Niacin (mg)	Vitamin B ₆ (mg)	Calcium (mg)	lron (mg)
Baked goods/dessert									
All items	27	65.4	0.5	0.2	0.2	24	0.2	63.4	21
Cookies	10	18.6	0.0	0.1	0.2	1.5	0.0	19.9	1.8
Doughnuts	2	20.4	0.1	0.1	0.2	1.0	0.0	33.6	1.0
Pastries (danish toaster pastries	2	20.4	0.1	0.2	0.2	1.4	0.0	55.0	1.5
nice)	7	200 6	0.0	0.2	0.4	5.2	0.5	126.6	2.5
Pies)	/	209.0	0.9	0.3	0.4	5.Z	0.5	61.2	2.5
Shack cakes	0	9.0	0.2	0.2	0.2	1.5	0.0	01.5	2.5
Beverages									
All items	66	96.9	57.6	0.1	0.2	1.0	0.2	197.6	0.7
Fruit juice, 100% juice	7	33.7	106.5	0.2	0.0	0.8	0.2	184.7	1.3
Iced tea drinks, fortified	2	0.0	60.0	0.0	0.0	4.0	0.4	100.0	0.0
Juice-based drink, fortified, 16-20									
fl oz	2	215.6	70.1	0.6	0.0	0.2	0.0	21.6	0.2
Juice-based drinks, 11.5-12 fl oz	4	3.3	40.1	0.0	0.0	0.1	0.0	26.4	0.7
Juice-based drinks, 16-20 fl oz	8	19.2	67.8	0.0	0.1	0.2	0.0	36.3	1.1
Juice-based drinks, fortified,									
11.5-12 fl oz	23	146.2	82.3	0.1	0.1	1.7	0.2	54.8	0.7
Milk, flavored, skim or 1% fat,									
fortified	2	284.4	17.3	0.2	0.8	0.6	0.2	684.0	1.3
Milk. flavored, whole or 2% fat.									
fortified	9	153.0	4.4	0.2	0.8	0.5	0.2	816.7	0.7
Milk, whole	1	151.3	4.6	0.2	0.8	0.4	0.2	582.7	0.2
Sports drinks	1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.3
Spring water	5	0.0	0.0	0.0	0.0	0.0	0.0	10.2	0.0
Yogurt drinks	2	51.1	60.0	0.1	0.3	0.7	0.4	250.0	0.7
Bread or grain products									
All items	21	20.2	2.0	0.2	0.1	1.8	0.1	58.0	2.0
Corool bara/aarool miyoo	21	29.2	2.0	0.2	0.1	1.0	0.1	00.0	2.0
Creakero/Creaker Sandwichoo	0	92.7	0.4	0.2	0.2	2.5	0.2	09.0	3.4
Cranala har	7	7.7	0.3	0.2	0.2	2.0	0.1	70.0	1.5
Brattala	5	0.4	0.4	0.1	0.0	0.6	0.1	22.1	1.2
Pielzeis Pieco or corp cakes	2	1.0	0.0	0.2	0.3	2.2	0.0	15.3	1.8
NICE OF COTTI CARES	I	1.0	0.0	0.0	0.0	1.0	0.0	2.3	0.5
Candy/gum									
All items	58	2.1	5.3	0.0	0.0	0.2	0.0	10.6	0.1
Candy with chocolate	9	12.9	0.2	0.0	0.0	1.1	0.0	45.1	0.5
Candy with chocolate, single	-			a -	a -				a -
piece or pop	3	2.2	0.0	0.0	0.0	0.0	0.0	15.8	0.0
Candy without chocolate	14	0.0	14.2	0.0	0.0	0.0	0.0	4.3	0.1
Candy without chocolate, single									
piece or pop	19	0.0	5.6	0.0	0.0	0.0	0.0	2.7	0.0
Chewing gum	13	0.0	0.0	0.0	0.0	0.0	0.0	3.7	0.0
Frozen desserts									
All items	33	50.8	1.8	0.0	0.1	0.3	0.0	71.6	0.5
Ice cream	28	58.8	0.6	0.0	0.2	0.4	0.0	83.7	0.5
Non-dairy frozen desserts	5	6.1	8.8	0.0	0.0	0.0	0.0	4.1	0.2
Salads with dressing									
All items	2	250.4	15.7	0.0	0.1	10	0.0	64.3	10
Side salads	2	250.4	15.7	0.0	0 1	1.0	0.0	64.3	1.0
	-			0.0			0.0	00	

Food Available Inside and Outside the School Cafeteria

Micronutrient Composition of Competitive Foods Available During the School Day, All Schools and Sources — Continued

		Mean amount per item										
	Number of items	Vitamin A (RE)	Vitamin C (mg)	Thiamin (mg)	Riboflavin (mg)	Niacin (mg)	Vitamin B ₆ (mg)	Calcium (mg)	Iron (mg)			
Snacks												
All items	57	6.8	7.4	0.0	0.0	1.0	0.1	32.9	0.7			
Chips	40	7.2	4.7	0.0	0.1	1.3	0.2	39.4	0.8			
Chips - baked or reduced fat	5	4.5	0.0	0.0	0.0	1.0	0.1	25.9	0.3			
Fruit roll-ups/fruit snacks	5	5.4	46.4	0.0	0.0	0.0	0.1	9.2	0.4			
Meat snacks (jerky)	2	0.0	0.0	0.0	0.0	0.3	0.0	3.7	1.0			
Popcorn	5	10.5	0.0	0.0	0.0	0.4	0.0	23.5	0.5			

Food Available Inside and Outside the School Cafeteria

Micronutrient Composition of Competitive Foods Available During the School Day, All Schools and Sources — Continued

					Mean amou	nt per item			
	Number of items	Vitamin A (RE)	Vitamin C (mg)	Thiamin (mg)	Riboflavin (mg)	Niacin (mg)	Vitamin B ₆ (mg)	Calcium (mg)	lron (mg)
Baked goods/dessert									
All items	25	46 7	0.5	0.2	0.2	23	0.1	66.4	20
Cookies	10	18.6	0.0	0.2	0.2	1.5	0.1	19.9	1.8
Doughnuts	2	20.4	0.1	0.1	0.2	1.0	0.0	33.6	1.9
Pastries (danish toaster pastries	-	20.1	0.1	0.2	0.2		0.0	00.0	1.0
nies)	5	174.0	11	03	03	5.6	0.5	180 5	21
Snack cakes	8	9.0	0.2	0.2	0.2	1.5	0.0	61.3	2.3
Beverages									
All items	63	101 4	54 5	01	02	10	02	205.3	07
Fruit juice 100% juice	7	33.7	106.5	0.2	0.0	0.8	0.2	184 7	1.3
Iced tea drinks fortified	2	0.0	60.0	0.0	0.0	4.0	0.4	100.0	0.0
Juice-based drink fortified 16-20	-	0.0	00.0	0.0	0.0		0		0.0
fl oz	2	215.6	70 1	0.6	0.0	02	0.0	21.6	02
Juice-based drinks 11 5-12 fl oz	4	3.3	40.1	0.0	0.0	0.1	0.0	26.4	0.7
Juice-based drinks, 16-20 fl oz	6	23.6	29.5	0.0	0.1	0.2	0.0	32.0	1.0
Juice-based drinks, fortified,	Ũ	20.0	20.0	0.0	0.1	0.2	0.0	02.0	1.0
11.5-12 fl oz	23	146.2	82.3	0.1	0.1	1.7	0.2	54.8	0.7
Milk, flavored, skim or 1% fat, fortified	2	284.4	173	0.2	0.8	0.6	0.2	684.0	13
Milk, flavored, whole or 2% fat.	2	204.4	17.0	0.2	0.0	0.0	0.2	004.0	1.0
fortified	9	153.0	4.4	0.2	0.8	0.5	0.2	816.7	0.7
Milk. whole	1	151.3	4.6	0.2	0.8	0.4	0.2	582.7	0.2
Sports drinks	1	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.3
Spring water	4	0.0	0.0	0.0	0.0	0.0	0.0	97	0.0
Yogurt drinks	2	51.1	60.0	0.1	0.3	0.7	0.4	250.0	0.7
Bread or grain products									
All items	16	17.6	0.9	0.2	0.1	1.5	0.1	40.4	1.3
Cereal bars/cereal mixes	2	111.9	5.1	0.1	0.1	1.7	0.2	3.0	0.7
Crackers/Cracker Sandwiches	6	9.0	0.3	0.2	0.2	2.0	0.1	82.9	1.5
Granola bar	5	0.4	0.4	0.1	0.0	0.6	0.1	22.1	1.2
Pretzels	2	0.0	0.0	0.2	0.3	2.2	0.0	15.3	1.8
Rice or corn cakes	1	1.8	0.0	0.0	0.0	1.6	0.0	2.3	0.5
Frozen desserts									
All items	33	50.8	1.8	0.0	0.1	0.3	0.0	71.6	0.5
Ice cream	28	58.8	0.6	0.0	0.2	0.4	0.0	83.7	0.5
Non-dairy frozen desserts	5	6.1	8.8	0.0	0.0	0.0	0.0	4.1	0.2
Salads with dressing									
All items	2	250.4	15.7	0.0	0.1	1.0	0.0	64.3	1.0
Side salads	2	250.4	15.7	0.0	0.1	1.0	0.0	64.3	1.0
Snacks									
All items	45	8.1	8.6	0.0	0.0	1.1	0.2	36.8	0.8
Chips	30	8.7	5.1	0.0	0.1	1.4	0.2	45.4	0.9
Chips - baked or reduced fat	5	4.5	0.0	0.0	0.0	1.0	0.1	25.9	0.3
Fruit roll-ups/fruit snacks	5	5.4	46.4	0.0	0.0	0.0	0.1	9.2	0.4
Popcorn	5	10.5	0.0	0.0	0.0	0.4	0.0	23.5	0.5

Food Available In the School Cafeteria

Micronutrient Composition of Competitive Foods Available During the School Day, All Schools and Sources — Continued

					Mean amou	int per item			
	Number of items	Vitamin A (RE)	Vitamin C (mg)	Thiamin (mg)	Riboflavin (mg)	Niacin (mg)	Vitamin B ₆ (mg)	Calcium (mg)	lron (mg)
Baked goods/dessert									
All items	2	298.6	03	03	04	41	0.4	27.0	3.6
Pastries (danish toaster nastries	2	200.0	0.0	0.0	0.4	4.1	0.4	27.0	0.0
pies)	2	298.6	0.3	0.3	0.4	4.1	0.4	27.0	3.6
Beverages									
All items	3	4.1	122.0	0.0	0.0	0.0	0.0	36.9	0.9
Juice-based drinks, 16-20 fl oz	2	6.2	182.9	0.1	0.1	0.1	0.0	49.4	1.3
Spring water	1	0.0	0.0	0.0	0.0	0.0	0.0	11.8	0.0
Bread or grain products									
All items	5	66.4	5.7	0.2	0.2	2.7	0.2	114.3	4.1
Cereal bars/cereal mixes	4	83.1	7.0	0.2	0.2	2.8	0.3	133.2	4.8
Crackers/Cracker Sandwiches	1	0.0	0.2	0.2	0.1	2.4	0.0	38.7	1.1
Candy/gum									
All items	58	2.1	5.3	0.0	0.0	0.2	0.0	10.6	0.1
Candy with chocolate	9	12.9	0.2	0.0	0.0	1.1	0.0	45.1	0.5
piece or pop	3	2.2	0.0	0.0	0.0	0.0	0.0	15.8	0.0
Candy without chocolate	14	0.0	14.2	0.0	0.0	0.0	0.0	4.3	0.1
Candy without chocolate, single									
piece or pop	19	0.0	5.6	0.0	0.0	0.0	0.0	2.7	0.0
Chewing gum	13	0.0	0.0	0.0	0.0	0.0	0.0	3.7	0.0
Snacks									
All items	12	2.1	2.8	0.0	0.0	0.9	0.0	18.3	0.7
Chips	10	2.5	3.4	0.0	0.0	1.1	0.0	21.2	0.6
Meat snacks (jerky)	2	0.0	0.0	0.0	0.0	0.3	0.0	3.7	1.0

Food Available Outside the School Cafeteria

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, All Schools and Sources

			Mean number o	f servings pe	r pyramid grou	ip	Mean amo	unt per item
	Number of items	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)
Baked goods/dessert								
All items	27	1.7	0.0	0.0	0.0	0.0	13.3	7.2
Cookies	10	1.2	0.0	0.0	0.0	0.0	13.8	6.1
Doughnuts	2	1.7	0.0	0.0	0.0	0.0	21.2	4.8
Pastries (danish, toaster pastries,								
pies)	7	2.4	0.0	0.1	0.0	0.0	9.8	7.7
Snack cakes	8	1.8	0.0	0.0	0.0	0.0	13.8	8.9
Beverages								
All items	66	0.0	0.0	0.5	0.4	0.0	2.4	7.1
Fruit juice, 100% juice	7	0.0	0.0	2.2	0.0	0.0	0.0	0.0
Iced tea drinks, fortified	2	0.0	0.0	0.0	0.0	0.0	0.0	4.3
Juice-based drink, fortified, 16-20								
fl oz	2	0.0	0.0	1.0	0.0	0.0	0.0	9.7
Juice-based drinks. 11.5-12 fl oz	4	0.0	0.0	0.4	0.0	0.0	0.0	8.7
Juice-based drinks, 16-20 fl oz	8	0.0	0.0	0.4	0.0	0.0	0.0	13.8
Juice-based drinks, fortified,	23	0.0	0.0	0.3	0.0	0.0	0.0	8.8
Milk flavored skim or 1% fat	20	0.0	0.0	0.0	0.0	0.0	0.0	0.0
fortified	2	0.0	0.0	0.0	20	0.0	49	6 1
Milk flavored whole or 2% fat	-	0.0	0.0	0.0	2.0	0.0	1.0	0.1
fortified	٩	0.6	0.0	0.0	19	0.0	14.0	65
Milk whole	1	0.0	0.0	0.0	2.0	0.0	15.4	0.0
Sporte drinke	1	0.0	0.0	0.0	2.0	0.0	0.0	0.0
Sports utilities	5	0.0	0.0	0.0	0.0	0.0	0.0	9.0
Yogurt drinks	2	0.0	0.0	1.4	0.0	0.0	0.0 4 1	0.0 7.4
	2	0.0	0.0	1.4	0.4	0.0	7.1	7.4
Bread or grain products								
All items	21	1.3	0.0	0.0	0.0	0.0	5.0	1.6
Cereal bars/cereal mixes	6	1.0	0.0	0.0	0.0	0.0	3.2	2.9
Crackers/Cracker Sandwiches	7	1.5	0.0	0.0	0.0	0.1	8.2	1.2
Granola bar	5	1.0	0.0	0.0	0.0	0.1	5.4	1.7
Pretzels	2	2.6	0.0	0.0	0.0	0.0	0.0	0.0
Rice or corn cakes	1	1.5	0.0	0.0	0.0	0.0	0.0	0.0
Candy/gum								
All items	58	0.0	0.0	0.0	0.0	0.0	2.9	5.8
Candy with chocolate	9	0.4	0.0	0.0	0.1	0.2	12.6	4.6
Candy with chocolate, single								
piece or pop	3	0.0	0.0	0.0	0.0	0.0	2.6	2.3
Candy without chocolate	14	0.0	0.0	0.0	0.0	0.0	2.0	11.8
Candy without chocolate, single								
piece or pop	19	0.0	0.0	0.0	0.0	0.0	1.1	4.8
Chewing gum	13	0.0	0.0	0.0	0.0	0.0	0.0	2.5
Frozen desserts								
All items	33	0.3	0.0	0.0	0.2	0.0	8.9	4.1
Ice cream	28	0.3	0.0	0.0	0.2	0.0	10.5	3.9
Non-dairy frozen desserts	5	0.0	0.0	0.2	0.0	0.0	0.0	5.3
Salads with dressing	_							<i>c</i> -
All items	2	0.2	1.5	0.0	0.0	0.2	12.0	0.3
Side salads	2	0.2	1.5	0.0	0.0	0.2	12.0	0.3

Food Available Inside and Outside the School Cafeteria

1 Also includes legumes, soybean products, nuts, and seeds.

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, All Schools and Sources — Continued

		I	Mean number o	of servings per	r pyramid grou	р	Mean amou	unt per item
	Number of items	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)
Snacks								
All items	57	1.1	0.5	0.0	0.0	0.0	11.6	0.5
Chips	40	1.3	0.7	0.0	0.0	0.0	14.5	0.0
Chips - baked or reduced fat	5	0.9	0.7	0.0	0.0	0.0	8.1	0.4
Fruit roll-ups/fruit snacks	5	0.0	0.0	0.8	0.0	0.0	0.0	4.5
Meat snacks (jerky)	2	0.0	0.0	0.0	0.0	0.5	3.3	0.0
Popcorn	5	1.2	0.0	0.0	0.0	0.0	7.2	0.0

Food Available Inside and Outside the School Cafeteria

1 Also includes legumes, soybean products, nuts, and seeds.

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, All Schools and Sources — Continued

			Mean number o	of servings per	r pyramid grou	р	Mean amo	unt per item
	Number of items	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)
Baked goods/dessert								
All items	25	1.6	0.0	0.0	0.0	0.0	13.8	7.2
Cookies	10	1.2	0.0	0.0	0.0	0.0	13.8	6.1
Doughnuts	2	17	0.0	0.0	0.0	0.0	21.2	4.8
Pastries (danish toaster pastries	2	1.7	0.0	0.0	0.0	0.0	21.2	4.0
	5	0.1	0.0	0.0	0.0	0.0	10.7	70
Pres/	5	2.1	0.0	0.0	0.0	0.0	10.7	7.0
Shack cakes	0	1.0	0.0	0.0	0.0	0.0	13.0	8.9
Beverages								
All items	63	0.0	0.0	0.5	0.4	0.0	2.5	7.0
Fruit juice, 100% juice	7	0.0	0.0	2.2	0.0	0.0	0.0	0.0
Iced tea drinks, fortified Juice-based drink, fortified, 16-20	2	0.0	0.0	0.0	0.0	0.0	0.0	4.3
fl oz	2	0.0	0.0	1.0	0.0	0.0	0.0	9.7
Juice-based drinks, 11.5-12 fl oz	4	0.0	0.0	0.4	0.0	0.0	0.0	8.7
Juice-based drinks, 16-20 fl oz	6	0.0	0.0	0.4	0.0	0.0	0.0	13.1
Juice-based drinks, fortified,	23	0.0	0.0	0.3	0.0	0.0	0.0	8.8
Milk, flavored, skim or 1% fat,	20	0.0	0.0	0.0	0.0	0.0	4.0	0.0
Milk, flavored, whole or 2% fat,	2	0.0	0.0	0.0	2.0	0.0	4.9	6.1
fortified	9	0.6	0.0	0.0	1.9	0.0	14.0	6.5
Milk, whole	1	0.0	0.0	0.0	2.0	0.0	15.4	0.0
Sports drinks	1	0.0	0.0	0.0	0.0	0.0	0.0	9.0
Spring water	4	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Yogurt drinks	2	0.0	0.0	1.4	0.4	0.0	4.1	7.4
Bread or grain products								
All items	16	1.4	0.0	0.0	0.0	0.0	5.1	1.3
Cereal bars/cereal mixes	2	0.8	0.0	0.0	0.0	0.0	2.9	3.4
Crackers/Cracker Sandwiches	6	1.6	0.0	0.0	0.0	0.0	8.1	1.0
Granola bar	5	1.0	0.0	0.0	0.0	0.1	5.4	1.7
Pretzels	2	2.6	0.0	0.0	0.0	0.0	0.0	0.0
Rice or corn cakes	1	1.5	0.0	0.0	0.0	0.0	0.0	0.0
Frozen desserts								
All items	33	0.3	0.0	0.0	02	0.0	89	4 1
lce cream	28	0.3	0.0	0.0	0.2	0.0	10.5	3.9
Non-dairy frozen desserts	5	0.0	0.0	0.2	0.0	0.0	0.0	5.3
Salads with dressing								
All items	2	02	15	0.0	0.0	02	12.0	03
Side salads	2	0.2	1.5	0.0	0.0	0.2	12.0	0.3
Snacks								
All items	45	1.3	0.5	0.0	0.0	0.0	11.7	0.5
Chips	30	1.6	0.6	0.0	0.0	0.0	15.0	0.0
Chips - baked or reduced fat	5	0.9	0.7	0.0	0.0	0.0	8 1	0.4
Fruit roll-ups/fruit snacks	5	0.0	0.0	0.8	0.0	0.0	0.0	4.5
Poncorn	5	12	0.0	0.0	0.0	0.0	7 2	0.0
· • • • • • • • • • • • • • • • • • • •	5	1.4	0.0	0.0	0.0	0.0	1.4	0.0

Food Available In the School Cafeteria

1 Also includes legumes, soybean products, nuts, and seeds.

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, All Schools and Sources — Continued

		1	Mean number o	of servings per	r pyramid grou	ıp	Mean amo	unt per item
	Number of items	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)
Bakad goods/dossart								
All items	2	32	0.0	0.1	0.0	0.0	74	77
Pastries (danish, toaster pastries,	-	0.2	0.0	011	010	010		
pies)	2	3.2	0.0	0.1	0.0	0.0	7.4	7.7
Beverages								
All items	3	0.0	0.0	0.2	0.0	0.0	0.0	10.7
Juice-based drinks, 16-20 fl oz	2	0.0	0.0	0.4	0.0	0.0	0.0	16.0
Spring water	1	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Bread or grain products								
All items	5	1.0	0.0	0.0	0.0	0.0	4.5	2.7
Cereal bars/cereal mixes	4	1.0	0.0	0.0	0.0	0.0	3.4	2.6
Crackers/Cracker Sandwiches	1	0.7	0.0	0.0	0.0	0.3	8.9	2.8
Candy/gum								
All items	58	0.0	0.0	0.0	0.0	0.0	2.9	5.8
Candy with chocolate	9	0.4	0.0	0.0	0.1	0.2	12.6	4.6
Candy with chocolate, single								
piece or pop	3	0.0	0.0	0.0	0.0	0.0	2.6	2.3
Candy without chocolate	14	0.0	0.0	0.0	0.0	0.0	2.0	11.8
Candy without chocolate, single								
piece or pop	19	0.0	0.0	0.0	0.0	0.0	1.1	4.8
Chewing gum	13	0.0	0.0	0.0	0.0	0.0	0.0	2.5
Snacks								
All items	12	0.4	0.8	0.0	0.0	0.0	11.2	0.1
Chips	10	0.5	0.9	0.0	0.0	0.0	12.7	0.2
Meat snacks (jerky)	2	0.0	0.0	0.0	0.0	0.5	3.3	0.0

Food Available Outside the School Cafeteria

1 Also includes legumes, soybean products, nuts, and seeds.

Competitive Foods Meeting FDA Regulations for Nutrient Content Claims: Foods Available During the School Day, All Schools and Sources

		Percent of items meeting FDA labeling criteria ¹									
	Number of items	Calorie free	Low calorie	Fat free (total fat)	Low fat (total fat)	Low saturated fat	Choleste- rol free	Low choleste- rol	Sodium free	Low Sodium	
Baked goods/dessert											
All items	27	0.0	0.0	0.0	0.0	185	25.9	29.6	0.0	22.2	
Cakes/cookies/doughnuts/pastries	27	0.0	0.0	0.0	0.0	18.5	25.9	29.6	0.0	22.2	
Beverages											
All items	66	7.6	0.0	80.3	3.0	83.3	78.8	4.5	1.5	75.8	
Fruit juice/juice-based drinks	47	0.0	0.0	97.9	2.1	100.0	100.0	0.0	2.1	91.5	
Milk and vogurt-based drinks	14	0.0	0.0	14.3	7.1	21.4	0.0	21.4	0.0	14.3	
Spring water	5	100.0	0.0	100.0	0.0	100.0	100.0	0.0	0.0	100.0	
Bread or grain products											
All items	21	0.0	0.0	0.0	23.8	52.4	61.9	28.6	0.0	23.8	
Cereal bars and cereal products	14	0.0	0.0	0.0	35.7	71.4	71.4	28.6	0.0	35.7	
Crackers/Cracker Sandwiches	7	0.0	0.0	0.0	0.0	14.3	42.9	28.6	0.0	0.0	
Candy/gum											
All items	58	0.0	0.0	48.3	3.4	81.0	81.0	19.0	22.4	72.4	
Candy with chocolate	12	0.0	0.0	8.3	0.0	8.3	8.3	91.7	0.0	75.0	
Candy without chocolate	33	0.0	0.0	42.4	6.1	100.0	100.0	0.0	0.0	100.0	
Chewing gum	13	0.0	0.0	100.0	0.0	100.0	100.0	0.0	100.0	0.0	
Frozen desserts											
All items	33	0.0	0.0	15.2	0.0	15.2	15.2	27.3	6.1	90.9	
Ice cream	28	0.0	0.0	0.0	0.0	0.0	0.0	32.1	0.0	96.4	
Non-dairy frozen desserts	5	0.0	0.0	100.0	0.0	100.0	100.0	0.0	40.0	60.0	
Salads with dressing											
All items	2	0.0	0.0	0.0	0.0	0.0	50.0	0.0	0.0	0.0	
Side salads	2	0.0	0.0	0.0	0.0	0.0	50.0	0.0	0.0	0.0	
Snacks											
All items	57	0.0	0.0	0.0	8.8	17.5	52.6	5.3	0.0	3.5	
Chips - baked or reduced fat	5	0.0	0.0	0.0	0.0	100.0	100.0	0.0	0.0	0.0	
Chips/Popcorn	45	0.0	0.0	0.0	0.0	0.0	44.4	6.7	0.0	0.0	
Fruit roll-ups/fruit snacks	5	0.0	0.0	0.0	100.0	100.0	100.0	0.0	0.0	40.0	
Meat snacks (jerky)	2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	

Food Available Inside and Outside the School Cafeteria

1 Items identified as *low-calorie* do not include items identified as *calorie-free*, and similarly for other categories.

Note: Some food categories have been combined due to small numbers of items.

Competitive Foods Meeting FDA Regulations for Nutrient Content Claims: Foods Available During the School Day, All Schools and Sources — Continued

				Perce	nt of items	meeting FD	A labeling	criteria ¹		
	Number of items	Calorie free	Low calorie	Fat free (total fat)	Low fat (total fat)	Low saturated fat	Choleste- rol free	Low choleste- rol	Sodium free	Low Sodium
Baked goods/dessert										
All items	25	0.0	0.0	0.0	0.0	12.0	20.0	32.0	0.0	24 0
Cakes/cookies/doughnuts/pastries	25	0.0	0.0	0.0	0.0	12.0	20.0	32.0	0.0	24.0
Beverages										
All items	63	6.3	0.0	79.4	3.2	82.5	77.8	4.8	1.6	74.6
Fruit juice/juice-based drinks	45	0.0	0.0	97.8	2.2	100.0	100.0	0.0	2.2	91.1
Milk and vogurt-based drinks	14	0.0	0.0	14.3	7.1	21.4	0.0	21.4	0.0	14.3
Spring water	4	100.0	0.0	100.0	0.0	100.0	100.0	0.0	0.0	100.0
Bread or grain products										
All items	16	0.0	0.0	0.0	25.0	43.8	50.0	37.5	0.0	25.0
Cereal bars and cereal products	10	0.0	0.0	0.0	40.0	60.0	60.0	40.0	0.0	40.0
Crackers/Cracker Sandwiches	6	0.0	0.0	0.0	0.0	16.7	33.3	33.3	0.0	0.0
Frozen desserts										
All items	33	0.0	0.0	15.2	0.0	15.2	15.2	27.3	6.1	90.9
Ice cream	28	0.0	0.0	0.0	0.0	0.0	0.0	32.1	0.0	96.4
Non-dairy frozen desserts	5	0.0	0.0	100.0	0.0	100.0	100.0	0.0	40.0	60.0
Salads with dressing										
All items	2	0.0	0.0	0.0	0.0	0.0	50.0	0.0	0.0	0.0
Side salads	2	0.0	0.0	0.0	0.0	0.0	50.0	0.0	0.0	0.0
Snacks										
All items	45	0.0	0.0	0.0	11.1	22.2	57.8	6.7	0.0	4.4
Chips - baked or reduced fat	5	0.0	0.0	0.0	0.0	100.0	100.0	0.0	0.0	0.0
Chips/Popcorn	35	0.0	0.0	0.0	0.0	0.0	45.7	8.6	0.0	0.0
Fruit roll-ups/fruit snacks	5	0.0	0.0	0.0	100.0	100.0	100.0	0.0	0.0	40.0

Food Available In the School Cafeteria

1 Items identified as *low-calorie* do not include items identified as *calorie-free*, and similarly for other categories.

Note: Some food categories have been combined due to small numbers of items.

Competitive Foods Meeting FDA Regulations for Nutrient Content Claims: Foods Available During the School Day, All Schools and Sources — Continued

				Perce	nt of items	meeting FD	A labeling	criteria ¹		
	Number of items	Calorie free	Low calorie	Fat free (total fat)	Low fat (total fat)	Low saturated fat	Choleste- rol free	Low choleste- rol	Sodium free	Low Sodium
Baked goods/dessert										
All items	2	0.0	0.0	0.0	0.0	100.0	100.0	0.0	0.0	0.0
Cakes/cookies/doughnuts/pastries	2	0.0	0.0	0.0	0.0	100.0	100.0	0.0	0.0	0.0
Beverages										
All items	3	33.3	0.0	100.0	0.0	100.0	100.0	0.0	0.0	100.0
Fruit juice/juice-based drinks	2	0.0	0.0	100.0	0.0	100.0	100.0	0.0	0.0	100.0
Spring water	1	100.0	0.0	100.0	0.0	100.0	100.0	0.0	0.0	100.0
Bread or grain products										
All items	5	0.0	0.0	0.0	20.0	80.0	100.0	0.0	0.0	20.0
Cereal bars and cereal products	4	0.0	0.0	0.0	25.0	100.0	100.0	0.0	0.0	25.0
Crackers/Cracker Sandwiches	1	0.0	0.0	0.0	0.0	0.0	100.0	0.0	0.0	0.0
Candy/gum										
All items	58	0.0	0.0	48.3	3.4	81.0	81.0	19.0	22.4	72.4
Candy with chocolate	12	0.0	0.0	8.3	0.0	8.3	8.3	91.7	0.0	75.0
Candy without chocolate	33	0.0	0.0	42.4	6.1	100.0	100.0	0.0	0.0	100.0
Chewing gum	13	0.0	0.0	100.0	0.0	100.0	100.0	0.0	100.0	0.0
Snacks										
All items	12	0.0	0.0	0.0	0.0	0.0	33.3	0.0	0.0	0.0
Chips/Popcorn	10	0.0	0.0	0.0	0.0	0.0	40.0	0.0	0.0	0.0
Meat snacks (jerky)	2	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0

Food Available Outside the School Cafeteria

1 Items identified as *low-calorie* do not include items identified as *calorie-free*, and similarly for other categories.

Note: Some food categories have been combined due to small numbers of items.

Weighted Tables

Food Energy and Macronutrient Composition of Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources

						Mean	amount p	er item			Mean p calorie	ercent of es from
	Number of items	Number sold	calories per item	Total fat (g)	Satura- ted fat (g)	Carbo- hydrate (g)	Protein (g)	Choles- terol (mg)	Sodium (mg)	Dietary fiber (g)	Total fat	Satura- ted fat
Baked goods/dessert												
All itoms	26	2580	3/2 0	17.0	4.6	46.0	3.8	16.3	268.0	17	11 1	11.8
Cookies	10	2056	33/ 0	17.0	4.0	40.0	3.0	80	200.0	1.7	44.1	13.4
Doughputo	0	2000	200 6	22.0	5.1	44.4	10	40.4	247.7 414.6	1.0	40.0 52.0	11.4
Dougilluis	2	12	360.0	22.9	5.0	41.5	4.5	40.4	414.0	1.5	55.9	11.0
rastiles (ualish, loaster pastiles,	0	000	225.0	11.0	0.1	EE 4	4.0	00.0	000 0	1 0	00.0	E 0
pies)	0	209	335.0	174	2.1	55.4	4.3	20.2	200.9	1.0	20.3	5.3
Shack cakes	0	243	406.3	17.4	3.8	60.8	3.8	14.7	248.2	1.2	37.9	8.3
Beverages												
All items	51	4220	183.3	0.3	0.2	45.6	0.6	1.1	69.7	0.4	0.8	0.5
Fruit juice, 100% juice	5	207	184.1	0.5	0.0	44.1	1.9	0.0	13.4	0.7	2.4	0.3
Iced tea drinks, fortified Juice-based drink, fortified, 16-20	2	137	130.0	0.0	0.0	33.0	0.0	0.0	9.9	0.0	0.0	0.0
fl oz	1	15	254.4	0.0	0.0	62.4	0.0	0.0	370.0	0.0	0.0	0.0
Juice-based drinks, 11.5-12 fl oz	1	9	166.9	0.0	0.0	42.3	0.0	0.0	78.1	0.4	0.0	0.0
Juice-based drinks, 16-20 fl oz	7	774	275.4	0.0	0.0	70.3	0.2	0.0	95.3	0.6	0.3	0.0
Juice-based drinks, fortified,												
11.5-12 fl oz	20	1671	186.7	0.0	0.0	47.2	0.5	0.0	60.1	0.4	0.1	0.0
Milk, flavored, skim or 1% fat,	•	05	011.0			54.0	10 7	00.4			10.1	
TORTIFIED	2	35	311.6	5.9	3.6	54.6	16.7	20.1	268.6	2.7	16.1	9.8
fortified	7	100	414.0	155	0.6	E4 0	15 /	E0 0	070.0	0.4	20 E	20.0
Milk whole	1	123	200.9	10.0	9.0	04.9	10.4	00.Z	273.2	2.4	33.5 40.0	20.0
WIIK, WIOIE	1	9	299.0	10.5	10.2	22.1	10.1	00.4	239.1	0.0	40.9	30.5
Sports uninks	0	1000							147			
Spring water	3	1233	0.0	0.0	0.0	0.0	0.0	10.0	77.0	0.0	0.0	0.0
rogurt drinks	2	/	240.0	0.0	0.0	59.5	4.9	16.0	11.3	2.9	0.0	0.0
Bread or grain products												
All items	18	1209	188.5	7.4	2.2	26.9	4.3	2.1	379.7	1.2	32.9	9.9
Cereal bars/cereal mixes	5	148	163.8	4.7	1.0	28.8	3.0	0.5	265.3	1.1	25.2	5.3
Crackers/Cracker Sandwiches	5	843	206.9	10.2	2.9	24.1	4.7	2.7	402.5	0.6	44.4	12.4
Granola bar	5	110	168.8	6.4	4.5	25.1	3.7	0.0	104.8	1.2	33.7	23.3
Pretzels	2	33	162.0	1.5	0.3	33.7	3.9	0.0	729.3	1.4	8.2	1.8
Rice or corn cakes	1	75	99.1	0.8	0.1	20.7	2.5	0.0	74.3	0.8	7.3	1.1
Candy/gum												
All items	58	5332	1117	23	07	24 1	05	04	24.2	01	12.0	37
Candy with chocolate	q	295	253.9	12.0	5.0	31.5	35	35	93.5	12	45.3	17.8
Candy with chocolate single	3	235	200.0	12.3	5.0	51.5	0.0	0.0	30.5	1.2	40.0	17.0
niece or non	3	100	63.3	10	11	11 0	0.4	13	8.4	0.1	26.8	16.2
Candy without chocolate	14	946	212.6	2.5	0.6	10.4	0.4	0.0	29.4	0.1	20.0	2 1
Candy without chocolate single	14	040	210.0	2.0	0.0	73.4	0.5	0.0	00.4	0.0	3.0	<u> </u> .1
piece or pop	10	2020	65.7	0.0	0.2	15.0	0.1	0.0	12.5	0.0	10.9	25
Chewing gum	13	162	68.3	0.0	0.2	22.0	0.0	0.0	1.5	0.0	1.1	0.2
Francis de conte												
Frozen desserts	07	057	100.0	10.0	F 4	05.0	0.4	04.0	FF0	0.7	45.0	04.0
All items	27	857	196.6	10.0	5.4	25.2	3.4	24.3	55.3	0.7	45.0	24.3
Ice cream	23	819	200.1	10.4	5.6	25.1	3.5	25.3	57.0	0.7	46.7	25.1
ivon-dairy trozen desserts	4	38	114.1	0.0	0.0	29.3	0.4	0.0	17.0	0.0	0.2	0.0

Food Available Inside and Outside the School Cafeteria

- Data not available.

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Food Energy and Macronutrient Composition of Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

	Number Nur of items s		Mean			Mean	amount p	er item			Mean p calorie	ercent of es from
	Number of items	nber Number ems sold	calories per item	Total fat (g)	Satura- ted fat (g)	Carbo- hydrate (g)	Protein (g)	Choles- terol (mg)	Sodium (mg)	Dietary fiber (g)	Total fat	Satura- ted fat
												*
Salads with dressing												
All items	2	35	137.6	11.9	1.9	6.5	2.4	11.1	405.6	1.7	77.8	12.3
Side salads	2	35	137.6	11.9	1.9	6.5	2.4	11.1	405.6	1.7	77.8	12.3
Snacks												
All items	54	3358	197.0	11.1	2.6	22.3	3.0	1.1	279.2	1.9	50.9	12.9
Chips	40	2149	227.5	13.3	3.0	25.1	3.1	0.4	291.6	2.0	53.1	12.5
Chips - baked or reduced fat	5	152	168.3	7.0	0.9	24.1	1.9	0.0	274.7	1.8	36.8	4.7
Fruit roll-ups/fruit snacks	2	264	180.3	1.7	0.5	41.5	0.1	0.0	135.5	1.7	8.2	2.1
Meat snacks (jerky)	2	610	76.1	4.8	2.0	2.0	6.2	8.9	410.5	0.3	56.2	23.8
Popcorn	5	183	142.8	8.4	2.3	14.8	2.9	2.5	237.6	2.8	52.8	14.2

Food Available Inside and Outside the School Cafeteria

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Food Energy and Macronutrient Composition of Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

	Numbor	Numbor	Mean			Mean	amount p	er item			Mean p calorie	ercent of es from
	of items	sold	calories per item	Total fat (g)	Satura- ted fat (g)	Carbo- hydrate (g)	Protein (g)	Choles- terol (mg)	Sodium (mg)	Dietary fiber (g)	Total fat	Satura- ted fat
Baked goods/dessert												
All items	24	2523	341.5	17.1	4.6	45.7	3.8	16.5	266.0	1.7	44.4	12.0
Cookies	10	2056	334.9	17.1	5.1	44.4	3.7	8.9	247.7	1.8	45.3	13.4
Doughnuts	2	72	380.6	22.9	5.0	41.5	4.3	40.4	414.6	1.5	53.9	11.6
Pastries (danish, toaster pastries,												
pies)	4	152	338.3	12.2	2.3	53.9	4.4	37.5	281.3	1.8	30.3	5.8
Snack cakes	8	243	406.3	17.4	3.8	60.8	3.8	14.7	248.2	1.2	37.9	8.3
Beverages												
All itomo	10	1000	175 0	0.2	0.0	407	0.6		66.0	0.4	0.0	0.5
Fruit iuioa 100% iuioa	40	4020	10/1	0.5	0.2	43.7	1.0	1.1	10.0	0.4	0.0	0.5
Fruit juice, 100% juice	5	207	104.1	0.5	0.0	44.1	1.9	0.0	13.4	0.7	2.4	0.3
luise based drinks, fortified	2	137	130.0	0.0	0.0	33.0	0.0	0.0	9.9	0.0	0.0	0.0
Juice-based drink, fortilied, 16-20	-	15	054.4	0.0	0.0	60.4	0.0	0.0	070.0	0.0	0.0	0.0
		15	204.4	0.0	0.0	02.4	0.0	0.0	370.0	0.0	0.0	0.0
Juice-based drinks, 11.5-12 fl oz	Ļ	9	166.9	0.0	0.0	42.3	0.0	0.0	78.1	0.4	0.0	0.0
Juice-based drinks, 16-20 fl oz	5	631	267.8	0.1	0.0	68.7	0.2	0.0	75.0	0.5	0.4	0.0
Juice-based drinks, fortified,		1071	400 7		~ ~	47.0			00.4			
11.5-12 fl OZ	20	1671	186.7	0.0	0.0	47.2	0.5	0.0	60.1	0.4	0.1	0.0
Milk, flavored, skim or 1% fat,												
fortified	2	35	311.6	5.9	3.6	54.6	16.7	20.1	268.6	2.7	16.1	9.8
Milk, flavored, whole or 2% fat,	_											
fortified	7	123	414.9	15.5	9.6	54.9	15.4	58.2	273.2	2.4	33.5	20.8
Milk, whole	1	9	299.8	16.3	10.2	22.7	16.1	66.4	239.1	0.0	48.9	30.5
Sports drinks	0	-	-	-	-	-	-	-	-	-	-	-
Spring water	2	1184	0.0	0.0	0.0	0.0	0.0	0.0	14.6	0.0	0.0	0.0
Yogurt drinks	2	7	240.0	0.0	0.0	59.5	4.9	16.0	77.3	2.9	0.0	0.0
Bread or grain products												
All items	13	1124	154.4	5.0	1.7	24.3	3.3	1.7	430.6	0.9	23.3	7.8
Cereal bars/cereal mixes	1	83	141.0	3.7	0.7	26.9	1.1	0.0	161.5	0.1	23.8	4.7
Crackers/Cracker Sandwiches	4	823	220.1	10.8	3.6	26.2	4.4	4.6	447.4	0.8	44.0	14.8
Granola bar	5	110	168.8	6.4	4.5	25.1	3.7	0.0	104.8	1.2	33.7	23.3
Pretzels	2	33	162.0	1.5	0.3	33.7	3.9	0.0	729.3	14	82	1.8
Rice or corn cakes	1	75	99.1	0.8	0.1	20.7	2.5	0.0	74.3	0.8	7.3	1.1
Frozen desserts												
All items	27	857	196.6	10.0	54	25.2	34	24.3	55 3	07	45.0	24.3
lee cream	23	810	200.1	10.0	5.6	25.2	3.5	24.0	57.0	0.7	46.7	24.0
Non-dairy frozen desserts	20	38	114 1	0.4	0.0	20.1	0.0	20.0	17.0	0.7	0.2	20.1
	-	00	114.1	0.0	0.0	20.0	0.4	0.0	17.0	0.0	0.2	0.0
Salads with dressing												
All items	2	35	137.6	11.9	1.9	6.5	2.4	11.1	405.6	1.7	77.8	12.3
Side salads	2	35	137.6	11.9	1.9	6.5	2.4	11.1	405.6	1.7	77.8	12.3
Snacks												
All items	42	2631	213.6	11.9	2.7	24.8	2.9	0.7	282.6	2.1	49.9	11.9
Chips	30	2032	227.0	13.3	3.1	25.0	3.1	0.4	290.1	2.0	53.4	12.9
Chips - baked or reduced fat	5	152	168.3	7.0	0.9	24.1	1.9	0.0	274.7	1.8	36.8	4.7
Fruit roll-ups/fruit snacks	2	264	180.3	1.7	0.5	41.5	0.1	0.0	135.5	1.7	8.2	2.1
Popcorn	5	183	142.8	84	2.3	14.8	2.9	2.5	237 6	2.8	52.8	14.2
· -F	•		5	.					_00		02.0	· · ·

Food Available In the School Cafeteria

Data not available.

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Food Energy and Macronutrient Composition of Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

	Number Number of items sold	Mean			Mean	amount p	er item			Mean po calorie	ercent of es from	
	of items	sold	calories per item	Total fat (g)	Satura- ted fat (g)	Carbo- hydrate (g)	Protein (g)	Choles- terol (mg)	Sodium (mg)	Dietary fiber (g)	Total fat	Satura- ted fat
Baked goods/dessert												
All items	2	57	408.9	10.6	1.6	74.0	4.9	0.0	436.0	2.2	23.4	3.5
Pastries (danish, toaster pastries, pies)	2	57	408.9	10.6	1.6	74.0	4.9	0.0	436.0	2.2	23.4	3.5
Beverages												
All items	3	192	216.3	0.0	0.0	54.8	0.0	0.0	105.8	0.5	0.0	0.0
Juice-based drinks, 16-20 fl oz	2	143	290.5	0.0	0.0	73.5	0.0	0.0	136.0	0.6	0.0	0.0
Spring water	1	49	0.0	0.0	0.0	0.0	0.0	0.0	17.8	0.0	0.0	0.0
Bread or grain products												
All items	5	85	188.2	6.5	1.5	28.6	4.9	0.9	366.4	1.7	30.8	6.8
Cereal bars/cereal mixes	4	65	186.6	5.6	1.3	30.6	4.9	0.9	369.2	2.1	26.6	5.9
Crackers/Cracker Sandwiches	1	20	193.7	9.6	2.1	22.0	5.0	0.8	357.6	0.4	44.7	9.9
Candy/gum												
All items	58	5332	111.7	2.3	0.7	24.1	0.5	0.4	24.2	0.1	12.0	3.7
Candy with chocolate	9	295	253.9	12.9	5.0	31.5	3.5	3.5	93.5	1.2	45.3	17.8
Candy with chocolate, single												
piece or pop	3	100	63.3	1.9	1.1	11.9	0.4	1.3	8.4	0.1	26.8	16.2
Candy without chocolate	14	846	213.6	2.5	0.6	49.4	0.3	0.0	38.4	0.0	9.6	2.1
Candy without chocolate, single	10	2020	65.7	0.0	0.0	15.0	0.1	0.0	10 E	0.0	10.0	0 5
Chowing gum	19	160	60.7	0.0	0.2	10.2	0.1	0.0	13.5	0.0	10.0	2.5
Chewing guin	15	102	00.5	0.0	0.0	22.0	0.0	0.0	1.5	0.0	1.1	0.2
Snacks												
All items	12	727	106.4	6.5	1.8	9.3	3.2	3.2	240.5	0.8	55.1	18.2
Chips	10	117	217.8	14.2	3.3	21.3	2.5	0.1	261.1	1.6	57.4	13.4
Meat snacks (jerky)	2	610	76.1	4.8	2.0	2.0	6.2	8.9	410.5	0.3	56.2	23.8

Food Available Outside the School Cafeteria

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools.

Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Micronutrient Composition of Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources

	Number of items Number sold Vitamin C Thiamin Riboflavin Vitamin Calcium									
	Number of items	Number sold	Vitamin A (RE)	Vitamin C (mg)	Thiamin (mg)	Riboflavin (mg)	Niacin (mg)	Vitamin B ₆ (mg)	Calcium (mg)	Iron (mg)
Reked seeds/deesert										
All items	26	2590	E2 6	0.2	0.0	0.0	16	0.0	20.0	2.0
Cookies	20	2000	28.0	0.3	0.2	0.2	1.0	0.0	30.9 21 4	2.0
Doughpute	2	2030	20.4	0.2	0.1	0.2	1.3	0.0	21.4	1.9
Doughnuts Dastrias (danish toastar pastrias	2	12	20.4	0.1	0.2	0.1	1.5	0.0	33.0	1.5
nies)	6	200	201.0	0.0	0.4	0.4	12	0.4	138.3	2.2
Snack cakes	8	243	7.1	0.2	0.2	0.2	1.7	0.0	62.9	2.6
Beverages										
All items	51	4220	39.4	46.8	0.0	0.0	0.6	0.0	64.2	0.7
Fruit juice, 100% juice	5	207	41.1	140.6	0.2	0.0	0.8	0.2	258.2	1.3
Iced tea drinks. fortified	2	137	0.0	60.0	0.0	0.0	4.0	0.4	100.0	0.0
Juice-based drink, fortified, 16-20	-									
fl oz	1	15	231.3	75.2	0.7	0.0	0.2	0.0	23.1	0.2
Juice-based drinks. 11.5-12 fl oz	1	9	3.6	105.1	0.0	0.0	0.0	0.0	28.4	0.8
Juice-based drinks, 16-20 fl oz	7	774	15.4	80.8	0.1	0.1	0.1	0.0	38.4	1.2
Juice-based drinks, fortified,										
11.5-12 fl oz	20	1671	76.6	88.2	0.1	0.1	1.3	0.1	48.7	0.7
Milk, flavored, skim or 1% fat,										
fortified	2	35	284.4	16.2	0.2	0.7	0.6	0.2	693.9	1.3
fortified	7	100	155.0	4 5	0.0	0.0	0.5	0.0	007.0	0.0
Milk whole	1	123	155.0	4.5	0.2	0.8	0.5	0.2	607.9 590.7	0.9
Sporte drinke	0	9	151.5	4.0	0.2	0.0	0.4	0.2	362.7	0.2
Sports unitiks	0	1000								
Yogurt drinks	2	7	51.1	60.0	0.0	0.3	0.0	0.0	250.0	0.0
Bread or grain products										
All items	18	1209	18.3	38	02	02	29	02	48 7	31
Cereal bars/cereal mixes	5	148	70.6	7.9	0.2	0.2	3.3	0.3	24.5	3.6
Crackers/Cracker Sandwiches	5	843	6.1	0.2	0.2	0.2	2.2	0.1	56.7	1.5
Granola bar	5	110	0.4	0.4	0.1	0.0	0.7	0.1	22.6	1.2
Pretzels	2	33	0.0	0.0	0.2	0.3	2.2	0.0	15.3	1.8
Rice or corn cakes	1	75	1.8	0.0	0.0	0.0	1.6	0.0	2.3	0.5
Candy/gum										
All items	58	5332	1.4	4.5	0.0	0.0	0.1	0.0	7.5	0.0
Candy with chocolate	9	295	12.8	0.2	0.0	0.1	1.1	0.0	45.9	0.5
piece or pop	3	100	1.7	0.0	0.0	0.0	0.0	0.0	12.0	0.0
Candy without chocolate	14	846	0.0	15.2	0.0	0.0	0.0	0.0	5.1	0.1
Candy without chocolate, single										
piece or pop	19	3929	0.0	4.1	0.0	0.0	0.0	0.0	1.9	0.0
Chewing gum	13	162	0.0	0.0	0.0	0.0	0.0	0.0	2.8	0.0
Frozen desserts										
All items	27	857	57.0	0.5	0.0	0.2	0.4	0.0	82.1	0.5
Ice cream	23	819	59.1	0.5	0.0	0.2	0.4	0.0	85.4	0.6
Non-dairy frozen desserts	4	38	0.7	2.6	0.0	0.0	0.0	0.0	2.0	0.1

Food Available Inside and Outside the School Cafeteria

- Data not available.

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Micronutrient Composition of Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

						Mean amou	int per item			
	Number of items	Number sold	Vitamin A (RE)	Vitamin C (mg)	Thiamin (mg)	Riboflavin (mg)	Niacin (mg)	Vitamin B ₆ (mg)	Calcium (mg)	lron (mg)
Salads with dressing										
All items	2	35	268.2	13.7	0.0	0.0	0.7	0.0	44.7	0.8
Side salads	2	35	268.2	13.7	0.0	0.0	0.7	0.0	44.7	0.8
Snacks										
All items	54	3358	7.0	4.2	0.0	0.0	0.9	0.1	34.6	0.7
Chips	40	2149	7.5	3.5	0.0	0.0	1.0	0.1	41.3	0.7
Chips - baked or reduced fat	5	152	2.3	0.0	0.0	0.0	1.1	0.1	32.9	0.3
Fruit roll-ups/fruit snacks	2	264	5.9	50.0	0.0	0.0	0.0	0.1	9.8	0.4
Meat snacks (ierkv)	2	610	0.0	0.0	0.0	0.0	0.3	0.0	3.7	1.0
Popcorn	5	183	10.5	0.0	0.0	0.0	0.4	0.0	23.5	0.5

Food Available Inside and Outside the School Cafeteria

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Micronutrient Composition of Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

	Number of items	Number sold	Mean amount per item							
			Vitamin A (RE)	Vitamin C (mg)	Thiamin (mg)	Riboflavin (mg)	Niacin (mg)	Vitamin B ₆ (mg)	Calcium (mg)	lron (mg)
Baked goods/dessert										
All items	24	2523	51 5	03	0.2	0.2	16	0.0	39.0	19
Cookies	10	2056	38.2	0.0	0.1	0.2	1.5	0.0	21.4	1.0
Doughnuts	2	72	20.4	0.1	0.2	0.1	1.3	0.0	33.6	1.9
Pastries (danish, toaster pastries,	_	. –			•					
pies)	4	152	167.6	1.2	0.4	0.4	3.9	0.3	144.7	1.8
Snack cakes	8	243	7.1	0.2	0.2	0.2	1.7	0.0	62.9	2.6
Beverages										
All items	48	4028	39.8	41.8	0.0	0.0	0.6	0.0	63.2	0.7
Fruit juice, 100% juice	5	207	41.1	140.6	0.2	0.0	0.8	0.2	258.2	1.3
Iced tea drinks, fortified	2	137	0.0	60.0	0.0	0.0	4.0	0.4	100.0	0.0
Juice-based drink, fortified, 16-20										
fl oz	1	15	231.3	75.2	0.7	0.0	0.2	0.0	23.1	0.2
Juice-based drinks, 11.5-12 fl oz	1	9	3.6	105.1	0.0	0.0	0.0	0.0	28.4	0.8
Juice-based drinks, 16-20 fl oz Juice-based drinks, fortified,	5	631	20.1	29.7	0.0	0.1	0.1	0.0	32.8	1.1
11.5-12 fl oz	20	1671	76.6	88.2	0.1	0.1	1.3	0.1	48.7	0.7
fortified	2	35	284.4	16.2	0.2	0.7	0.6	0.2	693.9	1.3
WIIK, flavored, whole or 2% fat,	7	100	155.0	4 5	0.0	0.0	0.5	0.0	007.0	0.0
tortified	1	123	155.0	4.5	0.2	0.8	0.5	0.2	807.9	0.9
WIIK, WHOIE	1	9	151.3	4.6	0.2	0.8	0.4	0.2	582.7	0.2
Sports driftiks	0	1104	_	_	_	_	_	_	- 07	_
Spring water	2	1184	0.0	0.0	0.0	0.0	0.0	0.0	9.7	0.0
Foguit drinks	2	/	51.1	60.0	0.1	0.5	0.7	0.4	250.0	0.7
Bread or grain products										
All items	13	1124	8.8	0.3	0.2	0.2	1.7	0.1	34.3	1.4
Cereal bars/cereal mixes	1	83	111.9	5.1	0.1	0.1	1.7	0.2	3.0	0.7
Crackers/Cracker Sandwiches	4	823	12.3	0.2	0.2	0.2	2.0	0.2	74.6	1.8
Granola bar	5	110	0.4	0.4	0.1	0.0	0.7	0.1	22.6	1.2
Pretzels Bice or corp cakes	2	33 75	0.0	0.0	0.2	0.3	2.2	0.0	15.3	1.8
		75	1.0	0.0	0.0	0.0	1.0	0.0	2.5	0.5
Frozen desserts										
All items	27	857	57.0	0.5	0.0	0.2	0.4	0.0	82.1	0.5
Ice cream Non-dairy frozen desserts	23	819 38	59.1 0.7	0.5 2.6	0.0	0.2	0.4	0.0	85.4 2.0	0.6
	-	00	0.7	2.0	0.0	0.0	0.0	0.0	2.0	0.1
Salads with dressing										
All items	2	35	268.2	13.7	0.0	0.0	0.7	0.0	44.7	0.8
Side salads	2	35	268.2	13.7	0.0	0.0	0.7	0.0	44.7	0.8
Snacks										
All items	42	2631	7.9	5.1	0.0	0.0	0.9	0.1	37.9	0.7
Chips	30	2032	7.4	4.0	0.0	0.0	1.1	0.2	40.2	0.8
Chips - baked or reduced fat	5	152	2.3	0.0	0.0	0.0	1.1	0.1	32.9	0.3
Fruit roll-ups/fruit snacks	2	264	5.9	50.0	0.0	0.0	0.0	0.1	9.8	0.4
Popcorn	5	183	10.5	0.0	0.0	0.0	0.4	0.0	23.5	0.5

Food Available In the School Cafeteria

- Data not available.

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.
Micronutrient Composition of Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

		Mean amount per item									
	Number of items	Number sold	Vitamin A (RE)	Vitamin C (mg)	Thiamin (mg)	Riboflavin (mg)	Niacin (mg)	Vitamin B ₆ (mg)	Calcium (mg)	lron (mg)	
Bakad goods/dossart											
All items	2	57	208.6	0.3	03	0.4	11	0.4	27.1	3.6	
Pastries (danish toaster pastries	2	57	290.0	0.5	0.5	0.4	4.1	0.4	27.1	5.0	
pies)	2	57	298.6	0.3	0.3	0.4	4.1	0.4	27.1	3.6	
Beverages											
All items	З	192	4.6	136.2	0.1	0.1	0.0	0.0	39.8	10	
Juice-based drinks 16-20 fl oz	2	143	6.2	182.9	0.1	0.1	0.0	0.0	49.4	1.3	
Spring water	1	49	0.0	0.0	0.0	0.0	0.0	0.0	11.8	0.0	
Bread or grain products											
All items	5	85	22.4	83	03	02	43	0.3	44 2	52	
Cereal bars/cereal mixes	4	65	29.2	10.7	0.3	0.2	4.9	0.0	45.9	6.4	
Crackers/Cracker Sandwiches	1	20	0.0	0.2	0.2	0.1	2.4	0.0	38.7	1.1	
Candv/gum											
All items	58	5332	1.4	4.5	0.0	0.0	0.1	0.0	7.5	0.0	
Candy with chocolate	9	295	12.8	0.2	0.0	0.1	1.1	0.0	45.9	0.5	
piece or pop	3	100	1.7	0.0	0.0	0.0	0.0	0.0	12.0	0.0	
Candy without chocolate	14	846	0.0	15.2	0.0	0.0	0.0	0.0	5.1	0.1	
Candy without chocolate single											
piece or pop	19	3929	0.0	4 1	0.0	0.0	0.0	0.0	19	0.0	
Chewing gum	13	162	0.0	0.0	0.0	0.0	0.0	0.0	2.8	0.0	
Snacks											
All items	12	727	1.9	1.2	0.0	0.0	0.5	0.0	14.7	0.6	
Chips	10	117	2.3	3.2	0.0	0.0	1.1	0.0	21.1	0.6	
Meat snacks (jerky)	2	610	0.0	0.0	0.0	0.0	0.3	0.0	3.7	1.0	

Food Available Outside the School Cafeteria

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources

			M	ean number of	Mean amount per item				
	Number of items	Number sold	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)
Baked goods/dessert									
All items	26	2580	1.4	0.0	0.0	0.0	0.0	15.5	6.5
Cookies	10	2056	1.3	0.0	0.0	0.0	0.0	15.8	6.4
Doughnuts	2	72	1.7	0.0	0.0	0.0	0.0	21.2	4.8
Pastries (danish, toaster pastries,									
pies)	6	209	2.0	0.0	0.0	0.0	0.0	9.2	7.3
Snack cakes	8	243	1.8	0.0	0.0	0.0	0.0	13.9	8.8
Beverages									
All items	51	4220	0.0	0.0	0.3	0.0	0.0	0.3	8.7
Fruit juice, 100% juice	5	207	0.0	0.0	2.2	0.0	0.0	0.0	0.0
Iced tea drinks, fortified	2	137	0.0	0.0	0.0	0.0	0.0	0.0	4.3
Juice-based drink, fortified, 16-20									
fl oz	1	15	0.0	0.0	1.0	0.0	0.0	0.0	10.4
Juice-based drinks, 11.5-12 fl oz	1	9	0.0	0.0	0.2	0.0	0.0	0.0	9.2
Juice-based drinks, 16-20 fl oz	7	774	0.0	0.0	0.4	0.0	0.0	0.0	14.1
Juice-based drinks, fortified,	20	1671	0.0	0.0	0.2	0.0	0.0	0.0	0 0
Nills flavorad akim at 1% fat	20	1671	0.0	0.0	0.3	0.0	0.0	0.0	0.0
fortified	2	25	0.0	0.0	0.0	2.0	0.0	16	6.2
Milk flavorod whole or 2% fat	2	35	0.0	0.0	0.0	2.0	0.0	4.0	0.2
fortified	7	123	0.4	0.0	0.0	10	0.0	14.6	6.6
Milk whole	1	9	0.4	0.0	0.0	2.0	0.0	15.4	0.0
Sporte drinke	0	_	0.0	0.0	0.0	2.0	0.0	-	0.0
Spring water	3	1233	0.0	0 0	0.0	0.0	0.0	0.0	0.0
Yogurt drinks	2	7	0.0	0.0	1.4	0.4	0.0	4.1	7.4
- Durad an analysis and durate									
All items	10	1000	1 5	0.0	0.0	0.0	0.0	6.0	1.0
Careal bara/aarool miyoo	10	149	1.5	0.0	0.0	0.0	0.0	0.0	1.0
Creakers/Creaker Sandwishes	5	146	1.1	0.0	0.0	0.0	0.0	3.0	2.3
Granala bar	5	043	1.3	0.0	0.0	0.0	0.2	0.9	1.0
Brotzola	5	22	1.1	0.0	0.0	0.0	0.1	5.0	1.7
Rice or corn cakes	1	75	1.5	0.0	0.0	0.0	0.0	0.0	0.0
Candy/gum									
All itoms	59	5333	0.0	0.0	0.0	0.0	0.0	2.2	47
Candy with chocolate	50	205	0.0	0.0	0.0	0.0	0.0	12.0	4.7
Candy with chocolate single	5	235	0.0	0.0	0.0	0.1	0.2	12.5	4.4
niece or pop	з	100	0.0	0.0	0.0	0.0	0.0	19	27
Candy without chocolate	14	846	0.0	0.0	0.0	0.0	0.0	23	11.8
Candy without chocolate, single	14	040	0.0	0.0	0.0	0.0	0.0	2.0	11.0
piece or pop	19	3929	0.0	0.0	0.0	0.0	0.0	0.8	3.7
Chewing gum	13	162	0.0	0.0	0.0	0.0	0.0	0.0	1.6
Frozen desserts									
All items	27	857	0.4	0.0	0.0	0.2	0.0	9.6	4.0
Ice cream	23	819	0.4	0.0	0.0	0.2	0.0	9.9	3.9
Non-dairy frozen desserts	4	38	0.0	0.0	0.0	0.0	0.0	0.0	6.9

Food Available Inside and Outside the School Cafeteria

1 Also includes legumes, soybean products, nuts, and seeds.

- Data not available.

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

		Mean number of servings per pyramid group						Mean amount per item		
	Number of items	Number sold	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)	
Salads with dressing										
All items	2	35	0.0	1.6	0.0	0.0	0.0	11.4	0.3	
Side salads	2	35	0.0	1.6	0.0	0.0	0.0	11.4	0.3	
Snacks										
All items	54	3358	1.1	0.3	0.0	0.0	0.0	9.5	0.2	
Chips	40	2149	1.4	0.4	0.0	0.0	0.0	11.6	0.0	
Chips - baked or reduced fat	5	152	0.4	0.9	0.0	0.0	0.0	7.0	0.6	
Fruit roll-ups/fruit snacks	2	264	0.0	0.0	0.9	0.0	0.0	0.0	4.9	
Meat snacks (jerky)	2	610	0.0	0.0	0.0	0.0	0.5	3.3	0.0	
Popcorn	5	183	1.2	0.0	0.0	0.0	0.0	7.2	0.0	

Food Available Inside and Outside the School Cafeteria

Also includes legumes, soybean products, nuts, and seeds.
Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

			Me	ean number of	servings pe	er pyramid gr	oup	Mean amount per item	
	Number of items	Number sold	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)
Baked goods/dessert									
All items	24	2523	1.3	0.0	0.0	0.0	0.0	15.6	6.5
Cookies	10	2056	1.3	0.0	0.0	0.0	0.0	15.8	6.4
Doughnuts	2	72	1.7	0.0	0.0	0.0	0.0	21.2	4.8
Pastries (danish, toaster pastries,									
pies)	4	152	1.8	0.0	0.0	0.0	0.0	10.5	7.5
Snack cakes	8	243	1.8	0.0	0.0	0.0	0.0	13.9	8.8
Beverages									
All items	48	4028	0.0	0.0	0.3	0.0	0.0	03	83
Fruit juice 100% juice	5	207	0.0	0.0	22	0.0	0.0	0.0	0.0
Iced tea drinks fortified	2	137	0.0	0.0	0.0	0.0	0.0	0.0	4.3
Juice-based drink, fortified, 16-20	-	107	0.0	0.0	0.0	0.0	0.0	0.0	40.4
fl oz	1	15	0.0	0.0	1.0	0.0	0.0	0.0	10.4
Juice-based drinks, 11.5-12 fl oz	1	9	0.0	0.0	0.2	0.0	0.0	0.0	9.2
Juice-based drinks, 16-20 fl oz	5	631	0.0	0.0	0.4	0.0	0.0	0.0	13.2
11.5-12 fl oz	20	1671	0.0	0.0	0.3	0.0	0.0	0.0	8.8
Milk, flavored, skim or 1% fat,									
fortified Milk flavored whole or 2% fat	2	35	0.0	0.0	0.0	2.0	0.0	4.6	6.2
fortified	7	123	0.4	0.0	0.0	10	0.0	1/ 6	66
Milk whole	1	125	0.4	0.0	0.0	2.0	0.0	15.4	0.0
Sporte drinke	0	_	0.0	0.0	0.0	2.0	0.0	- 10.4	0.0
Spring water	2	118/	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Yogurt drinks	2	7	0.0	0.0	1.4	0.4	0.0	4.1	7.4
Bread or grain products									
All items	13	1124	1.7	0.0	0.0	0.0	0.0	3.7	0.3
Cereal bars/cereal mixes	1	83	0.8	0.0	0.0	0.0	0.0	2.9	3.4
Crackers/Cracker Sandwiches	4	823	1.9	0.0	0.0	0.0	0.0	8.9	0.3
Granola bar	5	110	1.1	0.0	0.0	0.0	0.1	5.6	1.7
Pretzels	2	33	2.6	0.0	0.0	0.0	0.0	0.0	0.0
Rice or corn cakes	1	75	1.5	0.0	0.0	0.0	0.0	0.0	0.0
Frozen desserts									
All items	27	857	0.4	0.0	0.0	0.2	0.0	9.6	4.0
Ice cream	23	819	0.4	0.0	0.0	0.2	0.0	9.9	3.9
Non-dairy frozen desserts	4	38	0.0	0.0	0.0	0.0	0.0	0.0	6.9
Salads with dressing									
All items	2	35	0.0	1.6	0.0	0.0	0.0	11.4	0.3
Side salads	2	35	0.0	1.6	0.0	0.0	0.0	11.4	0.3
Snacks									
All items	42	2631	1.3	0.4	0.0	0.0	0.0	10.3	0.3
Chips	30	2032	1.3	0.5	0.0	0.0	0.0	11.5	0.0
Chips - baked or reduced fat	5	152	0.4	0.9	0.0	0.0	0.0	7.0	0.6
Fruit roll-ups/fruit snacks	2	264	0.0	0.0	0.9	0.0	0.0	0.0	4.9

Food Available In the School Cafeteria

1 Also includes legumes, soybean products, nuts, and seeds.

- Data not available.

Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

	Number of items			Me	an number o	f servings pe	er pyramid gr	oup	Mean amount per item			
		Number sold	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)			
Snacks Popcorn	5	183	1.2	0.0	0.0	0.0	0.0	7.2	0.0			

Food Available In the School Cafeteria

1 Also includes legumes, soybean products, nuts, and seeds. Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Food Guide Pyramid Servings for Competitive Foods Available During the School Day, Weighted by Sales, All Schools and Sources — Continued

		Mean number of servings per pyramid group							Mean amount per item	
	Number of items	Number sold	Grain	Vegetable	Fruit	Dairy	Meat ¹	Discre- tionary Fat (g)	Added Sugars (tsp)	
Baked goods/dessert										
All items Pastries (danish, toaster pastries,	2	57	3.2	0.0	0.1	0.0	0.0	7.3	7.7	
pies)	2	57	3.2	0.0	0.1	0.0	0.0	7.3	7.7	
Beverages										
All items	3	192	0.0	0.0	0.3	0.0	0.0	0.0	11.9	
Juice-based drinks, 16-20 fl oz	2	143	0.0	0.0	0.4	0.0	0.0	0.0	16.0	
Spring water	1	49	0.0	0.0	0.0	0.0	0.0	0.0	0.0	
Bread or grain products										
All items	5	85	1.2	0.0	0.0	0.0	0.2	5.4	1.6	
Cereal bars/cereal mixes	4	65	1.4	0.0	0.0	0.0	0.1	4.3	1.2	
Crackers/Cracker Sandwiches	1	20	0.7	0.0	0.0	0.0	0.3	8.9	2.8	
Candy/gum										
All items	58	5332	0.0	0.0	0.0	0.0	0.0	2.3	4.7	
Candy with chocolate	9	295	0.6	0.0	0.0	0.1	0.2	12.9	4.4	
piece or pop	3	100	0.0	0.0	0.0	0.0	0.0	1.9	2.7	
Candy without chocolate	14	846	0.0	0.0	0.0	0.0	0.0	2.3	11.8	
candy without chocolate, single	10	2020	0.0	0.0	0.0	0.0	0.0	0.0	27	
Chewing gum	13	162	0.0	0.0	0.0	0.0	0.0	0.0	1.6	
e			0.0	0.0	0.0	010	0.0	0.0		
Snacks										
All items	12	727	0.4	0.2	0.0	0.0	0.2	5.5	0.0	
Chips	10	117	0.5	1.0	0.0	0.0	0.0	13.7	0.2	
Meat snacks (jerky)	2	610	0.0	0.0	0.0	0.0	0.5	3.3	0.0	

Food Available Outside the School Cafeteria

1 Also includes legumes, soybean products, nuts, and seeds. Note: Number of items is the number offered at the three schools. Number sold is the total sales for the three schools. Estimates are the result of weighting items by the number sold within school, and then giving equal weight to schools. Weighted and unweighted results may be identical for subcategories with small numbers of items offered at each school.

Chapter 5 Methodology for Assessing the Impact of Competitive Food Policies

Competitive food policies determine the types of foods available to students during the school day, in addition to USDA reimbursable meals. These policies may concern the locations and time periods of availability, the types of foods prohibited or required, and the pricing of foods. Competitive food policies, together with policies regarding USDA reimbursable meal planning, define the school food environment.

This chapter discusses "an approach/methodology to be used on an ongoing basis by FNS and/or State Child Nutrition Agencies to monitor the impact of schools changing their policies on competitive foods on the nutritional quality of foods at school."¹

The impact of competitive food policies can be expressed by the following research questions:

- 1. What is the nutritional quality of competitive foods available at school?
- 2. Do competitive foods "crowd out" the selection of USDA reimbursable meals?
- 3. How much do competitive foods add to (or subtract from) total food energy, key nutrients, added sugars, and fat served to students during the school day?

Estimating the impact of changes in competitive food policies requires data collection sufficient to answer one or more of these research questions. The primary design considerations for a data collection plan are:

- Overall design of the study: cross-sectional or longitudinal data;
- Ability to collect comparable information about competitive food policies across schools; and
- Ability to collect comparable data on foods offered and served, across schools.

This chapter provides a discussion of the operational and methodological issues involved in collecting data and answering the research questions. The discussion provides recommendations for large-scale monitoring, and is informed by the findings of the data collection portion of this feasibility study.

Overall Design Considerations

A study of the impact of changes in competitive food policies requires two types of information: data about policies and data on outcome measures related to those policies. Because FNS is interested in monitoring the impact of competitive food policies over time, the monitoring effort will require ongoing data collection from individual schools.

¹ USDA/FNS Statement of Work, Task Order: Feasibility Study to Develop Methodology to Monitor Impact of Changes in Competitive Food Policies, January 2003.

There are two basic design options for data collection and evaluation of competitive food policies:

- Cross-sectional study of the variation in policies across schools, and
- Longitudinal study of the impact of policy changes within schools.

A cross-sectional design involves a recurring survey with a sample of schools that changes over time; a longitudinal design involves a recurring survey of a single sample of schools followed over time.² Cross-sectional data can provide a wealth of information describing policies and outcomes, but cannot support evaluation of the causal impact of school policies. Longitudinal data can be used to estimate the impact of changes in policies.

Cross-Sectional Design

From an operational standpoint, a cross-section design spreads the burden of ongoing monitoring over more schools; each year, a new sample of schools might be surveyed to provide estimates of the prevalence of different competitive food policies.

A cross-sectional study of a nationally representative sample of schools can provide a wealth of information describing school environments—including competitive food policies/practices and outcome measures. There is currently only limited information about the foods available to students, outside of USDA reimbursable meals. For example, SNDA-II found that a la carte items are available in nine out of ten NSLP schools and vending machines are in one-third of NSLP schools, but SNDA-II did not collect data to describe the nutritional quality of a la carte foods or the volume of purchases. A nationally representative cross-sectional study could be used to assess the current status of competitive food availability.

Cross-sectional studies are limited, however, in their ability to estimate the causal impact of policy changes. For example, a cross-sectional study can tell that particular competitive food policies are associated with particular nutritional outcomes, but cannot attribute causality to the competitive food policies. The inability to attribute causality in a cross-sectional study is due to the fact that competitive food policies are endogenous—these policies are not adopted in isolation, but are likely to be related to other school characteristics that influence student food selections.

As an illustration of the fact that competitive food policies are endogenous, consider that schools in low-income areas may be less likely to offer competitive foods because a large percentage of the student population receives free USDA meals and cannot afford to purchase vending and a la carte items in addition to, or in lieu of, reimbursable meals. Another example of endogenous policies is that a la carte food offerings can, in some schools, be limited by the physical characteristics of the school cafeteria.

Because competitive food policies are endogenous, the impact of policies cannot be determined from cross-sectional data without substantial effort to control for school characteristics that may influence both policies/practices and outcomes. With cross-sectional data, analyses of the impact of

² The Current Population Survey, March Demographic File is usually thought of as a recurring cross-section because the sample changes from year to year (although the survey has a rotating sample, so that a small portion of respondents appear in consecutive files). Examples of true longitudinal surveys are the Panel Study of Income Dynamics or National Longitudinal Surveys, which followed a single sample of respondents over time.

competitive food policies must include modeling that accounts for the selection of competitive food policies by schools. The analytic modeling required to account for endogenous policies imposes additional data collection burdens on participating schools, and analysis results are generally less certain when compared to a longitudinal study.

Longitudinal Design

A longitudinal design places all data collection burden on a single sample of schools that is surveyed on a recurring basis. A longitudinal approach may minimize total data collection burden, however; if respondents at the single sample of schools are stable, they may become familiar with the instruments, and hence more efficient with each round of data collection. In addition, data collection can be reduced after an initial baseline survey because some school characteristics are stable over time.³

A longitudinal study is the best design for studying the impact of changes in competitive food policies. With this design, changes in policies are related to changes in outcomes **within** schools. To support impact analyses, a longitudinal study would, at a minimum, collect information about competitive food policies and outcome measures to address at least one of the research questions listed at the start of this chapter. Schools with changes in competitive food policies would be identified from data collected at multiple points in time. Changes in outcomes would also be measured by comparing data collected at multiple points in time. In a longitudinal study, the impact of competitive food policies is identified from the within-school changes in outcomes in schools with changes in policies.

The one limitation of a longitudinal study design is that impact estimates can only be obtained if changes in policies are observed within the study sample. If competitive food policies are stable, a very large panel of schools would need to be followed in order to ensure that some schools will experience changes in policies. (Likewise, with a cross-sectional study design, the expected variation in competitive food policies is an important consideration in determining sample size.)

For analyses, longitudinal data allow researchers to estimate the impact of changes in policies, while controlling for school characteristics that are difficult (and costly) to measure, but may influence both school policies and the outcomes of interest. Many characteristics of schools may be considered "fixed effects" if they are relatively stable over time. Fixed effects would include the physical characteristics of the food service areas, demographic composition of the student body, and regional food preferences and dietary habits. This "fixed effects" approach with longitudinal data affords savings in data collection costs.

Although a longitudinal approach relies mainly on observation of within-school changes in policies and outcomes, this design also allows researchers to control for time trends. For example, changes in student food choices over time may reflect general trends in the popularity of particular food items, or the acceptance of more healthful food choices, or the reformulation of food products. These general trends will be observed across all schools, regardless of whether the schools experience policy changes.

³ One consideration in designing a monitoring effort is whether to use school staff as respondents or to use outside observers. This discussion of burden assumes school staff respondents.

The next sections of this chapter discuss options and recommendations for measuring competitive food policies and outcomes. The final section of the chapter provides a discussion of analysis methods.

Measuring Competitive Food Policies

Competitive food policies, whether implicit or explicit, govern the types of foods available, locations of sales, and hours of availability. The critical consideration in designing an approach for monitoring the impact of changes in competitive food policies is to develop data collection methods that collect comparable data across schools and over time.

This feasibility study collected data about competitive food policies and practices from two types of respondents: SFA directors and school principals. SFA directors were asked about district-level policies/procedures and SFA operations; school principals were asked school policies and procedures.

The surveys asked a wide range of questions about the availability of competitive foods (venues, locations, hours), types of competitive foods (required and/or prohibited foods), persons responsible for approving competitive foods, pricing strategies for competitive foods, and financial arrangements (who receives the revenue and how is it used). Some of the data collected via survey were validated during on-site training visits, or revised based on debriefing discussions with respondents. The survey instruments used for this study are too long to be used as a model for ongoing monitoring activities, but they provide a good basis for prioritizing survey methods and topic areas.

Three main problems were found regarding data collection on competitive food policies:

- 1. Identifying the appropriate respondent for the survey of school policies;
- 2. Defining food "sources"; and
- 3. Finding a lack of clear policies.

Identifying the Respondent

Competitive food venues do not fall within the jurisdiction of a single person. In the three schools surveyed, the school food service had responsibility for most of the food served in the cafeteria, whereas one or two other organizations had responsibility for foods available outside the cafeteria.⁴

One of the most important findings was that school principals, whose jurisdiction covers the entire school, did not always have knowledge of food policies and practices. This finding suggests that complete and accurate information about the school food environment cannot be obtained by a simple mail survey of school principals. Principals were not motivated to seek out the information requested on the surveys, or to delegate the task to others; they returned incomplete or inaccurate surveys.

There are two possible alternatives to a principal survey: (1) use a screener survey to identify respondents, and send multiple survey modules to multiple respondents within a school;⁵ or (2) rely

⁴ In each of the three schools surveyed, the food service is responsible for all foods available in the cafeteria with one exception: the vending machine in the cafeteria of School #1 is governed by a citywide beverage contract.

⁵ The respondents would be identified as persons responsible for each food venue within the school.

on cafeteria managers (or SFA directors) to report on competitive foods available throughout their schools. The first option would be costly to field because it adds a round of data collection and increases the number of respondents requiring follow-up. Recommendation (2) may be viable because this feasibility study found that cafeteria managers are concerned about the school food environment and are motivated to obtain information requested on surveys. Cafeteria managers, however, cannot be expected to shoulder the combined burden of a detailed cafeteria survey and a survey of other competitive food venues. It may be reasonable, however, to ask cafeteria managers to respond to a well-designed short-form monitoring survey.

Defining Food Sources

This study designed Principal and SFA Director surveys with a parallel structure to collect similar information about each possible source of competitive foods available during the school day: (a) a la carte, (b) snack bars, (c) vending machines, and (d) school stores. Visits to the schools, however, quickly revealed that a "food source" is not necessarily comparable across schools.

The main difficultly in collecting and examining data on foods "by source" is that different school food service operations may use different methods of food delivery to sell the same a la carte items to students. "Snack bar" may be just a name given to a separate a la carte line or window; vending machines may be used to sell a la carte beverages so as to save space on the line. It seems clear that the presence of different food sources in and near the cafeteria may be related to space and staff constraints faced by the school food service.

This study found that the structure of the Principal Survey resulted in missing data on a la carte-only operations in one school. That school has no a la carte-only service other than a snack bar. As a result, the respondent completed the snack bar portion of the survey and "skipped out" of questions about certain a la carte-only policies.⁶

An argument can be made to use the term "a la carte" or "food service foods" to identify all foods made available by the school food service in the cafeteria, and all policies related to those foods. Consistent with this argument, we grouped competitive foods by location inside and outside the cafeteria for the analyses presented in Chapter 4. An examination of individual items available from different sources in the cafeteria at different schools (Appendix E) supports this grouping for the three schools surveyed.

An important consideration is that the grouping of data that made sense for this study may not make sense in a larger study. Some school food service operations provide food outside the cafeteria—for example, in outdoor kiosks in warm-weather locations. For this reason, we recommend that the structure of data collection and analysis in future studies focus on the distinction between the following four categories of competitive foods:

- 1. School food service, during lunch
- 2. School food service, outside lunch
- 3. Non-food service, during lunch
- 4. Non-food service, outside lunch

⁶ These responses were subsequently obtained during the debriefing.

This structure is important for collecting information about competitive food offerings and for collecting information about competitive food policies and practices.

Lack of Clear Policies

carte items?).

Evidence from the three schools included in this feasibility study showed that schools do not necessarily have clear policies about competitive foods. As a result, we found that respondents did not interpret questions about policies consistently.

Two different types of questions were used to collect data about competitive school policies:

Type 1	Respondents were asked about observable practices (Do you have vending machines? How many? Where are they?). ⁷
Type 2	Respondents were asked about unobservable policies (Do you restrict the types of foods available in vending machines? What is your pricing strategy for a la

In many environments, questions about policies would be categorized as "observable". Individual schools, however (like other small organizations) may have many policies that are not written down. Because they are not codified in formal documents, the difference between a "policy" and a "practice" is blurred, and respondents may have difficulty responding to survey questions.

Type 1 and Type 2 questions provided a mix of factual questions about the school environment and questions that may have required subjective judgment. Responses to Type 1 survey questions were verified for this feasibility study through on-site visits, thus providing some validation of the reliability of responses. Responses to Type 2 questions are not easily verified, and may be interpreted differently by different respondents. For example, when we ask whether certain foods are prohibited from vending machines, some respondents will answer affirmatively only if they have clear written policies about certain food items; other respondents will answer affirmatively because they know that certain items are not in their machines (this example was revealed during our debriefings).

The types of survey questions must be considered together with the identity of respondents. This feasibility study used a Principal Survey because it was expected that only the principal would know the policies that affect multiple operations in the school. If instead of asking about policies, we ask only about observable practices, there are other potential respondents to the survey.

It can be argued that unobservable policies and procedures result in observable practices, and that a survey aimed at obtaining comparable data should limit questions to observable practices ("just tell us what you do, not what you think you might be doing"). For example, we can ask respondents about their pricing strategies, or we can collect data on prices and analyze their pricing practices. Likewise, we can ask whether certain items are prohibited from vending machines, or we can collect information about the contents of vending machines. The difficulty in this case is that FNS wants to know the impact of changes in competitive food policies, without first having evidence that district-and school-level policies (as opposed to practices) exist.

⁷ We use the term **observable** to mean "verifiable by an outside observer".

The first issue then is how to measure "policies". With the data collected for this study, there are several options for defining baseline policies; changes in those policies would require additional data from the same schools at a subsequent point in time. A series of policy-change variables could be used together in a regression analysis to estimate the marginal impact of changes in a number of competitive food policies. For example, policy variables could include indicators of:

- Any non-food service-operated food venues located in the cafeteria?
- Any non-food service-operated food venues located outside the cafeteria?
- Whether competitive food items have to meet written state or district nutrition standards?
- Whether competitive food items have to meet written state or district price standards?
- Any pouring rights contract governing beverage sales in the school?

These questions require little interpretation by respondents, and are likely to elicit consistently accurate responses across schools and over time. Additional policy questions could be posed in the same concise question style.

Measuring Nutritional Outcomes

The three research questions posed in the introduction to this chapter are shown below. Each research question corresponds to an outcome measure. Outcome measures are expressed as **changes** over time, because the overriding research question is the impact of changes in competitive food policies. These outcome measures may be used (separately or together) to evaluate the impact of changes in competitive food policies in schools.

Research Question

- 1. What is the nutritional quality of competitive foods available at school?
- 2. Do competitive foods "crowd out" the selection of USDA reimbursable meals?
- 3. How much do competitive foods add to (or subtract from) total food energy, key nutrients, added sugars, and fat served to students during the school day?

Outcome Measure

Change in average nutrient content of competitive foods offered

Change in participation in USDA reimbursable meal programs

- a. Change in nutrient content of all foods served/sold OR
- b. Change in marginal impact of competitive foods on total nutrients selected during the school day (using an expected value for the nutrient content of reimbursable meals)

This feasibility study collected much of the data that are analogous to baseline data that would be collected to address each research question.⁸

- Nutrient content of competitive foods offered—This outcome can be measured by collecting a list of all competitive foods offered in schools, and entering those food products in a nutrient database. This study fielded the Cafeteria Survey and Inventories of Competitive Foods to obtain lists of food products offered and data on number of products served. The reporting of servings data imposed a large burden on respondents, but servings data are not needed to evaluate the nutrient content of foods offered.
- Percent of students receiving reimbursable meals—This outcome can be measured by collecting data on daily meal counts and numbers of students eligible for free and reduced-price meals; the participation rates calculated from these numbers might be adjusted by daily attendance data. This study collected meal counts from cafeteria managers and attendance data from school principals.⁹ Reporting of attendance and meal counts did not impose a large burden on respondents. Future studies, however, must determine the most efficient method for collecting attendance data if school principals are not used as respondents for a survey of competitive food policies.
- Impact of competitive foods on nutrient content of all foods as selected—This outcome can be measured by collecting data on the nutrient content of competitive foods served. As discussed in the following section, simulation analyses are recommended to determine the impact of competitive foods on total nutrients served during the school day, rather than analyzing both competitive foods and reimbursable meals from the same sample of schools. For this feasibility study, the Cafeteria Survey and Inventories of Competitive Foods collected information about servings of competitive foods. Collection of servings data, however, imposed a substantial burden on respondents, and it became clear that a study attempting to collect servings of competitive foods would have to confront the problem of missing data (from non-food service-operated venues) and inaccurate data (cafeterias providing counts of portions prepared, rather than portions served).

The varying success of data collection on each outcome measure suggests that future studies may be able to evaluate only the first two outcomes for all schools. Missing data will likely result in a selected sample of schools for which the third outcome can be evaluated.

Analyzing Nutritional Outcomes

This section discusses the analysis methods that may be applied to evaluate each of the three research questions identified in this chapter. It is important to remember throughout the discussion that the three research questions identified above represent different requirements for data collection and different levels of precision for the true outcome measure of interest. The true outcome measure of

⁸ Data were collected at a single point in time to evaluate feasibility, and cannot be used to analyze changes in outcomes.

⁹ We did not recognize the need to test the ability to collect data on numbers of students eligible for free and reducedprice meals until after the data collection period. The data are not needed for the feasibility study, because we are not estimating the impact of policy changes with just one point of observation per school; it is expected that these data would not be difficult to collect.

interest is the impact of changes in competitive food policies on the average nutritional quality of foods selected by students during the school day. This true measure of interest, however, can only be precisely measured by conducting 24-hour recalls with students; all other means of data collection and analysis provide approximations for the true outcome measure of interest. Each of these approximations is discussed below.

What Is the Impact of Changes in Competitive Food Policies on the Nutritional Quality of Competitive Foods Available at School?

Changes in competitive food policies should be visible in the types of foods available to students at school. The types of foods available define the boundaries of the quality of food that students may select (i.e., students cannot purchase low-calorie items at school if the items are not offered for sale at school). In this sense, analysis of food offerings provides an assessment of the first order impact of a change in food policies.

Chapter 4 provided summary information about the nutritional quality of competitive foods offered at the three schools in this study. That chapter provided information about the average nutrient content of competitive food products, including macronutrients, micronutrients, and Food Pyramid servings.

Longitudinal data collection about competitive food policies and competitive food offerings could be used in the following model to analyze the impact of changes in policies:

$$(AvgCalories_{s,t} - AvgCalories_{s,t-1}) = \alpha + \delta X_s + \beta_i \Delta policy_{s,i} + ... + \beta_j \Delta policy_{s,j} + \epsilon$$

The dependent variable in this model is the change from time *t*-1 to time *t* in average calories for competitive food items offered in school *s*. The right hand side has an intercept term (α), an array of school characteristics for school s (X_s), and variables measuring the change in policies (Δ policy_{s,j} is the change in policy j at school s). In this model, the coefficients on the policy change variables ($\beta_i \dots \beta_j$) measure the marginal impact of changes in each competitive food policy on the average caloric content of competitive food items. The same form of the model would be estimated with additional measures of nutritional quality to examine the ways in which competitive food offerings are changing in response to changes in policy.¹⁰

Do Competitive Foods "Crowd Out" the Selection of USDA Reimbursable Meals?

The availability of competitive foods can have two effects on food selections: these foods can add to total food consumption during the school day, or they can substitute for the USDA reimbursable meals. As an example of the substitution effect, students with money for a reduced-price or full-price lunch may choose to spend this money on competitive foods instead. Analysis of the impact of changes in policies on meal counts for reduced-price and full-price lunch could be examined at low cost, and should be part of the analysis of longitudinal data. The following model demonstrates this approach:

(RP redemption_{s,t} - RP redemption_{s,t-1}) = $\alpha + \delta X_s + \beta_i \Delta \text{ policy}_{s,i} + \dots + \beta_j \Delta \text{ policy}_{s,j} + \epsilon$ (1)

¹⁰ As discussed previously, the average nutritional quality of competitive foods may change due to general trends in food preferences or reformulation of food products. These trends will be common to all schools, regardless of whether they have changed policies, and these trends will therefore not be attributed to changes in policy.

(FP redemption_{s,t} – FP redemption_{s,t-1}) =
$$\alpha + \delta X_s + \beta_i \Delta \text{ policy}_{s,i} + \dots + \beta_j \Delta \text{ policy}_{s,j} + \epsilon$$
 (2)

where

RP redemption = (avg daily # reduced-price lunch served) / (#students eligible for RP lunch) FP redemption = (avg daily # full-price lunch served) / (#students not eligible for free or reduced-price lunch)

The denominator for both measures of redemption might be adjusted by the average rate of absenteeism during the target week.

If changes in competitive food policies reduce or increase the number of items available as "meal alternatives", we would expect to see impacts on rates of reimbursable meal purchases among students who pay reduced and full price.

How Much Do Competitive Foods Add to (or Subtract from) Total Food Energy, Key Nutrients, Added Sugars, and Fat Served to Students During the School Day?

This is the most difficult research question to address, due to the difficulty of collecting data on servings of competitive foods, and due to the denominator problem discussed below. This question may be evaluated with two fundamentally different approaches.

- 1. Collect data on all foods served throughout the school day and evaluate the contribution of competitive foods to total nutritional quality of all foods served.
- 2. Collect data on competitive foods served and evaluate the marginal contribution of competitive foods to total nutritional quality, under various assumptions about the quality of reimbursable meals (i.e., using the NSLP standards or extant data from SNDA-II or the Integrated Study).

These approaches differ with respect to the amount of data to be collected. Both approaches, however, must confront the denominator problem inherent in a study of all foods served at schools.

The Denominator Problem

The denominator problem is that it is not possible to collect data about foods consumed on a perstudent basis from a cafeteria survey. School cafeterias monitor production of individual food items and servings of reimbursable meals. Prior studies of the nutritional quality of school foods have focused almost exclusively on the nutritional quality of the reimbursable meal, without estimating the nutrients consumed per student (for example, studies of meal quality do not account for a la carte purchases of "second helpings" or a la carte-only items).¹¹

For this study, we collected data about the competitive foods offered and the sales of individual items, where available. There is no way, however, to assign these sales to individual students to assess the percent of students purchasing competitive foods. The only way to use these data is to assess competitive food purchases, on average, using different assumptions about the denominator (i.e., the universe of students who purchase competitive foods). Table 5.1 demonstrates this approach for

¹¹ One exception was the analysis of 24-hour dietary recalls in SNDA-I.

Schools #2 and #3; School #1 is excluded from analyses in this chapter because the sales data for competitive foods were incomplete.

Table 5.1 shows the total number of competitive food items purchased at Schools #2 and #3 during the target week: 12,232 and 4,257, respectively. The average daily number of competitive food items purchased was 2,446 at School #2 and 851 at School #3. If we assume that all students are equally likely to purchase competitive foods, then the average number of items purchased per student per day is equal to the average daily number of items sold divided by average daily attendance: on average, students purchased 0.87 competitive food items per day at School #2 and 0.66 items per day at School #3. (Another way to look at this is that, on average, 87 percent of students purchased one competitive food item each day at School #2, and 66 percent of students purchased a competitive food item each day at School #2.)

Table 5.1

Measure	School #2	School #3
Competitive foods		
Number of items sold during target week	12,232	4,257
Average daily number of items sold	2,446	851
Average calories per item (weighted by servings)	161.97	214.16
Student counts		
Attendance, daily average	2,823	1,297
Reimbursable meal counts, daily average	1,472	600
Average number competitive food items purchased per student per day		
Denominator = all students	07	00
[lower bound estimate]	.87	.00
Denominator = (attendance – meal count) [higher bound estimate]	1.81	1.22
Average daily calories consumed from		
competitive foods per student		
Denominator = all students	140.01	1/1 35
[lower bound estimate per student]	140.91	141.55
Denominator = (attendance – meal count) [higher bound estimate per student]	293.16	261.28

Alternative Denominators for Assessing Competitive Food Sales on a Per Student Basis

Using total attendance as the denominator assumes that all students are equally likely to purchase competitive food items; this assumption yields a lower bound on the average number of competitive food items purchased per student per day. Alternative assumptions provide other bounds on the average consumption of competitive food items. A higher bound estimate on the average number of competitive foods purchased per student per day (among students purchasing any) might exclude from the denominator all students purchasing a reimbursable meal (i.e., those with bag lunch or a la

carte entrée purchase). The resulting higher bound estimates of the number of items per student purchasing competitive foods are 1.81 items per student in School #2 and 1.22 items per student in School #3.

Additional assumptions could be made about the denominator in order to provide a basis for comparison of competitive food purchases across schools. It should be clear, however, that analysis of the number of competitive food items purchased (or the average nutritional content of foods purchased) must necessarily rely on distributional assumptions about the number of students purchasing those items. Without 24-hour recalls with students, it is not possible to obtain data about per student consumption of foods that are served from multiple food venues within a school.¹²

Using these same assumptions about the average daily purchases of competitive foods items by students, the food energy contribution of competitive foods can be examined on a per student basis. Table 5.1 shows that if all students are equally likely to purchase competitive foods, then the average daily contribution of competitive foods to calorie intake is 141 calories in both Schools #2 and #3. (On average, competitive food items in School #3 have higher caloric content than in School #2, but students in School #3 purchase fewer items.) The higher bound estimates for the food energy contribution of competitive foods on a per student basis are 293 calories at School #2 and 261 calories at School #3.

Contribution of Competitive Foods to the Nutritional Quality of All School Foods

Our proposal recommended that nutrient analyses and servings data for reimbursable meals are not needed to estimate the impact of competitive foods on the overall nutritional quality of all school foods. The data collection plan included reimbursable meals information (food product descriptions and servings data), but these data were not entered in FIAS and reimbursable meal components were not included in the analyses presented in Chapter 4.

Building on the information presented in Table 5.1, simulation analyses could be conducted to determine upper and lower bounds for the overall nutritional quality of foods at school, given information collected about competitive foods. Table 5.2 provides an analysis of the impact of competitive foods on the total calories provided to students who receive reimbursable meals. The NSLP nutrition standards for reimbursable meals, and the SNDA-II findings from a nationally representative survey, provide two benchmarks against which to gauge the impact of competitive foods. Use of these benchmarks allows evaluation of the average impact of competitive foods.

When estimated average daily calories from competitive foods is combined with benchmarks for the content of reimbursable lunches, we see that competitive foods raise the food energy content of food selections during the school day to between 36 and 39 percent of total daily recommended energy allowance.

The combination of competitive food data from one sample with out-of-sample information—through simulation analysis—could greatly reduce the burden imposed on respondents to USDA surveys. The use of benchmarks for evaluating the range of average impacts is valid—if both SNDA-II and the competitive foods survey use nationally representative samples, we can combine the average characteristics of reimbursable meals and the average characteristics of competitive foods without

¹² Even if POS systems provided the data to derive per student consumption of food service items from the cafeteria lines, the prevalence of vending machines and off-line snack bars precludes derivation of per student servings data.

collecting reimbursable meal and competitive food data from the same sample. We cannot, however, say anything about the distribution of the combined data; for example, we cannot determine the percentage of students with total food selections during the school day above certain standards.

Table 5.2

Contribution of Competitive Foods to the Overall Nutritional Quality of Foods at School

Measure	Food Energy	Percent REA
NSLP standard for food energy	825	33.0%
SNDA-II findings for average food energy provided in USDA lunch in secondary schools	750	30.0
<i>Feasibility sample</i> Average daily food energy from competitive foods, per student	141	5.6
Total food energy from all foods, assuming NSLP standard met for USDA lunch	966	38.6
Total food energy from all foods, assuming school is average (i.e., SNDA-II findings)	891	35.6

Conclusions

For ongoing monitoring and analysis, we have three recommendations:

- 1. Maintain updated information about state policies on the FNS website.
- 2. Monitor competitive food practices with a survey of schools (fielded on a regular basis), to obtain information about school practices with regard to competitive foods. Limit data collection at the school-level to observable practices, so that collected data will be consistent across schools and over time.

This sample should be large enough to provide national estimates of the prevalence of various competitive food practices.

3. Begin a longitudinal survey with a subsample of schools from #2. Follow these schools to collect data over time about changes in school practices, types of competitive foods offered, and rates of participation in the USDA reimbursable meal programs. For the same schools, collect information about district-level competitive food policies. Districts are more likely than schools to have codified policies because policy-making at the district level must be formally communicated to multiple schools.

This panel of schools will provide data for examining the impact of changes in state and district competitive food policies on the nutritional quality of foods available to students, and the impact on rates of participation in USDA meal programs.

References

Cline, T. and G. White. Position of the American Dietetic Association: Local support for nutrition integrity in schools. *Journal of the American Dietetic Association* 100(1):108-111, 2000.

Fox, M.K., M.K. Crepinsek, P. Connor, and M. Battaglia. *School Nutrition Dietary Assessment Study-II: Final Report*. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Special Nutrition Programs Report no. CN-01-SNDAIIFR, April 2001.

Gleason, P. and C. Suitor. *Children's Diets in the Mid-1990s: Dietary Intake and Its Relationship with School Meal Participation*, CN-01-CD1. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, 2001.

National Research Council. *Diet and Health: Implications for Reducing Chronic Disease Risk.* Washington, DC: National Academy Press, 1989a.

National Research Council. *Recommended Dietary Allowances*, 10th ed. Washington, DC: National Academy Press, 1989b.

U.S. Department of Agriculture. *Changing the Scene: Improving the School Nutrition Environment.* Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, August 2000.

U.S. Department of Agriculture. *Foods Sold in Competition with School Meal Programs: A Report to Congress.* Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, August 2001.

U.S. Department of Agriculture. Testimony of Eric M. Bost, Undersecretary, Food, Nutrition and Consumer Services, Before the Senate Committee on Agriculture, Nutrition and Forestry, April 3, 2003. http://www.fns.usda.gov/cga/Speeches/CT040303.html.

U.S. Department of Agriculture, Agricultural Research Service. *Documentation: Pyramid Servings Database for USDA Survey Food Codes, October 2000.* Online at http://www.barc.usda.gov/bhnrc/ cnrg.

U.S. Department of Agriculture, Food and Nutrition Service. *Eat Smart. Play Hard*[™], 3000. http://www.fas.usda.gov/eatsmartplayhard. Accessed June 2003.

U.S. Department of Agriculture, Food and Nutrition Service. State Competitive Foods Policies, September 2002. http://www.fns.usda.gov/cnd/lunch/competitive foods/state_policies_2002.htm

U.S. Department of Health and Human Services. *Healthy People 2010: Understanding and Improving Health*, 2d ed. Washington, DC: U.S. Government Printing Office, 2000.

U.S. Departments of Health and Human Services and Agriculture. *Nutrition and Your Health: Dietary Guidelines for Americans*, 4th ed. Washington, DC: U.S. Government Printing Office, 1995.

U.S. Departments of Health and Human Services and Agriculture. *Nutrition and Your Health: Dietary Guidelines for Americans*, 5th ed. Washington, DC: U.S. Government Printing Office, 2000.

U.S. Food and Drug Administration, Center for Food Safety and Applied Nutrition. *A Food Labeling Guide*, Sept. 1984 (rev. June 1999).

U.S. Food and Drug Administration. Kurtzweil, P. "Daily Values" encourage healthy diet. *FDA Consumer Magazine*, 27(4): May 1993. Online at <u>http://www.fda.gov/fdac/special/foodlabel/dvs.</u> <u>html</u>. Accessed July 2003.

Wechsler, H., N.D. Brener, S. Kuester, and C. Miller. Food service and foods and beverages available at school: Results from the School Health Policies and Programs Study 2000. *Journal of School Health* 71(7):313-324, 2001.

Appendix A Instrument Design and Data Quality Issues

This appendix provides a list of specific issues regarding the quality of the data obtained for this feasibility study. The main sources of information include in-person discussions with respondents during training, technical assistance, and debriefing visits; on-site observations made by study staff; and review and coding of the surveys and competitive foods data. Data quality and instrument design issues are discussed for each instrument, where relevant; and some recommendations for future testing are provided.

Principal Survey

Data Quality

There were discrepancies between the principals' reports of the types of competitive foods available and the number and location of vending machines, and what was observed at the school by Abt Associates staff. For example, in two of the schools principals reported the existence of a vending machine that neither the SFA director nor Abt staff could locate. There were also some discrepancies between principal reports of the types of vending machines in the school (e.g., milk, water, snacks, etc.) and the data collected from the vending machine inventories. In part, this was because there is often a mix of items within the same machine, and the response categories to capture this on the questionnaire were limited.

Item Nonresponse

There were many questions that principals either could not answer, could not answer without consulting other school personnel with more knowledge about competitive food policies, or did not answer accurately (based on discussions during debriefing and comparisons with SFA director responses).

Principals frequently chose the "don't know" option for questions about the existence of competitive food policies, pouring rights contracts, and nutrition initiatives. One issue is that school-specific policies are not necessarily documented, especially if they have been in effect for a long time. Another is that some foods or practices are "discouraged" rather than prohibited, or not considered "nutrition policies." For example, at one of the high schools the principal has asked teachers to sell non-food items for fundraisers. This was communicated during an all-staff meeting at the beginning of the school year, but is not a documented rule. In addition, the discipline code for the middle school prohibits gum chewing, but there is no policy prohibiting the sale of gum. Another issue is that questions about foods prohibited were sometimes answered based on actual practice, rather than knowledge about explicit policy. For example, coffee or tea were marked as "prohibited" because they are never offered at the middle school, but the principal was unaware of a policy restricting their sale.

Principals were not knowledgeable about the financial arrangements concerning food sales at their schools. The principal of one school with beverage sales covered by a pouring rights contract was unaware of the contract. Principals also found it difficult to answer questions about the profit earned from competitive food venues, and did not always consult others to obtain the requested information.

Principals in two schools would have needed to collect this information from teachers and athletic directors and, in the third school, from the city business manager. (They were not asked to provide profit information for food service-operated food venues.) The only other school personnel principals consulted when completing the survey were cafeteria staff.

Instrument Design Issues

Information about the types of vending machines might be better collected with a checklist of items in each machine and post-coding of these data to assign category codes. This procedure would eliminate the discrepancies between the principal report of types of vending machines and the information contained on inventory forms.

Because of the manner in which the vending machine section of the questionnaire was structured, it was not possible to relate certain environmental characteristics or policy issues to individual vending machines. For example, the person or organization responsible could not be mapped to the type, location, and hours of availability of each machine (although, for this study, the mapping was obtained from the on-site observation and discussions with school personnel). Future data collection efforts might consider collecting detailed data for each individual vending machine.

A question asking if parents or teachers are "permitted" to offer students food or beverages in the classroom was misinterpreted by at least two of the three principals. All answered in the negative, but during the debriefing principals mentioned that teachers have discretion to use candy, snacks, and pizza parties as rewards, and parents can bring in food for special occasions. This question should be presented differently on future surveys.

The current instrument asked about restrictions on access to food venues, but did not ask about compliance. For example, information obtained during on-site visits suggests that vending machine timers may not always work.

Recommendations

Much of the information collected on the Principal Survey seemed better suited to a site-visit protocol that could combine an in-depth interview with on-site observation. The self-report survey resulted in poor quality data that would be of questionable value had we not been able to supplement and revise those data from information obtained on-site. Future studies might consider one of the following two data collection options:

- On-site data collection interviews and direct observation by study staff while training on foods forms, or
- SFA staff to complete competitive food checklists (types, locations, hours, etc.), followed up with telephone interview.

SFA Director Survey

Data Quality

Not surprisingly, SFA directors seemed to be knowledgeable about school food policies for their district and, compared with principals, had less difficulty answering these questions. SFA directors

also seemed motivated to consult with others in order to answer all survey questions. For example, the SFA director for the one school covered by an exclusive beverage contract obtained a copy of the contract, and conferred with the city business manager prior to completing that section of the survey.

Item Nonresponse

There were no problems.

Instrument Design Issues

SFA directors had some of the same difficulty interpreting questions about "policies" as the school principals. The SFA director in one of the high schools noted that the school committee "requested" that certain beverages (juice and water) be made available for sale to students throughout the school day. He implemented the request, but was not sure this was considered "policy." One director felt the survey needed clearer delineation between questions that asked about district practices/policies and those about the specific school participating in the study.

SFA directors could not readily provide data on total profit earned from vending machine or snack bar sales.¹ One problem is that a school food service is considered a nonprofit entity—it is not supposed to earn a profit, so the terminology used in the survey was confusing to SFA directors. Information on revenue from the various food service-operated competitive food venues is easy for directors to provide. Data on expenses, however, are much more difficult to obtain and usually cannot be separately estimated for the different sources of food (i.e., vending *versus* a la carte expenses).

As with the Principal Survey, the structure of the questions on vending machines was limiting. For example, it was difficult for one SFA director to respond to the question about the types of schools (elementary, middle, or high) in which the food service has primary responsibility for vending machines, because this varies by type of machine (e.g., beverage *versus* ice cream).

The one respondent who completed the section on food service management company (FSMC) involvement felt it was "a waste of time"; she reported that the SFA makes all decisions about the foods allowed to be served to students, and the FSMC simply implements them.

There were some minor issues with the survey items about components of reimbursable meals also offered on an a la carte basis, and about a la carte pricing. In one of the schools, students can only buy a full second meal a la carte, not the individual components. There was no way to indicate this on the survey. Another school charges less for an a la carte entrée than a full reimbursable meal, but if combined with milk, the prices are the same.

Recommendations

Although the SFA directors reported that the survey was "easy", there were still a significant number of survey items that they found difficult to interpret and answer. The survey might be improved by including more opportunities for open-ended narrative responses that can accurately capture

¹ They were not asked about the amount of profit earned from a la carte sales—only whether or not a la carte service was profitable.

information about individual SFA experiences (recognizing that these data require more resources to process and analyze).

Cafeteria Survey

The Cafeteria Survey consisted of seven data collection forms, as described in Chapter 2. Selected forms are discussed below, by topic.

Data Quality

Data quality problems were encountered for three data items: weekly a la carte sales dollars (Daily Meal Counts Form), number of food servings (Every Day and Cafeteria Foods Forms), and manufacturer product codes (Every Day and Cafeteria Foods Forms). In addition, there was a small amount of confusion about the identification of menu items as "a la carte only", "reimbursable only", and "both".

Respondents at two of the schools did not have easy access to information on the weekly a la carte sales total (in dollars). In one high school, a daily "Sales Activity Report" included a total sales amount that combined a la carte sales (adults and students), vending sales, and sales of a few non-food items.² At the second high school, revenues from food-service operated vending machines are added to the daily a la carte totals, and the POS system reports the combined amount (this amount does not include adult a la carte sales).³ In contrast, the middle school manager was able to obtain a la carte sales (in dollars) directly from daily POS reports; this figure usually included sales to no more than one or two adults.

Incorrect data on the number of servings were provided on the Every Day and Cafeteria Foods Forms by two of the three schools. These two respondents recorded the number of servings *prepared*, rather than the total number **served or sold**. One of the schools that provided incorrect data had access to the correct data because both the number prepared and number leftover are recorded on daily production records. In the second school, information on servings was not always available for individual menu items and not always in terms of portions.⁴ In addition, one school provides free meals for cafeteria staff, which are not tracked in the POS system or on production records. These problems did not affect the analysis of a la carte-only foods, but would have been significant for an analysis of reimbursable meal items.

The cafeteria manager in the third school provided correct servings data because she was able to simply transcribe the "number of portions used" from her daily production records to the "number of portions served" column on the cafeteria foods forms. This was not possible for milk, however,

² The manager subtracted vending and non-food sales each day, and totaled the remainder over the five days of the target week in order to provide total weekly a la carte sales; this amount included a la carte sales to teachers. The only way to exclude teacher sales would be to record their purchases by hand at point of sale. Doing this for one day during the target week confirmed the manager's report that a la carte sales to teachers was minimal.

³ For this school, it would be necessary to subtract vending machine from a la carte sales during analysis.

⁴ For example, the total portions of "chicken" served were provided, but this count was for chicken patties, chicken nuggets, and chicken fingers combined. The form included the amounts of each type of chicken **prepared**, but only in terms of the number of cases or pounds. In order to estimate the number of servings of each type of chicken sold, it would be necessary to request additional information on the number of portions per case/pound for these items.

because the production records differentiate milks by size (14 oz and 8 oz), but not by flavor or fat content. The inventory method was used to track sales of milk at this school.

One complication for servings data—common to all schools—is the difficulty of excluding adult servings from total servings counts. This does not appear to be feasible in any of the three schools without putting special, manual procedures in place. All managers reported that adult sales are minimal. The additional burden of asking schools to track adult sales is probably not worth the small gain in accuracy of the data.

The third item subject to data quality problems was product codes. Upon review of the survey forms, it was clear that most of the codes recorded were distributor codes (item order numbers), rather than the manufacturer codes. The task of providing manufacturer product codes for pre-prepared food items was problematic in two of the three schools. One respondent noted that identifying and recording product codes "was the hardest part" of the Cafeteria Survey. She found that some boxes had more than one code, and others did not have a code at all. Despite special attention to the issue during training and technical assistance visits, the middle school manager was unable (or possibly unwilling) to provide manufacturer product codes. She also had difficulty identifying which foods were processed commodity items, one of the types of preprepared foods for which product codes were requested. In the third school, product codes were provided, but for more foods than necessary and sometimes in addition to another code (UPC or distributor code). The cafeteria manager in this school had access to some product codes on preprinted production record forms and in a purchasing handbook used throughout the district.

All respondents thought it would be easier to record distributor item numbers, rather than product codes; although one felt the SFA director should be responsible for this task. One of the SFA directors pointed out, however, that distributors change items without changing the numbers, and they make substitutions. She recommended requesting SFA directors' purchasing specifications, at least for districts where purchasing is centralized. It seems that it would be difficult to obtain any single, standard document from all SFA directors that includes all of the foods served at the school.

A fourth item on the Cafeteria forms that showed data quality problems was the field of the Every Day and Cafeteria Foods Form that required indication of whether each menu item was available as "a la carte only," "reimbursable only", or "both a la carte and reimbursable." Respondents had some difficulty with this form field. For example, in two schools where a la carte purchases of fruit and vegetables are very minimal, managers marked these items as "reimbursable only." This did not affect the analysis of a la carte-only foods, and could probably be resolved with additional emphasis in the instruction manual and during training.

Finally, missing recipe data was a problem for sandwiches in two of the three schools, although components of sandwiches were recorded on cafeteria foods forms in both cases.

Item Nonresponse

There were two main problems of missing data on the Cafeteria Forms: (1) the respondent at one school omitted condiments altogether from the Cafeteria Survey, and (2) one cafeteria manager did not use the Inventory of Every Day Prepackaged Foods correctly, resulting in missing data. Both problems affected the data received from the manager for the middle school (School #1).

The starting inventory was missing from the Inventory of Every Day Prepackaged Foods Form (she recorded bulk packaging information in this column). During the technical assistance visit, study staff learned that the manager had been out on Monday of the target week, and therefore no starting inventory was taken. During follow-up, the manager reported that most items were sold out by the end of the week, and recalled having only ten bags of chips and six cases of Snapple left on Friday afternoon. Deliveries were received for some items on Monday, and others on Thursday. There were also some items recorded on the form for which no starting inventory, deliveries, or ending inventory were reported.

Instrument Design Issues

The Self-Serve Bar Form created some problems for respondents. Respondents in two schools had difficulty knowing which forms to use to record condiments. In one high school with a condiment bar (School #2), the respondent used the Self-Serve Bar Form to record these items. Instead of providing an inventory of amounts put out and left over for just one day, as described in the instructions, the form was used like the inventory forms, and accounted for all of the bulk containers used on the bar over the course of the week. In fact, this information would have been more useful during analysis than a one-day inventory, especially because the school was unable to say how many students actually used the condiment bar. In addition, as noted above, one respondent failed to use the Self-Serve Bar Form, and omitted condiments altogether from the Cafeteria Survey.⁵

The Inventory of Every Day Prepackaged Foods form was **not** used for prepackaged condiments offered every day, as intended. Instead, for the one high school that offered individually packaged condiments, the cafeteria manager recorded daily the number of full cases and fractions of a case (e.g., 2¹/₄ cases) taken by students each day on the Every Day Cafeteria Foods Form. It seems as though it would have been less burdensome to use the inventory method and count the cases (and fractions of cases) at the beginning and end of the week only, assuming deliveries of these items did not occur frequently. Although estimating fractions of cases is not as precise as counting individual packages; this seemed like a reasonable compromise, and preferable to obtaining no information on condiment use at all.

The issue of shared stockrooms for multiple sources of competitive foods came up in the high school that sells a la carte-only items from the cafeteria line *versus* a separate snack bar (School #3). Many of the same foods and beverages sold on the line are also stocked in the food service-operated vending machines in the cafeteria. Although it was assumed it would be too difficult to use the inventory method in schools with shared stock rooms, the cafeteria manager was instructed to take the a la carte inventory before beginning to fill the vending machines and complete both forms according to the instructions. The numbers of items added to the vending machines during the target week were then subtracted from the calculated number sold for the a la carte items during analysis. Although a bit tedious to identify the specific items that appeared in both places, in the future the forms could include a column for respondents to indicate the items served in multiple venues.

⁵ In subsequent discussions, the manager realized that the easiest way to record condiments for her school (bulk, self-serve, vary daily) would be to use the Self-Serve Bar Form for a daily inventory of the number of cans used.

Recommendations

Overall, many of the problems that were encountered with the Cafeteria Survey instruments might be alleviated with more and better-timed on-site technical assistance. Some of the problems, however, were likely to be due to the fact that cafeteria managers were overwhelmed with the amount of data requested, and shortcuts were attempted and mistakes made.

A separate form for recording condiments should be considered to help avoid the missing condiment data issue. Sample completed forms for all of the various ways in which offer condiments could be added to the instruction manual.

To avoid errors in the reporting of a la carte sales dollars, it would be advisable to request copies of the POS report for each day of the target week or, as one SFA director suggested, obtain this information as part of the SFA Director Survey. Having POS records would allow a check on the calculations performed by cafeteria managers.

Inventory of Snack Bar Prepackaged Food and Beverage Items

Only one of the three schools (School #2) had a snack bar in the cafeteria, and it was instructed to complete the snack bar forms. Because the school did not offer cafeteria-prepared items or reimbursable meals at the snack bar, the Inventory of Prepackaged Snack Bar Items (inventory method) was to be used alone. Unfortunately, the form designed for this study was not used.

Data Quality/Item Nonresponse

The SFA director for School #2 created a spreadsheet of all foods and beverages sold in the snack bar in an effort to save staff the time it would take to write food descriptions. The SFA director planned to enter food names and descriptions, bulk packaging information, and package sizes, so that the snack bar staff would only have to enter the inventory information. Unfortunately, there were two problems with this approach. First, the SFA director's form omitted some items that were available for sale during the target week (based on validation data collected by Abt Associates study staff).

Snack bar staff did not add these items to the form, so they were not inventoried and sales data were not available. Second, package size and packaging information were missing for a large number of foods and had to be obtained via follow-up contacts with the SFA director before the nutrient content could be analyzed. The significance of these problems in a larger study could be great, and points to the need for respondents to provide information on standardized forms.

Instrument Design Issues

The Inventory of Prepackaged Snack Bar Items was unable to accommodate damaged items that were included in the starting inventory but could not be sold. This could be a problem if the damaged items are either returned to the supplier or discarded. Apparently, this does not occur frequently, either for the snack bar or other competitive food sources, so it is probably not a major issue.

Inventory of School Store Food and Beverage Items

Two of the three schools surveyed had school stores. One school store supervisor willingly cooperated with the survey and provided high quality data (School #3). The second school store supervisor was unwilling to use the instrument provided (School #2); thus, the test of the instrument is based on only one respondent.

Data Quality

The faculty supervisor for the store at School #3 provided complete descriptions of each item, including manufacturer and brand names, and completed the starting and ending inventories. Because this was a testing week at the school, she did not receive any deliveries, and none were recorded on the form. The store supervisor did note that sales (in dollars) were fairly typical, so students were choosing other items when stock ran out. For example, they would buy Juicy Fruit gum instead of Big Red. Because her cash register was not working during the target week, sales were computed manually (students counted items sold and multiplied by price).

Item Nonresponse

The second school store supervisor provided invoice data in lieu of completing the survey form. The invoice provided some information on the items sold in the school store, but did not include full food descriptions or package sizes.⁶ Furthermore, based on information collected by Abt Associates staff as part of the validation, it was clear that the weekly school store invoice was not representative of all food items available or sold to students during a typical school week. At a minimum, one would want to collect invoices for a full month of orders, although follow-up would still be required to obtain package sizes. When study staff attempted to obtain package size information from manufacturers or their websites for certain candies, beef jerky, and gum, they found that several sizes were available and could not know which size was actually sold in the school store.

Inventory of Vending Machine Items

Data Quality

The Inventory of Vending Machine Items seemed to work quite well for the two high schools with food service-operated vending machines. Both managers reported that the forms were easy to complete. The only problem encountered with these forms was that respondents did not always fill out the vending inventory form as intended (e.g., items in multiple slots were combined on the form), although it usually did not result in any loss of data. Additional training or technical assistance earlier in the data collection week could alleviate these minor issues. In addition, one school (School #3) operated a shared stock room for a la carte sales and vending machines, and the data had to be adjusted to account for the shared inventory.

⁶ An order/price book from the distributor who supplies food items for the school store was provided, but it did not contain the needed package size information.

Item Nonresponse

The main issue for the collection of information on vending machine items was the feasibility of obtaining data on the number of items sold (and, for some items, complete food descriptions) from machines stocked by an outside distributor. As noted in Chapter 2, this information was available for only one of the three schools, from only one of four vendors. Principals and two of the three SFA directors were unable to assist in obtaining this information because they were not directly involved with the vending machines in any of the three schools.

Appendix B Validation Study Findings

This appendix presents results from the validation component of the feasibility study. Abt Associates staff collected information about the competitive foods available to students on one day during the target week at the three sample schools. The objectives of this on-site data collection were to: (1) corroborate the identity of competitive foods described by school staff on study forms; (2) check that respondents provided a comprehensive list of items available; and (3) document the ability of an outside observer to collect detailed data without disrupting school operations. The data collection process and analysis results are described in the sections that follow.

Collection of Validation Data

As mentioned in Chapter 2, during the on-site technical assistance visits with cafeteria staff, study staff collected a complete description of all competitive foods and beverages available, including manufacturer and product name, package or serving size, and price. For vending machines, the machine location and slot numbers were also included. They used the same study forms as those used by cafeteria and other school staff to record the main study data. All items offered a la carte-only, in vending machines, snack bars, and school stores were recorded on the forms. To provide a larger sample and variety of competitive food items, validation data were also collected for the vending machines that operate after school only at School #3.

In addition to providing data to assess the completeness of the information respondents recorded on study forms, study staff documented the time spent and their experiences collecting these data at the schools. Findings are discussed here, although it should be noted that time estimates include the purchasing of a sample of items from each competitive food venue. Labels from these items were compared to nutrient information obtained from the nutrient database used for the feasibility study (discussed in Appendix C).

A la Carte-Only and Snack Bar Items

The time spent completing the validation data collection for a la carte-only foods and beverages ranged from 20 minutes at School #1 (middle school) to 50 minutes at School #2 (snack bar). The mean completion time over all three schools was approximately 33 minutes. As anticipated, the data collection time increased with the total number of items offered.

Two main problems were encountered when collecting information on a la carte-only items. First, it was difficult for observers to spot or gain access to all of the a la carte-only offerings at all three schools. For the two schools where a la carte service is operated off the main cafeteria line, items were located in multiple areas of the cafeteria, under counters, in freezers, or on multiple serving lines. In the school with the snack bar, many items were not accessible due to the small and busy serving area. Without the assistance of cafeteria staff in all three schools, it would not have been possible to identify all of the a la carte-only items available, or record complete descriptions and package sizes.

The second problem was that prices were not always posted. Cafeteria staff assistance was needed at all three schools to obtain pricing information for a la carte-only food items.

Vending Machine Items

Recording the foods or beverages in each vending machine required from 10 to 25 minutes, or an average of 16 minutes per machine. The main difficulty with the collection of these data was obtaining the package size for some items and certain vending machines. For machines dispensing snack items, package sizes were not always visible due to the positioning of the packages in the machine. This was a particular problem for gum and candy in small packages with very small printing on the labels.

For most beverage machines and one ice cream machine, it was not possible to "see" into the machine to read the labels. Staff had to rely on the posted label for each slot for food descriptions and sizes (if available). On one beverage machine, a package size was posted on the outside of the vending machine, **between** two buttons, making it unclear which of two beverages was the posted size. Without actually purchasing the item, there may be no definitive way to accurately record the package size for some items. It was also not clear that the posted descriptions and sizes were always accurate. On two occasions, when study staff tried to purchase a sample of items, they found the machines did not vend the posted item.¹

Prices for individual vended items were posted on the machines.

School Store Items

Obtaining information on the food and beverage items available in the two school stores took an average of 36.5 minutes per store (30 and 43 minutes). The assistance of store workers in one school was required in order to record the foods and their descriptions, because most of the food items were located behind the counter. In the other school, although no assistance was needed to record information about the food items, access to the store had to be scheduled with the store supervisor. Operating hours for this store were limited to mornings, before classes started; due to space limitations, the inventory could not have been conducted while students were making purchases.

Prices were posted for most, but not all, food and beverage items offered for sale at the school stores.

Validation Results

The validation data collected by Abt Associates staff were compared to the information provided by cafeteria and other school staff on the Cafeteria Survey and other competitive food forms. Three main types of discrepancies were noted for the competitive food and beverage items recorded in the three schools: (1) missing items, (2) missing or incomplete food descriptions, and (3) missing or incomplete package sizes.² Food descriptions and package sizes were considered missing/incomplete

¹ Other issues that observers encountered when purchasing items from the vending machines included locked machines (machines were on timers) and non-functioning machines.

² Vending machine location and item slot numbers were also validated for the 15 vending machines included in the analysis reported here. Location was missing for three machines (20 percent), and slot numbers were omitted for eight machines (53 percent).

if the information provided was insufficient to code accurately the food in the FIAS nutrient database. Price data were not requested from respondents, and thus are not included in the analysis; as discussed previously, however, the data were collected by study staff to test the feasibility of doing so by direct observation.

Table B.1 presents the results of the validation study analysis, citing the extent of missing or incomplete data for each source of competitive foods (a la carte, snack bar, vending, and school store), and for all sources combined.³ Problems with food descriptions are reported separately for the various components of the description, that is, manufacturer name and product name (food type, brand, flavor, percent fat or other indication of nutritional value). Likewise, package size issues are indicated for both the amount and the unit of measure. For example, for a 20 ounce bottle of a juice-based drink, "20" represents the amount and "ounce" is the unit of measure.

Overall, across all sources of competitive foods in the sample schools, only 5 percent of the competitive foods observed by study staff were missing completely from the school foods forms.⁴ Missing data did vary somewhat by source of food, however. The respondent for the one school store included in this analysis provided a complete listing of all food/beverage items available (0 percent missing). Cafeteria managers for the two schools with a la carte-only items missed about 12 percent of all items available. This degree of missing data might have had an impact on conclusions drawn about the number, type, or nutritional characteristics of a la carte-only items offered in the these schools.⁵

The main source of missing or incomplete data for items reported by respondents was the package size. Almost one third of the items reported were missing or had incomplete information about the unit of measure, and 14 percent had problems with the amount. These problems can be attributed almost exclusively to the issue with the collection of snack bar data at School #2. The school food service staff used a spreadsheet set up by the SFA director in lieu of the study form, and the spreadsheet was missing package sizes altogether for over half of all items. Package-size reporting was also a problem for a small share of foods (6 percent) in vending machines. For example, school personnel omitted package sizes for Lifesavers, gum, and a few snack items at School #3, and Abt staff package sizes differed for some snack and ice cream items from descriptions provided by a vending distributor for School #2.

³ The analysis excluded (1) vended beverages at School #1—the only source of data on these items was the validation data collected by study staff; and (2) school store food and beverage items for School #2—data provided by school staff were not for the target week when validation data were collected.

⁴ Only items that appeared on study staff forms but not respondents' forms were identified as missing. It was not possible to validate items that appeared on respondents' forms but **not** on the validation forms; these items may have been available at some point during the target week, but were not in stock on the day of the validation data collection.

⁵ All nutrient analyses for this report combined data from respondents and the validation study to provide a more complete picture of the competitive foods available in the three schools.

[™] ↓ Table B.1

Extent of Missing or Incomplete Data on Competitive Foods, Overall and by Source

	All Sources		A la Carte		Snack Bar		Vending		School Store	
	Number	%	Number	%	Number	%	Number	%	Number	%
Items missing from study forms	15	5.1	5	11.5	2	3.4	8	5.3	0	0
Food description missing/incomplete										
Manufacturer	65	22.1	12	26.7	44	75.9	9	5.9	0	0
Product name	46	15.6	8	17.8	14	24.1	15	9.9	9	23.1
Package size missing/incomplete										
Amount	42	14.3	0	0	32	55.2	9	5.9	1	2.6
Unit	37	32.4	0	0	29	50.0	7	4.6	1	2.6
Total number of items ^a	294		45		58		152		39	

a Excludes vended items for School #1 and school store items for School #2.

Sources: Validation data, Cafeteria Survey, and Inventories of Competitive Foods.

Food descriptions were missing, or to some extent incomplete, among all sources of competitive foods. Overall, 22 percent of items were missing manufacturer name, and about 16 percent had incomplete product names. One possible explanation for missing manufacturer names may be that, with the more recognized branded foods such as Doritos or Cheetos, cafeteria managers may have thought that brand was sufficient to describe the product.⁶ Some examples of product names with problems are shown below. Missing information provided by the validation data collected by Abt study staff is shown in parentheses.

- Veryfine Orange juice (100% juice)
- Veryfine iced tea with lemon (Chillers, vitamin-fortified)
- Chipburger (Good Humor premium cookie sandwich)
- Hostess cupcakes (chocolate)
- Snapple (fruit punch)
- Strawberry shortcake ice cream bar (Hood)
- Sunbelt chewy chocolate chip (granola bar)
- Extra spearmint gum (sugar-free)
- Chex Mix (Betty Crocker traditional)
- Honey buns (Hostess iced)
- Hershey chocolate milk (2% fat, calcium-fortified)

Food description problems were greatest for the snack bar, with three quarters of items missing manufacturer and one quarter with incomplete product names. The vended items recorded by respondents had the fewest food description problems (less than ten missing manufacturer or with incomplete product names). Without the validation data collected for the feasibility study, follow-up telephone calls to cafeteria staff and SFA directors would have been required to obtain the data needed for an accurate nutrient analysis.

⁶ Manufacturer name is not always essential to enter a food into a nutrient analysis system, but it is difficult to provide respondents with "rules" about when they are and are not useful.

Appendix C Nutrient Database Issues

The Food Intake and Analysis System (FIAS), version 3.99, was used to code and analyze the food energy and nutrient content of the competitive foods offered in the three sample schools. FIAS was chosen for the feasibility study for three main reasons: (1) the database includes the nutrients and dietary components of most interest to FNS; (2) the database includes commonly consumed snack and beverage items, some of which can be specified by brand name; and (3) food codes can be linked to the USDA Pyramid Servings Database, which provides Food Guide Pyramid food group servings for each item.¹ In addition, FIAS is being considered for use in a large, national study of the USDA meal programs (Integrated Study of School Meal Costs and Outcomes). Thus, FNS was interested in learning about any problems and issues with the use of FIAS that might have implications for this and other future studies.

This appendix reports the experiences of the project nutritionists who entered the competitive food items into FIAS for nutrient analysis. They identified three key issues:²

- 1. FIAS did not always specify or include all brands or flavors of the foods—the implications of this finding are not clear.
- 2. Some foods could not be coded in FIAS—nutrient values for these items had to be "added" to the database.
- 3. Most ice cream novelties could not be entered in FIAS without a package weight weight information could not always be obtained, even from manufacturers.

Following a brief description of the FIAS entry procedures, each of these issues and related analyses are discussed in detail. The second part of this appendix presents findings from a comparison of FIAS nutrient values and Nutrition Facts labels for a sample of the competitive foods offered in the three schools.

Entry Procedures

Two project nutritionists edited and entered the competitive foods data using FIAS 3.99. The entry process involved searching the database, selecting an appropriate food, and assigning an amount to the food. Search attempts began with the brand name, when available. FIAS typically assigns several brands or varieties of a food to a single food code. For example, when "Pepsi" is entered as the search term, FIAS displays the following:

¹ Other features of FIAS that would be important for a larger scale analysis of competitive foods include the ability to (1) add new recipes and foods to the database, and (2) use USDA food codes to group foods for analysis into broad and specific categories. Because none of the competitive foods offered in the sample schools was prepared from recipes, this feature was not used here. Given time constraints and the relatively small number of items, it was more efficient to "replace" *versus* "add" nutrients for new foods (discussed below), and manually assign each item to food categories.

² Additionally, incomplete food descriptions and missing package sizes were important problems requiring some followup with respondents and manufacturers, and heavy reliance on the validation data (see Appendix B).
92410310 Soft drink, cola-type (Includes Coca Cola; Coke; Pepsi; RC Cola)

If the exact brand could not be found, or if FIAS did not list any brands for a food, the more general food name was used as the search term and an appropriate "generic" item selected. Specific food codes were selected from the database for foods whose description included a nutritional claim (e.g., 2% lowfat, sugar-free, baked or reduced-fat). If no separate code was available, the food was flagged for review.³ Nutritionists used their judgment to decide when information about the particular flavor of an item, which was not specified in FIAS, was important enough to warrant flagging an entry for review.

When competitive food items could not be matched to the description of any items in the FIAS database, the nutritionists selected the closest item and flagged these foods for review. The closest item was selected to provide a FIAS record to match to the Pyramid servings database. Nutrient information was obtained for all "missing" foods from on-line sources, calls to manufacturers, and when necessary, package labels. Nutritionists then compared the per-package food energy and (available) nutrient values with the FIAS nutrient analysis for these items.⁴ If values differed by more than the standards defined for this study (shown in Table C.1), the foods were flagged for replacement. Nutrient replacement is discussed in later sections of this appendix. The remaining "missing" items not flagged for replacement were analyzed using the FIAS values.

Calories	± 75 kcal
Protein	± 2.5 g
Carbohydrate	± 15 g
Total fat	± 3.5 g
Saturated fat	± 2 g
Sodium	± 120 mg
Calcium	± 80 mg
Vitamin C	± 15 mg
Niacin	± 4 mg
Vitamin B6	± 0.4 mg
Iron	± 3 mg

Criteria Used for Replacement of Selected Nutrients: Amount per Package

Table C.1

³ Few of the items offered at the sample schools fell into this category, and even fewer were flagged as missing for this reason.

⁴ This comparison had to be done outside the FIAS system, using analysis reports. The FIAS entry system only allows the user to view nutrient values for 100 grams of the food rather than the amount entered. It also required conversion of label nutrients to per-package values.

Entry Issues

Brand Name and Flavor Specificity

Brand name, flavor, and missing food entry issues for competitive foods are summarized in Table C.2. The number and percentage of food items for which FIAS did not allow coding by brand or flavor are presented, by type of competitive food venue and overall (all schools and sources combined). Results show that FIAS entry was not brand-specific for approximately 17 percent of foods overall, including corn and tortilla chips, milk, doughnuts, and some ice cream products. The database was not flavor-specific for 35 percent of all items: juice-based drinks, sports drinks, fruit snacks, trail mix, and some candy and ice cream products. FIAS was neither brand- nor flavor-specific for potato chips, popcorn, and chewing gum. Candy and gum accounted for the large share of items offered at the school stores (65 percent) for which FIAS did not differentiate by flavor.

Table C.2 also shows the proportion of items for which the exact brand or flavor of the competitive food was missing from the FIAS database (when FIAS **does** specify brand names or flavors for a similar food). The exact brand name was not included in FIAS for about one fifth (21 percent) of all items. This was somewhat more likely for a la carte-only foods (including snack bar items) than vended or school store items. Not surprisingly, FIAS did not include the brand names of locally made ice cream, but beverage brands such as Snapple and Veryfine were missing. Other items whose brand was not found in FIAS included all fruit snacks, and some cookies and granola bars. It was relatively rare that nutritionists could not find a FIAS code for a particular flavor of a food (6 percent of items overall, although, as noted above, 35 percent of items were coded with a FIAS entry that was not flavor-specific.

Missing Foods

Some foods could not be coded in FIAS, based on their description (i.e., brand and food name). They included foods that were new to the market since the last database update and vitamin- or mineral-fortified versions of products in the database.⁵ Project nutritionists identified 59 such items, representing 17 percent of all competitive foods analyzed (see Table C.2). Three quarters of these items were found in vending machines in the sample schools. Of the 59 items "missing" from the database, 15 (25 percent) were found to have nutrient values close to a similar food in FIAS, that is, values fell within the parameters of the criteria shown in Table C.1.⁶ The remaining 44 items had one or more nutrient value that met the criteria for replacing nutrients, when compared to a similar item in FIAS.

⁵ Spring water was another category of items missing from the FIAS database, although it was not counted here. The only water entry in FIAS is "water as an ingredient," which contains sodium—most spring water does not. This affected the analysis of items meeting the FDA labeling criteria for sodium where spring water was counted as "low sodium", but not "sodium free".

⁶ The criteria were designed to identify only the clear outliers, that is, foods with one or more nutrient values that differed substantially from the closest food in FIAS.

Table C.2

FIAS Entry Issues, Overall and by Source

	All Sou	irces	A la Ca	arte	Snack	Bar	Vend	ing	School	Store
	Number	%	Number	%	Number	%	Number	%	Number	%
FIAS not brand-specific	58	16.8	7	15.6	8	12.5	28	17.8	15	18.8
FIAS not flavor-specific	121	35.0	13	28.9	17	26.6	39	24.8	52	65.0
Exact brand not in FIAS	74	21.4	13	28.9	22	34.4	27	17.2	12	15.0
Exact flavor not in FIAS	22	6.4	3	6.7	7	10.9	9	5.7	3	3.8
No match on description in FIAS ^a	59	17.0	4	8.9	7	10.9	43	27.4	5	6.3
Types of nonmatches:										
New food/Not in FIAS ^b	13	22.0	3	75.0	3	42.8	6	14.0	0	0.0
Not fortified in FIAS ^b	31	52.5	1	25.0	2	28.6	26	60.5	2	40.0
Did not meet criteria for nutrient replacement	15	25.4	0	0.0	2	28.6	11	25.6	3	60.0
Total	59	100.0	4	100.0	7	100.0	43	100.0	5	100.0
Total number of items ^c	346		45		64		157		80	

a Nutrient values for each of these items were obtained from online sources or, when necessary, package labels.

b Items in these categories met criteria for replacing nutrients (see Table C.1).

c Includes six types of salad dressing offered with salads at the snack bar in School #2.

More than half of the foods that could not be coded in FIAS were highly fortified with vitamins A, C, and/or B vitamins, or with calcium or iron, and met the criteria for nutrient replacement (Table C.2). Twenty-two percent were newer food items, although many of these foods were also fortified (e.g., Honey Nut Cheerios Milk and Cereal Bar). Fortified items, especially beverages (e.g., Pink Lemonade Chiller), accounted for about 60 percent of vending machine items missing from FIAS.

Ice Cream Amounts

Ice cream novelties (e.g., ice cream bars, ice cream sandwiches, ice cream cones) posed a unique problem when nutritionists attempted to enter their package sizes in FIAS. The unit of measure for the package size printed on the wrappers of these items was "fluid ounces."⁷ For example, the label for a Twix Ice Cream Bar, described as "ice cream and caramel loaded with cookie crunch bits", indicated a package size of "2.5 fl. oz.". The main problem is that FIAS does not allow items that are not all fluid at room temperature to be entered in volume units.⁸ The options were either to obtain a weight (ounces, grams) for the item (or a similar item) or enter a food-specific unit, for example, "1 bar" or "1 cone". Neither alternative was without problems.

Study staff reviewed labels collected as part of the validation study and noted that only a few ice cream products included the weight (in grams) of the item on the Nutrition Facts portion of the label. They attempted to obtain the weights of the remaining ice cream items on-line and from manufacturers; weight information was not always available, however. Some of the food-specific units in FIAS included a conversion to volume units, for example, "1 bar (3 fl. oz.)". In these cases, the food-specific unit was an acceptable option for entering the package size. Nutritionists used these conversion factors and the available volume-to-weight information to determine amounts to enter for the remaining ice cream items.

Comparison of FIAS Nutrients with Replacement Values

Additional analyses were conducted to assess the relative impact of replacing nutrient values for foods missing from the database on the average nutrient composition of competitive foods. Nutrient values were generally replaced only for those nutrients that did not meet the replacement criteria. Exceptions included calories, protein, carbohydrate, total fat, and saturated fat, where adjustments were also made to the percentages of calories from fat and saturated fat. Because the main reason for replacing nutrients was fortification, it seemed reasonable to replace only those nutrients that differed substantially (i.e., according to the criteria used for this study).⁹

⁷ Respondents (including one vending distributor) tended to record the package sizes for ice cream products as if they were weighed ounces, e.g., 3 oz. instead of 3 fl. oz. Entry of these amounts as weight units would have introduced error.

⁸ FIAS does allow entry of volume units for ice cream and sundae cups, sherbet, Italian ice, and frozen juice bars/tubes these are typically fluid at room temperature.

⁹ Replacement of nutrients was the preferred method for addressing the missing food issue for the feasibility study for two reasons: (1) time constraints did not allow for the research or imputation that would be necessary to avoid missing nutrient values if the foods were simply added to the database, using easily accessible nutrient information (e.g., Nutrition Facts label); and (2) it was important to maintain the original food code for the most similar item in FIAS, to allow for linking with the Pyramid Servings database.

Table C.3 shows the number of items with replaced nutrient values in each of the major food categories, the number of replacement values higher or lower than FIAS values, and the mean nutrient content (as offered) for the items in each category before and after replacing nutrients. As a group, beverages were the items for which nutrients were replaced most often. Twenty-eight percent of all beverage items had replaced nutrients, and beverages comprised two thirds of all items with replaced values. The remaining third of items with replaced nutrients fell within the baked goods/dessert, bread or grain products, and frozen desserts categories. None of candy, snacks, or salads/salad dressing met the criteria for nutrient replacement.

Not surprisingly, the preponderance of replacement values was **higher** than FIAS values for the most similar foods in the database (Table C.3). Fortification of beverages with vitamins B6 and C, niacin, and calcium is the major contributor, but other breakfast and snack items fortified with calcium and iron have replacement values higher than FIAS nutrients. Also notable are the frozen desserts, where calorie, fat, and saturated fat replacement values were higher than FIAS values. This could be reflective of the difficulty encountered with the entry of package sizes for ice cream, or it might be that FIAS values for ice cream do not reflect the current mix available in the marketplace.

The dietary components most likely to be replaced with values **lower** than FIAS values were calories and vitamin C for beverages, and sodium for baked goods and frozen desserts. The higher vitamin C values in FIAS for beverages are for juice-based drinks fortified to a higher level of vitamin C than the fortified products observed in this study (Veryfine Chillers and Fruitworks).

The average nutrient composition of competitive foods as offered was compared using original (FIAS) and revised (replacement) nutrient values for the 44 foods "missing" from the nutrient database. Results are presented in the last two columns of Table C.3 for the four food categories containing items with replaced nutrients. The original and revised nutrient values differ surprisingly little, even among the beverage category, which experienced the most replacements. The largest and most consistent changes seen are for calcium. Revised mean values for calcium increased for baked goods/desserts, bread or grain products, and beverages, but by only 19 to 35 mg, or 2 to 4 percent of the calcium RDI.

Summary of FIAS Entry Issues

It was beyond the capacity of this study to predict precisely the implications of limited brand- and flavor-specificity and foods missing from the nutrient database for a larger study. Results of analyses presented here, however, do not indicate that the impacts would be great. Presumably, the reason the FIAS database does not differentiate by brand or flavor for some foods is because nutrient composition information available at the time of the last database documentation. The brand names observed at the three sample schools were found in the database for the majority of foods for which FIAS specifies brands. Twelve percent of foods could not be coded with FIAS, but when the nutrient values for these foods were "added" to the database (or replaced in the data set), the mean nutrient composition of the four affected food categories did not change substantially. Because it has been a number of years since the FIAS database was last updated, the proportion of missing foods should be reduced with a new release.

Table C.3

Comparison of FIAS Nutrients with Replacement Values

					Mean Nutrie	ent Value for
	Total	Number o	f Items with Repl	aced Nutrients	Cate	gory ^a
Food Category/	Number		Replacement	Replacement		
Nutrient	of items	Total	Value Higher	Value Lower	Original	Revised
Baked goods/Dessert	34	4	2	<u> </u>		o 4 - 4
Calories		2	2	0	342.2	347.4
l otal fat (g)		2	2	0	14.5	15.1
Saturated fat (g)		2	2	0	3.5	3.6
Vitamin B6 (mg)		1	1	0	0.1	0.1
Calcium (mg)		2	2	0	37.4	56.7
Niacin (mg)		1	1	0	2.0	2.2
Sodium (mg)		3	0	3	275.7	261.6
Beverages	104	29				
Calories		6	2	4	206.2	205.9
Carbohydrate (g)		3	3	0	46.8	47.1
Vitamin B6 (mg)		16	15	1	0.1	0.2
Vitamin C (mg)		19	11	8	43.4	44.7
Calcium (mg)		22	22	0	93.5	128.6
Niacin (mg)		16	15	1	0.4	1.1
Sodium (mg)		5	5	0	88.1	94.6
Bread or grain						
products	26	4				
Calories		3	2	1	172.7	172.9
Protein (g)		2	2	0	3.5	3.9
Total fat (g)		1	0	1	6.0	5.9
Saturated fat (g)		1	0	1	2.1	2.0
Iron (mg)		2	2	0	1.7	2.0
Calcium (mg)		3	3	0	29.8	52.7
Candy	72	0				
Frozen desserts	33	7				
Calories		5	4	1	184.8	188.3
Total fat (g)		5	4	1	9.2	9.5
Saturated fat (g)		5	4	1	5.5	5.5
Sodium (mg)		4	0	4	83.0	57.5
Salads/Salad dressing	8	0				
Snacks	69	0				
Total	346	44	104	27		

a Mean values include all competitive foods for which data were available, both during and after school.

The most serious problem encountered by this feasibility study was the inability to enter observed package sizes for ice cream in FIAS. The effort involved in obtaining the information needed to enter an accurate amount for ice cream novelties could be substantial in a larger study. It may be worthwhile to explore the potential for software and database modifications to allow ice cream to be entered according the label amount.

Comparison of FIAS Data to Label Nutrition Information

While collecting data for the validation component of the feasibility study, Abt Associates staff purchased a random one-third sample of all competitive food (non-beverage) items from each of the food venues. Information from the Nutrition Facts portion of the labels from these items was used for comparison with the nutrient information obtained from the nutrient database. The purpose of the comparison was to determine the extent of variation between the two sources of nutrition information for competitive foods offered in the sample schools. This comparison is relevant because school-children who are concerned about healthful choices rely on nutrient label information.¹⁰

The planned analyses were similar to the comparisons performed to identify foods that could not be accurately coded in the FIAS database (discussed above). For this reason, any items with replaced nutrients were excluded from the analysis (n=6). The resulting sample for the label comparison was 41 foods. Nutrient values were compared for the full package—the amount entered in FIAS for each food. For items with more than one serving per package, label nutrient values were multiplied by the number of "servings per container". Label values expressed as a percent of Daily Value (DV) were converted to absolute amounts of the nutrient. (This introduces some error due to the rounding of DV percentages.) The nutrients and dietary components included in the comparison are those for which label information was available for the majority of sample foods.¹¹

Two types of analysis were conducted. First, for each of the 12 nutrients, the label value was expressed as a percentage of the FIAS value and averaged over all foods for which both values were available. Second, because the mean ratio of label value to FIAS value does not adequately represent the distribution of differences between the two data sources, the relative numbers of foods with FIAS values that were lower, higher, or the same as the label nutrients were identified.

The mean ratio of label to FIAS nutrients ranged from 0.64 for calcium to 3.17 for vitamin A. This ratio was close (within 10 percent) to 1.0 for food energy (calories), carbohydrate, total fat, saturated fat, cholesterol, and dietary fiber (data not shown). The higher FIAS values for calcium can be explained by the fact that 13 of 32 labels (41 percent) reported a calcium value of 0 percent of DV (RDI), whereas FIAS had small amounts of calcium for these items. It seems reasonable to assume that these foods do contain calcium, albeit in small amounts. This finding for calcium is important because it points to the limitations of label nutrition information. On the other hand, none of the

¹⁰ The nutrient information from food labels was assumed to be more accurate than the FIAS values, although manufacturers can label their products using data from a nutrient database in place of actual laboratory analysis (USFDA, 1999). Furthermore, labeling regulations allow rounding of values and certain nutrients may be expressed as a percentage of DV, which detracts from their precision.

¹¹ All 41 labels included data for calories, protein, carbohydrate, total fat, and sodium. Data for saturated fat, cholesterol, dietary fiber, vitamins A and C, calcium, and iron were included on 32 to 35 labels (78 to 85 percent). The B vitamins thiamin, niacin, riboflavin, and vitamin B6 were excluded from the comparison because they were found on only 2 to 6 labels, or fewer than 15 percent.

foods for which FIAS shows calcium values are important sources of this nutrient in children's diets. Zero values on Nutrition Facts labels were also common for vitamin A (when FIAS had small values), but this was offset by especially high vitamin A values on the labels (relative to FIAS) of French dressing and fruit snacks (2,000 and 1,875 International Units, respectively).¹²

Other nutrients where the mean ratio of label values to FIAS values differed from 1.0 include vitamin C (1.79), protein (1.45), and sodium (1.29). The variation in values for these nutrients is attributable in large part to a few food items where the FIAS values are substantially lower than indicated on the label. These foods include vitamin C-fortified fruit snacks, fruit roll-ups, and candy; French dressing; and some ice cream novelties.¹³ For example, according to the Nutrition Facts label for fruit roll-ups, one package has 15 mg vitamin C, 1.5 g protein, and 55 mg sodium *versus* FIAS values of 0.9 mg vitamin C, 0.1 g protein, and 9 mg sodium. Note that none of the foods where differences between FIAS and label nutrients are great is typically considered an important dietary source of those nutrients.

Figure C.1 shows the share of foods with FIAS values that are lower, higher, or the same (within 1 percent) as the label nutrients for that food. Nutrient values were converted to percent of DVs for this comparison. The analysis finds close agreement between FIAS and label values for food energy, protein, carbohydrate, and cholesterol content, for two thirds or more of the sample foods. Total fat, vitamin A, and vitamin C values agree for a little over half of the foods (despite the large mean differences for vitamins A and C discussed above). For all other dietary components examined, a large share of sample foods have FIAS values that fall above or below the label values. Although the mean difference for this particular sample is relatively small (10 percent), saturated fat values are particularly likely to vary between the two data sources, with a greater tendency for FIAS values to be higher than labeled amounts.

When FIAS values differ from the label nutrients, more food items have higher values for food energy, protein, total fat, saturated fat, and calcium; and lower values for cholesterol, fiber, and vitamins A and C.

¹² Zero values on Nutrition Facts labels were also seen for vitamin C, although when FIAS had values for these foods, they tended to be very small (less than 0.5 mg vitamin C).

¹³ If beverages were included in this analysis, fortified juice-based drinks would also likely be a source of differences in vitamin C values between the label and FIAS.



Comparison of FIAS to Label Nutrient Information (as Percent of Daily Value)

C-10

Figure C.1

Table D.1

Reimbursable Meals Offered During Target Week

Meal Compon	ents	School #1	School #2	School #3
Milk		Skim, 2%, choc 1% (8 oz)	Skim, whole, 1%, choc 1% (10 oz)	Skim, 1%, choc 1% (8 oz) Strawb 1%, choc 1% (14 oz)
Entrée (availab every day)	le		Salads: tuna, ham, garden turkey, garden Sandwiches: cheeseburger; hamburger; veggie burger; hot dog; kielbasa dog; chicken patty; cheese; pb/jelly; pb/fluff Sub sandwiches: tuna; turkey; turkey salad; ham/cheese; Italian; bologna/cheese	Pizza: pepperoni or cheese Sandwiches: chicken filet; cheeseburger; pork rib Club sandwich: steak/ cheese Yogurt/bagel w/ cream cheese
Daily entrée	Mon	Turkey taco boat; chicken nuggets; chicken fingers; chicken cacciatore	Macaroni/cheese; chicken nuggets; sausage/egg sandwich	Turkey taco boat; chicken nuggets; chicken fingers; chicken cacciatore
	Tue	Chicken Caesar salad; minestrone; mozzarella sticks; chicken legs; nachos	Lasagna; egg salad sandwich	Chicken Caesar salad; minestrone; mozzarella sticks; chicken legs; nachos
	Wed	American chop suey; chicken strips; chicken fingers; mozzarella sticks/salsa	Beef taco/refried beans; tuna salad	American chop suey; chicken strips; chicken fingers; mozzarella sticks/salsa
	Thu	Breaded chicken; chicken nuggets (honey BBQ); macaroni/cheese	BBQ chicken; turkey/cheese sandwich	Breaded chicken; chicken nuggets (honey BBQ); macaroni/cheese
	Fri	Pizza; chicken nuggets; buffalo chicken; macaroni/cheese	Sweet/sour pork; Italian sub; cheese sandwich	Pizza; chicken nuggets; buffalo chicken; macaroni/cheese
Grains/Breads		Bread (w/ salads) ^a Dinner roll (2 days only) Rice (1 day)	Dinner roll, white bread (every day) Ziti (1 day) Goldfish crackers; biscuits; tortilla chips (1 day)	Roll (1 day) Rice (1 day)
Vegetable side (every day)		Tossed salad ^a Lettuce/tomato (w/ sandwich) ^a	French fries; carrot/celery sticks	French fries; tossed salad
Daily vegetable side	Mon	Lettuce/tomato; corn	Corn on the cob	Lettuce/tomato; corn
	Tue Wed	Stay Crisp fries Green beans; wax beans	Green beans; cole slaw Salsa/lettuce cup	Stay Crisp fries Green beans; wax beans

Table D.1

Reimbursable Meals Offered During Target Week

Meal Compon	ents	School #1	School #2	School #3
	Thu	Mashed potatoes; cole slaw; Stay Crisp fries; green beans; wax beans	Mashed potatoes; broccoli	Mashed potatoes; cole slaw; Stay Crisp fries; green beans; wax beans
	Fri	Wax beans; green beans	Corn on the cob	Wax beans; green beans
Fruit side		Fruit choice (fresh, canned, or juice) ^a	Apple (every day) Canned fruit (every day) Apple juice (1 day)	Peach cup Pears (canned); apple Fruit cup (canned); apple Grapefruit juice; apple Fruit cup (canned); apple
Dessert/Misc.		None	Italian ice (1 day) Potato chips & cheese curls (every day)	Peach crisp (1 day) Peach cup (frozen) (1 day)

a Information from printed menu but missing from Cafeteria Survey.

b Information from production records but missing from Cafeteria Survey.

carte
A la
~
Ħ,
ool #1,

ar	t per	ded jars sp)	5	0.8	8. 8.	Ņ	0.000		0.0	5.4		8.0	5.5 5.6
t & sug	i amoun item	- Ado / Sug		ω		ч	φ.φ	₽ '	0)	()		0	U) U)
Fai	Mean	Discre tionary Fat (g)	15.9	12.4	25.0 13.3	17.5	1 1 1	11	I	2.9	15.5 15.2 10.6 10.6 10.8	5.7	8.6 9.8
group		Meat ³	I	I	1 1	I	1 1 1	11	I	I	11111	I	I I
yramid		Dairy	I	I	1 1	I	1 1 1	11	I	I		I	1.1
js per p		Fruit	I	I	0.1	I	0.4 - 0.4	0.6	0.9	I	1111	I	0.1
¢ servinç		Veget- able	I	I	11	I	111	11	I	I	+	1.1	0.2 0.2
Mean #		Grain	1.7	2.0	1.4 1.8	2.0	1 1 1	11	I	0.8	2 	I	1.0 12
		lron (mg)	2.2	3.3	2.1 1.3	1.7	1.3 0.3 1.3	0.5	0.2	0.7	0.8 0.8 0.8 0.8	0.3	1. 1. 1. 2.
trients		um (mg)	19.5	72.4	29.8 59.8	37.4	49.6 - 49.6	18.7 10.0	20.0	3.0	11.9 28.8 76.4 20.6	39.9	34.6 43.7
Aicronut	197	C C (mg)	I	0.1	1.6	0.2	34.7 - 34.7	28.5 -	65.0	5.1	15.4 0.1 1 1 1	I	10.3 9.6
4	T 171	A (RE)	I	5.0	26.4 72.2	14.5	6.2 - 6.2	95.4 -	0.002	11.9	16.9 9.9 12.1	I	32.6 16.0
	nt of from	satura- ted fat	14.0	6.9	15.4 6.9	7.7	1 1 1	0.1	1	4.7	18.4 10.7 9.0 10.7	5.0	6.6 7.8
	Percer calories	Total 6 fat	42.3	34.7	58.9 36.3	49.0	1 1 1	0.6	I	23.8	58.1 55.9 47.1 55.9	34.5	30.2 35.4
		Dietary fiber (g)	2.0	0.8	1.7 1.9	1.3	0.6 - 0.6	0.2	I	0.1	0.0.0 0.0.0 0.2 0.4 0.4	1.5	1.1
		Sodi- um (mg)	245.6	421.7	364.9 349.6	464.4	136.4 240.0 136.4	16.4 15.0	320.0	161.5	294.7 520.9 262.0 262.0	292.5	270.9 275.8
trients	E	Chole- sterol (mg)	1	16.9	49.3 52.5	31.5	1 1 1	11	I	I		I	8.5 5.2
lacronu	ıt per iteı	Protein (g)	4.2	3.4	4.3 5.0	4.3	1 1 1	0.7 -	I	1.1	3.5 3.5 2.7	1.6	2.3 2.7
4	n amour	Carb- bhydr- F ate (g)	52.1	59.8	40.8 56.1	42.3	73.8 37.8 73.8	66.6 _	54.0	26.9	26.3 26.7 31.2 31.2 19.1	22.8	41.2 44.6
	Mea	satura- ed fat (g)	5.8	2.8	6.9 2.9	3.1	1 1 1	1 1	I	0.7	2.5 2.5 2.5 2.5	0.8	2.2 2.5
		Total S fat t (g)	17.6	14.4	26.4 15.2	19.5	1 1 1	0.2	I	3.7	17.2 17.1 13.0 12.2	5.8	9.7 11.1
		Calori- es	375	373	403 376	358	291 150 291	256 -	220	141	266 275 249 249 196	151	257 284
	FMNV & FDA	label criteri- a ²		0			0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0	9 9 9	0	0 0 0 0	0	
			0.75	0.85	0.75 0.85	0.75	1.25 1.25 1.25	1.25 1.00	1.25	0.75	0.75 0.75 0.75 0.75 0.75	0.75	0.91 0.89
	-dmb-	er1	120	36	36 36	36	na na 216	72 na	na	na	62 122 122 00	62	18 1102
	ackage N	size	2.8 oz	3.5 oz	3.0 oz 3.5 oz	3.0 oz	0.0 oz 0.0 oz 0.0 oz	0.0 oz 6.9 oz	6.0 oz	1.3 oz	1.8 oz 1.8 oz 1.8 oz 1.3 oz	1.1 oz	1 1
	<u>č</u>		Baked goods/dessert Grandma's Choc chip cookies	cupcakes	Hostess Choc frosted doughnuts Hostess Honey buns	doughnuts	Beverages Fruitopia Kiwi Berry 2 Gatorade Fruit punch 2 Snapple Fruit Punch 2	Spring water	Sunny Delight 1	Bread/grain products Rice Krispie Treat	Snacks BBQ potato chips Cheetos Curls Doritos Cool Ranch Funyuns	potato chips	Average of all items Unweighted

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol, ③ = sodium-free; **0** = low sodium;
Also includes legumes, soybean products, nuts, and seeds.
Succe: Cafeteria Survey and and Inventories of Competitive Foods.

School #1, Vending Machine #1⁴

								2	lacronu	trients					Σ	icronutr	ients	_	llean # s	ervings I	oer pyra	mid gro	dn	Fat & sı	ıgar
	Package	Numb-		FMNV & FDA			Me	an amour	ıt per iteı	۶			Percent calories f	io d									2	ean amo item	unt per
	size	e -	a	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- E um (mg)	Dietary fiber (g)	Total S _i fat	atura- ted fat	HE BE		mg)	ug)	rain Ve a	Die Fr		M	eat ³ Dis	icre- hary S at S	Added bugars (tsp)
Beverages																									
Minute Maid Apple juice (100%) Minute Maid	12.0 oz	па	0.65	2 6 4	175	0.4	I	43.5	0.2	I	11.2	0.4	2.1	0.4	I	3.4	56.0	1.4	I		0	1	1	I	I
CranApple Raspberry drink Minute Maid Eruitonia	12.0 oz	na	0.65	2 6 4	164	0.3	I	42.6	9.0	I	9.5	0.2	1.7	0.2	2.3	13.6	7.8	0.5	I			I	1	<u>0.1</u>	6.3
Fruit Integration Minute Maid Fruitopia	12.0 oz	па	0.65	2 6 4	175	I	I	44.3	I	I	81.8	0.4	I	I	3.7	20.8	59.8	0.8	I		0.2	1	1	I	9.6
Strawberry Passion [2]	12.0 oz	na	0.65	2 6 4	175	I	I	44.3	I	I	81.8	0.4	I	l	3.7 2	20.8	9.8	0.8	I			I	1	I	9.6
iviliate Maid Olarige juice (100%) [2]	12.0 oz	na	0.65	2 84	157	0.5	I	36.8	2.2	I	7.5	0.8	3.0	0.4 6	37.2 12	28.5 2	6.9	1.6	I		0.2	I	1	I	I
Average of all items Unweighted	I	5	0.65		169	0.2	I	42.3	0.6	I	38.4	0.4	1.3	0.2	5.4	37.4 2	26.6	1.0	I		0.1	I	1	1	5.1
Weighted	I	na	na		I	I	I	I	I	I	I	ı	I	I	I	I	I	I	I			1	1		I

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Data are not available. The number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol-free; **0** = low cholesterol-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol-fr

School #2, Snack bar⁵

								-	Macronu	trients					4	licronut	rients		Mean # s	ervings	per pyra	mid gro	dn	Fat & su	ıgar
	Package	-dmb-		FMNV & FDA			Me	an amou	nt per ite	ε			Percer calories	nt of from	111	101							Σ	ean amot item	unt per
	size	er1	LICE	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total ⁶ fat	Satura- ted fat	A (BE)	C C M	um (mg)	(mg)	àrain <	ible F	ruit Di	airy Me	eat ³ Dis tio	scre- nary S -at S g)	dded ugars (tsp)
Baked goods/dessert Grandma's Choc chip cookies	2.8 oz	76	0.75		375	17.6	5.8	52.1	4.2		245.6	2.0	42.3	14.0	I	I	19.5	2.2	1.7				-	5.9	7.1
Grandma's Fudge chip cookies Li'l Deb Coffee cakes	2.8 oz 2.1 oz	49 18	0.75		390 203	23.9 6.4	6.1 1.7	42.7 32.6	4.1 3.8	26.8 59.7	229.2 267.8	2.1 0.8	55.1 28.4	14.1 7.4	4.6 27.6	11	17.3 69.4	1.8	1.4 4.1	1 1	1 1			2.5 5.0	5.8 3.1
Eri Deb Fuage	2.5 oz	24	09.0	0	334	14.6	2.6	49.8	3.3	I	428.1	2.3	39.3	7.0	I	I	18.4	2.8	1.6	I	I		÷	3.0	7.1
Li'l Deb Star Crunch Li'l Deb Star Crunch Mini Oreo cookies Nutter Butter cookies	3.0 oz 2.2 oz 1.5 oz 1.8 oz	43 29 166 0	0.60 0.60 0.50 0.50	0 0 0 0	413 266 201 237	18.9 12.9 8.8 11.7	3.9 8.9 2.2	58.0 41.2 29.9 29.2	4.1 2.0 8.8	7.6	102.4 134.6 256.9 205.9	1.6 0.9 0.9	41.1 43.7 39.3 44.5	8.4 30.0 7.0 8.5	8.0 58.9 1 4.5	1 4 1 1	61.6 12.2 11.1 17.4	2.6 1.7 1.3	1.9 0.3 0.9				+ <u>6</u> + <u>6</u>	6.9 2.7 7.8 0.8	7.9 7.5 4.2 6.0
Pep.Farm Inilano cookies Whoopie Pie	1.5 oz 3.0 oz	38 38	0.75 0.75	8 8	165 358	4.8 14.4	1.3 4.0	29.2 57.6	1.9 3.4	10.1	103.6 142.9	0.4 1.7	26.0 36.1	7.1 10.1	8.2 0.9	ı F	36.0 39.1	0.7 2.2	0.8 0.7	1 1	1 1		· ·	3.6 3.6	4.7 10.7
Beverages Hershey's Choc milk (2%)	14.0 oz	<u>6</u>	1.00		312	8.7	5.4	45.4	14.0	29.7	263.0	2.2	25.2	15.6	249.0	4.0 8	175.0	1.1	I	I		Ľ.		7.2	4.8
Hersney's Strawberry milk (2%) Spring water	14.0 oz 16.0 oz	n *	1.00 1.00	X 0 0 0 0 0	336	7.9 -	5.3	54.5	13.2	54.5	210.1 14.2	1 1	21.1	14.1	124.2 _	3.9 3.9	375.0 9.5	0.4	0.9	1 1		9. 		2.7	6.0
V8 Splash berry blend	11.5 oz	213	0.75	2 6 4	385	0.2	I	97.9	0.8	I	69.9	÷.	0.5	1	727.2	87.4	24.5	0.5	I	0.5	0.3			I	10.0
blend	11.5 oz	*	0.75	264	385	0.2	I	97.9	0.8	I	69.9	1.1	0.5	1	727.2	87.4	24.5	0.5	I	0.5	0.3			I	10.0
Vo Optasti strawberry kiwi	11.5 oz	*	0.75	2 6 4	385	0.2	I	97.9	0.8	I	69.9	1.1	0.5	1	727.2	87.4	24.5	0.5	I	0.5	0.3			I	10.0
blend	11.5 oz	*	0.75	2 6 4	385	0.2	ı	97.9	0.8	ı	6.69	. .	0.5	1	727.2	87.4	24.5	0.5	I	0.5	0.3	Ì		I	10.0

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
K indicates fool of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: © = calorie-free; • = low calorie; © = cholesterol-free; • = low cholesterol; © = sodium-free; • = low sodium; as last incluses legumes, sovbean products, nuts, and seeds.
Total number sold was sometimes reported for all flavors of the same product. 'Number' appears in table for first flavor, followed by remaining flavors denoted by *.

с Ш

School #2, Snack bar⁵

								N	lacronut	rients					Mi	cronutrie	ents	Me	an # serv	ings per	pyramic	l group	Fat	& sugar
	Package	-dmb/		FDA FDA			Me	in amoun	t per iten	٦		0	Percent calories fr	ot									Mean	amount per tem
	size	er1	LICE	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- F ate (g)	rotein 6 (g)	Chole-	Sodi- D um 1 (mg)	ietary fiber (g)	Total Sa fat t	tura- ed (I	A A SE) (r			g) Gra	n Vegei able	Fruit	Dairy	Meat ³	Discre- tionary Fat (g)	Added Sugars (tsp)
Bread/arain products																								
Chewey choc chip granola bar	1.3 oz	31	0.50	8	161	6.2	4.5	23.6	3.5	1	98.5		34.9 2	5.2	0.4	0.4 2	1.3	÷ 	ı	I	I	0.1	5.2	1.6
granola bar	1.0 oz	80	0.50	88	129	5.0	3.6	18.9	2.8	I	78.8	0.9	34.9 2	5.2	0.3	0.3 17	0.0	0 0	۱ ۳	I	I	0.1	4.2	1.3
granola bar Granola bar	1.2 oz	æ	0.50	2 8 ¢	122	2.4	I	22.7	3.3	1	94.6		17.9	1	0.3	0.3 20	0.4 1	 	і О	I	I	0.1	5.0	1.6
sandwich crackers	1.5 oz	na	0.50	0	207	9.7	2.5	26.0	4.0	2.6 4	75.9	0.2	12.2	0.9	3.2	0.7 91	7.0 1.	 	1	I	I	I	8.2	0.1
cheddar	1.5 oz	585	0.50		214	10.8	4.0	24.8	4.3	5.5 4;	23.1	1.0	45.3 1	6.8	2.8	9	1.2 1.2	0	і О	I	I	I	8.8	I
cuaker Caramer com cake Rice Krispie Treat Ritz Bitz w/ cheese	0.9 oz 1.3 oz 1.8 oz	75 83 171	0.50 0.75 0.50	0 0 0	99 141 242	0.8 3.7 11.3	0.1 0.7 2.9	20.7 26.9 30.4	2.5 1.1 1.6	3	74.3 51.5 55.2	0.0 0.1 0.3 1 1 0.3 1 1 0.3 1 1 0.3 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	7.3 23.8 12.1	1.1 4.7 11 0.9	1.9 8.0.4		0.01	- 0 - 0	1 1 1	1 1 1	1 1 1	1 1 1	2.9 9.6	- 3.4 0.1
Ritz Bitz w/ peanut butter S'mores granola treat	1.8 oz 1.1 oz	47 38	0.50 0.50	© 0	246 135	12.2 5.2	2.7 3.8	27.9 19.9	6.3 2.9	0. 1. 1 4. 2	53.5 32.8	0.0	44.7 34.9 2	9.9 5.2	0.3	0.2 49	9.1 7.9 1.	40	۱ I ۳	1 1	1 1	0.4 0.1	11.3 4.4	3.6 1.4
Frozen desserts Choc Chip cookie sandwich	4.5 oz	92	1.00	9	207	8.1	4.7	31.3	3.8	28.5	52.4	0.8	35.1 2	0.3	5.8	0.4 86	3.5 0.	4	ı m	I	0.2	I	8.0	4.3
éclair	3.0 oz	58	0.75	9	219	11.0	5.0	29.2	3.4	34.4	70.0	0.8	45.2 2	0.5 3	8.0	0.1 8	1.	0.	1	I	I	I	7.5	4.1
Ice cream bar, suaw. shortcake Ice cream sandwich	3.0 oz 3.0 oz	[*]	0.75 0.75	© © ©	171 173	12.0 6.8	5.0 3.9	15.3 26.2	1.4 3.2	15.3 23.8	28.1 43.8	0.7	53.2 2 35.1 2	6.3 0.3 6	3 5 3 3 7 3	4.7 48 0.3 72	3.5	0 0 0 0		0.2	0.1 0.2	11	3.9 6.7	2.5 3.6
Nutry ruyale ice cream cone Star bar ice cream Strawberry sherbet Sundae cup, choc	3.0 oz 4.4 oz 3.0 oz 6.0 oz	61 71 2 1	0.75 0.75 0.75 0.75 0.75	8 8 9 8 8 9	170 343 73 242	10.7 25.3 - 7.9	4.4 19.4 - 4.1	17.1 29.1 18.1 38.5	3.5 3.4 5.6 5.6	20.6 39.4 - 17.0 1	42.8 72.0 3.6 24.5	1.1 - 1.1	56.5 2 56.5 5 1.1 29.2 1	3.1 5 1.0 10 - 5.0 4	5.0 2.7 8.5	0.3 78 0.5 116 8.5 24 15	3.5 0. 5.1 0. 1.5 0. 7.2 0.	0 0 m N m	-	0.2	0.2 0.3 0.5	0.1 1 1	10.5 25.4 - 6.5	2.6 3.0 7.3

Value is exactly zero.
 Value is exactly zero.
 Individual items within a source if information was obtained during validation but not reported on the survey.
 Information was obtained during validation but not reported on the survey.
 Information was obtained during validation but not reported on the survey.
 Intervet column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
 X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = low calorie; 0 = cholesterol-free; 0 = low cholesterol; 0 = sodium-free; 0 = low sodium;

3 Also includes legumes, soybean products, nuts, and seeds.
 5 Total number sold was sometimes reported for all flavors of the same product. 'Number' appears in table for first flavor, followed by remaining flavors denoted by *.
 Source: Cateteria Survey and and Inventories of Competitive Foods.

School #2, Snack bar⁵

									lacronu	trients					Σ	icronutr	ients	<	llean # s	ervings	per pyra	mid gro	dn	Fat & s	ugar
	Package	-dmb-		FDA FDA			Me	an amour	nt per iter	۶			Percent calories 1	t of from	101								2	lean amo iten	unt per
	size	er 1		label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total fat	atura- ted fat	RE)				rain Ve al	get-	ruit	airy Me	eat ³ Di tic	scre- mary Fat (g)	Added sugars (tsp)
Frozen desserts Sundae cup, strawberry	6.0 oz	*	0.75	80	242	7.9	4.1	38.5	5.6	17.0	124.5	1	29.2	15.0	48.5	2.4 15	57.2	0.3	1	1	-	0.5	1	6.5	7.3
Salads with dressing Shake-Up Caesar Salad	4.0 oz	0	2.00		176	14.6	2.8	6.4	5.7	41.3	279.5	1.5	74.8	14.1	13.8	19.9 10	14.7	4.1	0.3	1.3	-	1.0	0.3	3.3	I
snake-up ciarden Greens	4.0 oz	26	2.00	0	124	10.9	1.6	6.6	1.3	0.6	449.2	1.7	78.9	11.7 2	87.0	11.6	23.9	9.6		8.1	I	1	-	0.7	0.4
Snacks Cheetos Asteroid Flamin' Hot																									
snacks Cheetos Crunchy Doritos Asteroids 30's Nacho Mini	2.5 oz 1.8 oz	109 83	1.25 0.75	0	393 275	24.4 17.1	4.7 3.3	38.1 26.7	5.4 3.8	50 8	744.2 520.9	0.8	55.9 55.9	10.7	24.1 16.9	1.0.0	11.1 28.8	2.2	2.7 1.9		1 1			5.2	1 1
Doritos Cool Ranch Fritos corn chios	2.0 oz 1.8 oz 2.3 oz	27 368 32	1.25 0.75	0 0 0	284 249 344	14.9 13.0 21.3	0, 12, 0 0, 12, 0	35.7 31.2 36.3	9.5 0.7 0.7		299.4 262.0 401.9	3.2 3.2 1	47.1 47.1 55.8	9.0 9.0 7.6	11.3 9.9		37.3 76.4 81.0	6. 8. 8 0.8	22.7					2.1 0.6	
Fruit By the foot	0.8 oz	- 66 - j	0.50	000 000 X X	50	0.6	0 - 0	12.0	0.2		13.0 8.7	0.000	7.7 7.8	1.7	2.6	,	6.9	0.5		1 1	0.3	1 1			2.1 1.4
Fruit snacks, grape Fruit snacks, peach Fruit snacks, unhite	2.3 02 2.3 02	<u></u> *	0.75	9 9 0	238	5.2	0.0	54.4 54.4	1 1		202.2	5 7	8.4 8.4	2 Z	7.7	20.0		0.5	1 1		- <u>+</u>			1 1	6.4 6.4
Fruit Sriacks, write grape Funyuns	2.3 oz 1.3 oz	* 9 1	0.75 0.75	0 0	238 196	2.2 12.2	0.6 2.3	54.4 19.1	- 2.7	- - 4.	202.2 372.1	2.2 0.4	8.4 55.9	2.4 10.7	7.7 12.1	9.92	11.5 20.6	0.5 0.8	1.3	1 1	1.2	1.1		0.8	6.4
chips	1.5 oz	18	0.75		228	14.7	4.7	22.5	3.0		252.6	1.9	58.1	18.4	I	. 3.2	0.2	0.7	1	1.5	I	I	-	3.3	I
BBQ potato chips	1.5 oz	78	0.75		228	14.7	4.7	22.5	3.0	1	252.6	1.9	58.1	18.4	I	. 3.2	0.2	0.7	1	1.5	I	I	-	3.3	I
potato chips	1.2 oz	34	0.75	0	161	6.2	0.9	24.3	1.7	1	312.0	1.6	34.5	5.0	I	1	12.5	0.3		2 I	I	I	1	6.1	0.8

Value is exactly zero.
 Value is exactly zero.
 Individual items within a source if information was obtained during validation but not reported on the survey.
 Information was obtained during validation but not reported on the survey.
 Interface for a column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
 X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ② = cholesterol-free; ③ = low cholesterol; ③ = sodium-free; ④ = low sodium;

3 Also includes legumes, soybean products, nuts, and seeds.
 5 Total number sold was sometimes reported for all flavors of the same product. 'Number' appears in table for first flavor, followed by remaining flavors denoted by *.
 Source: Cateteria Survey and and Inventories of Competitive Foods.

School #2, Snack bar⁵

								¥	lacronut	rients					Σ	icronutr	ients		Mean # s	ervings	per pyr	amid gr	dno	Fat &	sugar
	ackage	-dmb/	Ċ	FMNV & FDA			Mea	in amoun	t per iten	c			Percent calories 1	t of irom										Mean an ite	iount per m
	size	e.		label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- chydr- F ate (g)	rotein (g)	Chole- sterol (mg)	Sodi- [um (mg)	Dietary fiber (g)	Total S. fat	atura- ted fat	A E	Jo ∭ Dia Dia Dia Dia Dia Dia Dia Dia Dia Dia	um (mg)	lion (jan	àrain V	eget-		Dairy	Meat ³	Discre- ionary Fat (g)	Added Sugars (tsp)
Snacks Lays Salt & Vinegar potato chips Lays Sour Cream &	1.5 oz	28	0.75		228	14.7	4.7	22.5	3.0	I	252.6	1.9	58.1	18.4	I	13.2	10.2	0.7	I	1.5	I	I	I	13.3	I
Onion baked potato chips Nachos Doritos Ruffles Cheddar &	1.2 oz 1.8 oz	19 305	0.75 0.75	00	161 249	6.2 13.0	0.9 2.5	24.3 31.2	1.7 3.5		312.0 262.0	1.6 3.2	34.5 47.1	5.0 9.0	9.9	1 1	42.5 76.4	0.3 0.8	2.4	1 2 1	1 1	1 1	1 1	10.6 10.6	11
Sour Cream potato chips	1.5 oz	124	0.75		228	14.7	4.7	22.5	3.0	1	252.6	1.9	58.1	18.4	I	13.2	10.2	0.7	I	1.5	I	I	I	13.3	I
Sun Chips Multicrain	1.0 oz	88	0.75		143	8.4	2.3	14.8	2.9	2.5	237.6	2.8	52.8	14.2	10.5	1	23.5	0.5	1.2	I	I	I	I	7.2	I
Snack Original flavor	1.5 oz	32	0.75	0	220	11.1	0.8	27.0	3.0	1	178.2	2.6	45.4	3.4	11.2	I	2.3	0.4	2.1	I	I	I	I	8.9	0.1
Average of all items Unweighted	1 1	58 3782	0.76		231 238	9.5 10.5	3.0 2.9	34.5 33.8	3.2 3.2	7.7 2	210.6 276.4	1.3 1.6	36.1 39.8	11.5	80.5 58.8	12.2 9.7	70.8 51.7	0.9 1.1	0.8 1.3	0.2	0.1	0.1	11	8.4 9.0	3.3 2.2

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
Number column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
A indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; 0 = low calorie; 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low calorie for all flavors of the same product. Number 'appears in table for first flavor, followed by remaining flavors denoted by *.
5 Total number selported for all flavors of the same product. 'Number' appears in table for first flavor, followed by remaining flavors denoted by *.
5 Total number selported for all flavors of the same product. 'Number' appears in table for first flavor, followed by remaining flavors denoted by *.

School #2, School store

								Z	lacronut	rients					M	cronutri	snts	Me	an # serv	ings pe	r pyrami	d group	Fat	& sugar
	ackage	Numb-	Ċ	FMNV & FDA			Mea	n amoun	t per iten	c		-	Percent calories f	rom rom		đ							Mean	amount per item
	size	er1		label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- chydr- ate (g)	rotein ((g)	Chole- sterol (mg)	Sodi- [um (mg)	Dietary fiber (g)	Total ^{Si} fat	atura- ted fat	A (I) A (I)	ی روت agin (jer	קר ד) ד) ד)	ig) Gra	in Vege able	Frui	t Dairy	Meat	Discre- tionary Fat (g)	Added Sugars (tsp)
Candy																								
Air rieads blue raspberry Air Heads cherry	0.6 oz 0.6 oz	180 144	0.25 0.25	0 0 0 0	<u>12</u> 12	2.6 2.6	0.6 0.6	25.9 25.9	0.3 0.3	1 1	33.1 33.1	1 1	18.8 18.8	4.3 4.3		0 0 0 0	2.3	, .	1 1	11	11	11	2.4 2.4	6.4 6.4
Air neads green apple Air Heads orange Air Heads strawberry	0.6 oz 0.6 oz 0.6 oz	108 72 144	0.25 0.25 0.25	000 000	122 122 122	2.6 2.6	0.6 0.6 0.6	25.9 25.9 25.9	0.3 0.3		33.1 33.1 33.1		18.8 18.8 18.8	4.3 4.3 6.3		5 5 3	 		111		111	111	2:4 2:4 2:4	6.4 6.4 6.4
Air Heads watermelon	0.6 oz	108	0.25	8	122	2.6	9.0	25.9	0.3	I	33.1	I	18.8	4.3	-	2.3	5.3		I	I	I	I	2.4	6.4
mystery Caramel apple pop Charme fluffu cotton	0.6 oz 1.0 pop	144 240	0.25	X 8 6 4 8	122 63	1 2.6	0.6	25.9 16.7	0.3	11	33.1 6.5	1 1	18.8 -	4.3		1 2.3	5.3 -		11	11		11	2.1	6.4 4.0
candy Charms sweet & sour	1.0 pop	240	0.25	×264	221	I	I	58.2	I	I	22.5	I	I	I	I	I	1.8	۱ N	I	I	I	I	I	13.8
pop	1.0 pop 1.4 oz 1.0 pop 1.0 pc	240 36 300 165	0.25 0.75 0.25 0.15	X X X 0 0 0 0 0 0 0 0	78 153 78 37	0. I I I	0.2	20.6 39.3 20.6 7.9	1 1 1 1	1 1 1 1	8.0 17.5 8.0 10.0		- - 18.7	+ + +.4	1111	N.		0 Q	N N N	1 1 1 1	1 1 1 1	1 1 1 1	- - 0.7	4.9 4.9 1.9
Lafty Taffy punch Lafty Taffy cherry Lafty Taffy punch Lifesaver pop swirl Lindor milk choc balls	1.0 pc 1.0 pc 1.0 pc 0.4 oz	165 165 165 48 48	0.15 0.15 0.15 0.25 0.35	0 0 0 0 0 X 0 0 0 0 0 X	37 37 52 62	0.8 0.8 1 - 3.7	0.2 0.2 2 - 2 2 - 2 2 2 - 2 2 - 2 2 2 - 2 2 2 - 2 2 2 - 2 2 - 2 2 2 2 - 2 2 2 - 2 2 2 2 - 2 2 2 - 2 2 2 2 - 2 2 2 - 2 2 2 2 - 2 2 2 - 2 2 2 2 - 2 2 2 2	7.9 7.9 7.9 7.1	0 0		10.0 5.3 9.8	1 1 1 0	18.7 18.7 18.7 - 53.8 (4.4 4.4 7.5 32.5		2. 2. 2. 2. 2.	1.6 1.6 1.6 0.0 0.0 0.0 0 0.0 0 0 0 0 0 0 0 0 0 0	· · · · øi			1 1 1 1 1	1 1 1 1 1	0.7 0.7 0.7 3.7	0. F F 6. F 0. 0. 0. 6. 4.
Lindor white choc balls M&M's Crispy M&M's Peanut Neon laser straw	0.4 oz 1.5 oz 1.7 oz 0.1 oz	48 24 1200 36	0.35 0.75 0.75 0.10	© © © © 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	65 209 11 175	9.0 12.3 0.0	0.02 0.05 0.0	7.1 30.3 30.3 3.0 80.6	0.7 1.8 1 - 5	6.05 2.00 2.00 2.00	10.8 25.9 1.1	- 9 o	53.6 45.8 2 - 8	32.4 18.0 18.0	0.1 22.5 11.3 21	0.2.0	3.9 7.5 7.5 7.5 7.5 7 7.5 7 7 7 7 7 7 7 7 7	່ຕ່ຕ່ໍດ 			0.2	0.2	9.0 16.1 16.1	- 4 0.7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7
Reese's Peanut butter cup	1.6 oz	38	0.75	0	244	14.1	5.0	24.6	- 4 6. 6		42.7	6. 1 . 4.	52.1	18.5	- 9.8 9.6	n rố	5.1.0	i ni I I		5 1	- 	0.4	13.5	4.4

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol-free; **0** = low

School #2, School store

								Z	acronut	rients					M	cronutri	ents	Σ	ean # sei	vings p	er pyram	id group	Fai	& sugar
	Dackage	-dmb-		FMNV & FDA			Mea	n amoun	t per iten	-		-	Percent calories fi	of Tom	10	Ċ							Mean	amount per item
	size	e.	L	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- bhydr- F ate (g)	rotein 6 (g)	thole-	Sodi- D um (mg)	lietary fiber (g)	Total Sé fat	ttura- ted fat	RE)		ng) (n ng) (n	ng (jg)	ain Veg abl	et- Fru	lit Dair	y Meat	3 Discre- tionary Fat (g)	Added Sugars (tsp)
Candy																								
Skittles Fruit candy Snickers	2.2 oz 2.1 oz	36 48	0.75 0.75	8 8 8	249 273	2.7 14.0	0.5 5.1	55.8 33.8	0.1 4.6	- 7.4 1(9.8 51.6	+ + 4.	9.7 46.2 1	1.9	2.2	1.2 0.3 5	3.6	4.0	5.	1 1	10	1 0	2.1 10.9	12.9 7.4
Sour patch kids	2.0 oz 2.0 oz	120 144	0.75 0.75	× % 8 8 4 8 4 8	219 219	1 1	1 1	56.1 56.1	11	1 1	25.0 25.0	11	1 1	1 1	1 1	1 1	1.7	<u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u>	ା । ସ୍ ସ୍	1 1		1 1	11	13.4 13.4
Starburst California Fruit Starburst Fruit Chews	2.1 oz 2.1 oz	72 36	0.75 0.75	0 0 0 0	231 231	4.8 8.8	22	49.1 49.1	0.6 0.6	0.2	62.7 62.7	1 1	18.8 18.8	4.2		0.0 0.0 1 1	0.0	<u></u>		11			4.6 4.6	12.1 12.1
Starburst Hard Tropical Starburst Sours Starburst Tropical Swedish fish red	1.6 oz 2.1 oz 2.1 oz 2.0 oz	24 72 72 72	0.75 0.75 0.75 0.75	X X 0 0 0 0 4 0 0 0	169 231 219	4.4. 8.8.	1 = = 1	44.5 49.1 49.1 56.1	- 0.6 0.6	1 0 0 1	17.2 62.7 62.7 25.0		1 8.8 1 8.8	4.2		- 3.3 - 3.3 - 1 - 1	4.1 0.0 7.1	1.0			1 1 1 1		4 4 9.4	10.6 12.1 13.4
Twix caramel cookie bar	2.0 oz	36	0.75	00	283	13.8	5.1	37.2	2.6	2.8	09.4	0.6	44.0	6.1	4.2	0.2 5	1.0	.5	9.	I	I	I	13.2	3.7
Chewy Centered	1.8 oz	96	0.75	X 264	187	I	I	49.3	I	I	19.1	I	I	I	I	I	1.5	-	1	I	I	I	I	11.7
Snacks Pringles Cheezums Pringles Original	1.8 oz 1.8 oz	<u>5</u> 5	0.75 0.75		277 277	19.1 19.1	4.7 4.7	25.3 25.3	2.9	00 	25.5 25.5	- - - 8, 8, 8, 8,	61.9 1.9	5.2	1 1	4.1 1.1 1	1.9		+ +		1 1	1 1	19.4 19.4	0.4 0.4
flavored	1.8 oz	12	0.75		277	19.1	4.7	25.3	2.9	ෆ 	25.5	1.8	61.9	5.2	I	4.1	1.9	.7	-	-	I	I	19.4	0.4
& onion Slim Jim	1.8 oz 1.0 stk	24 600	0.75 0.35		277 39	19.1 2.5	4.7 1.0	25.3 1.1	2.9 3.2	- 3 4.6 2	25.5 12.5	1.8 0.2	61.9 56.3 2	15.2 23.8	1 1	4.1	1.9		÷ 1			- 0.9	19.4 1.7	0.4
Average of all items Unweighted	11	41 5594	0.50 0.31		160 95	4.9 1.8	1.6 0.6	29.3 20.0	1.0 0.6	0.9	66.2 43.5	0.1	23.5 15.6	8.0 5.4	2.2 0.6	6.2 3.8 1	1.3 4.4		0	N	11	11	4.8 1.6	6.1 4.6

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; 0 = low calorie; 0 = conclesterol-free; 0 = low cholesterol; 0 = sodium-free; 0 = low sodium;
Also includes legumes, soybean products, nuts, and seeds.
Survey and and Inventories of Competitive Foods.

4
Ŧ
ē
Ē
ò
Ma
6
<u> </u>
Б
/e
-î
#
ō
2
õ
••

									Macron	utrients					2	Aicronut	rients		Mean # :	servings	per pyra	amid gr	dno	Fat & s	ugar
	Package	-dmuN	C	FMNV & FDA			Me	an amoui	nt per ite	Ę			Perce	nt of from	ļ	ļ								Mean am	ount per n
	size	e.	220	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total	Satura- ted fat	A (BE)	C C (mg)	(mg)	lron (mg)	àrain <	able F		Dairy N	feat ³ D	iscre- onary Fat (g)	Added Sugars (tsp)
Beverages Hood Fruit Punch (10% juice) [9]	16.0 oz	101	1.00	0 6 6	233	I	I	59.0	I	I	109.1	0.5	I	I	5.0	27.8	39.7	1.0	I	I	0.3	I	I	I	12.8
(10% juice) [9]	16.0 oz	150	1.00	2 8 4	261	0.3	I	68.0	0.4	I	16.3	0.6	1.0	0.1	14.4	25.6	17.1	1.1	I	I	0.4	I	I	0.1	10.4
(100%) [9] Nesquik Choc milk [5]	16.0 oz 16.0 oz	73 13	1.00	234	223 417	0.3 17.0	- 10.5	53.4 51.7	3.4 15.9	_ 61.0	5.0 298.0	1.0 4.0	1.2 36.6	0.1	39.6 145.0	92.7 4.6	300.1 300.0	0.5 1.2	1 1	1 1	2.7 -	2.0	1 1	- 16.1	- 6.7
milk [6]	16.0 oz	43	1.50		417	17.0	10.5	51.7	15.9	61.0	298.0	4.0	36.6	22.7	145.0	4.6	300.0	1.2	I	I	I	2.0	I	16.1	6.7
milk [4]	16.0 oz	28	1.50		442	15.4	9.6	62.4	15.1	62.4	240.5	I	31.4	19.5	142.2	4.4	300.0	0.4	.	I	I	1.9	I	14.5	6.9
milk [3]	16.0 oz	9	1.50		442	15.4	9.6	62.4	15.1	62.4	240.5	I	31.4	19.5	142.2	4.4	300.0	0.4	1.1	I	I	1.9	I	14.5	6.9
Average of all items Unweighted	1 1	7 414	1.29		348 284	9.3 3.7	5.7 2.2	58.4 60.6	9.4 1.1	35.3 13.4	172.5 93.5	1.5	19.7 8.1	12.1 4.7	90.5 44.7	37.7 5 51.0 2	551.0 295.6	0.8	0.3	1 1	0.5 0.7	1.1 0.4	1 1	8.7 3.4	7.2 8.4

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; • = low calorie; 0 = cholesterol-free; • = low cholesterol-free; • = low sodium;
All includes legumes, soybean products, nuts, and seeds.
A vending machine items occupying multiple slots are indicated with the number of slots shown in [] following the product name.

6-Ш

School #2, Vending Machine #2^{4,6}

								2	lacronu	trients					V	licronut	rients		Vlean # s	ervings	per pyra	amid gre	dno	Fat & s	ugar
	Package	Numb-		FMNV ^{&} FDA			Mea	in amoun	ıt per iter	۲		_	Percen calories	t of from	177									Mean amo iten	ount per 1
	size	er-1		label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- F ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- E um (mg)	Dietary fiber (g)	Total S fat	iatura- ted fat	A (RE)	C C M	(mg) (mg)	mg) (àrain V	eget-	Luit D	airy	feat ³ D	iscre- onary Fat (g)	Added Sugars (tsp)
								-	-				-						_	-		-			
Beverages Hood Apple juice (100%) [9]	16.0 oz	91	1.00	8 6	233	0.5	I	57.2	0.7	I	34.7	0.5	1.9	0.3	1	71.4 5	54.7	1.3	I	I	2.7	I	I	I	I
Hood Coffee milk (1%) [2]	16.0 oz	16	1.00		340	10.0	6.2	56.0	16.1	34.0 3	301.0	2.5	26.5	16.4 2	85.0	30.0	<u>68.0</u>	1.2	I	I	I	2.0	I	8.2	5.5
Hood Lemonade (10% juice) [9] Hood Whole Milk	16.0 oz 16.0 oz	92 9	1.00	2 6 4	261 300	0.3 16.3	_ 10.2	68.0 22.7	0.4 16.1	- 66.4 2	16.3 39.1	0.6	1.0 48.9	0.1 30.5 1	14.4 51.3	25.6 4.6	17.1 82.7	1.1 0.2	1 1	1 1	0.4	2.0	1 1	0.1 15.4	10.4 -
(fat free) [4]	16.0 oz	19	1.50	90	288	2.4	1.4	53.4	17.2	8.4 2	241.4	3.0	7.4	4.3	83.8	4.6	300.0	1.4	I	I	I	2.0	I	1.5	6.7
milk [6]	16.0 oz	6	1.50		417	17.0	10.5	51.7	15.9	61.0 2	98.0	4.0	36.6	22.7	45.0	4.6 8	300.0	1.2	I	I	I	2.0	I	16.1	6.7
milk [9]	16.0 oz	5	1.50		442	15.4	9.6	62.4	15.1	62.4 2	240.5	I	31.4	19.5	42.2	4.4 8	300.0	0.4	1.1	I	I	1.9	I	14.5	6.9
[3]	16.0 oz	na	1.50		442	15.4	9.6	62.4	15.1	62.4 2	240.5	I	31.4	19.5	42.2	4.4 8	300.0	0.4	1.1	I	I	1.9	I	14.5	6.9
Tropicana smootnies, mixed berry	11.5 oz	e	1.75	96 4	240	I	I	59.5	4.9	16.0	77.3	2.9	I	I	51.1	60.0	50.0	0.7	I	I	1.4	0.4	I	4.1	7.4
strawberry	11.5 oz	4	1.75	864	240	I	I	59.5	4.9	16.0	77.3	2.9	I	I	51.1	60.0	250.0	0.7	I	I	1.4	0.4	I	4.1	7.4
Average of all items Unweighted	1 1	10 254	1.35		320 272	7.7 2.9	4.7	55.3 59.6	10.6 4.6	32.6 1 10.4	76.6 86.9	1.6	18.5 7.7	11.3 4.3	26.6 62.5	37.0 E 75.1 3	542.3 191.1	0.9	0.2	1 1	0.6	1.3 0.5	1.1	7.8 2.5	5.3 5.3
dine is sulen –	0402																								

School #2, Vending Machine #3

								W	acronut	rients					M	cronutri€	ents	Mea	n # servi.	ngs per	pyramid	group	Fat &	sugar
	ackage	-dmb-	C	FMNV & FDA			Mea	an amoun	t per iten	_			Percent calories fi	jo me									Mean an ite	nount per m
	size	er1	Price	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- P ate (g)	rotein ((g)	thole- sterol ((mg) (Sodi- Di um f mg)	ietary liber (g)	Total ^{Sé} fat	atura- ted fat	RE) A RE)		n (mg) Grair	Veget- able	Fruit	Dairy	Meat ³	Discre- tionary Fat (g)	Added Sugars (tsp)
Prozen desserts American Glory sundae cone	4.3 oz	43	1.00	۵	256	16.1	6.6	25.8	5.2	31.1 (34.5	1.7	56.5 2	3.1 6	12.9	0.5 116	3.3 0.5	0.2	I	I	0.2	0.2	15.8	4.0
cream bar	3.0 oz	17	1.00	90	201	16.0	11.0	14.4	1.7	19.5	35.5	0.2	71.6 4	t9.3 E	51.8	0.3 5.	.4 0.	-	I	I	0.1	I	12.5	2.8
Sandwich	4.5 oz	14	1.50	9	207	8.1	4.7	31.3	3.8	28.5	52.4	0.8	35.1 2	20.3	75.8	0.4 8(.5 0.4	1 0.8	I	I	0.2	I	8.0	4.3
bar	4.0 oz	na	1.00		205	8.6	3.0	30.9	3.6	36.5 3	10.3	0.8	38.0 1	13.0 4	10.2	0.2 8(0.0 1.7	7 0.7	I	I	I	I	8.0	4.3
	4.5 oz	34	1.00	9	212	14.0	8.0	21.8	2.5	29.6	54.0	0.3	59.4 3	34.0 7	78.6	0.4 8.	.0 0.	1	I	I	0.2	I	19.0	4.2
Chy Dauy Exita Sour Cherry Italian Ice Ice cream sandwich	4.5 oz 3.5 oz	4 7 2	0.75 0.75	X © 6 6	139 168	6.6	3 I 3 I	35.4 25.4	0.4 3.1	23.1	23.9 12.5	0.6	- 35.1 2	20.3 6	- 31.5	1.1 2 0.3 70	0.0	- 0.6	11	11	- 0.2	11	- 6.5	8.8 3.5
Saidona Pic Cookle	4.2 oz	7	1.50	ø	175	6.8	4.0	26.5	3.2	24.1	44.4	0.7	35.1 2	20.3 6	34.2	0.3 7;	3.3 0.0	3 0.7	I	I	0.2	I	6.8	3.6
cream bar	3.0 oz	24	1.25	ø	280	19.5	12.2	24.8	4.5	27.4 (<u> 59.3</u>	0.9	62.6	39.1 7	71.8	0.4 10	.2	۱ ~	I	I	0.2	0.1	18.9	4.8
Lemonade	4.0 oz	9	1.25	234	98	0.1	I	24.1	1.4	I	4.8	I	1.1	1	3.6 1	1.3 (0.0	1	I	0.3	I	I	I	4.0
ice cream bar Supersicle	4.0 oz	26	1.00	ø	235	12.0	3.5	30.9	3.6	36.5	0.06	. 8.0	46.0	13.4 4	10.2	0.2 8(0.0	7 0.7	I	I	I	I	8.0	4.3
Firecracker Red, White & Blue Bar Twix ice cream bar	4.5 oz 2.5 oz	16 21	0.75 1.25	× 20 64	104 175	- 13.0	9.9	27.2 14.9	- 1.7	20.2	17.3 36.8	0.2	- 36.5	- 1.0	1.00	- 0.3 55		1 1	1 1	11	- 0.2	I I	- 13.0	6.9 2.9
Average of all items Unweighted	1.1	13 234	1.08 1.05		189 207	9.3 11.7	5.1 6.3	25.6 24.8	2.7 3.0	21.3	35.0 52.1	0.5	39.0 2	21.8 ² 25.7 5	18.0 36.8	1.2 6 [,] 0.7 7 <u>;</u>	1.1 0.5 5.1 0.5	0.3	1.1	11	0.1 0.2	I I	9.0 11.6	4.5 4.4

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
Microates for some information nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol; ③ = sodium-free; **0** = low sodium; as indicates low each and inventories of Competitive Foods.

School #2, Vending Machine #4

								-	Macronu	trients					Z	licronut	rients		Mean #	servings	per pyra	imid gro	dno	Fat & s	ugar
	Package	-dmb-		FMNV & FDA			Me	an amou	nt per ite	E			Percer calories	nt of from	1								2	<i>l</i> lean amc iten	unt per
	size	er1	Price	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total ⁶ fat	satura- ted fat	A (RE)	C C C	um (mg)	lron (mg)	Grain	eget-	-ruit D	airy	eat ³ Di	iscre- onary Fat (g)	Added Sugars (tsp)
Baked goods/dessert Grandma's Choc chip cookies	2.8 oz	40	0.75		375	17.6	5.8	52.1	4.2	1	245.6	2.0	42.3	14.0	I	I	19.5	2.2	1.7	I	I	I		15.9	7.1
Oatmeal Bites, strawberry Quaker Fruit &	2.0 oz	50	0.75	0	209	4.3	0.9	41.3	2.5	I	94.7	1.2	18.3	3.7	348.1	1	306.0	2.8	1.5	I	I	I	I	2.8	5.7
Uatmeal Toastables, iced strawb	3.5 oz	na	0.75	0	390	10.1	1.5	70.6	4.7	I	240.0	2.1	23.4	3.5	284.8	0.3 4	100.0	3.5	3.1	I	0.1	I	1	7.0	7.3
Bread/grain products Cheetos Curls	1.5 oz	na	0.75		214	10.8	4.0	24.8	4.3	5.5	423.1	1.0	45.3	16.8	12.8	I	64.2	2.0	2.0	I	I	I		8.8	I
Gold pretzels Gold pretzels	2.0 oz	18	0.75	0	216	2.0	0.4	44.9	5.2	I	972.4	1.8	8.2	1.8	I	I	20.4	2.5	3.4	I	I	I	I	I	I
granola bar, choc chip Quaker Crispums	2.0 oz 1.3 oz	25 20	0.75 0.75	80	257 156	10.0 3.7	7.2 0.6	37.8 28.3	5.6 2.5	1 1	157.6 223.0	1.8	34.9 21.5	25.2 3.2	0.6	0.6	34.0 110.0	1.8 4.1	1.6 1.6	1 1	1	1 1	0.2	8.4 2.1	2.6
Snacks Cheetos Crunchy, King Size	3.5 oz	9	1.25	0	550	34.1	6.5	53.4	7.5	4.0	1041.9	1.1	55.9	10.7	33.7	0.2	57.6	2.3	3.7	I	I	I		30.4	I
King Size	3.0 oz	93	1.25	0	426	22.3	4.3	53.5	6.0	I	449.1	5.5	47.1	9.0	17.0	1	131.0	1.3 1.3	4.0	I	I	I		18.2	I
Cheese, King Size	3.0 oz	თ	1.25	0	426	22.3	4.3	53.5	6.0	I	449.1	5.5	47.1	9.0	17.0	1	131.0	1.3	4.0	I	I	I		18.2	I
chips comorese chips comorese Funyuns, King Size	4.5 oz 2.5 oz	8	1.25 1.25	0 0	688 393	42.6 24.4	5.8 4.7	72.6 38.1	8.4 5.4	2.8	803.7 744.2	6.3 0.8	55.8 55.9	7.6 10.7	11.5 24.1	- 0.	162.0 41.1	1.7	5.3 2.7	11	11	1 1		37.2 21.7	11
BBQ potato chips	2.8 oz	5	1.25		418	27.0	8.5	41.2	5.5	I	463.1	3.5	58.1	18.4	I	24.3	18.7	1.3	I	2.7	I	I	1	24.4	I
Sour Cream	2.8 oz	ი	1.25		418	27.0	8.5	41.2	5.5	I	463.1	3.5	58.1	18.4	I	24.3	18.7	1.3	I	2.7	I	I	1	24.4	I

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol-free; **0** = low

#
ine
lach
≥ ⊡
din
Ven
ť,
Sc

								2	lacronu	trients					W	cronutri	ents	M	ean # ser	vings po	er pyrar	nid grou	p Fa	it & sugar
<u> </u>	ackage N	-dmb-		FMNV & FDA			Mea	an amour	ıt per iteı	F			Percen calories	t of from	101								Mea	ח amount per item
	size	er-1	2002	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total ^S fat	atura- ted fat	A A ()		alci- Im ng) (n	ng (jg)	ain Vege	e Eru	lit Da	iry Mea	at ³ Discr tional Fat (g)	 Added Sugars (tsp)
Snacks Sun Chips Multigrain Snack Harvest Cheddar	1.5 oz	ى ب	0.75	0	220	1.1	0.8	27.0	3.0	1	178.2	2.6	45.4	3.4	11.2	1	2.3	4.0	۱ ۲	I	I		8	0.1
Average of all items Unweighted	1 1	15 294	0.98 0.98		357 341	17.9 15.9	4.3 3.9	45.4 46.8	5.1 4.8	0.8	463.2 375.7	3.1 3.1	41.1 37.5	9.7	50.7 37.0	3.3 10 1.2 11	9.1 1.1	<u></u>	ις Ο Ο		1 1		13.2	1.7 2.3

マ
10
47
±±-
A
ω.
ē.
<u> </u>
~
<u> </u>
~
<u> </u>
~
<u> </u>
-
_
_
-
0,
<i>—</i>
<u> </u>
~
υ.
_
<u> </u>
()
>
_
-
2
#
_
^
U
^
U
_
-
()
~
n.
~

									Macronu	trients					W	cronutri	ents	N	ean # se	rvings p	er pyrar	nid grou	p Fa	t & sugar
<u> </u>	^o ackage	-dmb Numb	ć	FMNV & FDA			Me	an amour	nt per iter	۶			Percent calories f	rom rom									Mea	i amount per item
	size	e.	0011	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- [um (mg)	Dietary fiber (g)	Total Si fat	atura- ted fat	E B B B A R B	μο Έο Έο		ng (gr	ain Vec	leter E	nit Da	Me	at ³ Discre tionar Fat (g)	- Added Sugars (tsp)
Beverages Spring water [7]	16.0 oz	373	1.00	X 0 4 0	1	I	1	I	1	1	14.2	ı	I	1	1	1	9.5						I	I
Average of all items Unweighted	11	1 373	1.00		11	11	1.1	11	1.1	1.1	14.2 14.2	11	1.1	1.1	11	I I	9.5						11	1 1

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
Number column nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; • = low calorie; 0 = cholesterol-free; • = low cholesterol-free; • = low sodium;
Also includes legumes, sorbaan products, nuts, and seeds.
Vending machine tiemes occupying multiple slots are indicated with the number of slots shown in [] following the product name.
Cafeteria Survey and and Inventories of Competitive Foods.

School #2, Vending Machine #6⁴

								N	lacronut	trients					W	icronutr	ients	_	/lean # s	ervings	per pyra	amid gre	dno	Fat & s	ugar
	Package	-dmbN	Ċ	FMNV & FDA			Meê	an amoun	it per iter	٩			Perceni calories 1	t of rom									2	lean amo iten	ount per n
	size	e.		label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- F ate (g)	Protein (g)	Chole- sterol (mg)	um (mg)	Dietary fiber (g)	Total S fat	atura- ted fat	A (i			ng)	irain Ke	ible F	ruit D	airy	leat ³ Di	scre- nary Fat (g)	Added Sugars (tsp)
Beverages Vonding Arctic																									
Orange Chillers	11.5 oz	70	0.75	2 6 4	190	I	I	48.0	0.7	I	7.2	I	I	1	14.3	30.0 1(0.00	0.4	I	I	0.5	I	1	I	9.6
	11.5 oz	165	0.75	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	I	3.6 10	5.1 2	28.4	0.8	I	I	0.2	I	1	I	9.2
Veryine Icea tea w/ Lemon Chillers	11.5 oz	37	0.75	2 6 4	130	I	I	33.0	I	I	9.9	I	I	I		30.0 1(0.00	1	I	I	I	I	1	I	4.3
Veryine Orange Juice (100%)	11.5 oz	6	0.75	2 6 4	150	0.5	I	35.3	2.1	I	7.2	0.7	3.0	0.4	34.4 12	23.1	28.6	1.6	I	I	1.9	I	1	I	I
Chillers	11.5 oz	79	0.75	2 6 4	188	0.2	I	48.9	0.3	I	11.7	0.4	1.0	0.1	10.4	30.0 1(0.00	0.8	I	I	0.3	I	1	I	7.4
banana punch	11.5 oz	114	0.75	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	I	3.6 10)5.1 ²	28.4	0.8	I	I	0.2	I	1	I	9.2
veryine rropical Freeze Chillers	11.5 oz	80	0.75	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	I	3.6	30.0 1(0.00	0.8	I	I	0.2	I	1	I	9.2
Average of all items Unweighted	1 1	7 554	0.75		166 170	0.1	1 1	41.7 43.2	0.2	1 1	38.6 54.0	0.3	0.6 0.2	1.0	6.6 8	31.9 (33.7 (59.4 52.8	0.7 0.7	1 1	1 1	0.5 0.3	1 1	1 1	1 1	7.0 8.5

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
K indicates fool of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: © = calorie-free; • = low calorie; © = cholesterol-free; • = low cholesterol-free; • = low soluting a labor indicated with the number of slots shown in [] following the product name.
A vending machine items occupying multiple slots are indicated with the number of slots shown in [] following the product name.

<u>v</u>
Ń.
±
A \
w
Ē
<u> </u>
-
_
_
0
=
<u> </u>
>
5
ς,
_
=
σ
=
_
~
w
× .
~
~
<u>.</u>
±
_
ō
0
^
0
-
-
~
~
۲D.
~

								2	lacronut	rients					Mi	cronutri	ents	2	lean # su	srvings p	er pyrar	nid grou	dr	Fat & su	gar
	Package	-dmb/		FMNV & FDA			Me	an amoun	t per iten				Percent calories fr	o đ		đ							Me	ean amou item	int per
	size	Ē.	2100	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- F ate (g)	rotein (g)	Chole- (sterol ((mg) (Sodi- D (mg)	ietary fiber (g)	fat Se	ttura- ted fat	чц ЧЕ ЧЕ	ນ ເດີຍ ເດີຍ	ng) – n ng) – n ng)	on (gr	ain Ve	get- ole	uit Da	iry Me	eat ³ Disc tion (₍	cre- ary Si ()	dded ugars tsp)
Beverages Vendine Annle inice																									
(100%)	11.5 oz	22	0.75	264	168	0.4	I	41.6	0.2	I	10.7	0.4	2.1	0.4	I	3.2 2	5.0	6.1	I	-	ь В				I
Verfine Fruit punci [2]	11.5 oz	254	0.75	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	1	3.6 10	5.1 2	8.4 (9.8	I	0	Ņ				9.2
Grape Chillers	11.5 oz	66	0.75	2 6 4	162	I	I	41.3	I	1	21.6	1	I	1	ю I	0.0 10	0.0	9.6	I	0	Ņ				0.6
veryline iced tea w/ Lemon Chillers Vendine Kiwi	11.5 oz	100	0.75	264	130	I	I	33.0	I	I	9.9	1	I	1	9	0.0 10	0.0		I	I			ı 		4.3
Strawberry Chillers	11.5 oz	89	0.75	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	I	3.6	0.0 10	0.0	.8		0	ı Ņ				9.2
lemonade	11.5 oz	89	0.75	2 6 4	188	0.2	I	48.9	0.3	I	11.7	0.4	1.0	0.1	0.4 6	0.0 10	0.0).8	I	0	e.				7.4
veryine i ropical Freeze Chillers	11.5 oz	113	0.75	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	1	3.6 6	0.0 10	0.0	.8	I	0	Ņ				9.2
Average of all items Unweighted	1 1	7	0.75 0.75		164 164	1 1	1 1	41.7 41.7	1 1		41.2 53.6	0.3 0.3	0.4	1.0	3.5 7	8.3 7 3.9 7	9.1 2.9 (7.0	1 1	00	4 oi				6.9 8.0

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
K indicates fool of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: © = calorie-free; • = low calorie; © = cholesterol-free; • = low cholesterol-free; • = low soluting a labor indicated with the number of slots shown in [] following the product name.
A vending machine items occupying multiple slots are indicated with the number of slots shown in [] following the product name.

School #3, A la carte

	Macronutrients	Macronutrients	Macronutrients	Macronutrients	Macronutrients	acronutrients	ients				ercent of		Micr	onutrient	s	Mean	# servinç	js per py	ramid gi	dno	Fat & : Mean am	sugar nount per
attrate Catcher Fronte Catcher Fronte Catcher Fronte Catcher Fronte Catcher Ca	~	⁸ DA			Mea	n amount	per item	_		ca -	alories fror	< ح	Š	Calci							ite ite	
57 414 41 230 2855 19 820 103 633 15 210 27 210 27 210 27 210 27 210 27 210 27 210	teri- teri- a2 es ()	Calori- fi es ((F = =	at (g)	Satura- ted fat (g)	Carb- ohydr- Pr ate (g)	(g) (I	hole-Si terol L mg) (n	iodi- Die Jm fib ((stary To ber fɛ g)	ata Satu tec	ra- t t		(mg)	Iron (mg)	Grain	Veget- able	Fruit	Dairy	Meat ³ [Discre- ionary Fat (g)	Added Sugars (tsp)
4.9 73.4 5.2 9.6 129.6 2.1 4.11 8.4 10.1 7.80 3.2 1.6 1.6 1.0 10.0 10.7 7 - - - - - - - - - - - - - - - - - - - 10.0 10.7 - - - 150 - - - 100 - - - 100 10.7 - - 150 - - 2 100 - - 100 - - - 100 10.7 - 42.3 - - 2 10.1 0.1 20 0.2 2.1 10.4 10.7 10.7 - 42.3 - - 2 10.5 2.84 0.8 10.7 2.84 10.7 10.7 10.7 10.7 10.7 10.7	347 20.1 430 17.3	347 20.1 430 17.3	20.1 17.3		5.7 3.3	41.4 64.2	4.1 5.7 6	33.0 255 30.0 390	9.6 2.5 2	2.2 36	2.1 3.3 6.	8 109 82	5 4 - 0.	8 28.3 68.3	1.7 1.5	0.9 2.1	11	- 0.	11	0.2	19.1 15.2	6.3 9.0
- 10.0 - - 10.0 - - 10.0 - - 10.0 - - - 10.0 - - 10.0 - - 10.0 - - 10.0 - - 10.0 - - 10.0 - - 10.0 - - 10.0 - - 10.0 - - 10.0 - - 10.0<	6 524 23.9 8 460 22.0	524 23.9 460 22.0	23.9 22.0		4.9 5.0	73.4 65.2	5.2 3.2 1	9.6 12 16.3 22	9.6 0.0	2.1 41 0.4 43	1.1 8. 3.0 9.	8 5	+ + 0 0	1 78.0 1 45.9	3.2 1.3	2.5 1.6	11	11	11	1 1	21.4 10.0	10.0 10.7
$ \begin{array}{ c c c c c c c c c c c c c c c c c c $	۱ ۱ ۱	1	I		I	I	I	÷	5.0		I	1	I	10.0	I	I	I	I	I	I	I	I
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	(d) 254 –	254 –	I		I	62.4	I	- 37	- 0.0		I	231	3 75.	2 23.1	0.2	I	I	1.0	I	I	I	10.4
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	04 167 - 04 167 -	167 – 167 –	11		11	42.3 42.3	1 1		8.1 8.1 0	0.4 4.C		ოო	6 105. 6 105.	1 28.4 1 28.4	0.8 0.8	11	11	0.2	11	1 1	1 1	9.2 9.2
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	0 @ 162 -	162 –	I		I	41.3	I	1	1.6		1		60.	0 100.0	9.0	I	I	0.2	I	I	I	9.0
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	D @ 180 -	180 –	I		I	42.4	4.7	Ē.	0.8 C	- +.C	I	190	5 80.	9 18.0	1.0	I	I	1.0	I	I	I	4.7
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	B @ 167 –	167 –	I		I	42.3	I	- 7	'8.1 C	- 4.C	I	<i>ლ</i>	6 105.	1 28.4	0.8	I	I	0.2	I	I	I	9.2
	0 108 1.0	108 1.0	1.0		0.2	22.5	2.6	- 48	6.2	0.9	3.2 1.	-	I	10.2	1.2	1.7	I	I	I	I	I	I
2.6 2.6 2.6 3.1 31.4 80.0 0.7 38.0 13.1 34.6 0.1 74.0 1.4 0.6 - - 6 6 37 9.4 14.1 1.6 19.1 34.9 0.2 66.5 51.0 50.9 0.3 56.3 0.1 - - - 6 6 37 - 20.7 0.5 - 7.2 - - 24.1 23.0 8.1 0.4 - 0.3 5 36.3 3.7 9.4 14.1 1.6 19.1 34.9 0.2 66.5 51.0 50.9 0.3 56.3 0.1 - - 0.3 5 3.7 9.4 14.1 1.6 19.1 34.9 0.2 66.5 51.0 50.9 0.3 56.3 0.1 - - 0.3 -	0 207 8.1	207 8.1	8.1		4.7	31.3	3.8	28.5	2.4	35	5.1 20.	3 75	8	4 86.5	0.4	0.8	I	I	0.2	I	8.0	4.3
- 20.7 0.5 - 7.2 - - 24.1 23.0 8.1 0.4 - - 0.3 - - - 3.6 9.4 14.1 1.6 19.1 34.9 0.2 66.5 51.0 50.9 0.3 56.3 0.1 - - 0.1 - 12.3 2.7	0 176 7.4 00 166 12.3	176 7.4 166 12.3	7.4 12.3		2.6 9.4	26.6 14.1	3.1 3.1 1.6 1	31.4 8 19.1 3.	0.0 (4.9 0	0.7 36 0.2 66	8.0 13. 3.5 51.	1 34 0 50	9 0.0	1 74.0 3 56.3	1.4 0.1	0.6	11	11	0.1	1 1	6.9 12.3	3.7 2.7
9.4 14.1 1.6 19.1 34.9 0.2 66.5 51.0 50.9 0.3 56.3 0.1 0.1 - 12.3 2.7	B (4) 82 -	82 –	I		I	20.7	0.5		7.2 -		1	24	1 23.	0 8.1	0.4	I	I	0.3	I	I	I	3.6
	6 166 12.3	166 12.3	12.3		9.4	14.1	1.6	19.1 3	14.9 C	J.2 66	3.5 51.	0 50	9 0	3 56.3	0.1	I	I	I	0.1	1	12.3	2.7

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol-free; **0** = low

School #3, A la carte

									Macronu	trients					~	licronut	rients		/lean # s	ervings	per pyra	mid gro	dn	Fat & sı	ıgar
	^D ackage	Numb-		FMNV ^{&} FDA			Me	an amou	nt per ite	E			Perce	nt of from		10							2	lean amo item	unt per
	size	er1	9012	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total fat	Satura- ted fat	A (RE)	und Und	(mg)	lion mg)	rain ve	det-	uit De	airy	eat ³ Dis	scre- nary 5 -at 8	vdded ugars (tsp)
Frozen desserts																									
choc	4.0 oz	49	09.0	88	144	5.6	3.2	21.8	2.6	19.8	36.4	0.6	35.2	20.3	52.6	0.3	0.09	0.3	0.5			0.2		5.6	3.0
vanilla	4.1 oz	па	0.60	88	144	5.6	3.2	21.8	2.6	19.8	36.4	0.6	35.2	20.3	52.6	0.3	0.09	0.3	0.5	I		0.2		5.6	3.0
bar	5.0 oz	14	0.90	9	278	20.6	15.8	23.6	2.7	32.0	58.4	0.3	66.5	51.0	85.1	0.4	94.2	0.2	I	I		0.2	N 1	9.0	4.6
Cream cone	4.0 oz	45	0.60	9	217	13.6	5.6	21.9	4.4	26.4	54.7	1.4	56.5	23.1	70.3	0.4	00.3	0.7	0.2	I		0.2	0.2	3.4	3.4
bar	2.0 oz	72	0.90	00	164	11.4	7.1	14.5	2.6	16.0	40.5	0.5	62.6	39.1	42.0	0.2	59.2	0.2	I	I		0.1	-	1.0	2.8
orrawberry eclair ice cream bar	3.0 oz	36	0.60	9	176	7.4	2.6	26.6	3.1	31.4	55.0	0.7	38.0	13.1	34.6	0.1	74.0	1.4	0.6	I		I	1	6.9	3.7
Snacks Butter popcorn Cheese popcorn	1.0 oz 1.0 oz	22 43	0.60		143	8.4 8.4	2.3 2.3	14.8 14.8	2.9 2.9	5.5	237.6 237.6	2.8 2.8	52.8 52.8	14.2 14.2	10.5 10.5	1 1	23.5 23.5	0.5 0.5	1 2 1 2 2				1 1	7.2	1.1
chips	1.0 oz	55	0.60		152	9.8	3.1	15.0	2.0	I	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0		I	1	8.9	I
sour cream & onion potato chips	1.0 oz	37	0.60		152	9.8	3.1	15.0	2.0	l	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0	·	I		8.9	I
Average of all items Unweighted	1 1	27 3109	0.71 0.73		203	8.3 12.3	3.5 3.7	31.0 28.6	2.4 2.7	13.2 15.1	127.6 162.1	0.9 1.1	33.4 35.1	15.2 11.2	45.7 63.6	21.4 4.7	46.5 29.6	0.8 1.0	0.5 0.6	-			-	7.4 1.5	4.6 4.3
ı		1]]																					

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
K indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; 0 = low calorie; 0 = cholesterol-free; 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low calories. C = content set in the set is 0 = low calorie; 0 = cholesterol-free; 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low calories, 0 = low calories, 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low cholesterol-free; 0 = low calories, 0 = low cholesterol-free; 0 = low calories, 0

School #3, School store

								W	acronut	rients					2	icronuti	ients	~	llean # s	ervings	oer pyrai	nid grou	đ	Fat & su	gar
	Package	-dmb-		FMNV & FDA			Mea	n amoun	t per iterr	_			Percen calories	t of from									Me	an amou item	int per
	size	e	2106	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- hydr- P ate (g)	(g)	thole- tterol	Sodi- E um (mg))ietary fiber (g)	Total S fat	atura- ted fat	(RE)	iko∭ g@	(mg)	ron mg)	rain <	Die Fi	uit Da	iry Me	at ³ Disc tion (₍	sre- ary Ac (1) (1	dded ugars tsp)
Baked goods/dessert Frosted Poptarts, brown sugar/cinn	3.7 oz	18	1.00	0	409	10.6	1.6	74.0	4.9	1 4	36.0	N N	23.4	3.5	98.6	0.3	27.1	3.6	3.2	-		I		ကဲ	7.7
Frosted Poptarts, strawberry	3.7 oz	39	1.00	0	409	10.6	1.6	74.0	4.9	4	36.0	2.2	23.4	3.5 2	98.6	0.3	27.1	3.6	3.2	-	- 1.0		-	¢.	7.7
Beverages DejaBlue Purified water	20.0 oz	49	1.00	× 0 0 0	I	I	I	I	I	I	17.8	I	I	I	I	I	11.8	I	I	I	'				I
Hawaiian Punch Green Berry	20.0 oz	55	0.75	2 6 4	290	I	I	73.5	I	÷ I	36.0	0.6	I	I	6.2	82.9	49.4	1.3	I	-				-	0.9
Juicy Red	20.0 oz	88	0.75	2 6 4	290	I	I	73.5	I	÷	36.0	0.6	I	I	6.2 1	82.9	49.4	1.3	I	-	- 4.0			-	0.9
Bread/grain products Cheese crackers w/ peanut butter Chex Mix Traditional Cinnamon Toast	1.4 oz 1.8 oz	20	0.50	0 0	194 213	9.6 6.6	2.1	22.0 34.2	5.0	0.8 1.3 4	57.6 62.0	2.8	44.7 28.0	9.9 6.4	6. I	0.2	38.7 31.1	1.1 8.1	0.7	1 1			5.5 5.8	و، ن	2.8 0.7
Bar	1.6 oz	ო	0.75	0	192	4.6	0.9	33.2	0.9	÷	98.7	0.2	21.5	4.3	37.7	6.3 2	50.0	5.4	1.0	I			ю 	.5	4.1
Honey Nut Cheenos Milk & Cereal bar Rice Krispie Treat	1.4 oz 0.8 oz	ω <u>1</u>	0.75 0.50	0 0 0	171 85	4.0 2.2	0.8 0.4	29.0 16.2	6.0 0.7	 I I	73.9 96.9	1.0	21.1 23.7	4.2	20.5 67.2	5.5 3.1 3.1	50.0 1.8	5.4 0.4	0.9 0.5	1 1			ο ι	۱ .۲.	3.6 2.0
Candy Air Heads watermelon Big Red gum- 17s Big Red gum- 55	0.6 oz 17.0 stk 5.0 stk	74 4 15	0.25 1.00 0.25	6 8 8 6 9 8 6 9 9	61 145 43	1.3 2	0.3	13.0 13.4 13.4	0.2	1 1 1	16.6 3.1 0.9	1 1 1	18.8 1.1 1.1	4.3 0.1 0.2	1 1 1	6.2	2.7 4.8 1.4	111	1 1 1	1 1 1				<i>с</i> і	3 3 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
Doublemint gum- 5s	0.7 oz 17.0 stk 5.0 stk	25 5 14	0.25 1.00 0.25	x x x 0 0 0 0 0 0 0 0 0	69 145 43	0.2	111	18.1 45.6 13.4	111	1 1 1	7.0 3.1 0.9	1 1 1	122	- 0.1 0.2	111		0.6 4.8 1.4	1 1 1							4.3 4.3 1.3

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol-free; **0** = low

— Continued

Ð
Ľ.
0
÷
S
_
0
Ō
č.
7
ň
"
-
Ω
Ŧ
-
Ō
ο.
2
ΰ
ñ
_

ıgar	unt per	Added tugars (tsp)	I	6.0	3.0 4.3	1.3 4.4 2.2	I	4.4 12.9 12.1 12.1	3.7 4.3 1.3	11	111	I
Fat & sı	Vlean amo item	iscre- onary S Fat (g)	I	I	1 1	16.9 	I	13.5 2.1 - 4.7 - 1	13.2 -	9.8 6.1	6.1 10.3 4.9	8.9
dno	2	feat ³ Di	1	I		.0.3	I	. 0		1 1		I
amid gr		Dairy	I	I	1 1	0.2	I			1 1	111	I
s per pyr		Fruit	I	I	1 1		I			1 1	111	I
servings		'eget- able	I	I	1 1		I		111	1 1	111	1.0
Mean #		Grain	I	I	1 1		I	0.1	1. 1. 1. 6.	<u>د</u> ا دا دا دن	1.5 1.5	I
		(mg)	1	I	11	- 0.6	I	0.5	0.5	0.8 0.4	0.4 0.5 1.5	0.5
rients		um (mg)	6.0	6.7	3.4 4.8	1.4 49.8 0.8	5.6	35.1 - 1.4 0.5	51.0 4.8 1.4	18.5 43.7	43.7 45.0 5.5	6.8
Aicronut	121	(mg)	I	I	1 1	0.3	I	41.2 - 23.4	0.2	1 1		8.8
-	đ.	VII А (RE)	T	I	11	11.8	I	8 9	14.2	10.8 5.7	5.7 3.2 -	I
	nt of from	Satura- ted fat	0.2	0.1	0.2	0.2 18.0	0.2	18.5 1.9 1.9 1.3	16.1 0.1 0.2	10.7 9.0	9.0 7.6 23.8	18.4
	Percer calories	Total fat	1.3	1.1	22	1.1 45.8 -	1.3	52.1 9.7 1.1 18.8	44.0 1.1 1.1	55.9 47.1	47.1 55.8 56.2	58.1
		Dietary fiber (g)	I	I	11	1.7	I	1 4	0.6	0.4 1.8	1.8 1.7 0.5	1.3
		Sodi- um (mg)	2.1	4.4	2.2 3.1	0.9 23.7 15.0	2.0	142.7 9.8 0.9 63.0 6.5	109.4 3.1 0.9	334.9 149.7	149.7 223.3 608.6	168.4
itrients	E	Chole- sterol (mg)	I	I	11	- 4	I	0 1 2.3	2.1		13.2	I
Macronu	nt per ite	Protein (g)	I	I	11	- 4.7 -	I	4.6 0.1 0.6	2.6	2.0	2.0 9.1	2.0
-	an amoui	Carb- ohydr- ate (g)	28.4	64.3	32.2 45.6	13.4 29.8 34.7	26.5	24.6 55.8 13.4 49.3 16.7	37.2 45.6 13.4	17.2 17.8	17.8 20.2 3.0	15.0
	Mea	Satura- ted fat (g)	I	I	11	5.1	I	5.0 0.5 1.1	1.1	2.1 1.4	1.4 1.6 3.0	3.1
		Total (fat (g)	0.1	0.2	0.1 0.2	- 12.9 -	0.1	14.1 2.7 - 4.9	13.8 0.2 -	11.0 7.4	7.4 11.8 7.0	9.8
		Calori- es	80	205	102 145	43 254 134	75	244 249 43 232 63	283 145 43	177 142	142 191 113	152
	FMNV ^{&} FDA	label criteri- a ²	X 234	X 234	X 234 X 234	X 8 8 9 9 9 9	X 234	0 0 0 0 0 0 0 0 0 0	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	0 0	0 0	
		Price	1.00	1.00	1.00 1.00	0.25 0.75 0.75	1.00	0.75 0.75 0.25 0.25 0.25	0.75 1.00 0.25	0.50 0.50	0.50 0.50 1.00	0.50
	-dmb-	er1	Q	N	ب 0	4 1 6	24	20 7 8 8 38 8 4	36 12 61	ത ഗ	110	10
	ackage	size	15.0 stk	12.0 pc	12.0 pc 17.0 stk	5.0 stk 1.7 oz 1.3 oz	14.0 pc	1.6 oz 2.2 oz 5.0 stk 2.1 oz 0.6 oz	2.0 oz 17.0 stk 5.0 stk	1.1 oz 1.0 oz	1.0 oz 1.3 oz 1.0 oz	1.0 oz
	<u> </u>		Candy Extra spearmint gum, sugar-free	Cinnafusion gum	Mintensity gum Juicy Fruit gum- 17s	Juicy Fruit gum- 5s M&M's Peanut Mentos mixed fruit	Orbit peppermint gum, sugar-free	Skittles Fruit Candy	I wx caramel cookle bar Winterfresh gum- 17s Winterfresh gum- 5s	Snacks Cheetos Crunchy Doritos Cooler Ranch	Pontos Nacro Cheesier Fritos corn chips Giant Sim Jim	chips

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; 0 = low calorie; 0 = cholesterol-free; 0 = low cholesterol; 0 = sodium-free; 0 = low sodium;
Also includes legumes, soybean products, nuts, and seeds.
Source: Cafeteria Survey and and Inventories of Competitive Foods.

store
_
Q
0
Ć.
$\overline{0}$
Ñ
ť3,
0
ž
2
2
C
S

								-	Aacronu	trients					Σ	icronutri	ents	2	ean # se	rvings p	oer pyra	mid gro	dn	Fat & su	gar
	Package	Numb-	ć	FMNV & FDA			Me	an amour	t per iter	۶			Percent calories 1	of Iom									Σ	ean amou item	int per
	size	er1	РЛСе	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total Si fat	atura- ted fat	RE)	j − c li o li b		ug)	ain Vec	e et		airy	eat ³ Dis tio	cre- hary Ac at SL (1	dded ugars tsp)
Snacks Lays Classic potato	C	L			C L	c	Ċ	C	c			c T	Ċ			c	c	L	•	c				c	
cnips	1.U 0Z	ი	06.0		201	а. С	ς. Γ	0.61	0.2	I	108.4	<u></u>	DQ.1	ά.4	I	ά.α	0.0	c.0	-	, S	1	I	-	а.ч	I
Average of all items Unweighted	I	39	0.67		158 176	4.2 0	+- ¢	31.2 25 7	1.7	0.7	119.9	0.6	19.7	5.1	25.7	12.4	8.3	6.0 T	. 5.0			I	1	3.7	3.8 6
	I	667	<u>6</u> .0		2	0.0	2	1.00	<u>.</u>	c.o	0.10	0.0	0.0	4.r	- 0.	- 00	0.0	Ņ.	- 			1	1	0.0	0.0

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; • = low calorie; 0 = cholesterol-free; • = low cholesterol; 0 = sodium-free; • = low sodium;
Also includes legumes, soybean products, nuts, and seeds.
Sales includes legumes, soybean products, nuts, and seeds.

4
Ħ
θ
.⊆
~
$\overline{\mathbf{a}}$
ĕ
Ë
<
D
ĩ
.=
σ
È
ā
S.
-
~
Ψ.
7
×
2
2
C
S

								-	Macronu	trients						Micronu	trients		Mean #	serving	s per py	ramid g	roup	Fat &	sugar
	Package	Numb-		FMNV & FDA			Me	an amour	nt per ite	ε			Percer calories	nt of from		10,								Mean ar it	nount per em
	size	er1	L	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total ^S fat	satura- ted fat	A (RE)	(mg)	um (mg)	(mg)	Grain	/eget- able	Fruit	Dairy	Meat ³	Discre- tionary Fat (g)	Added Sugars (tsp)
Bavaradas																									
Veryfine Apple cherry berry punch	11.5 oz	6	0.70	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	I	3.6	105.1	28.4	0.8	I	I	0.2	I	I	I	9.2
	11.5 oz	108	0.70	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	I	3.6	105.1	28.4	0.8	I	I	0.2	I	I	I	9.2
Veryline Glacial Grape Chillers	11.5 oz	30	0.70	2 6 4	162	I	I	41.3	I	I	21.6	I	I	I	I	60.09	100.0	9.0	I	I	0.2	I	I	I	0.0
Verfine Orange Juce (100%)	11.5 oz	12	0.70	2 6 4	150	0.5	I	35.3	2.1	I	7.2	0.7	3.0	0.4	64.4	123.1	28.6	1.6	I	I	1.9	I	I	I	I
veryline Fineappie orange drink	11.5 oz	36	0.70	2 6 4	180	I	I	42.4	4.7	I	10.8	0.4	I		190.5	80.9	18.0	1.0	I	I	1.0	I	I	I	4.7
veryline orrawberry banana punch	11.5 oz	36	0.70	2 6 4	167	I	I	42.3	I	I	78.1	0.4	I	I	3.6	105.1	28.4	0.8	I	I	0.2	I	I	I	9.2
Average of all items Unweighted	1 1	6 231	0.70 0.70		165 167	1 1	1 1	41.0 41.8	1.1 0.8	1 1	45.6 56.6	0.3	0.5 0.2		44.3 35.4	96.5 96.4	38.6 36.1	0.9	11	1 1	0.6 0.4	1 1	11	1 1	6.9 8.0

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Data are not available. The number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
X indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol-free; **0** = low cholesterol-free; **0** = low calorie; ③ = cholesterol-free; **0** = low cholesterol-fr

School #3, Vending Machine $#2^{4,7}$

								-	Macronu	Itrients					E	licronut	rients		Mean # (ervings	s per pyı	amid gı	dno	Fat &	sugar
	ackage	Numb-		FMNV ^{&} FDA			Me	an amou	nt per ite	E			Perce calories	nt of s from	t,	1.1								Mean an ite	iount per m
	size	er1	1100	label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total fat	Satura- ted fat	A (BE)	C C (mg)	um (mg)	lron (mg)	Grain	able	Fruit	Dairy	Meat ³ [Discre- ionary Fat (g)	Added Sugars (tsp)
Baked goods/dessert Li'l Deb Honey Bun I'' Dob Comoor	4.0 oz	18	0.75		430	17.3	3.3	64.2	5.7	60.0	399.6	2.2	36.3	6.9	82.5	1.8	68.3	1.5	2.1	I	0.1	I	I	15.2	9.0
Lin Deb Calinea Cream Pie Li'l Deb Zebra Cake	3.8 oz 3.6 oz	0 0 0	0.75 0.75	0 0 0	524 460	23.9 22.0	4.9 5.0	73.4 65.2	5.2 3.2	9.6 16.3	129.6 220.0	2.1 0.4	41.1 43.0	8.4 9.8	10.1 5.1	0.1	78.0 45.9	3.2 1.3	2.5 1.6	1 1	1 1	1 1	11	21.4 10.0	10.0 10.7
Snacks Butter popcorn	1.0 oz	12	0.60		143	8.4	2.3	14.8	2.9	2.5	237.6	2.8	52.8	14.2	10.5	I	23.5	0.5	1:2	I	I	I	I	7.2	I
Cheese curls	1.0 oz 1.0 oz	8 7	0.60 0.60	0	157 143	9.8 4.4	1.9 2.3	15.3 14.8	2.2	1.1 2.5	297.7 237.6	0.3 2.8	55.9 52.8	10.7 14.2	9.6 10.5	1 1	16.4 23.5	0.7 0.5	1.1	1 1	1 1	1 1	1 1	8.7 7.2	11
Corn chips	1.4 oz	φç	0.60	00	210 142	13.0 7.4	1.8	22.2 17 8	2.6	1 1	245.6 149.7	1.9 8	55.8 47.1	7.6	3.5 7 7	1 1	49.5 43.7	0.5	1.6 6.1		1 1		1 1	11.4 6.1	1 1
Salt & Vinegar potato chips	1.0 oz	<u>ى</u> و	0.60)	152	9.8	3.1 .1	15.0	2.0	I	168.4	. .	58.1	18.4	5 1	8.8	6.8	0.5	2 1	1.0	I	I	I	8.9	I
Sour cream & onion potato chips [2]	1.0 oz	24	0.60		152	9.8	3.1	15.0	2.0	I	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0	I	I	I	8.9	I
Average of all items Unweighted	1 1	10 118	0.65 0.65		251 248	13.0 12.6	2.9 2.9	31.8 31.6	3.1 3.2	9.2 11.7	225.4 230.5	1.7 1.8	50.1 49.8	11.8 12.3	13.8 17.8	2.0	36.2 35.5	1.0	<u>+</u> + vi vi	0.2 0.2	1.1	1.1	1.1	10.5 10.6	3.0 2.9

Value is exactly zero.
Value is exactly zero.
Data are not available. The number sold was unavailable for some sources; number sold is 'na' for individual items within a source if information was obtained during validation but not reported on the survey.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
A indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; **0** = low calorie; ② = cholesterol-free; **0** = low cholesterol-free; **0** = low calories (and the survey is a sodium-free; **0** = low calories (and the survey and and Inventories of Competitive Foods.
</ul

Table F.1—Competitive Food and Beverage Products Available OUTSIDE the School Day, By School and Source

⁻at & sugar	an amount per item	re- Added ary Sugars t (tsp)		9.0	9.0	0.0	0.6	9.0	0.6	2 10.5	5.3	8.7	I
L	Me	Disci tione Far		1	I	I	1	I	I	0	I	1	
i group	Meat			I	I	I	I	I	I	I	I	I	I
Mean # servings per pyramio			I	I	I	I	I	I	I	I	I	I	
	Fruit			I	I	I	I	I	I	1.4	1.9	0.4	I
			I	I	I	I	I	I	I	I	I	I	
	Grain			I	I	I	I	I	I	I	I	I	I
Micronutrients	lron (mg)			0.3	0.3	0.3	0.3	0.3	0.3	0.8	1.2	0.5	I
	Calci- um (mg)			I	I	I	I	I	I	29.6	31.1	7.6	I
	(mg)			I	I	I	I	I	I	22.7	61.8	10.6	I
-	Vit A (RE)			I	I	I	I	I	I	3.8	34.5	4.8	I
Macronutrients	nt of s from	Satura- ted fat		I	I	I	I	I	I	0.2	0.4	I	I
	Mean amount per item calorie	Total fat		I	I	I	I	I	I	1.7	1.7	0.4	I
		Dietary fiber (g)		I	I	I	I	I	I	0.3	1.2	0.2	1
		Sodi- um (mg)		240.0	240.0	240.0	240.0	240.0	240.0	15.8	15.2	183.9	I
		Chole- sterol (mg)		I	I	I	I	I	I	I	I	I	I
		Protein (g)		I	I	I	I	I	I	1.0	2.0	0.4	I
		Carb- ohydr- ate (g)		37.8	37.8	37.8	37.8	37.8	37.8	71.1	62.7	45.1	I
		Satura- ted fat (g)		I	I	I	I	I	I	I	0.1	I	ı
		Total fat (g)		I	I	I	I	I	I	0.5	0.5	0.1	I
		Calori- es		150	150	150	150	150	150	272	250	178	I
	FMNV FDA FDA		24	24	24	24	2 4	24	2 6 4	2 6 4			
Price				1.00	1.00	1.00	1.00	1.00	1.00	1.25	1.25	1.06	na
	-dmbN		na	na	na	na	na	na	na	na	ω	na	
	ackage		20.0 oz	0.0 oz	0.0 oz	0.0 oz	20.0 oz	20.0.0	0.0 oz	0.0 oz	I	ı	
			Beverages Gatorade Cool blue	Catorade Strawberry	kiwi	melon	rush 20	Gatorade High tide 20	Gatorade Fruit punch 20	Tropicana Twister apple raspberry 2 Tronicana Twister	orange strawberry 2	Average of all items Unweighted	Weighted

School #3, Vending Machine #3 (outside school day)

Value is exactly zero. na Data are not available.
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
A indicates fool of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; • = low calorie; 0 = cholesterol-free; • = low cholesterol-free; • = low calories (a = cholesterol-free; • = low calories).
Also includes legumes, soybean products, nuts, and seeds.
Source: Inventories of Competitive Foods.

Mean amount per Added Sugars (tsp) 11.3 12.4 10.5 16.0 12.9 16.1 Fat & sugar I item Discretionary 0.2 0.1 (g) I. ī ī T 1 1 Meat³ Mean # servings per pyramid group T ī I I I T 1 1 Dairy 1 1 I I T T I 1 Fruit 5. 4 0.4 0.5 1.0 0.8 ı Veget-able 1 1 T I. 1 Т I T Grain 1 1 I ı ī 1 I T (mg) 0.6 0.8 <u></u>. 4 0.5 0.8 I I Micronutrients Calci-um (mg) 29.6 27.3 -11.9 30.0 49.4 21.4 21.2 125.0 150.0 150.0 150.0 150.0 150.0 u Cit I I 63.0 (BE) A ^Lİt 348.0 3.8 6.2 18.1 1.9 I Satura-0.2 0.2 0.1 ted fat calories from ı ī. T 1 1 Percent of Total fat 0.8 I 1.8 1.7 0.1 0.5 I ı. Dietary 0.4 1 0.6 0.3 0.6 0.7 (g) 0.2 I Sodi-148.8 175.0 175.0 175.0 17.8 175.0 175.0 (mg) I Macronutrients Cholesterol (mg) 1 1 I I I I I I Mean amount per item Protein 6. 1.0 0.5 1.0 0.7 (g I ī Carb-ohydr-64.5 71.1 73.5 85.0 81.1 76.2 (g) I ted fat Satura-Ø I I I T I I 1 1 fat (g) 0.6 0.3 0.5 0.4 0.2 I ī Calori-253 es 306 272 290 327 320 I & FDA label criteri-a² FMNV X 0 0 0 0 0 0 Q (4) © © © Price 1.00 1.00 na 1.0 9.1 1.00 1.00 1.00 , 0 1 1 9 Package Numb-size er¹ na na na na na na 20.0 oz 20.0 oz 20.0 oz 20.0 oz 20.0 oz 20.0 oz 1 1 Aquafina water Strawberry melon FruitWorks Tangerine Average of all items FruitWorks Peach FruitWorks Fruit FruitWorks Pink papaya lemonade punch ... citrus FruitWorks Beverages

School #3, Vending Machine #4 (outside school day)

na Data are not available. Value is exactly zero.

Value is exactly zero.
 na Data are not available.
 1 Number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
 1 Number column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
 2 x indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; 0 = low calorie; 0 = cholesterol-free; 0 = low cholesterol; 0 = sodium-free; 0 = low colosition;

3 Also includes legumes, soybean products, nuts, and seeds. Source: Inventories of Competitive Foods.
School #3, Vending Machine #5 (outside school day)

sugar	nount per em	Added Sugars (tsp)		16.1	16.7	16.7	16.6	7.4	I	16.7	I			Σ. LL	I
Fat &	Mean an ite	Discre- tionary Fat (g)		I	I	I	I	I	I	I	I			I	I
roup		Meat ³		I	I	I	I	I	I	I	I			I	I
rramid g		Dairy		I	I	I	I	I	I	I	I			I	I
s per p)		Fruit		I	I	I	I	I	I	I	I			I	I
serving		/eget- able		I	I	I	I	I	I	I	I			I	I
Mean #		Grain		I	I	I	I	I	I	I	I			I	I
		lron (mg)		0.2	0.4	0.4	0.4	0.1	0.2	0.4	I		0	0.3	I
trients		um (mg)		18.6	12.4	12.4	12.3	0.3	24.0	12.4	11.9			13.0	ı
Micronu	727	(mg)		I	I	I	I	I	I	I	I			I	I
	TUA .	A (RE)		I	I	I	I	I	I	I	I			I	I
	nt of s from	Satura- ted fat		I	I	I	I	I	I	I	I			I	I
	Perce calorie	Total fat		I	I	I	I	I	I	I	I			I	I
		Dietary fiber (g)		I	I	I	I	I	I	I	I			I	I
		Sodi- um (mg)		24.8	68.2	68.2	67.5	17.2	36.0	68.2	17.8			46.0	I
utrients	m	Chole- sterol (mg)		I	I	I	I	I	I	I	I			I	I
Macron	int per ite	Protein (g)		I	I	I	I	I	0.6	I	I			I	I
	an amou	Carb- ohydr- ate (g)		64.5	64.5	64.5	63.9	31.3	0.6	64.5	I			44.2	I
	Me	Satura- ted fat (g)		I	I	I	I	I	I	I	I			I	I
		Total fat (g)		I	I	I	I	I	I	I	I			I	I
		Calori- es		254	248	248	246	120	9	248	I		ļ		I
	FMNV & FDA	label criteri- a ²		X 2 6 4	X 264	X 264	X 2004	X 264	0264	X 264	X 02	8 4			
				1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00		00	00.1	na
	-dmuN	e.		na	na	na	na	na	na	na	na		c	x	na
	ackage	size		20.0 oz	20.0 oz	20.0 oz	20.0 oz	20.0 oz	20.0 oz	20.0 oz	20.0 oz			I	I
	<u> </u>							1 Tea 2					ns		
			Beverages	Pepsi	Pepsi Blue	Mountain Dew	7up	Lipton Brisk Iceo	Diet Pepsi	Dr. Pepper	Aquafina water .		Average of all iter	Unweighted	Weighted

School #3, Vending Machine #6 (outside school day)

Fat & sugar	san amount per item	cre- lary Sugars at (tsp)	2.6	- 4.4	- 7.8	- 9.9	- 9.9	- 9.7	1		7 2	+	
d	Me	at ³ Disc tion (g		1		I	1	1	1				
id groul		y Mea		I	I	I	I	I	Ι		l	I	
pyram		Dain		I	I	I	I	I	I			I	
ngs per		Fruit	1	I	0.3	I	I	I	I		I	I	
# servi		Veget- able	1	I	I	I	I	I	I		l	I	
Mean		Grain	I	I	I	I	I	I	I			I	
		lron (mg)	0.1	I	0.8	0.3	0.3	0.1	0.1		0	4. 0	
trients	:- 	um (mg)	11:2	0.2	12.9	7.4	7.4	11.2	14.4		Ċ	2.V	
licronu	101	C C M		I	19.2	I	I	I	I		r c	i.	
Z	100	A (RE)		I	10.8	I	I	I	I		u T	<u>.</u>	
	t of from	atura- ted fat	1	0.1	0.1	I	I	I	I			I	-
	Percen calories	Total S fat		I	1.0	I	I	I	I		•		
		lietary fiber (g)	1	I	0.4	I	ı	ı	I			1	
		Sodi- um (mg)	14.9	10.3	12.2	40.5	40.5	14.9	21.6		100	- 22	
trients	E	Chole- sterol (mg)		I	I	I	I	I	I		I	I	
lacronu	t per iter	rotein (g)		I	0.3	I	I	I	0.4			I	
N	n amoun	Carb- hydr- F ate (g)	38.7	18.8	51.0	38.3	38.3	38.7	0.4		0.00	07.0	
	Mea	atura- ed fat (g)		I	I	I	I	I	I		I	I	
		Total S fat t (g)		I	0.2	I	I	I	I			I	
		alori- es	152	72	196	147	147	152	4		101	14	
	:MNV & FDA	label a ² C	0 0 0	2 6 4	2 8 4	264	264	264	80 X	8 4			
		- 0 	0.75 ×	0.75 ×	0.75	0.75 ×	0.75 ×	0.75 ×	0.75		0 76	c	-
	-qun	er1	a D	na	na	na	na	na	na		٢	-	
	kage N	ize	ZO 0	ZO 0.	20 O.	.0 oz	.0 oz	.0 oz	.0 oz			1	
	Pac	ŝ	everages Pepsi	Lipton Brisk Iced Tea 12. Country Time	lemonade 12.	Slice Orange 12.	7up 12.	Dr. Pepper 12.	Diet Pepsi 12.		verage of all items	Oliweigineu	

School #3, Vending Machine #7 (outside school day)

School #3, Vending Machine #8 (outside school day) 4

								ž	acronutr	ients					Mic	ronutrier	ıts	Mea	n # servi	ngs per l	pyramid	group	Fat 8	t sugar
<u> </u>	^o ackage	-dmb-	C	FMNV & FDA			Mea.	n amount	per item			ö	Percent or alories fro	` چ چ									Mean a it	mount per em
	size	er1	Price	label criteri- a ²	Calori- es	Total (fat (g)	Satura- ted fat (g)	Carb- bhydr- Pi ate (g)	rotein C (g) (hole- terol (tiodi- bodi- Die mg) (etary iber (g)	otal Sat fat fa	ed (F		g) (mg) Grair	Veget- able	Fruit	Dairy	Meat ³	Discre- tionary Fat (g)	Added Sugars (tsp)
d goods/dessert m. Amos Oatmeal raisin cookies	2.0 oz	na	0.55	0	255	10.3	2.6	39.0	3.5	- 21	17.2	1.6	6.2			.2 21.	0	1.3	I	0.2	I	I	8.9	4.2
nilla flavored wafers ott's Raspberry	2.1 oz	na	0.75	0	308	14.6	2.2	42.2	2.5	ω I	38.6	0.4	2.8	4.0		- 10.	8	1.6	I	I	I	I	13.0	6.1
suctoread cookies so cookies ple danish	2.0 oz 2.0 oz 4.0 oz	na na	0.75 0.75 0.75	0	270 268 421	12.3 11.7 21.0	3.8 2.1 5.5	37.2 39.9 54.2	3.1 6.1 6.1	27.8 23 - 34 23.8 40	36.3 12.5 11.4	2.2 4 3 4	0.9 9.3 11 11	2.8 7.0 1.8 1.8 1.8	3.1 4	- 30. -4 14.	0 - 0	2.1.2	1 1 1	- 0.5	1 1 1	1 1 1	10.6 10.4 18.7	4.1 5.7 7.1
creme sandwich	4.5 oz 1.7 oz	na na	na 0.75	0	302 197	14.5 3.0	2.9 0.5	41.3 39.8	3.1 3.2	- 28 0.5 15	35.5 52.1	1.8 4 0.6 1	3.1 8	2.0 2.1 0	3.4 1 0.2 0	.4 12. .2 73.	0 4 	1.6	1 1	0.5	1 1	1 1	12.8 1.6	3.2 5.3
Algrain products tter snap pretzels urdough pretzels	1.5 oz 1.5 oz	na na	na 0.55	9 9 © 0	162 162	1.5 1.5	0.3 0.3	33.7 33.7	9.0 0.0		29.3 29.3	4. 1 . 4. 4.	8 8	<u></u>		 	0 0 	2.6	11	11	11	11	11	11
white cheddar ex Morning Mix eese crackers w/ peanut butter	1.5 oz 1.2 oz 1.3 oz	na na	0.55 0.75 0.65	0 0	214 140 175	10.8 4.4 8.7	4.0 1.0	24.8 22.5 20.0	4.3 3.8 4.5	5.5 42 0.8 30 0.7 32	23.1)3.6 23.9	1.0 1.8 0.4 2.4 4 2.4	5.3 16 8.0 6	6.8 9.9 9.9	1 1 2 8 1	- 64. .8 20.	- 20 - 20 -	2:0			111	0.3	8.8 3.3 8.1	2 0 - 2.6
y stle Butterfinger	2.3 oz	na	0.65	88	314	12.2	6.8	42.9	8.2	0.7 12	29.7	1.6	5.0 19	9.4		- 17.	7 0.	0.1	I	I	I	0.3	12.1	8.2
butter cup butter cup ickers Cruncher &M's Peanut	1.5 oz 1.6 oz 1.7 oz 2.0 oz	na na na	0.65 na 0.65 0.65	© © © © © © © © © ©	230 212 254 272	13.3 12.9 13.9	4.7 5.1 5.1	23.2 26.2 33.6	4.4 3.5 5.7 5.7	7.4.4 7.4 7.4 7.4 7.4 7.4 7.5 7.4 7.5 7.5 7.5 7.5 7.5 7.5 7.5 7.5 7.5 7.5	34.8 17.6 23.7 50.8	4 4 4 5	5.8 16 6.2 16 6.2 16 18 18	0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			0000	1 1 1 1	1 1 1 1	1 1 1 1	0.2	0.2 0.3 0.3	12.7 8.4 16.9	4.7 4.7 3.0 2.0
toppers milk balls stle Crunch with Caramel	1.8 oz 1.5 oz 45.0 g	na na	0.65 na 0.65	0 0 0	246 214 215	13.2 11.4 11.3	7.9 6.9 9.1	31.5 27.3 26.5	3.1 2.7 1.7	0.9	71.9 52.5 57.1	2.1 1.3	8.1 28 7.4 28 3.0	8.8.8 3.3 8.8 8.8 8.8 8.8 8.8 8.8 8.8 8.	5.0		-0 0 0 0	1 0.3	1 1 1	1 1 1	0.2	0.2	13.1 11.4 7.7	5.1 5.4 6.8
	1	1	1	1																				

Value is exactly zero. na Data are not available.
1 Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
2 A indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; ● = low calorie; ② = cholesterol-free; ● = low cholesterol; ③ = sodium-free; ● = low sodium;
3 Also includes legumes, soybean products, nuts, and seeds.
4 Vending machine items occupying multiple slots are indicated with the number of slots shown in [] following the product name.

School #3, Vending Machine #8 (outside school day) 4

								-	Macronu	Itrients					-	Aicronuti	ients		Vlean # s	ervings	per pyra	mid gro	dn	Fat & s	ugar
	Package	-dmb-		FMNV ^{&} FDA			Me	an amou	nt per ite	E			Perce calories	nt of trom	107								2	lean amc iten	unt per r
	size	er-1		label criteri- a ²	Calori- es	Total fat (g)	Satura- ted fat (g)	Carb- ohydr- ate (g)	Protein (g)	Chole- sterol (mg)	Sodi- um (mg)	Dietary fiber (g)	Total fat	Satura- ted fat	VII А (RE)	(mg)	um (mg)	(mg)	irain Vé a	ble F	ruit Da	airy Me	eat ³ Di tic	scre- mary Fat (g)	Added Sugars (tsp)
Candy Hershey's Choc bar w/ almonds	1.5 oz	па	0.65	90	216	14.1	7.0	21.9	3.7	7.8	30.4	2.6	58.9	29.1	5.8	I	92.1	0.7	I	I		0.2	0.1	3.8	4.2
Carefree Spearmint gum	6.0 stk	na	0.55	X 234	32	I	I	11.4	I	I	0.8	I	1.4	0.3	I	I	2.4	I	I	I		I	1	I	I
Icebreaker's Cool mint gum Cinnaburst gum	6.0 stk 7.0 stk	na na	0.55 0.55	× 2 3 4 3 4	32	11	11	11.4 27.1	11	11	0.8 1.7	11	1.4 0.8	0.3	11	1 1	2.4	1 1	1 1	1 1		1.1		1 1	-
Lifesaver's tropical fruit	0.9 oz	na	0.55	X 2 8 4	95	I	I	25.1	I	I	9.7	I	I	I	I	I	0.8	I	I	I		1	1	I	5.9
LITESAVET S peppermint	0.7 oz	na	0.55	X 264	76	I	I	20.0	I	I	7.8	I	I	I	I	I	0.6	1	I	I		I	1	I	4.8
Snacks Fritos corn chips Lay's Chicago Steak	1.3 oz	na	0.55	0	191	11.8	1.6	20.2	2.3	I	223.3	1.7	55.8	7.6	3.2	I	45.0	0.5	1.5	I		I		0.3	I
House potato chips	1.0 oz	na	0.55		152	9.8	3.1	15.0	2.0	I	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0		1	1	8.9	I
chips, jalapeno	1.0 oz	na	0.55		152	9.8	3.1	15.0	2.0	I	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0		I	1	8.9	I
Krunchers	1.0 oz	na	0.55		152	9.8	3.1	15.0	2.0	I	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0		I	1	8.9	I
Criecz Dooues, write cheddar Potato Stix	0.7 oz 1.0 oz	па па	0.55 0.55	©×♥	107 148	6.6 9.8	1.3 2.5	10.4 15.1	1.5 1.9	0.8	202.4 70.9	0.2 1.0	55.9 59.3	10.7 15.3	9.9	- 13.4	11.2 5.1	0.5 0.6	0.7 -	1.0			1 1	5.9 9.0	1 1
Vinegar potato chips	1.0 oz	па	0.55		152	9.8	3.1	15.0	2.0	I	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0			1	8.9	I
potato chips	0.9 oz	na	0.55		133	8.6	2.7	13.1	1.7	I	147.4	1 .1	58.1	18.4	I	7.7	6.0	0.4	I	6.0		I	1	7.8	I
potato chips	1.0 oz	na	0.55		152	9.8	3.1	15.0	2.0	I	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0		I	1	8.9	I
and Onion chips	1.0 oz	na	0.55		152	9.8	3.1	15.0	2.0	I	168.4	1.3	58.1	18.4	I	8.8	6.8	0.5	I	1.0	·		1	8.9	I

Value is exactly zero. na Data are not available.
1 Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
2 A indicates foods of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: ① = calorie-free; ● = low calorie; ② = cholesterol-free; ● = low cholesterol; ③ = sodium-free; ● = low sodium;
3 Also includes legumes, soybean products, nuts, and seeds.
4 Vending machine items occupying multiple slots are indicated with the number of slots shown in [] following the product name.

								Ŭ	acronuti	ients					Mic	ronutrie	nts	Me	an # ser	/ings pe	r pyram	id group	Fat	& sugar
	ackage	-dmb-		FMNV & FDA			Mean	amount	per item			0	Percent a	, o đ		ć							Mean	amount per item
	size	er1	2100	a ²	Calori-	Total Sc fat te (g)	atura-C bid fat ol (g)	barb- hydr- Pi ate (g)	otein C (g) (hole- terol mg)	Sodi- um (mg)	etary ber (g)	otal Sai	ed (F		ng n <u></u>	ی ہے ا میں ا) D Q	in Vege able		Dair	y Meat	3 Discre- tionary Fat (g)	Added Sugars (tsp)
Snacks Cheetos Crunchv	1.5 oz	ua Da	0.75	0	236	14.6	80	6.22	3.2		46.5	0.5 5	5.9	0.7	15	24	-		ا س	1	1	1	13.0	1
Trail mix unsalted [2]	2.3 oz	na	0.75	8	320	22.8	2.9	25.1	0.0	I	5.6	4.3	4.0		. 8.0	.0 54	 -	00	1	0.5	I	÷-	19.8	I
Average of all items Unweighted	1 1	38 na	0.62 na		195 -	9.5	3.1	25.9 -	3.0	- 2.8	38.1 _	1.2	0.1 1: -	6	8.	- 26	. 0.	6 -0 -	9. 1		11	1 1	8.5	2.8

School #3, Vending Machine #8 (outside school day) 4

Value is exactly zero. na Data are not available.
1 Number' column contains the number of individual items sold during the target week. The 'Number' in the 'All items' row is the number of items (unweighted) or the total number of sales (weighted).
2 A indicates food of minimum nutritional value (FMNV). Foods meeting FDA labeling criteria are indicated by: 0 = calorie-free; • = low calorie; 0 = low cholesterol-free; • = low cholesterol; 0 = sodium-free; • = low sodium;
3 Also includes legumes, soybean products, nuts, and seeds.
4 Vending machine items occupying multiple slots are indicated with the number of slots shown in [] following the product name.



Study of Foods in the School Environment

SURVEY OF PRINCIPALS

If you have questions or need assistance, please call 617-349-2674 or 617-349-2661

Spring 2003

The Study of Foods in the School Environment is being conducted for the:

Food and Nutrition Service US Department of Agriculture 3101 Park Center Drive Alexandria, Virginia 22302

By:

Abt Associates Inc. 55 Wheeler Street Cambridge, Massachusetts 02138

This questionnaire asks about all of the possible sources of foods and beverages available to students during the school day. Specifically, there are questions about:

- School lunch and school breakfast programs
- A la carte foods and beverages purchased separately from USDA reimbursable meals
- Vending machines
- Snack bars, including canteens, food carts, kiosks
- School stores
- Off-campus sources that students can visit when not required to be in some specific location like a class, study hall, or school cafeteria.
- Fundraisers selling food or beverages

Please note that the questionnaire will instruct you to skip over sections that do not apply to your school.

A. School Lunch and Breakfast Programs

1. Does your school offer breakfast?

 $\begin{array}{lll} \rho_1 & Yes \\ \rho_2 & No \ (\textit{Skip to }Q \ 5) \end{array}$

2. What time do you usually start serving breakfast?

____: ____a.m.

3. What time does the first class of the day usually start?

____: ____ a.m.

- 4. Where do students eat breakfast? (*Check all that apply.*)
 - ρ_1 Cafeteria
 - ρ_2 School buses
 - ρ_3 Classrooms
 - ρ_4 Outdoors
 - ρ₅ Other (Ple ase specify): _____

5. What (times are/time is) your lunch period(s)?

From	:	to	:
From	:	to	:
From	:	to	:
From	:	to	:

- 6. Where do students eat lunch? (*Check all that apply.*)
 - ρ₁ Cafeteria (*Specify number of cafeterias:* ____)
 - ρ_2 Classrooms
 - ρ_3 Outdoors
 - ρ₄ Other (*Please specify*): _____
- 7. Are students allowed to leave the area where meals are served after a certain point during their lunch period, for example, after the first 15 minutes?
 - ρ_1 Yes, all students may leave (*Skip to Q 8*)
 - ρ_2 Yes, some students may leave (either with or without special permission)
 - ρ_3 No, all students must stay in the area (*Skip to Q 8*)

7a. Which grades are allowed to leave after a set period? (Circle all that apply.)

- K 1 2 3 4 5 6 7 8 9 10 11 12
- 8. Are any students allowed to leave the area where meals are served **at any time during** their lunch period?
 - ρ_1 Yes, all students may leave (*Skip to Q 9*)
 - ρ_2 Yes, some students may leave (either with or without special permission)
 - ρ_3 No, all students must stay in the area (*Skip to Q 9*)
 - 8a. Which grades are allowed to leave at any time? (Circle all that apply.)

K 1 2 3 4 5 6 7 8 9 10 11 12

- 9. What types of payment are accepted for food and beverages sold through the school lunch and breakfast programs? (*Check all that apply.*)
 - ρ_1 Cash
 - ρ_2 Electronic debit
 - ρ_3 Prepaid account (cards, tickets, vouchers)
 - ρ₄ Other (*Please specify*): _____

B. A la Carte

"A la carte" refers to foods and beverages sold in or near the cafeteria(s) by the school food service, other than milk, full reimbursable meals, and vending machine items. For example, a la carte items may be sold on the cafeteria line with reimbursable meals or from a separate line or window within the cafeteria.

- 1. Are there a la carte food or beverage sales at your school?
 - ρ_1 Yes (Go to Q 2)
 - ρ_2 No (Please answer Q 1a, then skip to Section C)
 - 1a. What is the most important reason your school does not offer a la carte food/beverage sales?

- 2. Who decided to sell a la carte foods or beverages at your school? (*Check all that apply.*)
 - ρ_1 School board
 - ρ_2 Superintendent
 - ρ₃ School district business manager
 - ρ_4 Food service director
 - ρ_5 Principal
 - ρ_{6} Cafeteria manager
 - ρ₇ Other (*Please specify*): _____
- 3. Where may students buy food or beverages on an a la carte basis at your school? (*Check all that apply.*)
 - ρ_1 On the cafeteria line, along with USDA meals
 - $\rho_{\rm 2}$ $\,$ Separate line or location within the cafeteria
 - ρ_3 Outside but within 25 feet of the cafeteria
 - ρ₄ Other location (*Please specify*): _____

4. When may students buy a la carte foods/beverages? (*Check all that apply.*)

- ρ_1 No restrictions
- ρ_2 When student takes a reimbursable meal
- ρ_3 When student brings lunch from home
- ρ_4 When student has eaten meal
- ρ_{5} When all students have had the opportunity to take a reimbursable meal
- ρ_{6} Outside breakfast or lunch periods only
- ρ₇ Limited to specific day(s) (*Please specify*): _____
- ρ₈ Other restriction (*Please specify*): _____
- 4a. Are a la carte foods or beverages typically sold to students in the cafeteria **outside** of meal periods?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Q 5*)
 - ρ_3 Don't know (*Skip to Q 5*)
- 4b. Please specify hours when students may buy a la carte items outside of meal periods.

 :	to	_:	a.m. / p.m.
 :	to	<u>:</u>	a.m. / p.m.
 :	to	:	a.m. / p.m.
 :	to	:	a.m. / p.m.

In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, please tell us if your school restricts a la carte items based on price or nutrition criteria.

- 5. Do a la carte items have to fall within any particular price range?
 - ρ_1 Yes
 - ρ_2 No
 - ρ_3 Don't know

If yes, please explain or attach copy of price criteria.

- 6. Do a la carte items have to meet any particular nutrition standards?
 - $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \\ \rho_3 & Don't know \end{array}$

If yes, please explain or attach copy of nutrition criteria.

- 7. In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, does your school **prohibit** any specific foods or beverages from being sold a la carte?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 8)
 - ρ_3 Don't know (*Skip to Q 8*)
 - 7a. What foods or beverages are prohibited? (*Check all that apply.*)
 - ρ_1 Coffee or coffee-based beverages
 - ρ_2 Tea or tea-based beverages (including iced teas)
 - ρ_3 All caffeinated beverages (some coffees, teas, colas, energy drinks)
 - ρ₄ Carbonated soft drinks
 - ρ_{5} Juice drinks (less than 50% fruit juice, with added sweeteners)
 - ρ_{6} Other sweetened beverages (sports drinks, flavored waters with added sweetener)
 - ρ_7 All artificially sweetened beverages (diet soft drinks, Fruit₂0)
 - ρ_8 Fried vegetables (french fries)
 - ρ_9 High-fat snacks or baked goods
 - ρ_{10} High-sugar foods
 - ρ_{11} Candy
 - ρ_{12} Gum
 - ρ₁₃ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages prohibited from being sold a la carte at your school.

- 8. In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, does your school **require** any specific foods or beverages be sold a la carte?
 - $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \ (Skip \ to \ Q \ 9) \end{array}$

- ρ_3 Don't know (*Skip to Q 9*)
- 8a. What foods or beverages are required? (*Check all that apply.*)
 - ρ_1 Bottled water (unflavored, with no added sweeteners)
 - ρ_2 Juice (100% juice)
 - ρ_3 Milk (any type)
 - ρ₄ Milk—skim, nonfat or 1% lowfat
 - ρ_5 Low-fat snacks or baked goods
 - ρ_6 Low-sugar foods
 - ρ_7 Fresh fruit or vegetables
 - ρ_8 Lean meats
 - ρ_9 Whole grains
 - ρ_{10} Diet, "lite," or sugar-free beverages
 - ρ₁₁ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages required to be sold a la carte at your school.

- 9. What types of payment are accepted for food and beverages sold on an a la carte basis at your school? (*Check all that apply.*)
 - ρ_1 Cash
 - ρ_2 Electronic debit
 - ρ_3 Prepaid account (cards, tickets, vouchers)
 - ρ₄ Other (*Please specify*): _____

C. Vending Machines

1. Are there vending machines available for student use located on your school property?

 ρ_1 Yes (Go to Q 2) ρ_2 No (Please answer Q 1a, then skip to Section D)

1a. What is the most important reason your school does not have vending machines for students?

- 2. Who decided to install vending machines in your school? (*Check all that apply.*)
 - ρ_1 School board
 - ρ_2 Superintendent
 - ρ_3 School district business manager
 - ρ_4 Food service director
 - ρ_5 Principal
 - ρ_{6} Cafeteria manager
 - ρ₇ Other (*Please specify*): _____
- 3. How many of the vending machines located on school property are available for **student use**? (*If none, write "0" and skip to Section D.*)
- 4. Who operates (or has primary responsibility for) the vending machines in your school? *(Check all that apply.)*
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
 - ρ_8 Don't know

5. Please answer the following questions regarding the location and type of all vending machines available for student use at your school.

In your school, how many vending machines are located	How many vending mach each location?	ines of each type are available in
a. In the cafeteria?	Milk	Refrigerated foods*
	Water	Frozen foods**
	Other beverages	Combination
	Snacks	Other (<i>Specify</i>):
b. Outside but within 25 feet of	Milk	Refrigerated foods*
	Water	Frozen foods**
	Other beverages	Combination
	Snacks	Other (<i>Specify</i>):
c. In other indoor areas?	Milk	Refrigerated foods*
	Water	Frozen foods**
	Other beverages	Combination
	Snacks	Other (<i>Specify</i>):
d. Outside school buildings, on school grounds?	Milk Water Other beverages Snacks	Refrigerated foods* Frozen foods** Combination Other (Specify):
		3 mor (<i>speedy</i>)

*Examples of refrigerated foods include sandwiches, yogurt, pasta salad. **Examples of frozen foods include ice cream, frozen yogurt.

6. Availability of vending machines. (*Check all time periods when vending machines in your school are available for student use.*)

Location	Before classes begin	During breakfast	Before lunch	During lunch	After lunch	After last class
a. In the cafeteria.	O 1	0 ₂	O 3	O 4	0 ₅	0 6
b. Outside but within 25 feet of the cafeteria.	0 1	0 ₂	0 3	O 4	0 5	0 6
c. Other indoor areas.	O ₁	0 ₂	O 3	0 ₄	O 5	0 6
d. Outside school buildings, on school grounds.	0 ₁	0 ₂	0 ₃	O ₄	0 ₅	0 ₆

7. For times when vending machines are not available to students, how is access restricted? *(Check all that apply.)*

	In the cafeteria	Other locations	
a. Machines are turned off by timer.	0 ₁	0 ₂	$(Skip \ to \ Q \ 8)$
b. Machines are turned off with a key.	0 ₁	0 ₂	
c. Machines are in a room that is locked.	0 ₁	0 ₂	
d. Machines are visually monitored.	0 ₁	0 ₂	
e. Other (<i>Please specify</i>):	— 0 ₁	0 ₂	

7a. Who is responsible for turning off, locking, or monitoring vending machines? (*Please explain.*)

8. For the vending machines available to students at your school, who determines the location, hours of availability, and types of foods/beverages that may be sold? (*Please check all that apply.*)

	a. Location	b. Hours of availability	c. Types of foods/ beverages sold
School board	0 ₁	0 ₂	0 ₃
Superintendent	0 ₁	0 ₂	O 3
District business manager	0 ₁	0 ₂	O 3
Principal	0 ₁	0 ₂	O 3
Food service director	0 ₁	0 ₂	O 3
Cafeteria manager	0 ₁	0 ₂	O 3
Other (<i>Please specify</i>):	0 ₁	0 ₂	0 ₃

In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, please tell us if your school restricts vending machine items based on price or nutrition criteria.

9. Do vending machine items have to fall within any particular price range?

- ρ_1 Yes
- ρ_2 No
- ρ_3 Don't know

If yes, please explain or attach copy of price criteria.

10. Do vending machine items have to meet any particular nutrition standards?

- ρ_1 Yes
- ρ_2 No
- ρ_3 Don't know

If yes, please explain or attach copy of nutrition criteria.

11. In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, does your school **prohibit** any specific foods or beverages from being sold in vending machines?

- ρ_1 Yes
- ρ_2 No (*Skip to Q 12*)
- ρ_3 Don't know (*Skip to Q 12*)

11a. What foods or beverages are prohibited? (Check all that apply.)

- ρ_1 Coffee or coffee-based beverages
- ρ_2 Tea or tea-based beverages (including iced teas)
- ρ_{3} All caffeinated beverages (some coffees, teas, colas, energy drinks)
- ρ_4 Carbonated soft drinks
- ρ_{5} Juice drinks (less than 50% fruit juice, with added sweeteners)
- ρ_{6} Other sweetened beverages (sports drinks, flavored waters with added sweetener)
- ρ_7 All artificially sweetened beverages (diet soft drinks, Fruit₂0)
- ρ_8 Fried vegetables (french fries)
- ρ_9 High-fat snacks or baked goods
- ρ_{10} High-sugar foods
- ρ_{11} Candy
- ρ_{12} Gum
- ρ₁₃ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages prohibited from being sold in the vending machines in your school.

- 12. In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, does your school **require** any specific foods or beverages to be sold in vending machines?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 13)
 - ρ_3 Don't know (*Skip to Q 13*)

12a. What foods or beverages are required? (Check all that apply.)

- ρ_1 Bottled water (unflavored, with no added sweeteners)
- ρ_2 Juice (100% juice)
- ρ_3 Milk (any type)
- ρ₄ Milk—skim, nonfat or 1% lowfat
- ρ_5 Low-fat snacks or baked goods
- ρ_6 Low-sugar foods
- ρ_7 Fresh fruit or vegetables
- ρ_8 Lean meats
- ρ_9 Whole grains
- ρ_{10} Diet, "lite," or sugar-free beverages
- ρ₁₁ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages required to be sold in the vending machines in your school.

- 13. Please characterize the **purchasing** arrangements for vending machines at your school. *(Check all that apply.)*
 - ρ_1 Contract with single distributor who stocks machines
 - ρ₂ Contract with multiple distributors who stock machines
 - ρ_{3} Food service director orders all or some items for machines in district
 - ρ_4 School cafeteria staff orders all or some items for machines in school
 - ρ₅ Other (*Please specify*): _____
 - ρ_3 Don't know
- 14. Please characterize the **stocking** arrangements for vending machines at your school. (*Check all that apply.*)
 - ρ_1 Contract with single distributor who stocks machines
 - ρ₂ Contract with multiple distributors who stock machines
 - ρ_3 Food service director stocks all or some items for machines in district
 - ρ_4 School cafeteria staff stocks all or some items for machines in school
 - ρ₅ Other (*Please specify*): _____
 - ρ_3 Don't know
- 15. Do you have or can you obtain information on the number sold for each product in some or all vending machines at your school?
 - $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \ (Skip \ to \ Q \ 16) \end{array}$
 - 15a. For what time period is the information available? (Check all that apply.)
 - ρ_1 Per week
 - ρ_2 Per month
 - ρ_3 An average month
 - ρ_4 Per year
 - ρ₅ Other (*Please specify*): _____

Please attach a copy of your most recent invoice(s) or other documentation that includes this information.

- 16. Who receives the profits (i.e., revenue less costs) from the vending machines at your school? *(Check all that apply.)*
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
 - 16a. Approximately how much total profit is earned from **all** vending machines available for student use at your school (per year, month, or week)? Do not include profits that go to the school food service.

\$_____ per _____

- 16b. How are profits from vending machine sales used? (Check all that apply.)
 - ρ_1 General fund
 - ρ₂ Specific purpose (*Please specify*): _____
- 17. Are beverage sales in your school covered by a "pouring rights" contract (that is, a long-term contract with a beverage company that establishes the company as a sole-source vendor for some types of beverages)?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Section D*)
 - ρ_3 Don't know (*Skip to Section D*)
 - 17a. Who receives the commissions and/or incentives from the beverage contract?
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ_6 Other (*Please specify*): _____
 - 17b. Since starting the beverage contract, has the number of vending machines in your school increased?
 - ρ_1 Yes
 - ρ_2 No

D. Snack Bars (Including Canteens, Food Carts, Kiosks)

The term "snack bar" is used in this section to refer to places that sell food to students, other than cafeteria lines, a la carte windows, vending machines, or school stores. This may include canteens, food carts, kiosks, or concession stands open to students during the school day.

- 1. Are there any snack bars located on school property that sell food or beverages to students?
 - ρ_1 Yes (Go to Q 2)
 - ρ_2 No (*Please answer Q 1a, then skip to Section E*)
 - 1a. What is the most important reason your school does not operate snack bars for students?

- 2. Who decided to operate the snack bar(s) at your school? (*Check all that apply.*)
 - ρ_1 School board
 - ρ_2 Superintendent
 - ρ_3 School district business manager
 - ρ_4 Food service director
 - ρ_5 Principal
 - ρ_{6} Cafeteria manager
 - ρ₇ Other (*Please specify*): _____

3. Please answer the following questions regarding the location and type of all snack bars available for students during the school day at your school.

In your school, how many snack bars are located	How many snack bars of each type are available in each location?		
a. In the cafeteria?	Canteens	_ Concession stands	
	Food carts	Other (Specify):	
	Kiosks		
b. Outside but within 25 feet of the cafeteria?	Canteens	_ Concession stands	
	Food carts	_ Other (Specify):	
	Kiosks		
c. In other indoor areas?	Canteens	_ Concession stands	
	Food carts	Other (Specify):	
	Kiosks		
d. Outside school buildings, on school grounds?	Canteens Food carts	_ Concession stands _ Other (<i>Specify</i>):	
	Kiosks		

4. Availability of snack bars. (Check all time periods when snack bars in your school are available for student use.)

Location	Before classes begin	During breakfast	Before lunch	During lunch	After lunch	After last class
a. In the cafeteria.	O ₁	0 ₂	O 3	O ₄	O 5	0 6
b. Outside but within 25 feet of the cafeteria.	O ₁	0 ₂	0 3	O ₄	0 ₅	0 6
c. Other indoor areas.	O ₁	0 ₂	O 3	O ₄	O 5	0 6
d. Outside school buildings, on school grounds.	O ₁	0 ₂	O ₃	O ₄	O 5	0 6

- 5. Are the snack bar(s) usually open every school day?
 - ρ_1 Yes (Skip to Q 6)

 ρ_2 No

 ρ_3 Varies by snack bar type

Abt Associates Inc.

- 5a. Which days of the week are the snack bars usually open? (*Check all that apply.*)
 - ρ_1 Monday
 - ρ_2 Tuesday
 - ρ_3 Wednesday
 - ρ_4 Thursday
 - ρ_5 Friday
 - ρ_{6} Various days or no set schedule
- 6. What types of payment are accepted for food and beverages sold at the snack bar(s)? (*Check all that apply.*)
 - ρ_1 Cash
 - ρ_2 Electronic debit
 - ρ_3 Prepaid account (cards, tickets, vouchers)
 - ρ₄ Other (*Please specify*): _____
- 7. For the snack bar(s) at your school, who determines the location, hours of availability, and types of foods/beverages that may be sold? (*Check all that apply.*)

	a. Location	b. Hours of availability	c. Types of foods/ beverages sold
School board	0 ₁	0 ₂	0 ₃
Superintendent	0 ₁	0 ₂	0 ₃
District business manager	0 ₁	0 ₂	0 ₃
Principal	0 ₁	0 ₂	0 ₃
Food service director	0 ₁	0 ₂	0 ₃
Cafeteria manager	0 ₁	0 ₂	0 ₃
Other (Please specify):	0 ₁	0 ₂	0 ₃

In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, please tell us if your school restricts snack bar items based on price or nutrition criteria.

8. Do snack bar items have to fall within any particular price range?

 $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \\ \rho_3 & Don't know \end{array}$

If yes, please explain or attach copy of price criteria.

9. Do snack bar items have to meet any particular nutrition standards?

- ρ_1 Yes
- ρ_2 No
- ρ_3 Don't know

If yes, please explain or attach copy of nutrition criteria.

10. In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, does your school **prohibit** any specific foods or beverages from being sold at the snack bar(s)?

- ρ_1 Yes
- ρ_2 No (*Skip to Q 11*)
- ρ_3 Don't know (*Skip to Q 11*)

10a. What foods or beverages are prohibited? (Check all that apply.)

- ρ_1 Coffee or coffee-based beverages
- ρ_2 Tea or tea-based beverages (including iced teas)
- ρ_{3} All caffeinated beverages (some coffees, teas, colas, energy drinks)
- ρ_4 Carbonated soft drinks
- ρ_{5} Juice drinks (less than 50% fruit juice, with added sweeteners)
- ρ_{6} Other sweetened beverages (sports drinks, flavored waters with added sweetener)
- ρ_7 All artificially sweetened beverages (diet soft drinks, Fruit₂0)
- ρ_8 Fried vegetables (french fries)
- ρ_9 High-fat snacks or baked goods
- ρ_{10} High-sugar foods
- ρ_{11} Candy
- ρ_{12} Gum
- ρ₁₃ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages prohibited from being sold at snack bars in your school.

- 11. In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, does your school **require** any specific foods or beverages to be sold at the snack bar(s)?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 12)
 - ρ_3 Don't know (*Skip to Q 12*)
 - 11a. What foods or beverages are required? (Check all that apply.)
 - ρ_1 Bottled water (unflavored, with no added sweeteners)
 - ρ_2 Juice (100% juice)
 - ρ_3 Milk (any type)
 - ρ₄ Milk—skim, nonfat or 1% lowfat
 - ρ_5 Low-fat snacks or baked goods
 - ρ_6 Low-sugar foods
 - ρ_7 Fresh fruit or vegetables
 - ρ_8 Lean meats
 - ρ_9 Whole grains
 - ρ_{10} Diet, "lite," or sugar-free beverages
 - ρ₁₁ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages required to be sold at snack bars in your school.

- 12. Can students select a USDA reimbursable meal from any of the snack bar(s)?
 - ρ₁ Yes (*Please specify snack bar type(s)*): _____
 - ρ_2 No
 - ρ_3 Don't know
- 13. Who operates the snack bar(s) at your school? (*Check all that apply.*)
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____

- 14. Who receives the profits from the snack bar(s) at your school? (*Check all that apply.*)
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
 - 14a. Approximately how much profit is received **by the school** from the snack bar(s) (per year, month, or week)? *Do not include profits that go to the school food service*.

\$_____ per _____

- 14b. How are the school's profits from snack bar sales used? (*Check all that apply.*)
 - ρ_1 General fund
 - ρ₂ Specific purpose (*Please specify*): _____

E. School Stores

1. Are there any school stores located on school property that sell food or beverages to students?

 ρ_1 Yes (Go to Q 2) No (Plane sum Q la

- ρ_2 No (Please answer Q 1a, then skip to Section F)
- 1a. What is the most important reason your school does not operate a school store that sells food or beverages?

- 2. Who decided to operate the school store(s) in your school? (*Check all that apply.*)
 - ρ_1 School board
 - ρ_2 Superintendent
 - ρ_3 School district business manager
 - ρ_4 Food service director
 - ρ_5 Principal
 - ρ_{6} Cafeteria manager
 - ρ₇ Other (*Please specify*): _____
- 3. How many school stores sell food or beverages to students? (If more than one, please answer Questions 4 through 7 in this section for the store that generates the most revenue from food and beverage sales to students.)
- 4. Where is the school store located?
 - ρ_1 In the cafeteria
 - ρ_2 Outside but within 25 feet of the cafeteria
 - ρ_3 Other indoor area
 - ρ_4 Outside school buildings, on school grounds
- 5. Is the school store usually open every school day?
 - ρ_1 Yes (Skip to Q 6)
 - $\rho_2 \quad No$

- 5a. Which days of the week is the school store usually open? (*Check all that apply.*)
 - ρ_1 Monday
 - ρ_2 Tuesday
 - ρ_3 Wednesday
 - ρ_4 Thursday
 - ρ_5 Friday
 - ρ_{6} Various days or no set schedule
- 6. When is the school store usually open? (*Check all that apply.*)
 - ρ_1 Before classes begin
 - ρ_2 During the breakfast period
 - ρ_3 Before lunch
 - ρ_4 During lunch periods
 - ρ_5 After lunch
 - ρ_6 After last class
 - ρ₄ Other (*Please specify*): _____
- 7. What types of payment are accepted for food and beverages sold at the school store? (*Check all that apply.*)
 - ρ_1 Cash
 - ρ_2 Electronic debit
 - ρ_3 Prepaid account (cards, tickets, vouchers)
 - ρ₄ Other (*Please specify*): _____
- 8. Thinking about all of the school store(s) at your school, who determines the location, hours of availability, and types of foods/beverages that may be sold? (*Check all that apply.*)

	a. Location	b. Hours of availability	c. Types of foods/ beverages sold
School board	0 ₁	0 ₂	O 3
Superintendent	0 ₁	0 ₂	O 3
District business manager	0 ₁	0 ₂	O 3
Principal	0 ₁	0 ₂	O 3
Food service director	0 ₁	0 ₂	O 3
Cafeteria manager	0 ₁	0 ₂	O 3
Other (Please specify):	0 ₁	0 ₂	0 ₃

In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, please tell us if your school restricts school store foods and beverages based on price or nutrition criteria.

- 9. Do school store food and beverage items have to fall within any particular price range?
 - ρ_1 Yes ρ_2 No
 - ρ_3 Don't know

If yes, please explain or attach copy of price criteria.

10	Do school	store food	and beverage	items have to	o meet anv	particular	nutrition	standards?
10.	D0 school	store 1000	and beverage	nulls have u	J meet any	particular	nuunuon	stanuarus :

 $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \\ \rho_3 & Don't know \end{array}$

If yes, please explain or attach copy of nutrition criteria.

- 11. In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, does your school **prohibit** any specific foods or beverages from being sold in the school store?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 12)
 - ρ_3 Don't know (*Skip to Q 12*)

11a. What foods or beverages are prohibited? (Check all that apply.)

- ρ_1 Coffee or coffee-based beverages
- ρ_2 Tea or tea-based beverages (including iced teas)
- ρ_{3} All caffeinated beverages (some coffees, teas, colas, energy drinks)
- ρ₄ Carbonated soft drinks
- ρ_{5} Juice drinks (less than 50% fruit juice, with added sweeteners)
- ρ_{6} Other sweetened beverages (sports drinks, flavored waters with added sweetener)
- ρ_7 All artificially sweetened beverages (diet soft drinks, Fruit₂0)
- ρ_8 Fried vegetables (french fries)
- ρ_9 High-fat snacks or baked goods
- ρ_{10} High-sugar foods
- ρ_{11} Candy
- $\rho_{12} \quad \text{Gum}$
- ρ₁₃ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages prohibited from being sold in the school store at your school.

- 12. In addition to regulations, laws, or guidelines put in place by USDA, state, or school district authorities, does your school **require** any specific foods or beverages to be sold in the school store?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Q 13*)
 - ρ_3 Don't know (*Skip to Q 13*)
 - 12a. What foods or beverages are required? (Check all that apply.)
 - ρ_1 Bottled water (unflavored, with no added sweeteners)
 - ρ_2 Juice (100% juice)
 - ρ_3 Milk (any type)
 - ρ_4 Milk—skim, nonfat or 1% lowfat
 - ρ_5 Low-fat snacks or baked goods
 - ρ_6 Low-sugar foods
 - ρ_7 Fresh fruit or vegetables
 - ρ_8 Lean meats
 - ρ_9 Whole grains
 - ρ_{10} Diet, "lite," or sugar-free beverages
 - ρ₁₁ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages required to be sold in the school store at your school.

- 13. Who operates the school store at your school?
 - ρ_1 Food service department (*Skip to Q 15*)
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
- 14. Do you have or can you obtain information on the number sold for each food and beverage item in the school store at your school?
 - $\begin{array}{ll} \rho_1 & \text{Yes} \\ \rho_2 & \text{No} (Skip \ to \ Q \ 15) \end{array}$
 - 14a. For what time period is the information available? (Check all that apply.)
 - ρ_1 Per week
 - ρ_2 Per month
 - ρ_3 An average month
 - ρ_4 Per year
 - ρ₅ Other (*Please specify*): _____

Please attach a copy of your most recent invoice(s) or other documentation that includes this information.

- 15. Who receives the profits from food and beverage sales at the school store(s) in your school? *(Check all that apply.)*
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
 - 15a. Approximately how much profit is received **by the school** from food and beverage sales at the school store in total (per year, month, or week)? *Do not include profits that go to the school food service*.

\$_____ per _____

- 15b. How are the school's profits from the school store food and beverage sales used? (*Check all that apply.*)
 - ρ_1 General fund
 - ρ₂ Specific purpose (*Please specify*): _____

F. Off-Campus Eating Establishments

- 1. Are students allowed to leave the school during their **lunch** period?
 - ρ_1 Yes, all students may leave (*Skip to Q 2*)
 - ρ_2 Yes, some students may leave
 - ρ_3 No, no students may leave the school (*Skip to Q 2*)
 - 1a. Which grades are allowed to leave during their lunch period? (*Circle all that apply.*)

K 1 2 3 4 5 6 7 8 9 10 11 12

- 2. Are students allowed to leave the school during their **free** period(s)? (*Check one*)
 - ρ_1 Yes, all students are allowed to leave (*Skip to Q 3*)
 - ρ_1 Yes, some students are allowed to leave
 - ρ_2 No, no students are allowed to leave (*Skip to Q 3*)
 - ρ_3 Not applicable, students do not have free periods (*Skip to Q 3*)
 - 2a. Which grades are allowed to leave during their free period(s)? (*Circle all that apply.*)
 - K 1 2 3 4 5 6 7 8 9 10 11 12
- 3. Approximately how many of the following eating establishments are close enough for students to walk or drive to from campus during a lunch period?

Source	Number Within Walking Distance	Number Within Driving Distance
Fast food restaurants		
Other restaurants, cafeterias, or diners		
Food markets, grocery, convenience, or other stores		
Lunch wagons or push carts		
Food courts (e.g., in a shopping mall)		
Other (please specify):		
G. Other Sources of Food or Beverages

- 1. In this school, are parents or teachers permitted to offer students foods or beverages in the classroom?
 - $\begin{array}{ll} \rho_1 & \text{Yes} \\ \rho_2 & \text{No} (Skip \ to \ Q \ 4) \end{array}$
- 2. Does your school provide guidelines to parents or teachers for foods that can be offered to students in the classroom? (*Check all that apply.*)
 - ρ_1 Guidelines are provided to parents
 - ρ_2 Guidelines are provided to teachers
 - ρ_3 No guidelines are provided to parents or teachers
- 3. Are there any restrictions on the types of food or beverages that parents or teachers may offer students in the classroom?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Q 4*)
 - 3a. What foods or beverages are not permitted? (*Please specify or attach a copy of the document that lists the restricted items.*)

- 4. Are fundraisers that offer baked items, candy, or other food or beverages for sale to students ever held at this school?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Section H*)
- 5. Which organizations hold fundraisers that offer food or beverages for sale to students? (*Check all that apply.*)
 - ρ_1 Athletic department
 - ρ_2 Student clubs or service organizations
 - ρ_3 Music or art department
 - ρ_4 Parent-teacher organizations
 - ρ₅ Other (*Please specify*): _____
 - ρ_8 Don't know

6. On average, how often do the following types of activities take place during the breakfast period? (*If your school does not serve breakfast, skip to Q 7.*)

	Every day	3-4 times per week	1-2 times per week	1-2 times per month	Less than 1 time per month	Not appli- cable
a. Bake sales	O ₁	0 ₂	O 3	O 4	O 5	0 ₈
b. Candy sales	0 ₁	0 ₂	O 3	O 4	O 5	0 ₈
c. Other fundraisers selling food (<i>please specify</i>):	O ₁	O ₂	O ₃	0 ₄	O 5	0 ₈

7. On average, how often do the following types of activities take place during the lunch period?

	Every day	3-4 times per week	1-2 times per week	1-2 times per month	Less than 1 time per month	Not appli- cable
a. Bake sales	O ₁	O ₂	O 3	O 4	0 ₅	0 ₈
b. Candy sales	O ₁	0 ₂	O ₃	O 4	0 ₅	0 ₈
c. Other fundraisers selling food (<i>please specify</i>):	O ₁	0 2	0 3	O ₄	O 5	0 ₈

- 8. Are there any restrictions on the types of food or beverages that may be sold to students for fundraising purposes?
 - $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \ (Go \ to \ Section \ H) \end{array}$
 - 8a. What foods or beverages are not permitted for sale? (*Please specify or attach a copy* of the document that lists restricted items.)

H. Nutrition Initiatives

- 1. Other than initiatives that target USDA school meals, is your school currently implementing any of the following activities to promote the availability of healthful food choices for students at school? (*Check all that apply.*)
 - ρ_1 The school has a health or nutrition advisory council made up of school staff, students, and parents that provides input about the types of foods available at school.
 - ρ_2 School food service staff participate in making decisions and policies that affect the kinds of foods available to students outside of school meal programs.
 - ρ_3 School groups are encouraged to raise funds by selling non-food items.
 - ρ_4 Food is not used by teachers or other school staff as a reward or punishment (e.g., withholding snacks for misbehavior) for students.
 - ρ_5 Parents are encouraged to provide healthy food choices for bag lunches brought from home.
 - ho_6 The school has nutrition criteria for foods and beverages offered at parties, celebrations, and social events held after school or on weekends.
 - ρ_7 Foods and beverages sold outside the school meal program are not more highly marketed than the reimbursable school meals.
 - ρ_{8} The school conducts promotional activities to encourage healthy food choices, and does not permit advertising of less nutritious foods and beverages.
 - ρ_{9} The school bans particular types of soda, other soft drinks, or sweetened fruit beverages (less than 100% juice) that may be sold to students in the school or on school grounds (including vending machines).
 - ρ_{10} Other school food initiatives (*Please specify*):

I. Daily Attendance

Please provide the total daily attendance for the week of May 19, 2003 – May 23, 2003.

Number of Students

Monday, May 19	
Tuesday, May 20	
Wednesday, May 21	
Thursday, May 22	
Friday, May 23	

J. Survey Information

- 1. Did you need to consult other school personnel to complete this survey?
 - $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \ (Skip \ to \ Q. \ 2) \end{array}$
 - 1a. For each section of the survey, please check if other personnel were consulted and indicate their title.

ρ₁ School lunch and breakfast programs _____

- ρ₂ A la carte _____
- ρ₃ Vending machines _____

ρ₄ Snack bars _____

- ρ₅ School stores _____
- ρ₆ Off-campus eating establishments _____
- ρ_7 Other sources of food or beverages _____
- ρ₈ Nutrition initiatives _____
- ρ₈ Daily attendance _____
- 2. Please estimate the time it took you to complete the following:

Gathering information:

Consulting with other staff:

Completing questionnaire: _____

Thank you for taking the time to complete this survey. Your cooperation is very much appreciated.



Study of Foods in the School Environment

Survey of School Food Authority (SFA) Directors

If you have questions or need assistance, please call 617-349-2674 or 617-349-2661

Spring 2003

The Study of Foods in the School Environment is being conducted for the:

Food and Nutrition Service US Department of Agriculture 3101 Park Center Drive Alexandria, Virginia 22302

By:

Abt Associates Inc. 55 Wheeler Street Cambridge, Massachusetts 02138

This questionnaire asks about sources of foods available to students in your district during the school day, other than USDA reimbursable meals. If the SFA serves multiple school districts or a district as well as independent charter schools, please answer for the district containing [SCHOOL]. The questions are about school district policies regarding:

- A la carte foods and beverages
- Vending machines
- Snacks bars, including canteens, food carts, kiosks
- School stores
- Exclusive beverage contracts

The questionnaire also includes questions about the school food environment at [SCHOOL]. Please note that the questionnaire will instruct you to skip over sections that do not apply to your district or [SCHOOL].

A. A la Carte

For the purposes of this survey, "a la carte" refers to foods and beverages sold in or near the cafeteria(s) by school food service staff, other than milk, full reimbursable meals, and vending machine items. For example, a la carte items may be sold on the cafeteria line with reimbursable meals, or from a separate line or window within the cafeteria.

- 1. Are a la carte food and beverage sales offered during lunch to students in any of the schools in your district?
 - ρ_1 Yes ρ_2 No (*Skip to Section B*)
 - 1a. Which types of schools offer a la carte food and beverage sales during lunch? (*Check all that apply.*)
 - ρ_1 Elementary schools
 - ρ_2 Middle or junior high schools
 - ρ_3 High schools

In addition to regulations, laws, or guidelines put in place by USDA or state authorities, please tell us if your school district restricts a la carte items based on price or nutrition criteria.

- 2. Do a la carte items have to fall within any particular price range?
 - ρ_1 Yes
 - ρ_2 No
 - ρ_3 Don't know

If yes, please explain or attach copy of price criteria.

- 3. Do a la carte items have to meet any particular nutrition standard(s)?
 - ρ_1 Yes
 - ρ_2 No
 - ρ_3 Don't know

If yes, please explain or attach copy of nutrition criteria.

- 4. In addition to regulations, laws, or guidelines put in place by USDA or state authorities, does your school district **prohibit** any specific foods or beverages from being sold a la carte?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 5)
 - ρ_3 Don't know (*Skip to Q 5*)
 - 4a. Which foods or beverages are prohibited? (*Check all that apply.*)
 - ρ_1 Coffee or coffee-based beverages
 - ρ_2 Tea or tea-based beverages (including iced teas)
 - ρ_{3} All caffeinated beverages (some coffees, teas, colas, energy drinks)
 - ρ_4 Carbonated soft drinks
 - ρ_{5} Juice drinks (less than 50 percent fruit juice, with added sweeteners)
 - ρ_{6} Other sweetened beverages (sports drinks, flavored waters with added sweetener)
 - ρ_7 All artificially sweetened beverages (diet soft drinks, Fruit₂0)
 - ρ_8 Fried vegetables (french fries)
 - ρ_9 High-fat snacks or baked goods
 - ρ_{10} High-sugar foods
 - ρ_{11} Candy
 - ρ_{12} Gum
 - ρ₁₃ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages prohibited from being sold a la carte in your school district.

- 5. In addition to regulations, laws, or guidelines put in place by USDA or state authorities, does your school district **require** any specific foods or beverages to be sold a la carte?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 6)
 - ρ_3 Don't know (*Skip to Q 6*)
 - 5a. Which foods or beverages are required? (*Check all that apply.*)
 - ρ_1 Bottled water (unflavored, with no added sweeteners)
 - ρ_2 Juice (100 percent juice)
 - ρ_3 Milk (any type)
 - ρ_4 Milk (skim, nonfat, or 1 percent low-fat)
 - ρ_5 Low-fat snacks or baked goods
 - ρ_6 Low-sugar foods
 - ρ_7 Fresh fruit or vegetables
 - ρ_8 Lean meats
 - ρ_9 Whole grains
 - ρ_{10} Diet, "lite," or sugar-free beverages
 - ρ₁₁ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages required to be sold a la carte your school district.

- 6. In schools with a la carte food and beverage sales, are components of the USDA reimbursable lunch, other than milk, offered on an a la carte basis?
 - $\begin{array}{ll}\rho_1 & Yes\\\rho_2 & No\end{array}$
- 7. Is there a single standard price for a la carte entrees in schools with a la carte food or beverage sales?
 - ρ_1 Yes
 - ρ_2 No
 - ρ_3 Not applicable; entrees not sold a la carte in the district (*Skip to Q 11*)
- 8. How does the price of an a la carte entree compare to the full price of a complete reimbursable meal? (*Check all that apply.*)
 - ρ_1 A la carte entree price is always higher
 - ρ_2 A la carte entree price is sometimes higher
 - ρ_{3} A la carte entrees and full reimbursable meals are priced the same
 - ρ_4 A la carte entree price is sometimes lower
 - ρ_5 A la carte entree price is always lower

9. Do any schools in your district offer "branded foods," that is, foods from national or regional brand-name or chain restaurants (McDonald's, Taco Bell, Domino's, Subway, etc.) or from a local restaurant?

$$\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \ (Skip \ to \ Q \ 11) \end{array}$$

- 10. How does the price of branded entrees compare to the price of other entrees ...
 - 10a. When offered in reimbursable meals? (*Check all that apply.*)
 - ρ_1 Branded entree price is always higher
 - ρ_2 Branded entree price is sometimes higher
 - ρ_3 Branded entrees and other entrees are priced the same
 - ρ_4 Branded entree price is sometimes lower
 - ρ_5 Branded entree price is always lower
 - ρ_6 Not applicable
 - 10b. When offered a la carte? (Check all that apply.)
 - ρ_1 Branded entree price is always higher
 - ρ_2 Branded entree price is sometimes higher
 - ρ_3 Branded entrees and other entrees are priced the same
 - ρ_4 Branded entree price is sometimes lower
 - ρ_5 Branded entree price is always lower
- 11. Thinking about all foods and beverages that are offered a la carte, what factors does your school district consider in setting a la carte prices? (*Check all that apply.*)
 - ρ_1 Food costs
 - ρ_2 Product costs (wages, benefits, utilities, equipment, supplies, etc.)
 - ρ₄ Administrative or indirect costs
 - ρ_{5} Incentive for student participation in reimbursable meal program
 - ρ_{6} Incentive for choosing more healthful foods/beverages (e.g., fruit priced lower than baked goods)
 - ρ_7 Ease of collecting payment
 - ρ₈ Other (*Please specify*): ____
- 12. How would you characterize the profitability of a la carte food and beverage sales in your school district?
 - ρ_1 Generate a profit
 - ρ_2 Break even (*Skip to Section B*)
 - ρ_3 Generate a loss (*Skip to Section B*)

- 12a. Are the profits (i.e., revenue less costs) from a la carte sales used to subsidize the reimbursable meal program?
 - ρ_1 Yes
 - ρ_2 No
 - ρ_3 Varies by school

B. Vending Machines

- 1. What types of schools in your district have vending machines available for student use?
 - ρ_1 Elementary schools
 - ρ_2 Middle or junior high schools
 - ρ_3 High schools
 - ρ_4 No schools have vending machines for students (*Skip to Section C*)
- 2. In what types of schools does the school food service operate or have primary responsibility for vending machines available for student use?
 - ρ_1 Elementary schools (Go to Q 3)
 - ρ_2 Middle or junior high schools (Go to Q 3)
 - ρ_3 High schools (Go to Q 3)
 - ρ_4 No schools have vending machines for which the school food service is responsible (*Please answer Q 2a., then skip to Section C*)
 - 2a. What is the most important reason the school food service does not operate vending machines for students in some or all schools in your district?

In addition to regulations, laws, or guidelines put in place by USDA or school district authorities, please tell us if your school district restricts vending machine items based on price or nutrition criteria.

- 3. Do vending machine items have to fall within any particular price range?
 - ρ_1 Yes ρ_2 No ρ_3 Don't know

If yes, please explain or attach copy of price criteria.

- 4. Do vending machine items have to meet any particular nutrition standards?
 - $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \\ \rho_3 & Don't know \end{array}$

If yes, please explain or attach copy of nutrition criteria.

- 5. In addition to regulations, laws, or guidelines put in place by USDA or state authorities, does your school district **prohibit** any specific foods or beverages from being sold in vending machines?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 6)
 - ρ_3 Don't know (*Skip to Q 6*)
 - 5a. Which foods or beverages are prohibited? (*Check all that apply.*)
 - ρ_1 Coffee or coffee-based beverages
 - ρ_2 Tea or tea-based beverages (including iced teas)
 - ρ_{3} All caffeinated beverages (some coffees, teas, colas, energy drinks)
 - ρ₄ Carbonated soft drinks
 - ρ_{5} Juice drinks (less than 50 percent fruit juice, with added sweeteners)
 - ρ_{6} Other sweetened beverages (sports drinks, flavored waters with added sweetener)
 - ρ_7 All artificially sweetened beverages (diet soft drinks, Fruit₂0)
 - ρ_8 Fried vegetables (french fries)
 - ρ_9 High-fat snacks or baked goods
 - ρ_{10} High-sugar foods
 - ρ_{11} Candy
 - $\rho_{12} \quad \text{Gum}$
 - ρ₁₃ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages prohibited from being sold in the vending machines in your school district.

- 6. In addition to regulations, laws, or guidelines put in place by USDA or state authorities, does your school district **require** any specific foods or beverages to be sold in vending machines?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Q 7*)
 - ρ_3 Don't know (*Skip to Q 7*)
 - 6a. Which foods or beverages are required? (*Check all that apply.*)
 - ρ_1 Bottled water (unflavored, with no added sweeteners)
 - ρ_2 Juice (100 percent juice)
 - ρ_3 Milk (any type)
 - ρ₄ Milk (skim, nonfat, or 1 percent low-fat)
 - ρ_5 Low-fat snacks or baked goods
 - ρ_6 Low-sugar foods
 - ρ_7 Fresh fruit or vegetables
 - ρ_8 Lean meats
 - ρ_9 Whole grains
 - ρ_{10} Diet, "lite," or sugar-free beverages
 - ρ₁₁ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages required to be sold in the vending machines in your school district.

The following questions refer only to vending machines available for students for which the school food service has primary responsibility.

- 7. Please characterize the **purchasing** arrangements for vending machines in your district. *(Check all that apply.)*
 - ρ_1 Contract with single distributor who stocks machines
 - ρ_2 Contract with multiple distributors who stock machines
 - ρ_{3} Food service director orders all or some items for machines in district
 - ρ_4 School cafeteria staff order all or some items for machines in their schools
 - ρ₅ Other (*Please specify*): _____
- 8. Please characterize the **stocking** arrangements for vending machines in your district. (*Check all that apply.*)
 - ρ_1 Contract with single distributor who stocks machines
 - ρ_2 Contract with multiple distributors who stock machines
 - $\rho_{\rm 3}$ $\,$ Food service director stocks all or some items for machines in district
 - ρ_4 School cafeteria staff stock all or some items for machines in their schools
 - ρ₅ Other (*Please specify*): _____

- 9. Which of the following factors does your school district consider in setting prices for food and beverage items sold in vending machines? (*Check all that apply.*)
 - ρ_1 Food costs
 - ρ_2 Product costs (wages, benefits, utilities, equipment, supplies, etc.)
 - ρ₄ Administrative or indirect costs
 - ρ₅ Incentive for student participation in reimbursable meal program
 - ρ_{6} Incentive for choosing more healthful foods/beverages (e.g., milk priced lower than soft drinks)
 - ρ_7 Other (*Please specify*): _
- 10. Do you have nutrition information for foods and beverages sold in vending machines in your school district? That is, information on nutrient content from the manufacturer or distributor?
 - ρ_1 Yes, for all foods/beverages
 - ρ_2 Yes, for some foods/beverages
 - ρ_3 No

Please answer the following questions for the vending machines for which the school food service is primarily responsible at [SCHOOL].

- 11. Do you have or can you obtain information on the number sold for each product in vending machines at [SCHOOL]?
 - $\begin{array}{ll} \rho_1 & \text{Yes} \\ \rho_2 & \text{No} \ (Skip \ to \ Q \ 12) \end{array}$
 - 11a. For what time period is the information available? (Check all that apply.)
 - ρ_1 Per week
 - ρ_2 Per month
 - ρ_3 An average month
 - ρ_4 Per year
 - ρ₅ Other (*Please specify*): _____

Please attach a copy of your most recent invoice(s) or other documentation that includes this information.

- 12. Who receives the profits (i.e., revenue less costs) from the vending machines for which the school food service is responsible at [SCHOOL]? (*Check all that apply.*)
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
 - 12a. Approximately how much total profit is received **by the school food service** from **all** vending machines available for student use at [SCHOOL] (per year, month, or week)? Do not include profits that go to the principal or other school departments or funds.

\$_____ per _____

- 12b. How are the school food service profits from vending machine sales used? (*Check all that apply.*)
 - ρ_1 General fund
 - ρ₂ Specific purpose (*Please specify*): _____

11

C. Snack Bars (including Canteens, Food Carts, Kiosks)

The term "snack bar" is used in this section to refer to places that sell food to students, other than cafeteria lines, a la carte windows, vending machines, or school stores. This may include canteens, food carts, kiosks, or concession stands open to students during the school day.

- 1. Does the school food service in your district operate any snack bars that sell food or beverages to students during the school day?
 - ρ_1 Yes, district-wide (Go to Q 2)
 - ρ_3 Yes, some schools (*Please answer Q 1a, then go to Q 2*)
 - ρ_2 No (*Please answer Q 1a, then skip to Section D*)
 - 1a. What is the most important reason the school food service does not operate snack bars for students in some or all schools in your district?

- 2. In what types of schools does the school food service operate snack bars?
 - ρ_1 Elementary schools
 - ρ_2 Middle or junior high schools
 - ρ_3 High schools
- 3. What types of snack bars does the school food service operate in your district? (*Check all that apply.*)
 - ρ_1 Canteens
 - ρ_2 Food carts
 - ρ_3 Kiosks
 - ρ_4 Concession stands
 - ρ₅ Other snack bar types (*Please specify*): _____

In addition to regulations, laws, or guidelines put in place by USDA or state authorities, please tell us if your school district restricts snack bar items based on price or nutrition criteria.

- 4. Do snack bar items have to fall within any particular price range?
 - ρ_1 Yes
 - ρ_2 No
 - ρ_3 Don't know

If yes, please explain or attach copy of price criteria.

|--|

 $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \\ \rho_3 & Don't know \end{array}$

If yes, please explain or attach copy of nutrition criteria.

- 6. In addition to regulations, laws, or guidelines put in place by USDA or state authorities, does your school district **prohibit** any specific foods or beverages from being sold at snack bars?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 7)
 - ρ_3 Don't know (*Skip to Q 7*)
 - 6a. Which foods or beverages are prohibited? (*Check all that apply.*)
 - ρ_1 Coffee or coffee-based beverages
 - ρ_2 Tea or tea-based beverages (including iced teas)
 - ρ_{3} All caffeinated beverages (some coffees, teas, colas, energy drinks)
 - ρ_4 Carbonated soft drinks
 - ρ_{5} Juice drinks (less than 50 percent fruit juice, with added sweeteners)
 - ρ_{6} Other sweetened beverages (sports drinks, flavored waters with added sweetener)
 - ρ_7 All artificially sweetened beverages (diet soft drinks, Fruit₂0)
 - ρ_8 Fried vegetables (french fries)
 - ρ_{9} High-fat snacks or baked goods
 - ρ_{10} High-sugar foods
 - ρ_{11} Candy
 - ρ_{12} Gum
 - ρ₁₃ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages prohibited from being sold at snack bars in your school district.

- 7. In addition to regulations, laws, or guidelines put in place by USDA or state authorities, does your school district **require** any specific foods or beverages to be sold at snack bars?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 8)
 - ρ_3 Don't know (*Skip to Q 8*)
 - 6a. Which foods or beverages are required? (*Check all that apply.*)
 - ρ_1 Bottled water (unflavored, with no added sweeteners)
 - ρ_2 Juice (100 percent juice)
 - ρ_3 Milk (any type)
 - ρ₄ Milk (skim, nonfat, or 1 percent low-fat)
 - ρ_5 Low-fat snacks or baked goods
 - ρ_6 Low-sugar foods
 - ρ_7 Fresh fruit or vegetables
 - ρ_8 Lean meats
 - ρ_9 Whole grains
 - ρ_{10} Diet, "lite," or sugar-free beverages
 - ρ₁₁ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages required to be sold at snack bars in your school district.

- 8. Which of the following factors does your school district consider in setting prices for snack bar items? (*Check all that apply.*)
 - ρ_1 Food costs
 - ρ_2 Product costs (wages, benefits, utilities, equipment, supplies, etc.)
 - ρ_3 Administrative or indirect costs
 - ρ_4 Incentive for student participation in reimbursable meal program
 - ρ_{5} Incentive for choosing more healthful foods/beverages (e.g., fruit priced lower than baked goods)
 - ρ_{6} Ease of collecting payments
 - ρ₇ Other (*Please specify*): _____
- 9. Do you have nutrition information for foods and beverages sold in the snack bars in your school district? That is, information on nutrient content from the manufacturer or distributor?
 - ρ_1 Yes, for all foods/beverages
 - ρ_2 Yes, for some foods/beverages
 - ρ_2 No
- 10. Does the school food service operate a snack bar that sells food or beverages to students at [SCHOOL]?

 ρ_1 Yes

ρ_2 No (*Skip to Section D*)

Please answer the following questions for the snack bar(s) operated by the school food service at [SCHOOL].

- 11. Who receives the profits (i.e., revenue less costs) from the snack bar(s) at [SCHOOL]? (*Check all that apply.*)
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
 - 11a. Approximately how much total profit is received **by the school food service** from the snack bar(s) at [SCHOOL] (per year, month, or week)? Do not include profits that go to the principal or other school departments or funds.

\$_____ per _____

- 11b. How are the school food service profits from snack bar sales used? (*Check all that apply.*)
 - ρ_1 General fund
 - ρ₂ Specific purpose (*Please specify*): _____

D. School Stores

- 1. Does the school food service in your district operate any school stores that sell food or beverages to students during the school day?
 - ρ_1 Yes, district-wide
 - ρ_2 Yes, some schools
 - ρ_3 No (*Skip to Section E*)
- 2. In what types of schools does the school food service operate school stores that sell food or beverages?
 - ρ_1 Elementary schools
 - ρ_2 Middle or junior high schools
 - ρ_3 High schools

In addition to regulations, laws, or guidelines put in place by USDA or state authorities, please tell us if your school district restricts school store food/beverage items based on price or nutrition criteria.

- 3. Do school store food/beverage items have to fall within any particular price range?
 - ρ_1 Yes
 - ρ_2 No
 - ρ_3 Don't know

If yes, please explain or attach copy of price criteria.

- 4. Do school store food/beverage items have to meet any particular nutrition standards?
 - ρ_1 Yes
 - ρ_2 No
 - ρ_3 Don't know

If yes, please explain or attach copy of nutrition criteria.

- 5. In addition to regulations, laws, or guidelines put in place by USDA or state authorities, does your school district **prohibit** any specific foods or beverages from being sold at the school store?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 5)
 - ρ_3 Don't know (*Skip to Q 6*)
 - 5a. Which foods or beverages are prohibited? (*Check all that apply.*)
 - ρ_1 Coffee or coffee-based beverages
 - ρ_2 Tea or tea-based beverages (including iced teas)
 - ρ_{3} All caffeinated beverages (some coffees, teas, colas, energy drinks)
 - ρ₄ Carbonated soft drinks
 - ρ_{5} Juice drinks (less than 50 percent fruit juice, with added sweeteners)
 - ρ_{6} Other sweetened beverages (sports drinks, flavored waters with added sweetener)
 - ρ_7 All artificially sweetened beverages (diet soft drinks, Fruit₂0)
 - ρ_8 Fried vegetables (french fries)
 - ρ_9 High-fat snacks or baked goods
 - ρ_{10} High-sugar foods
 - ρ_{11} Candy
 - ρ_{12} Gum
 - ρ₁₃ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages prohibited from being sold at school stores in your school district.

- 6. In addition to regulations, laws, or guidelines put in place by USDA or state authorities, does your school district **require** any specific foods or beverages to be sold at the school store?
 - ρ_1 Yes
 - ρ_2 No (Skip to Q 7)
 - ρ_3 Don't know (*Skip to Q 7*)

- 6a. Which foods or beverages are required? (Check all that apply.)
 - ρ_1 Bottled water (unflavored, with no added sweeteners)
 - ρ_2 Juice (100 percent juice)
 - ρ_3 Milk (any type)
 - ρ_4 Milk (skim, nonfat, or 1 percent low-fat)
 - ρ_5 Low-fat snacks or baked goods
 - ρ_6 Low-sugar foods
 - ρ_7 Fresh fruit or vegetables
 - ρ_8 Lean meats
 - ρ_9 Whole grains
 - ρ_{10} Diet, "lite," or sugar-free beverages
 - ρ₁₁ Other (*Please specify*): _____

Please attach a copy of the document that specifies the foods or beverages required to be sold at school stores in your school district.

- 7. Which of the following factors does your school district consider in setting prices for food and beverage items sold in the school store(s)? (*Check all that apply.*)
 - ρ_1 Food costs
 - ρ_2 Administrative or indirect costs
 - ρ_3 Incentive for choosing more healthful foods/beverages (e.g., spring water priced lower than soft drinks)
 - ρ_4 Ease of collecting payments
 - ρ_5 Other (*Please specify*): _____
- 8. Do you have nutrition information for foods and beverages sold in the school stores in your district? That is, information on nutrient content from the manufacturer or distributor?
 - ρ_1 Yes, for all foods/beverages
 - ρ_2 Yes, for some foods/beverages
 - ρ_3 No
- 9. Does the school food service operate any school stores that sell food and/or beverages to students during the school day at [SCHOOL]?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Section E*)

Please answer the following questions for the school store(s) operated by the school food service at [SCHOOL].

10. Do you have or can you obtain information on the number sold for each food and beverage item in the school store(s) at [SCHOOL]?

 $\begin{array}{ll} \rho_{1} & Yes \\ \rho_{2} & No \, (\textit{Skip to Q 11}) \end{array}$

Please attach a copy of your most recent invoice(s) or other documentation that includes this information.

- 11. Who receives the profits (i.e., revenue less costs) from food and beverage sales at the school store at [SCHOOL]? (*Check all that apply.*)
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
 - 11a. Approximately how much total profit is received by the school food service from food and beverage sales at the school store(s) in [SCHOOL] (per year, month, or week)? Do not include profits that go to the principal or other school departments or funds.

\$_____ per _____

- 11b. How are school food service profits from food and beverage sales at the school store in [SCHOOL] used? (*Check all that apply.*)
 - ρ_1 General fund
 - ρ₂ Specific purpose (*Please specify*): _____

E. Exclusive Beverage Contracts

- 1. Is your school district, or are any schools in your district, engaged in a "pouring rights" contract (that is, a long-term contract with a beverage company that establishes the company as a sole source vendor for some types of beverages? *(Check one.)*
 - ρ_1 Yes, district-wide
 - ρ_2 Yes, some schools
 - ρ_3 No (*Skip to Section F*)
 - 1a. Who decides whether or not to enter into a pouring rights contract in your school district? (*Check all that apply.*)
 - ρ_1 School board
 - ρ_2 Superintendent
 - ρ_3 School district business manager
 - ρ_4 Food service director
 - ρ_5 Principal
 - ρ_6 Cafeteria manager
 - ρ₇ Other (*Please specify*): _____
 - 1b. Which types of schools are covered by the pouring rights contract? (*Check all that apply.*)
 - ρ_1 All elementary schools
 - ρ_2 All middle or junior high schools
 - ρ_3 All high schools
 - ρ₄ Other (*Please explain*): _____
- 2. Who is the contract with?
- 3. For how many years does the contract last?

_____ years

- 4. What types of beverages are covered under the contract? (*Check all that apply.*)
 - ρ_1 Carbonated soft drinks
 - ρ_2 Naturally or artificially flavored fruit juices and fruit juice-containing drinks
 - ρ_3 Fruit-flavored drinks
 - ρ₄ Ready-to-drink tea products
 - ρ_5 Sports drinks
 - ρ_{6} Bottled water (all types)
 - ρ₇ Other (*Please specify*): _____
- 5. What is the guaranteed annual minimum commission provided by the beverage contract?

\$_____commission

 ρ Not applicable

- 6. What types of non-monetary benefits does the contract provide? (*Check all that apply.*)
 - ρ_1 Athletic equipment
 - ρ_2 Scoreboard
 - ρ_3 Music department equipment
 - ρ_4 Product for fundraisers
 - ρ₅ Other (*Please specify*): _____
 - ρ_{6} No non-monetary benefits provided
- 7. What percentage of beverage sales dollars are paid to the district and/or schools?

____%

- 8. Who receives the commission from the beverage contract? (*Check all that apply.*)
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
- 9. What types of incentive payments are included in the contract? (*Check all that apply.*)
 - ρ_1 Payment for exceeding target sales
 - ρ_2 Payment for each vending machine placed
 - ρ₃ Other (*Please specify*): _____
 - ρ_4 No incentive payments (*Skip to Q 11*)

- 10. Who receives the incentives from the beverage contract? (*Check all that apply.*)
 - ρ_1 Food service department
 - ρ_2 School
 - ρ_3 Athletic department
 - ρ_4 Other school department or fund
 - ρ_5 Student club
 - ρ₆ Other (*Please specify*): _____
- 11. Since starting the beverage contract, has the number of vending machines in your school district increased?
 - $\begin{array}{ccc} \rho_1 & Yes \\ \rho_2 & No \end{array}$
- 12. Since starting the beverage contract, has the number of other in-school sites in your district selling beverages, such as snack bars, school stores, or concession stands, increased?
 - $\begin{array}{ll}\rho_1 & Yes\\\rho_2 & No\end{array}$
- 13. Is your school district, or are any schools in your district, engaged in a long-term contract that establishes a company as the sole source vendor for any food items (other than beverages)?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Section F*)
 - ρ_3 Don't know (*Skip to Section F*)
 - 13a. Who is the contract with?
 - 13b. What types of food items are covered under the contract?

F. Menu Planning and Computer Systems

- 1. Which of the following menu-planning options is currently used for [SCHOOL]?
 - ρ₁ Nutrient Standard Menu Planning (NuMenus)
 - ρ₂ Assisted Nutrient Standard Menu Planning (Assisted NuMenus)
 - ρ₃ Enhanced Food-Based Menu Planning
 - ρ₄ Traditional Food-Based Menu Planning
 - ρ₅ Other (*Please specify*): _____
- 2. Is a computerized system used for any of the following functions for [SCHOOL]? (*Check all that apply.*)
 - ρ_1 Nutrient analysis of menus
 - ρ_2 Nutrient analysis of a la carte items
 - ρ_3 Point-of-sale payment/meal counts
 - ρ_4 Food inventory
 - ρ_{5} None of the above (*Skip to Section G*)
 - 2a. Which software system is used for each of the function(s) indicated above? (*If more than one system is used, please check the one used most often.*)

	Nutrient analysis menus	Nutrient analysis a la carte	POS	Food inventory
Bon Appetite	O ₁	O ₂	0 ₃	O ₄
B.O.S.S. (Back Office Software Solutions)	0 ₁	0 ₂	O ₃	O 4
CAFS (Computer-Assisted Food Service)	0 ₁	0 ₂	O ₃	O 4
CAFE Terminal	0 ₁	0 ₂	O 3	O 4
CompuHelp	O ₁	O ₂	O 3	O ₄
CNP Manager	O 1	O ₂	O 3	O 4
Keeping TRAC	0 ₁	0 ₂	O 3	0 ₄
KYRUS (formerly AccuSERIES)	O ₁	0 ₂	0 ₃	O 4
Lunch Box (Kyrus Corp.)	O ₁	O ₂	0 ₃	O ₄
Mealtime (Energetics)	O ₁	O ₂	O 3	O ₄
Mealtime (CLM Group, Inc.)	0 ₁	0 ₂	O 3	O 4
Meal Tracker (Accu-Scan)	O ₁	O ₂	0 ₃	O 4
Netel	0 ₁	0 ₂	0 ₃	0 ₄
NutriKids	O ₁	O ₂	0 ₃	O ₄

	Nutrient analysis menus	Nutrient analysis a la carte	POS	Food inventory
NutriMenu 2001	O ₁	0 ₂	O 3	0 ₄
Nutrition Data Systems	O ₁	0 ₂	O 3	O ₄
PCS Revenue Control Systems	O ₁	0 ₂	O 3	O ₄
SNAP Systems	O ₁	O ₂	O 3	O 4
WinSNAP/WebSMARTT	O ₁	O ₂	O 3	O 4
Custom-developed system	O ₁	O ₂	0 ₃	O 4
Other software system (<i>Please</i> specify):	0 ₁	0 ₂	0 3	0 ₄
No software for this function	O ₁	0 ₂	O ₃	O ₄

G. Food Service Management

- 1. Is the food service operation in your school district currently under the direction of a food service management company?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Section H*)
 - 1a. Are you employed by the food service management company or by the school district? (*Check one.*)
 - ρ_1 Management company
 - ρ_2 School district
 - ρ_3 Consultant
 - ρ₄ Other (*Please specify*): _____
 - 1b. Who employs the food service cafeteria workers at [SCHOOL]?
 - ρ_1 Management company
 - ρ_2 School district
 - ρ₃ Other (*Please specify*): _____
- 2. Is the food service management company involved in providing a la carte service at [SCHOOL]?
 - ρ_1 Yes
 - ρ_2 No (*Skip to Section H*)
 - 2a. For each of the a la carte food service functions listed below, indicate if it is performed by the school district, the food service management company, or jointly by the school district and the food service management company. *(Check one answer for each line.)*

Food service functions	School district	Food service management company	Jointly performed
Planning a la carte menus	O ₁	0 ₂	O ₃
A la carte food purchases, including:			
Vendor selection	O ₁	0 ₂	O ₃
Food selection	0 ₁	0 ₂	0 ₃
Determining quantities ordered	0 ₁	0 ₂	O 3
Preparing a la carte items	O ₁	0 ₂	0 ₃
Serving a la carte items	O ₁	O ₂	0 ₃

3. Other than reimbursable meals or a la carte service, is the food service management company involved in providing other sources of food at [SCHOOL]?

 $\begin{array}{ll} \rho_1 & Yes \\ \rho_2 & No \, (Skip \ to \ Section \ H) \end{array}$

- 3a. For which other sources of foods is the management company involved?
 - ρ_1 Vending machines
 - ρ₂ Snack bars (including canteens, food carts, kiosks)
 - ρ_3 School store
 - ρ₄ Other (*Please specify*): _____
H. Survey Information

- 1. Did you need to consult other district or school personnel to complete this survey?
 - $\begin{array}{ll} \rho_1 & \text{Yes} \\ \rho_2 & \text{No} \left(Skip \ to \ Q \ 2 \right) \end{array}$
 - 1a. For each section of the survey, please check if other personnel were consulted and indicate their title.
 - ρ₁
 A la carte ______

 ρ₂
 Vending machines ______

 ρ₃
 Snack bars ______

 ρ₄
 School stores _______

 ρ₅
 Exclusive beverage contracts ________

 ρ₆
 Menu planning and computer systems ________

 ρ₇
 Food service management _______

2. Please estimate the time it took you to complete the following:

Gathering information:	
Consulting with other staff:	
Completing questionnaire:	

Thank you for taking the time to complete this survey. Your cooperation is very much appreciated.



Study of Foods in the School Environment

Cafeteria Survey

Target Week:

Monday ____/ ___/ ____/ ____

through

Friday _____ / ____ / ____

If you have questions or need assistance, please call 617-349-2674 or 617-349-2661

Spring 2003

The Study of Foods in the School Environment is being conducted for the:

Food and Nutrition Service US Department of Agriculture 3101 Park Center Drive Alexandria, Virginia 22302

By:

Abt Associates Inc. 55 Wheeler Street Cambridge, Massachusetts 02138

Daily Meal Counts Form

School Name: _____

For each day of the target week, please write in the **number of USDA** *reimbursable meals* **served** in your school. Please write the number of free meals, reduced-price meals, and full-price meals, as well as the total number of meals. Do **not** include meals for which you do not claim reimbursement, for example, second lunches sold to students on an a la carte basis.

Number of Reimbursable Lunches Served

Day of Week	Free	Reduced- Price	Full-Price	Total
Monday				
Tuesday				
Wednesday				
Thursday				
Friday				

Number of Reimbursable Breakfasts Served

Day of Week	Free	Reduced- Price	Full-Price	Total
Monday				
Tuesday				
Wednesday				
Thursday				
Friday				

Every Day Cafeteria Foods Form

MEAL (circle): Breakfast Lunch

A.	B. خ*	C. E	D. E	E.	F. G. H. Number (reimbut contraction)			Number (reimburs	I. er of Portions Served irsable and a la carte)				
Food Name	How Offered	Inventory forr used	Self-serve for used	Food Description (If pre-prepared, include brand name and manufacturer)	Commodity?	Product Code**	Recipe?	Portion Size (include units)	Mon	Tues	Wed	Thu	Fri

* R = Reimbursable only A = A la carte only B = Both

** Product code (and manufacturer) is required for (a) pre-prepared foods, (b) all foods with a nutritional claim (e.g., vitamin-fortified juice, Super Donut), and (c) processed commodities.

Inventory of Every Day Prepackaged Foods

Use this form for prepackaged foods offered every day and requiring no preparation, and prepackaged condiments.

A.	B.	C.	Starting	D. Inventory	(Specif	y units, e.g.	= each)	F. Ending Inventory			
Food Name/Description (include brand name and manufacturer)	Bulk Packaging Information (Type and number per pkg)	Individual Package Size (e.g., # ounces)	# Cases/ Pkgs	# Indiv. Pkgs	Mon	Tues	Wed	Thur	Fri	# Cases/ Pkgs	# Indiv. Pkgs

Cafeteria Foods Form

MEAL (circle): Breakfast	Lunch		DAY OF WEEK (circle): Mon	T	ues Wed Thu Fri			
A.	В.	C.	D.	E.	F.		G.	H.
Food Name	How Offered?*	Self-serve form used	Food Description (If pre-prepared, include brand name and manufacturer)	Commodity?	Product Code**	Recipe?	Portion Size (include units)	Number of Portions Served (reimbursable and a la carte)

* R = Reimbursable only A = A la carte only B = Both

** Product code (and manufacturer) is required for (a) pre-prepared foods, (b) all foods with a nutritional claim (e.g., vitamin-fortified juice, Super Donut), and (c) processed commodities.

Self-Serve Bar Form

Name of Bar:		Nui	mber of Servings fr	om B	ar:			
MEAL (circle): Breakfast	Lunch DAY OF WE	EK (cir	c <i>le):</i> All	Mon	Tues Wed	Thu Fri		
A.	В.	C.	D. Additional Informat	tion	E.	Pro	F. oduction Informa	tion
Food Name	Food Description (If pre-prepared, include brand name and manufacturer)	Commodity?	Product Code*	Recipe?	Portion Size (if pre-portioned or portioned by server)	Starting Amount	Amount Added	Amount Left Over

* Product code (and manufacturer) is required for (a) pre-prepared foods, (b) all foods with a nutritional claim (e.g., vitamin-fortified juice, Super Donut), and (c) processed commodities.

Recipe Form

Recipe/Food I	Recipe/Food Name:											
Meal (circle):	Breakfast	Lunch	Day of Week (circle):	All	Mon	Tues	Wed	Thu	Fri			
No. servings prepared:			Size of one serving:			-						

Check the box next to the option you select for completing this form:

□ Side 1 and side 2 of Recipe Form completed

□ Copy of recipe attached to side 1 of Recipe Form. Side 2 of form completed.

Note: If you attach a copy of a printed recipe, remember to edit the recipe, if necessary, to show how the recipe was prepared *in your school*.

A.	В.	C.	D. Additional Informa	ation	E.
Ingredient Name	Ingredient Description (If pre-prepared, include brand name and manufacturer)	Commodity?	Product Code*	Recipe?	Amount in Recipe <i>(include units)</i>

* Product code (and manufacturer) is required for (a) pre-prepared foods, (b) all foods with a nutritional claim (e.g., vitamin-fortified juice, Super Donut), and (c) processed commodities.

Recipe Form

(Side 2)

Please check (T) the boxes below to describe the procedures used in preparing this recipe.

1.	If recipe was cooked, what cooking metho	od did you u	use? (Che	eck all that apply.)
	□ Bake/roast □ Pan f □ Oven heat □ fl □ Microwave/warmer □ b □ Broil/grill □ Deep □ fl □ b	fry/sauté oured attered fry oured attered		 Boil Steam Other (<i>Please specify</i>):
2.	If recipe contains meat or poultry, was ar	nount meas	sured raw o	or cooked?
	 Does not apply to recipe (Skip to Q4) Raw Cooked 			
3.	If recipe contains meat or poultry did you	(Chec	k all that a	pply.)
	Trim the visible fat?	□ Yes	🗆 No	\Box Does not apply to recipe
	Drain fat after cooking?	□ Yes	🗆 No	\Box Does not apply to recipe
	Rinse with hot water, drain fat, and then rinse again?	□ Yes	🗆 No	□ Does not apply to recipe
	Remove skin before cooking?	□ Yes	🗆 No	\Box Does not apply to recipe
4.	If recipe contains noodles, rice, or vegeta	bles did yo	u add salt	to the cooking water?
	Noodles/pasta or rice	□ Yes	🗆 No	Does not apply to recipe
	Vegetables	□ Yes	□ No	\Box Does not apply to recipe
5.	If recipe contains canned vegetables or c	anned fruit,	did you dr	rain off all of the liquid?
	□ Yes □ No □ [Does not ap	ply to reci	pe

COMMENTS

Food Outside Meal Periods Checklist

Instructions: Answer Question 1 below. If your answer is "no," you do not need to complete this checklist. If you answer "yes," continue to Question 2 and complete the checklist.

- 1. Are students allowed to come to the cafeteria to purchase food or beverages between meals?
 - $\Box_1 \text{ Yes (Continue)}$ $\Box_2 \text{ No (Do not complete this form)}$
- 2. What hours may students purchase food or beverages between meals? (*Check all that apply and specify hours.*)
 - \square_1 Before breakfast, from ____: ____ to ____: a.m.
 - \square_2 After breakfast and before classes start, from ____ : ___ to ____ : ___ to ____ : ____ a.m.
 - \square_3 After start of classes and before lunch, from _____: ____ to ____:
 - \square_4 After lunch and before school ends, from ____ : ___ to ____ : ___ p.m.
 - \Box 5 After last class, from _____: ____ to _____: p.m.

Place a check mark in the box corresponding to each food and/or beverage for sale outside the regular breakfast and lunch periods during the target week. For example, if students can purchase muffins between breakfast and lunch, or if any foods are available all day, check them on this form. Do not include foods sold only to teachers or other adults.

- A. Beverages
 - Carbonated soft drinks (cola sweetened, cola diet, non-cola sweetened, non-cola diet)
 - \square_2 Juice (ρ 100% juice / ρ 50% juice)
 - □ ₃ Juice drinks (cranberry drink, fruit blends, Hi-C, lemonade, punch)
 - \square_4 Water (ρ spring water, sparkling water, mineral water, seltzer water / ρ flavored water with sweetener or fruit juice)
 - \square_5 Coffee or coffee-based (ρ Regular / ρ Decaffeinated)
 - \square_6 Tea or tea-based (ρ Regular / ρ Decaffeinated)
 - \Box_7 Hot chocolate
 - \square_8 Yogurt drinks
 - Energy and sports drinks (Gatorade, Powerade, Red Bull)
 - \square_{10} Other (*Please specify*): _____

- B. Dairy
- \square_1 Whole milk
- \square_2 Low fat (2%) white milk
- \square_3 1% white milk
- \square_4 Fat-free milk (skim, nonfat)
- \square_5 Flavored milk (ρ Whole or 2% / ρ Fat-free or 1%)
- \Box_6 Yogurt
- \square_7 Cheese (ρ Part-skim, string / ρ Regular fat)
- C. Baked Goods Dessert
 - \Box_1 Cake-type (brownies, cupcakes, Twinkies)
 - \square_2 Cookies
 - \square_3 Pastries (pies, turnovers)
 - \square_4 Other (*Please specify*): _____
- D. Bread or Grain Products
 - \square_1 Regular bread (bread, rolls, bagels)
 - \square_2 Other bread (biscuits, croissants, hot pretzels)
 - \square_3 Muffins
 - \Box_4 Granola bars
 - \Box_5 Pretzels
 - \square_6 Crackers/Cracker sandwiches (ρ peanut butter / ρ cheese)
 - \Box_7 Cereal/Cereal bars
 - □₈ Other (*Please specify*): _____
- E. Frozen Desserts
 - \square_1 Frozen non-dairy (fruit bars, Jello Pops, Popsicles)
 - \square_2 Ice cream (Bars, cups, Fudgsicles, sundaes)
 - \square_3 Low-fat frozen desserts (frozen yogurt, ice milk, sherbet)
- F. Fruit and Vegetables
 - \square_1 Canned, cooked fruit
 - \square_2 Fresh fruit
 - \square_3 Fruit salad (ρ fresh / ρ canned)
 - \Box_4 Dried fruit
 - \square_5 Vegetables, side salad
 - \square_6 Other fresh vegetables

- G. Snacks
 - \square_1 Chips (corn, potato, puffed cheese, tortilla)
 - \square_2 Nuts and seeds (almonds, peanuts, sunflower seeds, trail mix)
 - \square_3 Fruit roll-ups or fruit snacks
 - \Box_4 Popcorn
 - □ 5 Meat snacks (jerky, pork rinds)
 - \square_3 Candy (ρ with chocolate / ρ without chocolate)
 - \square_6 Energy bars (Balance Bars, Luna Bars, Power Bars)
 - \Box_7 Other (*Please specify*): _____
- H. Pre-prepared Entrees
 - \square_1 Hamburgers or cheeseburgers
 - \square_2 Hot dogs
 - \square_3 Other sandwiches
 - \square_4 Meal-side salad
 - 🛛 5 Pizza
 - \Box_6 Pasta

I. Other (*Please specify*):

Snack Bar Foods Form

Location of Snack Bar: _____

A.	B.	C.	D.	E.	F. Additional Informatic	on	G.			Н.				
	sred?*	, pe		ity?	ity?			Number of Items Sold						
Food Name	How Offe	Inventory Form Use	Food Description (If pre-prepared, include brand name and manufacturer)	Commod	Product Code**	Recipe?	Portion Size (include units)	Mon	Tues	Wed	Thu	Fri		
									-	-				

* R = Reimbursable only A = A la carte only B = Both

** Product code (and manufacturer) is required for (a) pre-prepared foods, (b) all foods with a nutritional claim (e.g., vitamin-fortified juice, Super Donut), and (c) processed commodities.

Inventory of Snack Bar Prepackaged Foods

Use this form for prepackaged snack bar foods requiring no preparation and prepackaged condiments.

A.	B.	C.	Starting	D. Inventory	(Specif	y units, e.g.,	= each)	F. Ending Inventory			
Food Name/Description (Include brand name and manufacturer)	Bulk Packaging Information (Type and number per pkg)	Individual Package Size (e.g., no. ounces)	# Cases/ Pkgs	# Indiv. Pkgs	Mon	Tues	Wed	Thur	Fri	# Cases/ Pkgs	# Indiv. Pkgs

Starting Inventory Taken:	:	a.m./p.m.
Ending Inventory Taken:	:	a.m./p.m.

Inventory of School Store Food and Beverage Items

Location of School Store: _____

A.	B.	C.	Starting	D. Inventory	E. Deliveries (Specify units, e.g., cs = case, pk = pkg, ea = each)					F. Ending Inventory	
Food Name/Description (Include brand name and manufacturer)	Bulk Packaging Information (Type and number per pkg)	Individual Package Size (e.g., no. ounces)	# Cases/ Pkgs	# Indiv. Pkgs	Mon	Tues	Wed	Thur	Fri	# Cases/ Pkgs	# Indiv. Pkgs

Inventory of Vending Machine Items

Location of Vending Machine: _____

Vending Machine # (if applicable): _____

A.	В.	C.	Number of Items in Vending Machine						
Vending Slot Food Name/Description		Individual Package Size	D.	E. Number Added				_ F.	
(if applicable)	if applicable) (Include brand name and manufacturer. If prepared item, (e.g., no. ounces)	(e.g., no. ounces)	Number	Mon	Tues	Wed	Thur	Fri	Ending Number
-									